Report by ICRIER, New Delhi for DfID, India

EXPLORING TRADE AND INVESTMENT OPPORTUNITIES BETWEEN INDIA AND SELECT AFRICAN AND ASIAN ECONOMIES

Anirudh Shingal, Neha Gupta, Minakshee Das, Akshaya Aggarwal, Varsha Jain

Final version: 14 October 2019

TABLE OF CONTENTS

	PAGE	NO.
1	EXECUTIVE SUMMARY	3
2	BACKGROUND AND INTRODUCTION	5
3	REVIEW OF LITERATURE	8
4	EMPIRICAL METHODOLOGY AND DATA SOURCES	11
5	TRADE AND INVESTMENT BARRIERS BETWEEN INDIA AND SELECT AFRICAN AND ASIAN COUNTRIES	16
6	MAPPING MERCHANDISE EXPORTS OF AFRICA AND ASIA IN INDIA	27
7	MAPPING SERVICES EXPORTS OF AFRICA AND ASIA IN INDIA	42
8	SUPPLY CONSTRAINTS IN SELECTED AFRICAN AND ASIAN COUNTRI	ES 50
9	INDIA'S OUTWARD INVESTMENT IN AFRICAN AND ASIAN COUNTRIE	S 73
10	QUADRANT ANALYSIS	81
11	CONCLUSION AND POLICY RECOMMENDATIONS	93
RI	EFERENCES	98
AN	NNEX A	100
Αľ	NNEX B	142
Αľ	NNEX C	180
Αľ	NNEX D	187

EXECUTIVE SUMMARY

Using descriptive statistical analysis, this study examines trade and investment opportunities between India and 41 African and Asian economies (mostly LDCs) by focusing on the latter's export opportunities in the Indian market and on India's investment opportunities in the selected countries. It also discusses barriers to realizing the identified trade and investment opportunities between India and the selected economies, based on a review of the existing literature.

Amongst the sample countries, the top sources of merchandise imports for India over time include Nigeria and Angola from Africa, and Myanmar, Bangladesh and Nepal from Asia. India's main imports from Africa include petroleum oils, precious stones and metals, mineral ores, coal, natural gas, and agricultural products. India's imports from Asia have also been concentrated in few traditional product lines such as jute, fish, gold, petroleum oil, copper ores, ferro-silico, beans, nuts, and non-alcoholic beverages. Thus, the export baskets of the sample countries studied in this report are still not diversified enough to meet the rising demand for consumer and capital goods in India, which points to the immense untapped potential for enhancing these countries' trade and investment relationship with India.

While the value of these countries' services exports to India is much lower than that of their merchandise exports, India's services imports have increased over time from all the 38 countries for which services trade data are available. This shows these countries' growing services trade linkages with India, which must also be strengthened with the right polices. Over time, Nigeria and Nepal have remained the top African and Asian services exporters to India, respectively. India's major services imports from these countries include travel, transportation and other business services (OBS). This sectoral distribution also points to the need for enhancing these countries' export capacities in other sectors especially computer and information services, where there is potential for value-chain integration with India.

Limited capacity in LDCs of Africa, and even Asia, for supplying goods and services to India has been regarded as one of the main obstacles in full realization of their bilateral trade. India can thus potentially invest in those sectors in the sample countries which have supply constraints (i.e., where India's global imports exceed the concerned country's global exports as well as the associated region's global exports). Results show that none of the Asian LDCs have a supply constraint in any of the goods and services, due to the Asian region's huge global exports. In Africa, in contrast, almost all manufacturing goods and financial and computer-related services are identified as lacking supply capacity and such constraints seem to be evident in all the sample countries from the continent.

Meanwhile, India's cumulative announced greenfield investment over the 2010-2018 time period was \$19.2 billion and \$10.3 billion in the selected African and Asian countries, respectively. In Africa, the top recipients include Mozambique, Nigeria and Ghana with a cumulative 58.8% share, while in Asia, Bangladesh alone received around 72% of the total greenfield investment. Sector-wise, India has invested majorly in the communications sector in Africa (\$2.7 billion in the sample countries), followed by real estate and software & IT services; and chemicals and automotive and non-automotive OEM. This is in contrast to the Asian countries where India has invested heavily in sectors like textiles (\$2.3 billion) and automotive OEM (\$1.8 billion), followed by coal, oil and gas, business services, renewable energy, financial services and even wood products.

Finally, the study has also undertaken a quadrant analysis to identify the products and services where the selected countries still have low shares in Indian imports despite a revealed comparative advantage in exporting, suggesting the need for active trade promotion policies. Across both regions, most of these products belong to the categories of general, electric and precision machinery followed by wood articles, paper articles, as well as fruits and vegetable products, and also some textiles items in the case of Asian LDCs. For services, transportation, other business services, financial, and computer and information services have been identified as the target sectors for pursuing active trade promotion policies.

CHAPTER 1

Background and introduction

The importance of least developed countries (LDCs) has been growing in India's trade and investment flows over time. In fact, India has been strongly promoting trade linkages with LDCs, particularly those in Africa. Illustratively, African countries accounted for 8.3% of India's merchandise exports of US\$294 billion in 2017 up from 8.1% in 2010 (global exports of US\$220 billion); the respective shares of African LDCs were significant at 3.3% and 2.3%. In contrast, although the share of Africa in India's total merchandise imports declined from 8.9% in 2010 to 8% in 2017, the share of African LDCs increased from 1.9% to 2.3% (India's global merchandise imports were US\$350 billion and US\$444 billion in 2010 and 2017, respectively)¹.

On the investment front, according to UNCTAD's World Investment Report (WIR) of 2018, total foreign direct investment (FDI) into the African continent declined from US\$46.8 billion in 2011 to US\$41.8 billion in 2017. Consistent with this trend, the stock of Indian FDI in Africa also declined from US\$16 billion in 2011 to US\$14 billion in 2016. Even so, UNCTAD (2018b) puts India as the eighth largest investor in Africa which is indicative of India's rising investment potential in Africa (including LDCs). Moreover, the flow of FDI has not been unidirectional. Between 2010 and 2014, the stock of African FDI in India increased from US\$57 billion to US\$73.3 billion, with FDI from Africa in India accounting for almost 23% of the country's FDI stock in 2014 (UNECA & CII, 2018).

The picture is somewhat different for Asia in terms of both trade and investment. Illustratively, Asian developing countries accounted for 47.5% of India's merchandise exports in 2017 but down from 50% in 2010; the respective shares of Asian LDCs were only 5.3% and 3.1%. Notably, the shares of South Asian and South-East Asian economies as a whole in India's merchandise exports in 2017 were 7.7% and 12%, respectively. In contrast, South Asian LDCs (Afghanistan, Bangladesh, Bhutan and Nepal) accounted for 4.7% and 2.5% in India's merchandise exports in 2017 and 2010, respectively; in contrast, the respective shares of South East Asian LDCs (Cambodia, Lao PDR, Myanmar and Timor-Leste) were very low at 0.41% and 0.39%. Meanwhile, the high share of Asia in India's total merchandise imports increased from 54% in 2010 to 56% in 2017, while that of Asian LDCs declined from 1.3% to 0.6%. However, in the case of South Asian LDCs, the shares were a meagre 0.37% in both these years (while the share of all South Asia remained around 3%); the shares declined for South East Asian LDCs from 0.32% in 2010 to 0.24% in 2017 (the share of all South East Asian economies increased from 10% to 11% over the same time-period).

Meanwhile, UNCTAD data on bilateral investment show that total FDI in South Asia declined from US\$31.3 billion in 2010 to US\$27.2 billion in 2012 and India's share in this total became much smaller over these years (reduced from a share of 4% to a share of 0.6%). Thus, India's outward FDI flows in South Asia declined from US\$1.3 billion in 2010 to US\$0.16 billion in 2012, though the stock of Indian FDI in South Asia increased from US\$0.52 billion to US\$1.07 billion between 2010 and 2012, respectively. Over the same time period, outward FDI from South Asia nearly halved from \$16.1 billion to \$8.7 billion, though the value of India as a

¹ Trade data are sourced from UNCTADStat (https://unctadstat.unctad.org/wds/ReportFolders/reportFolders.aspx?sCS_ChosenLang=en); FDI data are sourced from UNCTAD FDI Statistics and WIR 2018.

destination fell only slightly from \$7 million to \$6 million and the share, though insignificant, nearly doubled from 0.04% to 0.07%. At the same time, India's outward FDI flows in South East Asia also declined from US\$4.13 billion in 2010 to US\$1.98 billion in 2012, though again, the stock of Indian FDI increased from US\$20.8 billion to US\$ 22.3 billion over these years. For outward FDI from South East Asia, the value of India as a destination jumped from US\$1.6 billion in 2010 to US\$1.9 billion in 2012 (though shares declined slightly from 10.6% to 10.3% over the same period).

These stylized facts also highlight a major constraint in working with FDI data. Publicly available data on FDI from UNCTAD and in different editions of the WIR do not report India's inward and outward FDI stock or flows bilaterally to enable a calculation of the value of investment originating in or destined for LDCs in either Africa or Asia. The required data are neither available in aggregate nor at sectoral levels, which renders any meaningful analysis of bilateral investment opportunities involving India and African economies and Asian LDCs virtually impossible.

A review of existing literature undertaken in the following chapter suggests that India's investment engagement in all Asian LDCs has still not been studied at the disaggregated sectoral level. Moreover, while several studies have examined India's trading potential in Africa and even in South and South-East Asia, including in LDCs in these regions, existing work has not yet provided a comprehensive, comparative analysis. Besides, all existing studies draw inferences on bilateral investment opportunities on the basis of aggregate FDI data, which is a serious limitation and has obvious implications for the associated policy conclusions.

Against this background, the objective of this study is to examine trade and investment opportunities between India and African and Asian economies (mostly LDCs) by focusing on the latter's export opportunities in the Indian market and on India's investment opportunities in the selected Asian and African countries. The study further aims at discussing the barriers to realizing the identified trade and investment opportunities between India and selected Africa and Asian economies, based on a review of the existing literature. The overall analysis involves secondary research based on literature and data on trade and investment available from different trade and investment databases (details in Chapter 3.2).

The selected 41 African and Asian economies covered in this study are as follows²:

- Africa (33): Angola, Benin, Burkina Faso, Burundi, Central African Republic, Chad, Democratic Republic of Congo, Djibouti, Eritrea, Ethiopia, Gambia, Ghana, Guinea, Guinea-Bissau, Kenya, Lesotho, Liberia, Madagascar, Malawi, Mali, Mauritania, Mozambique, Niger, Nigeria, Rwanda, Senegal, Sierra Leone, Somalia, Sudan, Togo, Uganda, Tanzania, and Zambia (*LDCs+3*)
- **Asia** (8): Bangladesh³, Bhutan, Cambodia, Lao People's Democratic Republic (Lao PDR/Laos), Myanmar, Nepal, Timor-Leste (East Timor), and Yemen (*all are LDCs*)

² The choice of countries was made in consultation with DfID on the basis of their importance in the SITA framework. In the case of Africa, for instance, this included 30 of the 33 LDCs in the continent and three non-LDCs - Ghana, Kenya and Nigeria.

³ Note that LDCs were classified based on UNCTAD's latest 'The Least Developed Countries Report 2018', released in November 2018 (UNCTAD, 2018c). While this report includes Bangladesh amongst Asian LDCs (see Page 9 of the report), Bangladesh met the eligibility criteria for graduation from LDC status for the first time in 2018 only, but it is unlikely to leave the LDC status before 2024. Thus it may be noted that "three countries meet

On the trade front, the study aims to analyse the merchandise trade potential at the bilateral and disaggregated HS 6-digit-level, with data sourced from UN Comtrade at two different points in time – period average 2010-2012 and period average 2015-2017 – to enable a dynamic comparison of trade patterns and potentials over time. The analysis of services trade patterns and potential is carried out at the bilateral and sectoral levels using the recently released OECD-WTO Balanced Trade in Services Statistics (BaTIS) database for the period average 2000-2002 and 2010-2012, which, despite the lack of availability of more recent bilateral services trade data between India and the countries included in this study, will constitute significant value-added relative to existing literature.

Apropos to FDI, unfortunately, UNCTAD FDI data do not provide a disaggregated bilateral breakdown of inward or outward investment to enable more granular analysis. The study therefore seeks to examine India's investment opportunities in African countries and the Asian LDCs using data from 'fDI Markets', which is a private database on cross-border announced greenfield investment projects maintained by the Financial Times. These data cover 182 source and destination countries and 39 goods and services sectors over 2003-2018, enabling a comprehensive analysis of India's investment patterns and potential in African and Asian economies, most of which are covered in this database. The use of fDI Markets for analysis thus constitutes a significant value-addition to existing literature on this subject in the context of India.⁴

The layout of the report is as follows. Chapter 2 provides a brief review of existing literature followed by a chapter detailing methodology and data sources. Chapter 4 discusses the barriers faced by the Asian and African countries in realizing trade and investment opportunities vis-àvis India. Chapters 5 and 6 discuss results from an analysis of bilateral merchandise and services trade, respectively. Chapter 7 identifies supply constraints in the selected 41 countries in exporting to India while Chapter 8 examines India's outward investment in these countries. Chapter 9 discusses results from the quadrant analysis undertaken to highlight sectors of export potential in these countries with respect to India. Finally, Chapter 10 concludes the study with policy recommendations.

the graduation eligibility criteria for the first time: Bangladesh and Myanmar meet all three criteria; the Lao People's Democratic Republic meets both GNI and human assets index criteria. The countries were duly notified of that finding and will be considered for graduation at the next triennial review in 2021...In its triennial review of the list of least developed countries, the Committee found that Bangladesh, the Lao People's Democratic Republic and Myanmar fulfilled the criteria for graduation for the first time. They will be considered for graduation at the next triennial review in 2021, with a view to graduating in 2024.", as per UN's Committee for Development Policy's Report at its twentieth session (12–16 March 2018 – pages 10 and 23) (see, UNCTAD, 2018a).

⁴ However, one limitation of fDI Markets is that it only covers greenfield investment. According to World Investment Report (2018), announced greenfield investment (US\$720 billion) accounted for 50% of total global FDI (US\$1.43 trillion) in 2017, with the value of net cross-border mergers & acquisitions (M&A; US\$694 billion) making up for another 48.5%. M&A activity is thus not included in the analysis that follows, which is likely to influence some of the patterns that are observed in this report. To that extent, the analysis that follows may not be representative of the complete FDI potential that India may have in these countries, but that is a data limitation constraint that we cannot circumvent.

CHAPTER 2

Literature review

Several studies have examined India's trading potential in Africa, including in LDCs in that continent. For instance, Nayyar and Aggarwal (2014) show rising trade flows from India to Africa in the decade of the 2000s. India has been providing support for African development through its Exim Bank, while Africa too is ready to emerge as a strong trading partner for India. Afreximbank and Exim India (2018) highlight the importance of their rising economic partnership to strengthen South-South trade. Africa generally exports primary commodities to India, while the latter mainly exports manufacturing goods to the continent. However, as per Afreximbank and Exim India (2018), Indian-African trade is likely to double by the year 2021 if infrastructure bottlenecks and tariff and non-tariff barriers are properly addressed and potentials are fully realized. UNECA and CII (2018) have also analysed the pattern of Africa-India trade, emphasizing on better utilization of India's duty-free tariff preference scheme for African LDCs. This study also examines the effects of the Regional Comprehensive Economic Partnership (RCEP) agreement involving Australia, India, ASEAN+3 and New Zealand on African-Indian trade relations, but at the same time stipulates that the African Continental Free Trade Area (CFTA) would help to remove negative effects and provide more market access to Indian firms.

Existing work also notes that while China and India have been the main donors for Africa, their capacity to grant and areas of intervention differ. For instance, McCormick (2008) portrays China as major donor of both monetary and non-monetary aid to Africa in areas of trade, infrastructure, plants and equipment, etc. In contrast, India provides lesser aid but more of non-monetary grants in the form of technical assistance and mainly in the field of ICT. Despite China's greater economic engagements, the growing activities of India have also led to advancement in many African countries, besides emphasizing the need for improvement in Africa's reforms for trade (Broadman, 2008).

Banga and Saha (2016) follow a somewhat similar approach as ours of integrating LDCs into India's global value chains (GVCs) by looking at both trade and investment opportunities. The study identifies inputs into India's identified lead products which India could source from LDCs in a more competitive manner compared to existing sources as well as sectors for potential Indian investment in African and South Asian LDCs. Banga and Saha (2016) suggest, for instance, that 20 LDCs, including Uganda and Tanzania that have the highest potential market access into India, could export about U\$12 billion of inputs for exports of India's identified lead products.

In contrast, the literature is relatively thin in exploring India's trade opportunities at the disaggregated level in the South Asian region. For instance, UNCTAD, Commonwealth Secretariat, and CWTO (2010) show the strong potential of the textiles and clothing sector in developing regional supply chains in South Asia using case studies of India, Sri Lanka, Pakistan and Bangladesh. In terms of LDCs in the region, most studies largely pertain to Bangladesh. KPMG and DFID (2016), for instance, analyze the export and import profile of Bangladesh with respect to changes in its trade basket, revealed comparative advantages, trade balances, and engagement mainly with the UK. Thus, India's trade potential with all the South Asian LDCs at the disaggregated level has not been properly examined, which would be one of the important contributions of this study.

Regarding LDCs in South East Asia, namely, Laos, Myanmar and Cambodia, several studies in the last few years focus on India's bilateral trade relations with them and suggest measures for enhancing trade as well as promoting growth in LDCs via India's policies. Illustratively, Exim Bank (2014) documents the rising growth graph of these countries with a number of economic reforms already underway, with international trade recognized as one of the main contributors. The study shows how India has developed linkages with ASEAN countries since its 'Look East Policy' of 1992, especially its growing trade with LDCs along with Vietnam (CMLV) which has risen from US\$1 billion to almost US\$11 billion over the period of 2004-13. But Exim Bank (2014) also mention that India's exports mainly consisted of pharmaceuticals, machinery, motor vehicles, plastic articles and cotton. Notably, there exists a lot of potential to enhance bilateral relations where border trade between India and CMLV is of higher importance, linking through Myanmar. Particularly, in the case of Myanmar and Bangladesh, Exim Bank (2018a) provides an analysis of India's export potential in these countries using HS 6-digit level commodities.

In 2008, India introduced the Duty Free Tariff Preference Scheme (DFTP) for providing unilateral tariff preferences to all LDCs in their merchandise exports to India. However, LDCs' date of joining the scheme differs. Over the period 2008-12, the scheme covered almost 85% of India's total tariff lines in which LDCs were allowed duty-free market access in India, followed by another 9% lines where Margin of Preference (MOP) benefits were allotted (ranging between 10-100%), while only 6% of lines remained in the exclusion list with applicable MFN duty rates. Interestingly, in 2008 itself, the scheme covered about 92% of the tariff lines of LDCs export interest. The DFTP scheme was extended in 2014 by India's Department of Commerce in terms of coverage of countries and tariff lines (6-digit level). Consequently, the scheme now provides duty-free or preferential market access to about 98.2% of tariff lines.

As on October 2017, the following LDCs were notified as beneficiaries of the DFTP scheme: Benin, Burkina Faso, Burundi, Central African Republic, Chad, Comoros, Eritrea, Ethiopia, Gambia, Guinea, Guinea Bissau, Lesotho, Liberia, Madagascar, Malawi, Mali, Mozambique, Niger, Rwanda, Senegal, Somalia, Sudan, Uganda, Tanzania, Togo, Zambia, Afghanistan, Bangladesh, Cambodia, Timor-Leste, Lao PDR, Myanmar and Yemen. The number of tariff lines under MoP scheme have also been reduced from 458 to 114, and from 326 to 97 under the exclusion list (Ministry of Commerce, 2017; International Trade Centre, 2015). The present study also provides an opportunity to examine the export potential that the select African and Asian countries have in the Indian market, given the DFTP scheme and its extension.

As compared to existing and growing literature which examines India's trade potential in selected LDCs, the lack of available data FDI from UNCTAD and in WIR also accounts for the comparatively thin literature studying India's investment potential in Africa and Asia. Even in those few studies, the focus is not entirely on understanding the investment dynamics; rather they look at growing Chinese and Indian economic engagements in the regions both in terms of bilateral trade and investment. For instance, Schiere (2011) mentions about China's involvement in financing of development projects in Africa and investing in their infrastructure. At the same time, China also stands to gain from Africa in terms of getting access to raw materials, finding a market for its manufactured exports and investment in labour-intensive manufacturing.

In other work, Nayyar and Aggarwal (2014) show not only rising trade flows from India to Africa in the decade of the 2000s, but also growing outward investment. While Africa is all set

to emerge as a big investment partner for India, the study also mentions potential hindrances that can affect this rising trend. In this respect, more recently, Afreximbank and Exim India (2018) have highlighted how both the Exim Banks are working to deal with the issue of lack of finance, which majorly disrupts trade and investment flows between India and Africa.

At the same time, growing bilateral investment flows are also indicated in this decade, where both India and Africa are continuously offering each other as preferable investment destinations. In another recent joint report, UNECA and CII (2018) also analyse Africa-India investment patterns along with trade. Lack of information and basic infrastructure are reported as big concerns preventing the required growth in both trade and investment.

Overall, existing literature largely looks at aid for trade, trade-investment linkages, as well as India's comparison with China, the major donor in Africa. However, there are very few studies which suitably portray India's investment opportunities in African and Asian LDCs. Only Banga and Saha (2016) identify potential areas (such as packaging) for Indian investment in African LDCs which could productively increase Africa's capacity to export and help them integrate into India's GVCs. However, there is not much work specific to South Asian LDCs. KPMG and DFID (2016) particularly focus on understanding the investment climate of Bangladesh including a historical perspective on FDI, as well as sector- and country-wise analyses. Kumar and Singh (2009) identify impediments to regional integration in South Asia in terms of promoting trade and investment and suggest a more proactive role for India to lead on this front.

Regarding South East Asian LDCs, India through its Exim Bank provides a lot of support to its companies investing in CMLV via lines of credit, project exports, etc. (Exim Bank, 2014). As these countries suffer from infrastructural deficiencies, underdeveloped financial and education sectors (but with growing modernization in ICT), these areas provide plenty of opportunities. Investments in LDCs' agriculture and allied fields of fisheries and forest products (teak) are other potential areas for Indian companies to consider. However, with Myanmar and Bangladesh in particular, a more detailed investment analysis has been done sector-wise by Exim Bank (2018a) using fDI markets database, along with comprehensive trade potential analysis. The report documents that over the period 2003-2018 (March), both the countries obtained most of India's greenfield investment in their coal, oil and natural gas sectors, followed by automotive OEM, financials services, etc. In other work, Dixit (2015) emphasizes the obstacles that Indian investors have to face in CMLV in terms of low connectivity due to latter's weak road and rail infrastructure, along with the need to deal with presence of already strong competitor-investors therein.

On the whole, India's investment engagement in all Asian LDCs has still not been studied at the disaggregated sectoral level particularly for South Asia. Even in terms of African LDCs, exploring India's investment potential is limited to a few selected sectors as in Banga and Saha (2016). Against this background, the present study would also add value by examining India's investment opportunities in 41 selected African and South Asian economies using data from fDI Markets.

CHAPTER 3

Empirical methodology and data sources

This chapter outlines the methodology used to identify export opportunities of selected Asian and African economies in India and identify those sectors with investment potential for India in these selected economies, mostly LDCs. The data sources used for analysis are described in the second section of this chapter.

3.1 Methodology

This section outlines the methodology adopted in the study.

3.1.1 Identifying export opportunities in India

The methodology for identifying Africa and other Asia's export opportunities in India utilizes a two-pronged approach.

3.1.1.1 Mapping African and Asian exports

The first approach involves mapping existing exports of selected African and Asian countries to India over time with India's total imports from the rest of the world (ROW) based on disaggregated trade data at the HS-6 digit level sourced from UN Comtrade (in case of goods).

Note that the study uses mirrored statistics, i.e., India's imports of goods from the select 41 countries in all this analysis, due to the former's much better coverage given the limited availability of data on these countries' exports to India, especially at the HS-6 digit level.

3.1.1.2 Quadrant analysis

The second approach involves the use of quadrant analysis (based on Widodo, 2009 and KPMG and DFID, 2016) to classify existing bilateral merchandise exports of the selected countries to India at the HS-6 digit level on the basis of their competitiveness and shares in the latter's exports.

Quadrant analysis entails the use of the following two indices:

(i) The Bilateral Revealed Comparative Advantage or the BRCA Index: The BRCA measures a country's comparative advantage in exporting products to partner markets. An index value exceeding one suggests that the exporting country is competitive in the concerned product in the partner market, while a value less than one suggests the converse.

Formally,
$$BRCA = \frac{X_{ij}^k/\sum X_{ij}}{X_{iw}^k/\sum X_{iw}}$$

[where X = export value, i is the exporting country, j is the destination market (India), k is the product at the HS6-digit level, and w is the rest of the world.]

To make intertemporal and international comparisons meaningful, the BRCA index has been normalized by using the formula (BRCA-1)/(BRCA+1). A positive value of the normalized BRCA index reveals a comparative advantage while a negative value denotes comparative disadvantage.

(ii) Share in Indian Imports (Share): This share measures the demand for foreign goods in the Indian market and is constructed as the ratio of India's import of a product from a selected country to India's import of that product from the world.

Formally,
$$Share = \frac{M_{ij}^k}{M_{iW}^k}$$

[Where M = import value, i is reporting country (India), j is partner (selected 41 Asian and African countries), k is the product at the HS6-digit level, and W is world.]

For the purpose of analysis, we delineate the share into below-average and above-average where the average (Avg) is calculated over all the HS6 products that India imports from a particular country. A product share below India's calculated average for a particular country suggests that the product has a low demand in India's import basket, and vice-versa.

Share in India's Imports

Share <Avg
Share >Avg

Group B:

Group A:

BRCA>0 and Shares>Avg

BRCA>0 and Shares>Avg

Group D:

BRCA<0 and Shares>Avg

BRCA<0 and Shares>Avg

Figure 3.1: Exposition of quadrant analysis

The quadrant analysis thus classifies each country's HS-6 digit products' exports to India into four groups based on the BRCA index and share in India's imports (see Figure 3.1). Group A comprises products whose BRCA values are positive and the share in India's total imports is above-average; this group thus represents already significant exports which are being competitively supplied by the selected African and Asian countries to meet the prevalent demand in India for such products. The policy response for these products would be to preserve the status quo.

Group B comprises otherwise competitive products but where the concerned country is not able to capture a significant share of the Indian market (below average share). This group thus

identifies products/sectors where comparative advantage exists but where more opportunities can be created by appropriate policy measures such as export promotion in the selected 41 countries.

In contrast, Groups C and D include HS6 products where the countries lack comparative advantage, but at least possess a significant share in India's total imports in the former case, which warrants some policy response towards building comparative advantage in the concerned exporting country.

Note that the analysis of goods trade will be carried out over two time periods - 2010-12 and 2015-17 – to provide a dynamic picture of the export potential. We average the trade data over three year periods to minimize fluctuations in reporting practices.

Finally, the analysis is replicated for 'services trade' using bilateral and sectoral data from OECD-WTO Balanced Trade in Services Statistics (BaTIS) over two different time periods: 2000-02 and 2010-12. The subscript "k" in the BRCA and Share measure then denotes the services sub-sector as reported in BaTIS (see Chapter 3.2 for details).

3.1.2 Identifying sectors with investment potential for India in the selected countries

Inadequate supply-side capacity is one of the main reasons for selected African and Asian economies' (i) limited export competitiveness at the sectoral level; (ii) lack of diversified production and export structure; and (iii) limited integration into regional and global value chains. This suggests the need for both regional and extra-regional investment in sectors where these countries exhibit competitiveness but lack capacity.

The methodology for identifying sectors in these 41 countries with investment potential for India is based on a two-pronged approach:

- Identifying existing bilateral investment sectors between India and the chosen countries over time based on data from fDI Markets; and
- Building on Banga and Saha (2016), considering sectors where India's global imports exceed the concerned country's global exports as well as the associated region's global exports, thereby indicating a supply constraint in both the country and the geographical neighbourhood.

This two-pronged approach aims to provide a complete identification of existing and potential investment opportunities for India in the chosen 41 countries based on both aggregate and bilateral investment data, which is an original contribution to existing literature. The analysis is carried out during the cumulative time period 2010 to 2018 to enable a wider coverage of countries and sectors from fDI Markets.

3.2 Data sources

This section describes the data sources used in the study.

3.2.1 Trade data

Bilateral merchandise trade data between India and the selected 41 countries are sourced from UN Comtrade at the disaggregated HS 6-digit level at two different period averages – 2010-12

and 2015-17 – to enable a dynamic analysis. The study works with mirrored statistics in the case of goods trade analysis, i.e. looks at India's imports from the selected countries instead of their exports to India, due to the much better coverage of import data given the limited availability of data on these countries' exports to India, especially at the HS-6 digit level.

The analysis of services trade patterns and potential are carried out at the bilateral and sectoral levels using the recently released OECD-WTO BaTIS database for the period average 2000-2002 and 2010-2012. While 2012 is the last year for which BaTIS reports these data, the analysis in this report still constitutes significant value-added relative to existing literature, given the absence of more recent bilateral and sectoral services trade data for the countries covered in this study. Direct exports are used in the case of services trade analysis as the final balanced value of both exports and imports is reported in this database.

The coverage of trade data for the sample countries in the study is reported in Table 3.1.

Table 3.1: Coverage of goods and services trade data for 41 sample countries

	Country c	overage		Time period
Data source	African countries	Asian LDCs	Sector coverage	covered in database
UN Comtrade for data on aggregate and bilateral trade in goods	All 33	All 8	All HS-6	1988-2017
OECD-WTO BaTIS for data on aggregate and bilateral trade in services	Angola, Benin, Burkina Faso, Burundi, Central African Republic, Chad, Democratic Republic of the Congo, Djibouti, Ethiopia, Gambia, Ghana, Guinea, Guinea-Bissau, Kenya, Lesotho, Liberia, Madagascar, Malawi, Mali, Mauritania, Mozambique, Niger, Nigeria, Rwanda, Sengal, Sierra Leone, Sudan, Togo, Uganda, Tanzania, Zambia	Bangladesh, Bhutan, Cambodia, Laos, Myanmar, Nepal, Yemen	Transportation; Travel; Communication Services; Construction; Insurance Services; Financial Services; Computer and Information Services; Royalties and License Fees; Other Business Services; Personal, Cultural and Recreational Services; Government Services. n.i.e.	1995-2012

3.2.2 Investment data

Data on aggregate FDI are sourced from UNCTAD/WIR though these data are not available at the sectoral level or even bilaterally beyond 2012. More recent bilateral and sectoral investment data are thus sourced from 'fDI Markets' which is a private database on cross-border greenfield

investment maintained by the Financial Times. These data cover 182 source and destination countries and 39 goods and services sectors over 2003-2018, enabling a comprehensive analysis of India's investment patterns and potential in African and Asian economies, most of which are covered in this database. The use of fDI Markets for analysis thus constitutes a significant value-addition to existing literature on this subject in the context of India.⁵ The coverage of FDI data for the 41 sample countries is reported in Table 3.2.

Table 3.2: Coverage of investment data for sample countries

	Country	coverage		Time a mania d	
Data source	African countries	Asian LDCs	Sector coverage	Time period coverage	
UNCTAD/World Investment Report for data on aggregate FDI	All 33	All except Timor-Leste	None	1970-2017	
fDI Markets for data on bilateral and sectoral greenfield investment	Angola, Burkina Faso, Burundi, Chad, Democratic Republic of Congo, Ethiopia, Ghana, Kenya, Liberia, Madagascar, Malawi, Mozambique, Niger, Nigeria, Rwanda, Senegal, Sierra Leone, Sudan, Tanzania, Togo, Uganda, Zambia	Bangladesh, Bhutan, Cambodia, Laos, Myanmar, Nepal, Yemen	Automotive OEM; Beverages; Building Materials; Business Services; Chemicals; Coal, Oil and Natural Gas; Communications; Consumer Products; Electronic Components; Financial Services; Food & Tobacco; Healthcare; Hotels & Tourism; Industrial Equipment; Leisure & Entertainment; Metals; Minerals; Non- automotive Transport OEM; Paper, Printing & Packaging; Pharmaceuticals; Plastics; Real estate; Renewable Energy; Rubber; Software & IT services; Textiles; Warehousing; Wood Products	2003-2018	

⁵ However, one limitation of fDI Markets is that it only covers greenfield investment. According to World Investment Report (2018), announced greenfield investment (US\$720 billion) accounted for 50% of total global FDI (US\$1.43 trillion) in 2017, with the value of net cross-border mergers & acquisitions (M&A; US\$694 billion) making up for another 48.5%. M&A activity is thus not included in the analysis that follows, which is likely to influence some of the patterns that are observed in this report. To that extent, the analysis that follows may not be representative of the complete FDI potential that India may have in these countries, but that is a data limitation constraint that we cannot circumvent.

CHAPTER 4

Trade and investment barriers between India and select African and Asian countries

A review of the existing literature indicates challenges in both African and Asian countries mainly LDCs in accessing foreign markets for their products and in attracting foreign investment: huge infrastructure deficit, lack of finance, excessive regulation, etc. But the extent and impact varies between regions, also because there are a greater number of LDCs or low-income countries in Africa (33 LDCs out of total 54 African countries) which are still at much lower levels of development. Meanwhile, India also faces sector-specific constraints while trading with and investing in the selected African and Asian countries. This chapter not only identifies bilateral non-tariff barriers (NTBs), but also discusses NTBs imposed by selected African and Asian countries (including those by India too) in general. These barriers include those identified in the Europe Commission's (EC) Market Access Database (European Commission, n.d.) and the United States Trade Representative's National Trade Estimate Report 2019 on Foreign Trade Barriers (USTR, 2019), which are mainly with respect to the EU and the US, but are generally applicable to a number of other countries including India who are trading with and investing in the selected countries.

4.1 Barriers to India-Africa bilateral trade and investment

It is expected that India-Africa trade will grow up to USD 117 billion by 2021 (Afreximbank & Exim India, 2018). Despite such immense trade as well as investment potential as gauged between India and Africa (mainly LDCs), there continue to be both tariff and NTBs preventing the realization of this trade and investment potential. Based on a review of the existing literature, the major barriers to India-Africa trade and investment include:

- a) Exim Bank (2018b), UNECA and CII (2018), Afreximbank and Exim India (2018), along with WTO and CII (2013), etc. have clearly stated that Africa is still plagued with poor infrastructure. This factor ultimately raises transaction costs for both traders and investors. Exporters need proper roads, railways and other transport to access domestic markets for their goods, along with proper storage and warehousing facilities. They require support of regular power and water supply whose supply is under crisis in many parts of Africa. For instance, huge infrastructural bottlenecks, persistent energy crisis and limited cross-border linkages in West Africa have been specifically highlighted (Exim Bank, 2018b). Linked to this is the huge constraint of capacity on the supply side. The market is not developed enough in Africa for services to support and promote trade-investment activities, such as, development finance, marketing and so on. Further, with India, Africa's bilateral trade becomes even more challenging as India too has weak infrastructural set up with high production, transaction and logistics costs. As a result, most Indian exporters prefer to sell on free-onboard basis to avoid rising insurance and shipping charges and to shift risks of loss to potential buyers in Africa (Afreximbank & Exim India, 2018; Nayyar & Agarwal, 2014; WTO & CII, 2013). So, while both sides face similar issues, the gap is more in the case of African countries. In fact, as per Afreximbank and Exim India (2018), Indian-African trade is likely to double by the year 2021 if infrastructure bottlenecks and tariff and NTBs are properly addressed and potentials are fully realized.
- b) Another major characteristic feature of Africa which affects its expansion and connection with many countries including India is the huge trade finance gap. There is almost a complete absence of proper financing mechanism to support exporting in a

- sustainable way. The functioning of the capital markets in Africa is inefficient (Afreximbank & Exim India, 2018; Broadman, 2008).
- c) FDI is a highly important external source of finance due to the still low generation of fiscal revenues in many low-income African countries. But FDI flow to Africa declined in 2017⁶ due to various macroeconomic reasons such as weak domestic demand, bad governance, political instabilities, along with lower public spending (particularly in West Africa). This has further led to limited availability of finance (Exim Bank, 2018b).
- d) Lack or asymmetry of information between Africa and India mainly for Indian exporters is reported as one of the major barriers. For instance, UNECA and CII (2018) point out the lack of online data to provide necessary information to Indian firms regarding business in Africa. It is found that many African countries also face difficulty in exporting as they lack awareness about existence of duty-free tariff preference scheme offered by India to LDCs (Afreximbank & Exim India, 2018). This is highly crucial for SMEs operating on both the sides, but more so for African LDCs. Due to limited information, many Indian exporters find it difficult to access markets for their products in Africa. In fact, many a times, they find it tough to search for prospective buyers while connecting to firms in Africa. This is even more complex in the case of services exports as the sector is still not well developed in Africa. African firms too face problems in tracking potential buyers in India for their products (WTO & CII, 2013).
- e) UNECA and CII (2018) have also emphasized the under-utilization of India's duty-free tariff preference (DFTP) scheme for African LDCs. One reason for this is the inability of African firms to meet Indian product standards. For instance, Banga and Saha (2016) stress that "the 'LDC package' offered by India (Duty-free Tariff Preference scheme) provides an opportunity to strengthen India's GVCs and increase its number of trading partners. However, existing supply constraints, both in India and in partner LDCs, have led to underuse of the scheme, restricting India's engagement with potential trading partners" (including Asian LDCs).
- f) While there are growing vertical trade complementarities between India and Africa in recent times, such as in the cotton-textile garment value chains, the complementarities that exist are not yet fully explored and remain weak when the entire manufacturing sector is considered. Africa's product diversification is still at a very low level compared to Asian countries including India. It mostly supplies primary goods and natural resources to India, while the latter exports manufactured goods, refined petroleum and pharma products to Africa (UNECA & CII, 2018).
- g) PHD Research Bureau (2017) lists other key barriers which are faced by Indian exporters while trading with Africa. They include lower transparency in African regulations such as custom procedures and enquiry points; fluctuating export prices; demanding competition policies and licensing norms required for foreign firms in Africa; preference towards other countries (due to lower prices offered) or to local suppliers. Rules of origin criterion and variable import levies also affect India's exports to Africa. Importantly, custom clearance procedures including inspection and testing, custom valuation and classification procedures too are reported as major issues while exporting to Africa. While India has streamlined its custom procedures and trade facilitation with a risk management system for easy clearance of consignments, the

⁶ "FDI flows to Africa decreased by 21.5 percent to US\$ 41.8 billion in 2017 from US\$ 53.2 billion in 2016." (Exim Bank 2018b).

- complexity of India's import regime also tends to affect African exports and investments into India (WTO & CII, 2013).
- h) Most African nations continue to be predominantly agrarian economies. This further restricts linkages into GVCs where trade has been mostly in manufacturing and now progressively in services. In fact, the world is moving to a digitalized and innovation-driven set up which require more capacities, high productivity, better infrastructure and communications, etc. which Africa currently lacks. Thus, Africa is still at a lower rung of development with a lack of skilled labour and lower levels of human capital development (Broadman, 2008; WTO & CII, 2013; Lucey, Schoeman, & Makokera, 2015).
- i) A less conducive business environment with problems in getting visas or permit, along with complex processes in obtaining them, limit setting up of firms in Africa. There is almost an absence of bilateral investment agreements which is one main reason why not all African countries get quality FDI from India. India too has some stringent investment criteria, which prevent African firms to set up business in India (Nayyar and Aggarwal, 2014; UNECA & CII, 2018; WTO & CII, 2013).

Thus, concerns of India and Africa including LDCs, while exporting and investing in each other's territories are majorly pertaining to NTBs. Both face similar sets of issues relating to complex custom regulations and huge infrastructure deficit making transaction costs larger. Poor regional market integration in Africa is reported to affect easy movement of goods across borders. Hurdles in tracing reliable local business partners or buyers continue to persist, with connectivity issues on the both sides inhibit the trade potential between India and Africa.

4.2 General barriers in Africa: products and regulations

There are several sector- and country-specific barriers in Africa that prevent Indian firms from exporting to and investing in that continent, and African firms in turn from integrating into value chains, including with India. These barriers relate to the area of complex custom and registration procedures, restrictions or ban on imports and exports, technical barriers to trade (TBT) and sanitary and phytosanitary (SPS) barriers, discriminatory treatment, taxation, etc. (USTR, 2019; European Commission, n.d.). For instance, Nigeria prohibits imports of a number of products since 2002 including poultry, beer, medicaments, fruit juices in retail pack, soaps, mosquito repellant coils, paper board, bagged cement, carpets, footwear, used motor vehicles, ball point pens, etc. (absolutely prohibited items include arms and second hand clothing). On the other hand, although imports are not banned for more than 40 items⁸, since 2015, the Central Bank of Nigeria (CBN) restricts importers of goods and services to access foreign exchange at Nigerian Foreign Exchange markets for promoting local production. The items range from rice, cement, processed meat and vegetable products, steel pipes, iron rods, wood board, to kitchenware, textiles and clothing, rubber and plastics, cosmetics, furniture and now fertilizers too from December 2018. Ethiopia also puts a number of restrictions while importing used clothing, arms and ammunitions (except those done by its Ministry of Defense), goods of a commercial nature, and those which cannot be undertaken via formal banking payment system. Ghana continues to restrict the imports of poultry products since 2014 by limited issue of import permits to domestic traders for promoting this sector. Also, imports of tilapia are also banned for protecting fishermen. For importing of all food, drugs, cosmetic and

 $\frac{https://www.cbn.gov.ng/Out/2015/TED/TED.FEM.FPC.GEN.01.011.pdf}{https://www.cbn.gov.ng/Out/2018/TED/TED.FEM.FPC.GEN.01.006.pdf}$

⁷https://www.customs.gov.ng/ProhibitionList/import.php (Import Prohibition List by Nigeria)

⁸ Items restricted by CBN for Imports available at

agricultural products, the country needs certificates. In fact, as an SPS measure, Ghana bans the import of meat if it contains more than certain specified fat content, such as, more than 25 percent for beef and pork, 30 percent for mutton, etc. as well as restricts imports of milk powder items if contains milk fat of less than certain limit. In Angola, importers need to get valid general import license from the government as well as specific license by subject-concerned Ministries, such as, from the Ministry of Health in case of pharmaceutical substances; from the Agriculture Ministry for agricultural goods, plants, fruits, seeds, etc.; from the Ministry of Post and Telecommunications in the case of postal stamps, radios, transmitters and such devices; and from the Ministry of Customs, etc. All such ministerial authorization in Angola delays the process with additional costs to bear. Further, the country restricts imports of genetically engineered (GE) components, except as food aid, and usage of agricultural biotechnology in production. An SPS certificate is also required from the government while importing animal and plant products. Even in the case of Ethiopia, there is a burdensome process of approval for commercial imports of GE grains and oilseeds. Those processed food items including cereals, soybean oils, etc. prepared by using GE ingredients must meet labeling requirements when imported. Furthermore, Kenya has also banned import of GE food and feed, even in case of food aid shipments. In addition, certain products which pose a threat to the health, safety or environment, face various quantitative restrictions. Kenya also bans the import of second-hand clothing and footwear by imposing heavy import duty rates. On the export front too, Kenya limits the exports of scrap metal as well as copper waste, bans exports of raw agricultural produce including macadamia, cashew nuts, pyrethrum, etc.

Exporters to Nigeria have also mentioned burdensome customs procedures in terms of lengthy clearance, physical inspections and uneven valuation methods which can create delays in movement of goods and losses for importers in case of perishable items. Weak ports with heavy congestion further contribute to rising costs. There is another key barrier reported relating to mandatory pre-shipment registration with Standard Organisation of Nigeria (SON) of several categories of products, whether domestically produced or imported. This also includes e-Registration of machinery and electronic equipment since 2013. Not only fees rise for mixed shipments, but certificates' validity is also for a short duration ranging between 6 months to 1 year. Hurdles are reported while using Nigerian online platforms, where most information/documents are found to be missing. Particularly, in case of agriculture, food items and drugs/pharmaceutical sectors, exporters need to register all their products (and pay separate fees for different sizes of same type of goods) at National Agency for Food and Drug Administration (NAFDAC). The problem indicated is that representatives of NAFDAC need to physically inspect all the production facilities of companies for all new items meant for selling to Nigeria. Secondly, foreign exporters need to find Nigerian partner/distributor to register the product who ultimately gets imports rights. Moreover, lack of capacity is reported for reviewing import certificates, doing inspections and testing meant for clearance. Even in case of Angola, technical regulations and procedures for certification and testing for imports are said to be documented in a poor manner. Its custom cervices have although improved recently due to easing regulation on pre-shipment inspection of imports, but there are still barriers to trade due to low transparency and biasedness. Illustratively, shipment equal to or more than US\$1000 requires clearing agents, who have actually increased in number but continue to charge higher fees due to competition.

Meanwhile, Ethiopia imposes the condition of obtaining letter of credit (LoC) for import transactions and many a times importers are not always granted import permits which must be applied for before placing the contract. Ethiopia also has several custom barriers in terms of regular logistic backlogs owing to much paper-based process and structural inefficiencies, high

shipping and freight costs, etc. Ghana's customs practices are also still reported to be lengthy with unpredictable applications and use of system of pre-arrival risk assessment and reporting, and after-arrival inspections, etc. (despite introduction of electronic cargo clearance at ports since 2017). On top of it, Ghana has its own national standards (more than 2400) on food, agriculture, electrical, pharma goods, etc. under Ghana Standards Authority (GSA). GSA officials also inspect imports of goods classified as high-risk goods (HRGs); however such classifications are found to be confusing for exporters (viz. alcoholic and non-alcoholic goods could be in more than one category of beverages, drugs, industrial goods, etc.). Further, importers of HRGs need to register themselves and get approval of GSA first. For this, they need to submit a sample of goods, along with conformance certificate obtained from a laboratory in the country of export. Labeling requirements are also checked along with physical verification, followed by huge fess for registering number of HRGs. Kenya too has number of requirements which are complicated, expensive and non-transparent such as in case of imports of meat, dairy and poultry goods, along with standardized sanitary certification and no objection letter from its Department of Veterinary Services. In particular, all the importers in the country need to pay import declaration fee (which is 2.25% of import value) with documents, else their goods are liable to be rejected or prone to extra inspection. Subsequently, importers need to get Import Standardization Mark for each imported item based on national standards. Other NTBs are in case of Uganda's toys (wheeled toys, dolls, etc.) sector, where strict safety and quality standards are enforced upon products to be sold in the domestic market, which can create further barriers.

In mining/oil and gas sectors, Nigeria since 2010 imposes discriminatory treatment where Nigerian investors are given preference over the foreign bidders on the basis of meeting local content requirements. Thus, there are barriers to investment in African countries. For instance, Nigeria and Ethiopia practice significant foreign exchange controls, mainly in the case of the latter where foreign exchange permits are required for imports and exports, plus there are foreign exchange shortages in the country which causes delays. Foreign companies also need to register themselves with Investment Promotion Commissions/Centres such as in Nigeria, Ghana, etc. Nigeria's investments climate is significantly affected by huge intervention by government in terms of corruption and restrictions on foreign exchanges as well as continuous changes in regulations, weak infrastructure mainly power, and lack of finance access for small and medium enterprises along with number of social sector drawbacks. The process to acquire land in Kenya is cumbersome and non-transparent. In fact, in 2010, foreigners were prohibited to get freehold land titles in Kenya (only leasehold ones were allowed). Investors also face issues of fake or multiple titles due to complex procedures for land transactions, inadequate urban planning as well as under-investment in land demarcation.

Due to local content requirements (LCR) and sector-specific laws, investment in Angola is barred in petroleum, diamonds and financial services, subject to conditions sector-wise (such as forming joint ventures (JVs) with Angolan companies and acquiring low capital goods and services from domestic market in case of petroleum sector). In the case of Ethiopia, ban is on investments in banking, insurance and micro-finance industries, with prevention in the field of retail, wholesale, and service enterprises involved in printing, restaurants, transport services, shipping, broadcasting, beauty shops, etc. However, for telecommunication and defense industries, foreign investment is allowed only in partnership with Ethiopia's government. Ghana also prescribes foreign investors to form JVs with Ghanaian partner. Investment is prohibited in Ghana's sectors of petty trading, operation of taxi and automobile rental services with fleets of lesser than 25 vehicles, lotteries, operation of beauty salons and barber shops, printing of recharge scratch cards for subscribers to telecommunications services, production

of exercise books and stationery, retail of finished pharmaceutical products, and production and supply of drinking water in sealed pouches. Ban is also on obtaining small-scale mining license for mining operations for area less than 25 acres. In the case of the oil and gas sector, there is local sourcing requirement and need for creating consortium with state enterprise to get licenses. In fact, in many African countries, state-owned enterprises (SOEs) are given preference over private firms in a number of important sectors such as telecommunications, power, banking, insurance, transport, etc. which affect the playing field for foreign investors.

Although there have been significant improvements in the IPR rules in these countries, but there still remains lack of transparency, unpredictability and stringency or loopholes in terms of enforcement and protection. Capacity is limited with unnecessary delays in infringement cases, and also coordination becomes difficult among various governmental units such as in case of Ethiopia and Ghana. There is still widespread trading in counterfeit and pirated goods.

4.3 Barriers to India-Asia bilateral trade and investment

Asian LDCs (in South and South-East Asia) tend to have limited availability of capital, poor infrastructure (including bad roads, absence of adequate railway or port networks, power shortage), more of unorganized and small enterprises, etc. Many of them are still agrarian economies or continue to be rich in raw materials and resources. For instance, Myanmar has rich mineral and energy resources, Cambodia has abundant reserves of iron ore, manganese etc. (Exim Bank, 2014), which inhibits the integration of their economies in regional and global value chains, most of which are in the manufacturing sector.

Even so, many Asian LDCs including Bhutan, Nepal, and Bangladesh either enjoy preferential treatment into India or have signed free trade agreements. But they still have limited access to the penetration of technologies; their entire manufacturing line is not well-developed except in a few sectors such as apparels in the case of Bangladesh; they are afflicted by lack of trained and skilled manpower, etc. (KPMG & DFID, 2016 for Bangladesh). This prevents their linkages with other Asian countries including India who can help them in meeting infrastructure requirements, sharing of best practices for instance in setting up of SME clusters, etc. (Exim Bank, 2014). Some of the barriers between India and Asian LDCs are summarized as follows:

a) Although Asian LDCs are geographically closer to India, especially Bangladesh, Nepal and Myanmar, they continue to face issues of connectivity and/or disturbed border trade or weak border-related infrastructure. Illustratively, most of the bilateral trade between India and Bangladesh is carried through land, viz. using trucks mainly through land custom stations (LCS) and integrated check posts along the border. But there is significant unofficial or informal trade which tends to underestimate the actual trade

⁹ In terms of railway and waterways network between India and Bangladesh, there are issues of poor connectivity such as absence of proper water transport infrastructure. As per Exim Bank (2015), "India's National Waterways 2 (NW 2), cutting across Bangladesh, links the NER with West Bengal. The absence of all-weather navigation facilities, coupled with inadequate water depth, obstructs high-speed vessels from passing through national waterways, so these waterways can make little contribution towards enhancing merchandise trade between the two countries." and "Bangladeshi locomotives are unable to haul air-braked freight wagons used by India, and are restricted to BCX wagons (covered, vacuum braked eight-wheeler wagons); holding capacity of loops, yards and terminals are inadequate on the Bangladesh side, even at the current low volume of traffic; container traffic and open wagons are currently restricted by Bangladesh Railway; speed of freight trains is impeded by mechanical signaling."

potential. LCS even lack proper services in terms of power back up, internet connectivity, working hours of operational personnel, etc. which must be upgraded (Exim Bank, 2015). Four Border Haats are however currently operational to curb informal trade between India-Bangladesh. There is more need of such efforts in the case of India and Myanmar border trade. The road near their border is in bad shape which makes connectivity between the nations problematic (Exim Bank 2018a). There are currently challenges at LCS, such as, at LCS Moreh in terms of absence of operational warehousing and storage facilities, lack of internet connection, electricity issues, manual handing of goods, and lack of laboratory testing facility for imported food items, etc. All this acts as a hurdle in smooth border trade between Myanmar and India.

- b) India's trade with ASEAN has been rising particularly after signing of free trade agreement in 2009 which has led to development of some production networks in sectors of electronics and automobiles, etc. But still India's trade and investment linkages are more with Indonesia, Thailand and Singapore, rather than with LDCs such as Lao PDR, Cambodia, etc. Regarding investments, it is found that "despite being an emerging economy in Asia, India accounted for a measly three percent of ASEAN's total FDI from 2003 to 2016, and two percent of CLMV's" The reasons could include more dependence by Asian LDCs on China and developing/developed ASEAN. The connections therefore get limited with India who also has been focusing more on EU and US markets. Challenges remain in trading with ASEAN LDCs/CMLVs in terms of NTBs (such as stringent safety and health regulations or technical barriers), poor physical connectivity, lack of reliable power supply, weak quality and coverage of infrastructure; inadequate level of liberalization of services trade (Bhogal, 2018; Mazumdar, 2017; Assocham, 2016; Exim Bank, 2014; Dixit, 2015; etc.).
- c) Some Asian LDCs prefer sourcing from other Asian countries than India which is another main cause of lower imports and FDI coming from India. For instance, while India imports garments from Bangladesh, the latter imports fabrics mainly from China and converts them to garments for selling to India at duty-free rates (Gupta, 2018). This is because Bangladesh's readymade garments sector is more developed than that of India owing to low manufacturing costs, higher investment cap in MSMEs as well as preferential access in EU and US markets (Exim Bank, 2015).
- d) Restrictions under internal laws and policies of LDCs to attract FDI act as another obstacle. Despite liberal FDI reforms in some Asian LDCs in recent times, they are still underperforming. For instance, "Bangladesh has a welcoming stance towards FDI. Despite that, its FDI which has been increasing over the years, has still not grown in accordance with the flow in other low-income countries." (DFID & KPMG, 2016). The limited scope and coverage mentioned under Bangladeshi Foreign Private Investment Promotion and Protection Act (FPIPPA) is said to be the reason along with negative attitude towards welcoming FDI in high-performing garments, pharma and telecom sectors (UNCTAD, 2013).

4.4 General Barriers in India: products and regulations

India imposes a number of tariffs and NTBs, including TBT and SPS measures, which restrict exporters, including from Africa and Asia, in supplying easily to India. India's tariff rates on several items are very high such as 50% or more on motorcycles, flowers, rubber, automobiles, etc. and even more than 100% in case of coffee and alcoholic beverages. For agricultural goods

¹⁰ CLMV is Cambodia, Laos, Myanmar and Vietnam.

mainly processed ones, tariff changes at any time impact the exporters, such as, import duties had increased (during November 2017-March 2018 period) from almost nil to about 30-50% or more on chickpeas, peas, lentils, etc. In 2018, tariffs were raised on some fruit juices, edible vegetable oils, etc. India increased custom duties on textile and clothing products after introduction of Goods and Services Tax (GST) in 2017, particularly on those imported from the EU. India has been continuously increasing custom duties on ICT and telecom products, affecting imports of mobile phones and their components, static converters, base stations, and printers. In September 2018, India also increased duties on imports of non-essential items including polished diamonds, gemstones and jewellery, as well as electronics goods such as AC, refrigerators, washing machines, speakers; radial car tyres, and many household and office items made of plastics.

India also bans imports of furskins, reptile leather, tanned mink skins, etc. since 2017 owing to protection of wildlife (subject to prior consultations). In the case of agriculture and fisheries sectors, India imposes SPS measures which restricts imports of plant and its products relating to fumigation treatments (for alternatives to methyl bromide). On imports of dairy products, India puts condition that animals providing such products must not consume feeds that contain internal organs, blood meal, tissues, etc.

On a number of products, namely, electronics and telecom equipment, solar power plants, iron and steel, medical devices, there are measures introduced based on certain local content and value addition requirements, i.e., give preference to domestically manufactured goods in government procurements. Such measures have heightened particularly after launching of 'Make in India' initiative by the Government of India in 2015.

Other significant NTBs include: (i) Restrictions on import of ethanol (ban for fuel use since 2018) via its tendering system which sets blending of ethanol from indigenous feedstock with transport fuel. (ii) Controlling of prices of medical devices such as cardiac stents, knee transplants, etc. without much consideration to technological innovations which restricts access of many countries to India. Exports of refurbished medical devices are also prohibited.

Although imports of second-hand capital goods are allowed in India, the import licenses are required in case of remanufactured goods which is again a lengthy procedure. As per USTR (2019), "Refurbished computer spare parts can only be imported if an Indian chartered engineer certifies that the equipment retains at least 80 percent of designed operating life, while refurbished computer parts from domestic sources are not subject to this requirement. India subjects boric acid imports to stringent restrictions."

To summarize, India bans import of items of animal origin; restricts items related to livestock, chemicals, etc. which need license; but allows those which can be imported only by government and with cabinet approval such as medicines. Moreover, there are number of export subsidy programs in India for encouraging exports of its textiles and apparel, steel, paper, rubber, leather products, etc. which are being questioned by the US in particular. On the other side too, India has been levying export duties, since 2011, on the raw and semi-finished steel products and iron ore, raw hides and skins and certain semi-finished leather, followed by chromium and aluminum ores.

There are standards and other technical requirements as well as certification and registration procedures in the field of telecom equipment, cosmetics, wines and spirits, automotive, leather, steel, etc. (mainly since 2016). Concerns relate to deviation from international standards and

following in-country standards or schemes which may lack reliability as desired by certain countries. For instance, along with higher import duties, India has introduced number of standards for automotive manufacturing items (i.e. motor cars) and also placed mandatory certification for pneumatic tyres - much of these do not comply with the UNECE Agreement. However, over the years, requirements have risen for importers in terms of additional burden for fees, lengthy procedures, factory inspections, short validity of licences, and introduction of license marking scheme which are not notified to the WTO as a TBT. As per Department of Telecommunications, Government of India, all telecom equipment including mobile devices are required to undergo prior mandatory testing and certification¹¹ by Bureau of Indian Standards (BIS) even if the product is already internationally certified. There are concerns relating to rise in costs and delays for companies due to complex and long registration process, along with doubts over availability of suitable Indian laboratories for such purpose. Such incountry testing is also risky in the case of Highly Specialized Equipment (HSE) which include servers, printers, etc. requiring professionals to install and operate. Further, there are conditions attached for re-exporting within the stipulated time in terms of duty-free imports for repairs or testing (for electronic goods including mobile phones, colour TV, Xerox/ printing machines, LCD/LED panels, and certain medical devices).

In case of cosmetics sector (HS 33-essential oils, perfumery and cosmetics), there is complex registration process for importers with rising fees which has become US\$2000 for each brand plus extra fee as per variants. Domestic companies are eligible for application for perpetual licensing, unlike those products which are imported. Importers also face issues while becoming associated with national standards (which are also not found much in line with ISO). Moreover, India has also prohibited imports of animal-tested cosmetics. In 2018, the food safety standards were mentioned by Food Safety and Standards Authority of India (FSSAI) for distilled alcoholic beverages, wine and beer. However, this also concerns many exporters in terms of technical specifications such as content of alcohol and sugar to be incorporated which are mostly not as per international standards making it difficult for foreign companies to enter into India, followed by strict labeling requirements for prepackaged foods to list information on place of manufacturing/processing/storing or even residue of additives in the product. Without notifying to WTO as a possible TBT measure, Government also introduced in-country testing requirement for imported toys since 2017. While importing finished leather products, trade authorities in India also require veterinary certificates.

Above all, huge worries have been raised over India's complex custom procedures which still needs extensive documentation for clearance, plagued with uncertain and complex tariff structure. It is mentioned that "India's valuation procedures allow Indian customs officials to reject the declared transaction value of an import when a sale is deemed to involve a lower price than the ordinary competitive price, effectively raising the cost of exporting to India beyond applied tariff rates." (USTR, 2019).

India's IPR regime has issues relating to enforcements and stability due to continuous inadequate actions by IPR authorities in fighting against infringements. Inventors have mentioned problems in securing patents protection in India owing to certain flaws in IPR laws in terms of too restrictive criteria for patentability or too broad criteria to grant compulsory licences; lack of proper system to protect undisclosed test or data, in the field of pharmaceutical and agricultural chemical products, which are meant to get market approval against unfair

¹¹ Rules for Testing and Certification of Telecom Equipment are available at http://www.tec.gov.in/certification-approval-procedure/.

commercial use. Furthermore, India already has strong presence in many services sector, thereby it prohibits foreign entry in legal services, and restriction in case of financial and insurance services and in retail. India also puts several data localization requirements to restrict digital trade.

4.5 General Barriers in selected Asian countries: products and regulations

There are several sector- and country-specific barriers in Asian LDCs that prevent Indian firms from exporting to and investing in those countries, and their firms in turn from integrating into value chains, including with India. Myanmar mainly bans imports of certain goods, such as, narcotic drugs, playing cards, antiques, endangered species, counterfeit money and goods, arms and ammunitions, etc. This country still imposes import licenses on more than 4000 items, for entering into the country, including agricultural products mainly meat. Laos too put licenses on products of motor vehicles, refined petroleum fuels, oil, natural gas, timber and so on. The custom clearing and valuation processes of many of selected Asian LDCs are criticized on the ground of following non-transparent, irregular and arbitrary practices along with corruption. In the case of Myanmar, there are additional requirements for testing and inspections of shipments (not much in line with internationals standards for inspection) when agricultural products are imported, besides laboratory testing done during the process of approval of licenses.

Complex paperwork is involved in Cambodia, as reported, which causes delays in processing faced by both domestic and foreign businesses (duties also tend to change on the same kind of products shipped at different times in a year). Bangladesh largely puts registration requirement with the Chief Controller of Imports and Exports in its Ministry of Commerce on commercial importers and private industrial consumers. Further, all exporters and importers in Bangladesh need to be a member of recognized chamber of commerce or trade organization. Most importantly, Letter of Credit (LOC) is required for almost all the imports (except machinery and raw materials meant for industrial use), along with its authorization form, bill of lading, packing list, certification of origin, and import permits in case of health or security matters. With respect to SPS measures, Bangladesh places fumigation requirement on imported US Cotton at the entry port. Myanmar has also introduced more product labeling requirements for consumer goods since 2018, where some information needs to be translated into Burmese language too (USTR, 2019). On the other hand, Bangladesh provides export subsidies to certain exporters; however also ban exports in sectors like leather (European Commission, n.d.), such as on exports of raw hides and skins since 1990s which is still active (only finished leather is allowed to be exported).

In the services sector too, a number of hurdles are faced, such as foreign companies are generally prohibited in Bangladesh in those services which are reserved for government sector. Telecommunication, banking and insurance services are generally more regulated in Bangladesh thereby placing number of restrictions on new service providers in terms of equity share, having local partners, or getting license from government. In Laos' financial, medical, postal, telecommunications, leasing, media, transportation services, etc. there are issues in providing services by foreign companies. These two countries also impose barriers on digital trade and electronic commerce in terms of challenges for internet platforms which are dependent on user-generated content.

A challenging investment climate prevails in almost all Asian LDCs due to a number of reasons. To illustrate, foreign investors face issues in Bangladesh relating to administrative formalities and bureaucratic inefficiencies with frequent transfers of government officials,

corruption, as well as non-transparent as well as overlapping regulatory frameworks. For Laos too, some of the reasons are the same but with focus on weak judicial system, lack of finance services to support, complications while enforcing contracts, and varying requirements for ownership and partnerships which are generally inconsistent with administrative procedures. Investors in Myanmar and Cambodia are reportedly facing issues of lack of required finance and infrastructure, high electricity costs, rising logistics and transport costs, lack of skilled man force, and protection of minority investors (in case of Myanmar). In particular, Myanmar had issued negative list for investments in 2017 which prohibited investors to enter in certain sectors or allowed only domestic investments. However, "in May 2018, Burma's Ministry of Commerce opened the retail and wholesale sectors for permitted products to foreign-owned companies and joint ventures that meet minimum capital requirements. The full list of permitted products was published in July 2018 and includes consumer goods, agricultural products, household and kitchen goods, pharmaceuticals, electronics, machinery, and motor vehicle parts, among other products." (USTR, 2019).

Meanwhile, Cambodia continues to place a number of restrictions on foreign-owned land and buildings. Although many new laws and regulations are coming up in these countries for allowing foreign companies, but challenges continue in terms of implementation such as in case of Myanmar. Cambodia although allows 100% foreign investment in number of sectors but still places equity restrictions in case of movie production, rice milling, gemstone processing, publishing and printing, radio and television, wood and stone carving production, and silk weaving. Investors also face long and unpredictable government's tax procedure or more favouring of local investors or industry over foreign ones. For better protection of IPRs, there are number of steps being taken in Laos, Bangladesh and Cambodia, including those with the assistance of the US government. But there are concerns relating to low public awareness for use of counterfeit and pirated goods. Bangladesh is reported to still have limited resources for IPR enforcement (USTR, 2019).

CHAPTER 5

Mapping merchandise exports of Africa and Asia in India

This chapter details the top commodities at the HS 6-digit level imported by India from select African (33) and Asian (8) countries. The data have been sourced from UN Comtrade using mirrored data. In order to study the evolution of India's merchandise imports from these countries over time, two time periods have been considered and have been averaged over three-year periods from 2010 to 2012 and 2015 to 2017 to minimize the effect of fluctuations in data reporting practices and to account for macroeconomic and other external shocks that may affect trade.

UN Comtrade covers all the 41 countries examined in this study: Angola, Benin, Burkina Faso, Burundi, Central African Republic, Chad, Democratic Republic of Congo, Djibouti, Eritrea, Ethiopia, Gambia, Ghana, Guinea, Guinea-Bissau, Kenya, Lesotho, Liberia, Madagascar, Malawi, Mali, Mauritania, Mozambique, Niger, Nigeria, Rwanda, Senegal, Sierra Leone, Somalia, Sudan, Togo, Uganda, Tanzania, Zambia, Bangladesh, Bhutan, Cambodia, Laos, Myanmar, Nepal, Timor-Leste and Yemen.

In terms of structure, the chapter reports data on total merchandise imports by India from these 41 countries averaged over 3 years from 2010 to 2012 and 2015 to 2017 (Table 5.1) as well as data on the top three African and top five Asian commodities imported by India from these countries, again averaged over the same time periods (Table 5.2). The chapter also reports the top five commodities imported by India from the Rest of the World (ROW) but not from the select African or Asian countries under study in 2015-2017 (avg.) (Table 5.3) to illustrate products where there is demand in the Indian market but possible supply side constraints on the side of the exporting countries.

5.1 India's total merchandise imports from selected countries

An analysis of the data reported in Table 5.1 provides the following key results:

1. Top African countries from where India imported during 2010-2012

Nigeria (USD 12.6 billion), Angola (USD 6.3 billion), Tanzania (USD 432 million), Ghana (USD 343 million), Senegal (USD 339 million), Guinea (USD 334 million), Benin (USD 256 million), Guinea- Bissau (USD 228 million), Sudan (USD 179 million), and Zambia (USD 173 million)

2. Top African countries from where India imported during 2015-2017

Nigeria (USD 8.7 billion), Angola (USD 3.2 billion), Ghana (USD 2.5 billion), Zambia (USD 698 million), Mozambique (USD 586 million), Guinea (USD 454 million), Senegal (USD 371 million), Burkina Faso (USD 366 million), Sudan (USD 278 million), and Benin (USD 236 million)

3. Top Asian countries from where India imported during 2010-2012

Yemen (USD 1.3 billion), Myanmar (USD 1.3 billion), Bangladesh (USD 533 million), and Nepal (USD 468 million)

4. Top Asian countries from where India imported during 2015-2017

Myanmar (USD 996 million), Bangladesh (USD 690 million), Nepal (USD 460 million), and Laos (USD 217 million)

Column 4 of Table 5.1 reports the difference in value of merchandise imports of India from the select African and Asian countries between 2010-2012 (avg.) and 2015-2017 (avg.) to see whether India's imports from them have increased or declined (as highlighted in the Table 5.1) over the two time-periods. The results show that India's imports actually declined in the case of seven African countries, namely, Angola, Benin, Central African Republic, Guinea-Bissau, Kenya, Niger, and Nigeria, as well in the case of five Asian countries, namely, Bhutan, Myanmar, Nepal, Timor-Leste and Yemen.

To elaborate, imports decreased majorly for Nigeria (by \$3.9 billion) and Angola (by \$3.1 billion) from 2010-12 to 2015-17. However, both countries continue to dominate the Indian market as the top two African countries from where India has been importing goods. Secondly, Tanzania and Ghana were the third and the fourth largest exporters of merchandise to India during 2010-12, but the former is not listed amongst the top 10 countries during 2015-17 despite a rise in India's imports from it. Notably, India's merchandise imports from Ghana have increased by more than 7 times over the two time-periods (keeping the country still as the third largest exporter). Zambia has climbed the ladder with a rise in its exports to India, reaching the fourth position amongst the top countries during 2015-17. New entries in the top 10 during 2015-17 as compared to 2010-12 have been Mozambique and Burkina Faso.

In the case of Asian LDCs, although Yemen was the top country from where India imported much during 2010-12, that country has now lost the position with a major fall in (India's) import value by \$1.15 billion. In fact, India's imports from all Asian LDCs, except from Bangladesh, Cambodia and Laos, have declined though the difference is not as large as Yemen. For instance, Myanmar's exports have reduced by almost \$297 million despite it being the top Asian country in our sample during 2015-17. On the other hand, Nepal's position improved from fourth to third over time even though it witnessed a slight fall in (India's) imports value by \$7.6 million. Bangladesh too experienced a rise in ranking to the second position during 2015-17 accompanied by a rise in (India's) imports value by \$156 million.

Meanwhile, India imported the least in both the two time-periods from Burundi, Central African Republic and Rwanda (in case of selected Africa) and Cambodia and Timor-Leste (in case of selected Asia).

Table 5.1: Total merchandise imports of India from select African (33) And Asian (8) countries, 2010-2012 (avg.) and 2015-2017 (avg.)

Countries	Value (avg. 2010-2012, US\$ million)	Value (avg. 2015-2017, US\$ million)	Difference = (avg. 2015-2017) – (avg. 2010-2012)
1	2	3	4
Angola	6290.6	3187.0	(3103.6)
Benin	256.4	236.6	(19.8)
Burkina Faso	15.8	366.6	350.8
Burundi	1.1	11.2	10.1

Central African Republic	2.1	1.6	(0.5)
Chad	3.5	213.0	209.5
Congo	23.1	178.2	155.1
Djibouti	4.4	5.4	1.0
Eritrea	12.6	117.4	104.7
Ethiopia	41.6	63.2	21.7
Gambia	27.8	44.1	16.3
Ghana	343.3	2492.8	2149.4
Guinea	334.3	454.3	120.1
Guinea-Bissau	228.6	226.5	(2.1)
Kenya	138.8	107.2	(31.7)
Lesotho	3.0	31.0	28.0
Liberia	27.7	60.7	32.9
Madagascar	68.1	138.4	70.3
Malawi	37.5	46.5	9.0
Mali	16.3	153.1	136.8
Mauritania	4.9	23.5	18.6
Mozambique	159.0	586.7	427.8
Niger	76.3	20.0	(56.3)
Nigeria	12645.6	8671.4	(3974.2)
Rwanda	0.6	2.5	2.0
Senegal	339.8	371.9	32.1
Sierra Leone	7.3	20.8	13.6
Somalia	8.6	13.0	4.4
Sudan	179.3	278.2	98.9
Togo	151.6	223.1	71.5
Uganda	25.4	57.3	31.9
Tanzania	432.5	987.6	555.1
Zambia	173.2	698.1	524.8
Bangladesh	534.0	690.3	156.3
Bhutan	198.5	195.9	(2.6)
Cambodia	11.1	51.2	40.1
Laos	78.8	218.0	139.1
Myanmar	1293.4	996.8	(296.7)
Nepal	468.2	460.7	(7.6)
Timor-Leste	12.8	0.2	(12.5)
Yemen	1339.4	191.6	(1147.8)

Source: UN Comtrade; own calculations,

[*Note:* The values highlighted in the table report those African and Asian countries whose merchandise exports to India in 2010-2012 (avg.) were higher than in 2015-2017 (avg.).]

5.2 Top commodities imported by India from selected countries

An analysis of the data reported in Table 5.2 provides the following key results:

1. Top commodities imported from African countries (2010-2012)

Petroleum oils, cashew nuts, ferrous and aluminium waste, wood, cotton, manganese and copper ores, oil seeds, wood, photographic flashlight, wrist watches, cathodes, tanned/crust skin of sheep/lamb, precious stones, iron ores, kidney beans, coal, wool, cocoa beans, cloven, natural gas, sesamum seeds, and dried leguminous vegetables

2. Top commodities imported from African countries (2015-2017)

Petroleum oils, cashew nuts, gold, diamonds, copper ores, iron ores, coal, unwrought lead, natural gas, sesamum seeds, phosphoric seeds, cloves, tankers and dried leguminous vegetables

3. Top commodities imported from Asian countries (2010-2012)

Jute, fish, nuts, ferro-silicon, copper wire, palm oil, natural rubber, trousers, wood, beans, copper ores, petroleum oil, natural gas, and non-alcoholic beverages

4. Top commodities imported from Asian countries (2015-2017)

Jute, gold, beans, natural gas, nuts, non-alcoholic beverages, ferro-silicon, petroleum oil, and copper ores

Thus, India largely imports a number of agricultural products, mineral ores, petroleum oils, and some precious metals from the selected African and Asian countries though the import basket from both sets of countries seems to be more concentrated during 2015-17. The absence of some prominent items, including labour-intensive ones, in the list of top commodities imported may be attributed to India's ban on imports of items of leather and those of animal origin. India also puts a number of import restrictions and/or high tariff rates on electronic and ICT goods, machinery items, plastic articles, automobiles, alcoholic beverages, coffee, steel, cosmetics, pharmaceutical products (many of these also have been subjected to huge certification and testing requirements by India if they are to be imported). This, together with supply side constraints in the sample countries, could be the reason for the continuous low diversification of these countries' export basket to India.

Detailed tables are included in 'Annexure A' for each of the 41 countries highlighting the top 10 products imported by India (with their shares in India's imports) as well as India's imports of that commodity from ROW to show the importance of each product in the Indian market. Colour coding is used to highlight those products which are covered under India's duty-free preference scheme given for LDCs¹² – orange coded indicates products covered in the list of duty-free imports, while purple ones indicate those covered in the Margin of Preferences list (products in bold are not covered under the scheme). A look at the data reported in the Annex

¹² To highlight the products covered under the DFTP scheme for the time period 2010-2012, those countries have been considered which had joined till 2012. These include Benin, Burkina Faso, Burundi, Central African Republic, Eritrea, Ethiopia, Gambia, Lesotho, Liberia, Madagascar, Malawi, Mali, Mozambique, Rwanda, Senegal, Somalia, Sudan, Uganda, Tanzania, Zambia, Bangladesh, Cambodia, Timor-Leste, Lao PDR and Myanmar. The top ten commodities imported by India during 2010-2012 (avg.) and 2015-2017 (avg.), reported in Annex A, have been highlighted in different colours according to whether they belong to the exclusion list (where no tariff concessions are given), Margin of Preference (MOP) list (some tariff concession is given) or if they are under the duty-free list. This coding is not done for both the years in case of Angola, Bhutan, Congo, Djibouti, Ghana, Kenya, Mauritania, Nepal, Nigeria, and Sierra Leone, and for 2010-12 in the case of Chad, Guinea, Niger, Togo and Yemen, as these countries are still ineligible for India's preference scheme.

tables suggests that some countries may have benefitted more from India's tariff preferences given under the DFTP, for instance Central African Republic, Eritrea, Ethiopia, Madagascar, Mozambique, Tanzania, Laos and Myanmar.

Getting back to Tables 5.1 and 5.2, out of the selected 33 African countries, the top three sources for India's imports during 2010-12 were Nigeria, Angola and Tanzania, and in 2015-17, these included Nigeria, Angola and Ghana. Moreover, the shares of Nigeria's top three commodities in India's total merchandise imports during 2010-2012 and 2015-2017 were 98.7% and 98.4%, respectively, of which petroleum oil alone stood at 94.6% and 88% respectively. Other major commodities imported by India from Nigeria over the two time periods include natural gas, cashew nuts, aluminum waste and alloys, unwrought lead, and wood.

Analogous to Nigeria, the shares of Angola's top three commodities in India's total merchandise imports over the two time periods were 99.9% and 99.4% respectively; with the respective shares of petroleum oil alone being 99.2% and 82%. In contrast, Ghana's top commodity imported by India in 2015-2017 was gold (91% of India's total merchandise imports) as opposed to the top imports of cashew nuts during 2010-12 (with a share of 29% in India's imports which declined to just 3.2% in 2015-17). Meanwhile, Tanzania too exported cashew nuts and gold to India, along with dried leguminous vegetables in both the periods, but the combined share of the first two products increased from 52.4% to 82.2% over the two time periods.

In both these time periods, India imported the least from Central African Republic and Rwanda. While the value of India's merchandise imports from the former showed a decline from \$2.14 million in 2010-12 to \$1.64 million in 2015-17, India's imports from Rwanda displayed an increase of \$2 million. The top commodity imported by India in 2010-12 from Central African Republic in 2010-2012 was topical wood (47.3% of India's merchandise imports) but the share of this product declined to 12% in 2015-2017 and the top imported commodity became diamonds with a share of 58.4% in India's total imports. In the case of Rwanda, the top commodities imported by India in 2010-2012 and 2015-2017, respectively were coffee (29.8% of India's merchandise imports) and unwrought lead (47.8%).

In the case of the Asian LDCs, the top three sources of India's merchandise imports in 2010-2012 were Yemen, Myanmar and Bangladesh, whereas in 2015-17 these were Myanmar, Bangladesh, and Nepal. Commodity-wise imports' data provides some interesting results. Illustratively, in the case of Myanmar, the share of beans (urad, mung, black or green gram - HS 071331) and dried leguminous vegetables in India's total imports increased from 40% in 2010-12 to 65% in 2015-17, respectively. Meanwhile, kidney beans (0.6%), human hair (0.5%), petroleum oils (0.4%) (during 2010-2012), and chickpeas (1.5%), ferro-nickel (1.3%), nuts (1.2%) (2015-2017) were the least preferred commodities imported from Myanmar by India (please refer to Annex A tables for more details).

In the case of Bangladesh, it is evident that the share of jute and other textiles in India's total merchandise imports declined from 13.3% in 2010-2012 to 8.8% in 2015-2017, despite this being the top imported commodity. Sacks and bags (10.6%) was another item of India's import interest in both these periods, but its share too declined recently. In fact, during 2015-17, men's/boy's trousers (7% share) and jute yarn (6.5%) came to occupy the second and third positions among the top 5 commodities imported by India from Bangladesh. Fish was also a major import during 2010-12, but not in 2015-17.

Meanwhile, India imported the least from Cambodia and Timor-Leste in both the time periods. While the value of merchandise imports from the latter declined significantly from \$13 million in 2010-12 to just \$0.22 million in 2015-17, Cambodia showed an increase in value by a substantial margin of \$40 million. The top commodity imported from Timor-Leste in 2010-2012 was commodities not specified (HS 99999) (82%), whereas in 2015-2017, the market has become product-specific with imports of nuts (43.4% in India's total imports). Similarly, in the case of Cambodia, the top commodity imported by India in 2010-2012 was palm oil (34.6%), but in 2015-2017, topical wood and natural rubber shared the top positions with almost equal shares (14%).

Table 5.2: Top three (five) commodities imported by India from select African (Asian) countries, 2010-2012 (avg.) and 2015-2017 (avg.)

Countries	HS6 code	Product descripti on	Avg. 2010- 2012 (\$ million)	Share in total imports (%)	HS6 code	Product description	Avg. 2015-2017 (\$ million)	Share in total imports (%)
	1	•	AFRICA	N COUN	TRIES		•	
Angola	270900	Petroleu m oils	6238.9	99.2	270900	Petroleum oils	2622.0	82.3
	720449	Ferrous waste & scrap	36.4	0.6	710231	Diamonds	293.8	9.2
	760200	Aluminiu m waste & scrap	5.1	0.1	271111	Natural gas, liquefied	251.9	7.9
Benin	80131	Cashew nuts	137.4	53.6	80131	Cashew nuts	145.0	61.3
	440349	Topical wood	17.7	6.9	520100	Cotton	23.0	9.7
	760200	Aluminiu m waste & scrap	16.4	6.4	440349	Topical wood	19.8	8.4
Burkina Faso	80131	Cashew nuts	7.9	50.2	710812	Gold	311.5	85.0
	520100	Cotton	4.2	26.6	80131	Cashew nuts	31.5	8.6
	260200	Mangane se ores & concentra tes	2.1	13.2	520100	Cotton	22.1	6.0
Burundi	120799	Oil seeds & oleaginou s fruits	0.3	27.8	780110	Unwrought lead, refined	7.5	67.0
	410221	Raw skins of sheep/la mbs	0.2	15.9	790112	Zinc	1.3	11.7
	410510	Tanned/c rust skins of sheep/la mbs	0.1	9.4	330129	Essential oils	1.1	9.8

Central	440349	Topical	1.0	47.3	710231	Diamonds	1.0	58.4
African Republic	440399	wood Wood	0.2	10.7	440399	Wood	0.3	15.2
Republic	720449	Ferrous waste & scrap	0.2	9.2	440349	Topical wood	0.2	12.0
Chad	520100	Cotton	2.6	74.5	270900	Petroleum oils	181.3	85.1
	900669	Photogra phic Flashligh t	0.1	4.0	310530	Diammoniu m hydrogenorth ophosphate	19.8	9.3
	910119	Wrist- watches	0.1	3.8	520100	Cotton	8.1	3.8
Democratic Republic of Congo	740319	Unwroug ht products	12.1	52.2	260300	Copper ores	63.4	35.6
0011.50	260500	Cobalt ores	6.4	27.8	270900	Petroleum oils & oils obt. from bit	37.9	21.3
	740311	Cathodes	1.5	6.5	710231	Diamonds	30.9	17.4
Djibouti	850433	Electrical transform ers	1.0	22.5	170310	Cane molasses	1.7	32.1
	720449	Ferrous waste & scrap	0.9	19.6	120799	Oil seeds & oleaginous fruits (excl	0.7	12.1
	410530	Tanned/c rust skins of sheep/la mbs	0.7	15.9	120100	Soya beans	0.5	9.1
Eritrea	260112	Iron ores	9.7	76.7	260300	Copper ores	103.6	88.3
	710310	Precious stones	1.1	8.9	261690	Precious metal ores	5.8	5.0
	720449	Ferrous waste & scrap	0.8	6.2	710310	Precious stones	3.4	2.9
Ethiopia	71333	Kidney beans	6.2	14.9	71333	Kidney beans	15.7	24.8
	410530	Tanned/c rust skins of sheep/la mbs	5.0	12.1	120100	Soya beans	15.4	24.3
	71331	Beans	4.9	11.8	71320	Chickpeas	8.0	12.6
Gambia	80131	Cashew nuts	25.3	91.0	80131	Cashew nuts	39.3	89.1
	760200	Aluminiu m waste & scrap	0.8	3.0	720449	Ferrous waste & scrap (excl. of 720	2.3	5.3
	330190	Essential oils	0.3	1.0	854810	Waste primary cells	1.4	3.1
Ghana	80131	Cashew nuts	98.2	28.6	710812	Gold	2261.7	90.7
	260300	Copper ores	48.7	14.2	80131	Cashew nuts	78.6	3.2

	440349	Topical wood	47.9	13.9	440349	Topical wood	51.2	2.1
Guinea	270900	Petroleu m oils	232.4	69.5	710812	Gold	243.5	53.6
	260300	Copper ores	63.3	18.9	270900	Petroleum oils	47.8	10.5
	80131	Cashew nut	14.9	4.5	260600	Aluminium ores	45.0	9.9
Guinea- Bissau	80131	Cashew nut	131.9	57.7	80131	Cashew nut	220.7	97.4
	270900	Petroleu m oils	57.7	25.2	260600	Aluminium ores	5.2	2.3
	260300	Copper	21.1	9.2	720449	Ferrous waste	0.3	0.1
Kenya	283620	Disodium carbonate	30.1	21.7	71331	Beans	21.6	20.1
	270119	Coal	14.5	10.4	283620	Disodium carbonate	20.4	19.0
	71331	Beans	12.2	8.8	90240	Tea, black (fermented)	11.5	10.7
Lesotho	510119	Wool, greasy	2.3	74.9	710231	Diamonds	30.3	97.8
	510129	Wool, degrease d	0.3	10.8	510119	Wool, greasy	0.6	1.9
	853620	Automati c circuit breakers	0.1	2.1	903300	Parts & accessories	0.1	0.2
Liberia	271011	Light petroleu m oils	7.9	28.7	890120	Tankers	43.3	71.4
	890400	Tugs & pusher craft	4.0	14.5	720449	Ferrous waste	5.0	8.2
	180100	Cocoa beans	3.5	12.6	890110	Ships	4.0	6.6
Madagascar	90700	Cloves	23.4	34.4	90700	Cloves	78.2	56.5
	720449	Ferrous waste	13.2	19.4	750210	Nickel	14.2	10.2
	71339	Beans	5.5	8.1	71339	Beans	8.9	6.5
Malawi	71390	Dried legumino us vegetable s	26.9	71.7	71390	Dried leguminous vegetables	38.1	81.9
	520100	Cotton	3.3	8.8	120100	Soya beans	3.5	7.6
	90240	Tea, black (fermente d)	1.1	3.0	71339	Beans	1.5	3.2
Mali	520100	Cotton	10.4	63.7	710812	Gold	78.0	51.0
	260111	Iron ores & concentra tes	1.3	8.0	520100	Cotton	65.2	42.6
	80131	Cashew nuts	1.1	6.5	130120	Gum Arabic	5.0	3.2

Mauritania	760200	Aluminiu m waste	1.5	31.3	260111	Iron ores	9.5	40.5
	720449	Ferrous waste	1.4	29.2	230120	Flours, meals & pellets of fish	3.7	15.8
	740400	Copper waste	0.8	17.1	720449	Ferrous waste	3.7	15.8
Mozambique	270119	Coal	87.3	54.9	270119	Coal	390.9	66.6
	71390	Dried legumino us vegetable s	35.8	22.5	71390	Dried leguminous vegetables	92.8	15.8
	80131	Cashew nuts	27.9	17.5	80131	Cashew nuts	44.8	7.6
Niger	271111	Natural gas, liquefied	72.3	94.8	271111	Natural gas, liquefied	19.5	97.5
	999999	Commod ities not specified	3.5	4.6	780199	Unwrought lead	0.1	0.6
	390120	Polyethyl ene	0.1	0.1	760120	Aluminium alloys	0.1	0.5
Nigeria	270900	Petroleu m oils	11959.2	94.6	270900	Petroleum oils	7700.8	88.8
	271111	Natural gas, liquefied	504.3	4.0	271111	Natural gas, liquefied	775.7	8.9
	841940	Distilling /rectifyin g plant	15.8	0.1	80131	Cashew nuts	56.8	0.7
Rwanda	90111	Coffee	0.2	29.8	780199	Unwrought lead	1.2	47.8
	90240	Tea, black (fermente d)	0.2	28.4	760200	Aluminium waste & scrap	0.4	16.2
	90230	Tea, black (fermente d)	0.1	18.0	261100	Tungsten ores	0.2	8.1
Senegal	280920	Phosphor ic acid	264.8	77.9	280920	Phosphoric acid	286.3	77.0
	720449	Ferrous waste & scrap	18.7	5.5	80131	Cashew nuts	38.8	10.4
	80131	Cashew nuts	15.0	4.4	720449	Ferrous waste & scrap	18.3	4.9
Sierra Leone	720449	Ferrous waste	2.9	39.5	720449	Ferrous waste	6.8	32.7
	760200	Aluminiu m waste	1.2	16.0	710231	Diamonds	4.6	22.3
	740400	Copper waste	0.6	8.5	260600	Aluminium ores	3.6	17.5
Somalia	120740	Sesamum seeds	3.9	45.0	120740	Sesamum seeds	10.6	81.4
	844790	Knitting machines	1.0	11.1	130190	Natural gums	0.8	6.1

	410190	Raw	0.6	7.4	410190	Raw hides &	0.4	3.0
		hides & skins of				skins of bovine		
Sudan	270900	bovine Petroleu m oils	134.7	75.1	270900	Petroleum oils	135.5	48.7
	440349	Topical wood	10.5	5.8	71390	Dried leguminous vegetables	33.9	12.2
	720449	Ferrous waste	9.0	5.0	120799	Oil seeds & oleaginous fruits	31.1	11.2
Togo	251020	Natural calcium phosphat es	70.8	46.7	270900	Petroleum oils	65.5	29.4
	251010	Natural calcium phosphat es	31.0	20.5	251020	Natural calcium phosphates	42.5	19.1
	720449	Ferrous waste	11.1	7.3	251010	Natural calcium phosphates, natural	27.8	12.5
Uganda	90111	Coffee	9.2	36.1	90111	Coffee	17.9	31.2
	520100	Cotton	3.6	14.2	71390	Dried leguminous vegetables, n.e.s.	7.4	12.9
	180100	Cocoa beans	3.1	12.1	520100	Cotton	6.8	11.8
Tanzania	80131	Cashew nuts	131.2	30.3	710812	Gold	477.5	48.3
	710812	Gold	95.7	22.1	80131	Cashew nuts	236.0	23.9
	71390	Dried legumino us vegetable s	46.0	10.6	71390	Dried leguminous vegetables	85.4	8.6
Zambia	740200	Unrefine d copper	94.5	54.6	740200	Unrefined copper	606.5	86.9
	740311	Cathodes	20.3	11.7	710310	Precious stones	67.9	9.7
	260300	Copper ores	20.1	11.6	740311	Cathodes	8.7	1.2
			ASI	IAN LDC	S			
Bangladesh	530310	Jute & other textile	71.0	13.3	530310	Jute & other textile	61.1	8.9
	630510	Sacks & bags	56.7	10.6	620342	Men's/boys' trousers	49.7	7.2
	30269	Fish	52.5	9.8	530710	Yarn of jute/of other textile	44.7	6.5
	80290	Nuts	40.6	7.6	630510	Sacks & bags	32.6	4.7
	530710	Yarn of jute/of	32.2	6.0	780199	Unwrought lead	25.3	3.7

		other						
Bhutan	720221	textile Ferro- silicon	86.7	43.7	720221	Ferro-silicon	93.0	47.5
	740819	Copper wire	20.7	10.4	271600	Electrical energy	38.5	19.7
	284910	Carbides, of calcium	16.7	8.4	284910	Carbides, of calcium	14.0	7.1
	740319	Unwroug ht products	10.9	5.5	284920	Carbides, of silicon	9.5	4.8
	721430	Bars & rods	10.1	5.1	392099	Plates, sheets, film, foil & strip,	4.3	2.2
Cambodia	151110	Palm oil, crude	3.8	34.6	440349	Topical wood	7.4	14.4
	400122	Natural rubber	1.4	12.6	400121	Natural rubber (excl. latex)	7.2	14.1
	400121	Natural rubber (excl. latex)	0.9	8.0	440399	Wood	6.6	12.9
	620342	Men's/bo ys' trousers	0.4	3.3	250590	Natural sands	2.8	5.4
	80131	Cashew nuts	0.3	2.9	110814	Manioc (cassava) starch	2.6	5.2
Laos	260300	Copper ores	77.4	98.2	710812	Gold	85.2	39.1
	848190	Parts of the appliance s	0.3	0.4	260300	Copper ores	58.5	26.8
	846210	Forging/d ie- stamping machines	0.1	0.2	440399	Wood	13.3	6.1
	720449	Ferrous waste	0.1	0.2	710813	Gold	11.1	5.1
	740939	Copper plates, sheets & strip	0.1	0.1	710691	Silver	8.5	3.9
Myanmar	71331	Beans	326.0	25.2	71331	Beans	431.5	43.3
	440349	Topical wood	289.5	22.4	71390	Dried leguminous vegetables	226.0	22.7
	440399	Wood	268.2	20.7	71339	Beans	56.3	5.6
	71390	Dried legumino us vegetable s	196.3	15.2	880240	Airplanes & other aircraft	50.4	5.1
	71339	Beans	65.5	5.1	440839	Sheets for veneering	41.0	4.1

Nepal	392690	Articles of plastics & articles	41.1	8.8	220290	Non- alcoholic beverages	75.8	16.5
	220290	Non- alcoholic beverage s	30.7	6.6	90830	Cardamoms	41.1	8.9
	721049	Flat- rolled products	30.2	6.4	392690	Articles of plastics & articles	34.3	7.4
	721041	Flat- rolled products	21.5	4.6	640419	Footwear	21.5	4.7
	550921	Yarn other than sewing thread	20.5	4.4	630510	Sacks & bags	17.2	3.7
Timor-Leste	999999	Commod ities not specified	10.5	82.0	80290	Nuts	0.1	43.4
	851770	Parts of telephone sets	0.9	7.0	90700	Cloves	0.1	24.7
	382490	Chemical products	0.8	6.0	847990	Parts of Machines	0.0	16.3
	720449	Ferrous waste	0.3	2.7	760200	Aluminium waste	0.0	9.6
	760200	Aluminiu m waste	0.1	1.0	940180	Seats	0.0	3.2
Yemen	270900	Petroleu m oils	1165.3	87.0	271111	Natural gas, liquefied	118.7	62.0
	271111	Natural gas, liquefied	104.5	7.8	270900	Petroleum oils	59.9	31.3
	271019	Petroleu m oils	46.9	3.5	780199	Unwrought lead	2.9	1.5
	780199	Unwroug ht lead	7.3	0.5	890190	Vessels for the transport of goods	1.9	1.0
	740811	Copper wire	2.6	0.2	780200	Lead waste & scrap	1.3	0.7

5.3 India's top merchandise imports exclusively from rest of the world

An analysis of the detailed data reported in Annex A tables reveals that India imports positive values of the top 10 commodities at the HS 6-digit level (of every African and Asian country in the study) from the Rest of World (ROW), though the magnitudes differ considerably. To summarize, India's global merchandise imports are generally concentrated in petroleum oils (270900), gold (710812), diamonds (710231), coal (270119), parts of telephone sets (851770), natural Gas, liquefied (271111), palm oil, crude (151110), telephones for cellular networks (851712) and commodities not specified (999999).

However, at the level of the individual African and Asian countries included in this study, there are specific products where the concerned country does not export at all to India. These

sectors/products are reported in Table 5.3 for the years 2015-2017 (avg.) to illustrate cases where there is demand in the Indian market but possible supply side constraints on the side of the exporting countries. This said, India also imposes high import duties on some of the products mentioned in the table such as diamonds, along with mandatory testing and certification requirements for telecommunication equipment mainly telephone sets. But the main reason why India imports these products from RoW in significant amounts but not at all from the selected countries is very likely the presence of supply-side constraints in that particular country and sector. These constraints have been reported in the literature as more severe in the case of a number of African economies mainly LDCs.

Note that Table 5.3 reports information for all the 41 countries with selected products/sectors where such a supply-side constraint may exist, i.e., where India's imports from ROW are significantly positive but those from the concerned country are zero.

Table 5.3: India imports from ROW but not from selected 41 countries (2015-2017)

Angolo	Benin	Burkina Faso	Burundi
Angola	Бешп	Burkina r aso	Burunai
Gold (710812), Diamonds (710231), Coal (270119), Parts of telephone sets (851770), Telephones for cellular network (851712)	Petroleum oils (270900), Gold (710812), Diamonds (710231), Coal (270119), Parts of telephone sets (851770	Petroleum oils (270900), Diamonds (710231), Coal (270119), Parts of telephone sets (851770), Natural gas, liquefied (271111)	Petroleum oils (270900), Gold (710812), Diamonds (710231), Coal (270119), Commodities not specified (999999)
Central African	Chad	Dem. Rep. of Congo	Djibouti
Republic Petroleum oils (270900), Gold (710812), Coal (270119), Parts of telephone sets (851770), Natural gas, liquefied (271111)	Gold (710812), Diamonds (710231), Commodities not specified (999999) Parts of telephone sets (851770), Natural gas, liquefied (271111)	Gold (710812), Coal (270119), Parts of telephone sets (851770), Natural gas, liquefied (271111), Telephones for cellular networks (851712)	Petroleum oils (270900), Gold (710812), Diamonds (710231), Coal (270119), Commodities not specified (999999)
Eritrea	Ethiopia	Gambia	Guinea
Petroleum oils (270900), Gold (710812), Diamonds (710231), Coal (270119), Commodities not specified (999999)	Petroleum oils (270900), Gold (710812), Diamonds (710231), Coal (270119), Parts of telephone sets (851770)	Petroleum oils (270900), Gold (710812), Diamonds (710231), Coal (270119), Parts of telephone sets (851770)	Coal (270119), Parts of telephone sets (851770), Telephones for cellular networks (851712), Diamonds (710239), Palm oil, crude (151110)
Guinea-Bissau	Lesotho	Liberia	Madagascar
Petroleum oils (270900), Gold (710812), Diamonds (710231), Coal	Petroleum oils (270900), Gold (710812), Coal (270119), Parts of telephone sets (851770),	Petroleum oils (270900), Gold (710812), Coal (270119), Parts of	Petroleum oils (270900), Gold (710812), Diamonds (710231), Coal

(270119),	Natural gas, liquefied	telephone sets	(270119), Parts of
Commodities not	(271111)	(851770), Natural gas,	telephone sets (851770)
specified (999999)		liquefied (271111)	
Malawi	Mali	Mauritania	Ghana
Petroleum oils	Petroleum oils (270900),	Petroleum oils	Petroleum oils
(270900), Gold	Diamonds (710231), Coal	(270900), Gold	(270900), Coal
(710812), Diamonds	(270119), Natural gas,	(710812), Diamonds	(270119), Natural gas,
(710231), Coal	liquefied (271111),	(710231), Coal	liquefied (271111),
(270119), Parts of	Telephones for cellular	(270119), Commodities	Telephones for cellular
telephone sets	networks (851712)	not specified (999999)	networks (851712),
(851770)			Diamonds (710239), Palm oil, crude
			(151110)
Kenya	Mozambique	Niger	Nigeria
Petroleum oils	Petroleum oils (270900),	Petroleum oils	Gold (710812),
(270900), Gold	Gold (710812),	(270900), Gold	Diamonds (710231),
(710812), Diamonds	Diamonds (710231),	(710812), Diamonds	Coal (270119), Parts of
(710231), Coal	Parts of telephone sets	(710231), Coal	telephone sets
(270119), Natural	(851770), Natural gas,	(270119), Parts of	(851770), Diamonds
gas, liquefied (271111)	liquefied (271111)	telephone sets (851770)	(710239)
Rwanda	Senegal	Sierra Leone	Somalia
Petroleum oils	Petroleum oils (270900),	Petroleum oils	Petroleum oils
(270900), Gold	Gold (710812),	(270900), Gold	(270900), Gold
(710812), Diamonds	Diamonds (710231), Coal	(710812), Coal	(710812), Diamonds
(710231), Coal	(270119), Parts of	(270119), Commodities	(710231), Coal
(270119), Parts of	telephone sets (851770)	not specified (999999),	(270119), Commodities
telephone sets (851770)		Parts of telephone sets (851770)	not specified (999999)
Sudan	Tanzania	Togo	Uganda
Diamonds (710231).	Petroleum oils (270900),	Gold (710812),	Petroleum oils
Coal (270119), Parts	Coal (270119), Parts of	Diamonds (710231),	(270900), Gold
of telephone sets	telephone sets (851770),	Coal (270119), Parts of	(710812), Diamonds
(851770), Natural	Natural gas, liquefied	telephone sets	(710231), Coal
gas, liquefied	(271111), Telephones for	(851770), Natural gas,	(270119), Natural gas,
(271111), Telephones for	cellular networks	liquefied (271111)	liquefied (271111)
cellular networks	(851712)		
(851712)			
Zambia	Bangladesh	Bhutan	Cambodia
Petroleum oils	Petroleum oils (270900),	Petroleum oils	Petroleum oils
(270900), Gold	Gold (710812),	(270900), Gold	(270900), Gold
(710812), Diamonds	Diamonds (710231), Coal	(710812), Diamonds	(710812), Diamonds
(710231), Coal	(270119), Natural gas,	(710231), Natural gas,	(710231), Coal
(270119), Parts of	liquefied (271111)	liquefied (271111),	(270119), Parts of
* * * * * * * * * * * * * * * * * * * *		Tolombores for 11-1.	tolombons asta (051770)
telephone sets (851770)		Telephones for cellular networks (851712)	telephone sets (851770)

Petroleum oils	Petroleum oils (270900),	Petroleum oils	Petroleum oils
(270900), Diamonds	Diamonds (710231), Coal	(270900), Diamonds	(270900), Gold
(710231), Coal	(270119), Natural gas,	(710231), Coal	(710812), Diamonds
(270119), Natural	liquefied (271111),	(270119), Parts of	(710231), Coal
gas, liquefied	Telephones for cellular	telephone sets	(270119), Parts of
(271111),	networks (851712)	(851770), Natural gas,	telephone sets (851770)
Telephones for		liquefied (271111),	
cellular networks		Telephones for cellular	
(851712)		networks (851712)	

Yemen

Gold (710812), Diamonds (710231), Coal (270119), Parts of telephone sets (851770), Telephones for cellular networks (851712)

Source: UN Comtrade; own calculations

CHAPTER 6

Mapping services exports of Africa and Asia in India

Analogous to the analysis for goods, this chapter details the top services exported by select African (31) and Asian (7) countries to India. The data have been sourced from the OECD-WTO BaTIS ("Balanced Trade in Services") database and cover eleven services sectors namely, transportation; travel; communication services; construction; insurance services; financial services; computer and information services; royalties and license fees; other business services (OBS); and personal, cultural and recreational services. In order to study the evolution of these countries' services exports to India over time, two time-periods have been considered - 2000-02 and 2010-12 - and the data have been averaged over these three year time periods to minimize the effect of fluctuations in data reporting practices and to account for macroeconomic and other external shocks.

While 2012 is the last year for which BaTIS reports these data, the analysis in this report still constitutes significant value-added relative to existing literature, given the absence of more recent bilateral and sectoral services trade data for the countries covered in this study. Direct exports are used in the case of services trade analysis as the final balanced value of both exports and imports is reported in this database.

BaTIS covers 38 of the 41 countries examined in this study: Angola, Benin, Burkina Faso, Burundi, Central African Republic, Chad, Democratic Republic of Congo, Djibouti, Ethiopia, Gambia, Ghana, Guinea, Guinea-Bissau, Kenya, Lesotho, Liberia, Madagascar, Malawi, Mali, Mauritania, Mozambique, Niger, Nigeria, Rwanda, Senegal, Sierra Leone, Sudan, Togo, Uganda, Tanzania, Zambia, Bangladesh, Bhutan, Cambodia, Laos, Myanmar, Nepal and Yemen. Unfortunately, data are not available for three countries, namely, Eritrea, Somalia and Timor-Leste.

The data in this chapter are arranged as follows. Table 6.1 reports data on total services exported to India by these 38 countries averaged over 3 years from 2000 to 2002 and 2010 to 2012. Table 6.2 reports data on the top two services exported to India by these countries, again averaged over the same time periods. Table 6.3 reports the services imported by India from the Rest of the World (ROW) but not from the selected African or Asian countries under study in 2010-2012 (avg.) to illustrate cases where there is demand in the Indian market but possible supply side constraints on the side of the exporting countries.

6.1 Total services exports of Africa and Asia to India

The results for selected African and Asian countries based on an analysis of data reported in Table 6.1 are as follows:

1. **Top African countries which exported to India during 2000-2002**Nigeria (USD 82 million), Liberia (USD 38 million), Senegal (USD 36 million), Tanzania (USD 34 million), and Kenya (USD 30 million)

2. **Top African countries which exported to India during 2010-2012**Nigeria (USD 542 million), Angola (USD 364 million), Senegal (USD 141 million), Kenya (USD 130 million), and Tanzania (128 million)

3. Top Asian countries which exported to India during 2000-2002

Nepal (USD 61 million), Myanmar (USD 23 million), Yemen (USD 15 million), Bangladesh (USD 13 million), and Bhutan (USD 9 million)

4. Top Asian countries which exported to India during 2010-2012

Nepal (USD 183 million), Yemen (USD 131 million), Myanmar (USD 104 million), Bangladesh (USD 86 million), and Bhutan (USD 59 million)

The value of services exported to India by the African and Asian countries covered in this study export is much lower than that of their merchandise exports to India. However, unlike merchandise exports where some countries have shown a decline in the value of their exports to India over time, the value of total services exported to India by all the selected African and Asian countries in this study have increased over the period 2000-02 and 2010-12 (see Column 4 of Table 6.1). The table also highlights those countries that have experienced the highest growth in the value of their services exports to India between the two time-periods. In the case of Africa, these countries include Nigeria (rise in value by \$460 million), Angola (by \$361 million), and Senegal (\$105 million), while in the case of Asia, these countries include Nepal (\$122 million), Yemen (\$116 million), and Myanmar (\$81 million).

To elaborate, Nigeria and Nepal have retained their positions as the top African and Asian services exporters to India, respectively, in both the periods 2000-02 and 2010-12. As already stated, they have also increased the value of their services exports to India. For Africa, the second position was held by Liberia in the earlier period, followed by Senegal. Not only have their exports increased over the years, but their relative positions in the ranking of top services exporters to India has also changed. For instance, while Senegal remained the third largest services exporter to India even in 2010-12, Liberia's second position in 2000-02 was lost and replaced by Angola. This interchange also happened in the relative positions of Tanzania and Kenya in the case of Africa and between Yemen and Myanmar in the case of Asian LDCs, with a higher rise in their services export values to India. Meanwhile, Bangladesh and Bhutan have also been exporting significantly to India.

In contrast, the worst African performers in terms of services exports to India during 2000-02 (avg.) were Lesotho (with services exports of just \$.07 million), Burundi (\$0.1 million), Mauritania (\$0.17 million), Rwanda (\$0.18 million), Niger (\$0.49 million), and Malawi (\$0.64 million). Although these countries' services exports to India have risen in absolute terms from this period to 2010-12, they still lie in the bottom half of the distribution with Burundi, Lesotho and Rwanda as the lowest services exporters in the latter period.

Meanwhile, the worst Asian performers in 2000-02 were Laos (services exports to India of \$0.58 million), Cambodia (\$1.3 million), and Bhutan (\$8.8 million). By 2010-12, their services exports to India increased rapidly which is indicative of the growing India-LDCs trade relationship in the case of Asia, as also documented by existing literature on this subject. At the same time, as compared to top Asian performers, these countries still remained at the bottom rung of the services export ladder.

Table 6.1: Total services exports by select African (31) and Asian (7) countries to India, in 2000-2002 (avg.) and 2010-2012 (avg.)

Countries	Value (avg. 2000-	Value (avg. 2010-	Difference = (avg. 2010-
	2002, US\$ million)	2012, US\$ million)	2012) - (avg. 2000-2002)

1	2	3	4
Angola	3.2	364.4	361.2
Benin	14.9	62.2	47.4
Burkina Faso	6.4	17.1	10.7
Burundi	0.1	1.1	1.0
Central African Republic	1.6	4.7	3.1
Chad	1.8	3.4	1.6
Congo	2.7	54.2	51.5
Djibouti	5.2	24.1	18.9
Ethiopia	9.9	40.5	30.6
Gambia	3.1	30.2	27.1
Ghana	8.3	75.8	67.5
Guinea	1.4	21.5	20.1
Guinea-Bissau	4.3	25.3	21.0
Kenya	29.9	129.8	99.9
Lesotho	0.1	1.6	1.5
Liberia	38.3	90.8	52.5
Madagascar	2.6	35.5	32.9
Malawi	0.6	5.5	4.9
Mali	7.3	15.8	8.6
Mauritania	0.2	4.0	3.8
Mozambique	6.7	33.0	26.3
Niger	0.5	6.6	6.1
Nigeria	82.6	542.4	459.7
Rwanda	0.2	2.0	1.8
Senegal	36.6	141.6	105.0
Sierra Leone	1.0	5.5	4.5
Sudan	5.7	23.2	17.5
Togo	4.1	26.2	22.1
Uganda	3.0	30.5	27.4
Tanzania	34.4	128.1	93.7
Zambia	5.6	48.7	43.1
Bangladesh	13.4	85.9	72.5
Bhutan	8.8	58.8	50.0
Cambodia	1.3	13.9	12.6
Laos	0.6	34.2	33.6
Myanmar	22.8	103.8	81.0
Nepal	61.2	183.5	122.3
Yemen	15.4	131.8	116.3

Source: OECD-WTO BaTIS; own calculations

[*Note:* The figures highlighted report those African and Asian countries which experienced the highest growth in the value of their services exports to India between 2010-2012 (avg.) and 2015-2017 (avg.).]

6.2 Top services exported by selected countries to India

An analysis of the data reported in Table 6.2 reveals the following:

- 1. **Top services exported by African countries to India (2000-2002)** Travel, Transportation and Government Services
- 2. **Top services exported by African countries to India (2010-2012)** Travel, Transportation and Other Business Services
- 3. Top services exported by Asian countries to India (2000-2002) Travel and Transportation
- 4. **Top services exported by Asian countries to India (2010-2012)** Travel and Transportation

Detailed information is reported in the tables in 'Annex B', which show the top ten services export sectors for each of the African and Asian country along with their values and shares, and India's imports of services in that sector from ROW.

Table 6.2 shows that of the 31 African countries, Nigeria, Angola and Senegal were the top three services exporters to India in 2010-12 (Liberia was in place of Angola in 2000-02). Sector-wise, the share of Nigeria's top two sectors in its total services exported to India in the two time-periods was 77% and 67.5%, respectively; the share of transportation services exports alone stood at 43% and 48%, respectively. Other services such as construction, travel, and other business services (OBS) had shares of 19.3% (34.2%), 18.1% (7%) and 11.3% (11.8%) during 2010-12 (2000-02), respectively.

The share of Senegal's top two sectors in its total services exported to India in 2000-02 and 2010-12 was 64% and 65%, respectively; the share of travel services exports alone was about 43% and 41%, respectively. Meanwhile, Liberia's top service exported to India has been transportation (86% of its total exports to India in 2000-02 which later declined to 82.5%). Angola's top service in 2010-2012 was OBS (36.8% - risen from 16% during 2000-02), followed by construction whose share was almost reduced to half.

In contrast, African countries which have exported their services the least to India in both time-periods include Burundi, Lesotho and Rwanda. The top service exported by Burundi in 2000-02 was computer and information services (22.4% of its total exports), but in 2010-2012, the top sector was OBS (46%). In the case of Lesotho and Rwanda, the top services exported to India in the two time periods were travel (86% in 2000-02 and 84% in 2010-12) and financial services (whose share increased from 29% to 36% over the two periods), respectively.

In the context of the selected Asian LDCs, top services exporters to India in the two timeperiods were the same - Nepal, Yemen and Myanmar. All three countries have been significant exporters of travel services to India, though the importance of the sector in total services exports to India did not change much over time. For instance, the share of travel services in Nepal's total services exports to India increased slightly from 71.9% in 2000-02 to 73% during 2010-12, while in the case of Yemen, the sector's share increased from 53% to 58% over the same time period. For Nepal, communication services have been the second most exported services sectors, though the sector's share in total services exports to India declined from 17% to 14% during the two periods. In the case of Yemen, government services (share of 20.7%) were the second most exported service to India in 2000-02, but in 2010-12, these were replaced by OBS (with a share of 17%). In the case of Myanmar, the share of transportation in total services exports was 49% and 46% in 2000-02 and 2010-12, respectively. There were almost similar shares in the exports of travel services to India with respective shares of 44.4% and 44.7%.

Asian countries with the lowest value of services exported to India in both the time periods included Laos, Cambodia and Bhutan. However, there was a huge rise in the absolute values of their services exports to India, with respective growth of over 5500%, 900% and 500% (from 2000-02 to 2010-12). The top service exported by all these three countries to India in both the periods was travel, except for Cambodia in 2010-2012 where the top exported service was transportation.

Table 6.2: Top services exported by select African and Asian countries to India, 2000-2002 (avg.) and 2010-2012 (avg.)

Countries	Sector description	Avg. of 2000- 2002 (\$ million)	Share in total services exports (%)	Sector description	Avg. of 2010-2012 (\$ million)	Share in total services exports (%)
		AFRI	CAN COUN	TRIES		
Angola	Construction	1.63	50.47	Other business services (OBS)	134.21	36.83
	OBS	0.51	15.85	Construction	102.17	28.04
Benin	Travel	7.08	47.68	Travel	24.37	39.17
	OBS	2.55	17.17	OBS	16.04	25.78
Burkina Faso	Travel	3.51	54.83	Travel	6.03	35.29
	OBS	1.36	21.19	OBS	5.77	33.74
Burundi	Computer and information services	0.02	22.35	OBS	0.49	45.82
	Transportation	Transportation 0.02 19.40 Government		Government services, n.i.e.	0.21	19.70
Central	Travel	1.37	85.18	Travel	3.83	80.95
African Republic	Government services, n.i.e.	0.22	13.60	Government services, n.i.e.	0.79	16.78
Chad	Transportation	0.83	46.98	Transportation	1.52	45.34
Cliad	Government services, n.i.e.	0.51	28.86	Travel	1.00	29.72
Congo	Construction	0.84	30.77	OBS	11.90	21.96
C	OBS	0.62	22.65	Construction	8.74	16.14
Djibouti	Travel	2.06	39.24	Travel	8.44	35.02
թյանա	Government services, n.i.e.	1.46	27.76	Transportation	7.86	32.59

Ethiopia	Transportation	3.31	33.35	Transportation	18.41	45.41
Ешюріа	Government	2.96	29.86	Travel	10.71	26.43
	services, n.i.e. Travel	2.69	86.02	Travel	24.50	81.04
Gambia	Communicatio	0.22	6.90	Communications	3.19	10.55
_	ns services			services		
Ghana	Travel	5.96	71.79	Travel	40.13	52.96
	Transportation	1.61	19.35	Transportation	21.17	27.94
Guinea	Transportation	0.41	29.39	Transportation	6.01	27.95
	Government services, n.i.e.	0.31	22.37	Communications services	4.25	19.75
Guinea-	Travel	2.05	48.04	Travel	11.02	43.55
Bissau	Transportation	0.93	21.74	Transportation	7.56	29.88
Vanua	Travel	13.38	44.72	Transportation	60.83	46.86
Kenya	Transportation	12.15	40.62	Travel	47.48	36.58
Lesotho	Travel	0.06	86.13	Travel	1.36	83.88
Lesotho	Transportation	0.01	9.04	Transportation	0.20	12.44
Liberia	Transportation	32.60	85.12	Transportation	74.99	82.56
Liberia	Travel	2.86	7.46	Government services, n.i.e.	8.27	9.11
Madagascar	Travel	1.49	57.74	Travel	18.26	51.40
1.1mangus vui	Transportation	0.63	24.38	Transportation	11.32	31.85
M-1:	Travel	0.38	58.80	Travel	2.96	53.67
Malawi	Transportation	0.18	28.52	Transportation	1.72	31.20
Mali	Communicatio ns services	1.99	27.49	OBS	4.90	30.99
	OBS	1.63	22.54	Travel	4.37	27.60
Mauritania	Travel	0.07	43.58	Travel	1.39	34.85
Mauritania	Government services, n.i.e.	0.04	24.47	Government services, n.i.e.	1.05	26.34
Mozambique	Transportation	4.63	69.05	Transportation	22.65	68.68
•	Travel	1.38	20.60	Travel	6.09	18.47
Niger	Transportation	0.41	83.42	Transportation	5.97	90.13
- 1-8	Travel	0.04	7.30	Travel	0.23	3.42
Nigeria	Transportation	35.52	42.99	Transportation	261.56	48.23
<i>U</i>	Construction	28.25	34.19	Construction	104.57	19.28
Rwanda	Financial services	0.05	28.82	Financial services	0.71	36.22
	Travel	0.05	28.77	Travel	0.44	22.36
	Travel	15.78	43.07	Travel	58.01	40.97
Senegal	Government services, n.i.e.	7.80	21.29	OBS	33.93	23.96
Sierra Leone	Travel	0.43	42.37	Travel	2.45	44.46

	Transportation	0.40	39.67	Transportation	2.35	42.80
	Transportation	1.92	33.66	Transportation	9.19	39.55
Sudan	Travel	1.68	29.54	Travel	5.17	22.26
Togo	Travel	2.20	53.41	Travel	14.64	55.85
U	Construction	0.68	16.57	Transportation	4.81	18.35
Uganda	Travel	1.74	57.30	Travel	15.12	49.65
S	OBS	0.42	13.72	OBS	4.79	15.74
Tanzania	Travel	19.10	55.56	Travel	78.30	61.15
	Transportation	11.84	34.44	Transportation	39.52	30.86
Zambia	Travel	3.29	58.54	Transportation	22.17	45.54
	Transportation	1.73	30.74	Travel	21.74	44.66
		ASIA	N COUNT	RIES		
Bangladesh	Transportation	3.66	27.35	OBS	35.09	40.86
Ü	OBS	3.52	26.34	Transportation	26.41	30.76
Bhutan	Travel	8.57	96.93	Travel	56.89	96.74
Dilutan	Government services, n.i.e.	0.11	1.29	Transportation	0.94	1.60
Cambodia	Travel	0.66	49.04	Transportation	7.00	50.25
	Transportation	0.65	48.56	Travel	6.38	45.80
Laos	Travel	0.46	78.13	Travel	27.77	81.20
	Transportation	0.08	13.54	Transportation	3.16	9.24
Myanmar	Transportation	11.27	49.36	Transportation	47.38	45.65
,	Travel	10.14	44.40	Travel	46.38	44.68
Nepal	Travel	44.04	71.93	Travel	134.00	73.01
тераг	Communicatio ns services	10.43	17.03	Communications services	25.43	13.85
	Travel	8.14	52.71	Travel	76.75	58.24
Yemen	Government services, n.i.e.	3.2	20.71	OBS	22.68	17.21

Source: OECD-WTO BaTIS; own calculations

The reasons for these countries exporting only a limited number of services to India such as travel, transportation, government services, and even communications services and OBS can be linked to India's higher restrictions on imports of financial and insurance services, and a number of professional services such as legal, accounting, etc., followed by some barriers to entry in telecommunication services where the government still holds a major equity share. Foreign ownership limit also exists in the audio-visual and media sectors. Changes in policy, especially on FDI, could lead to a diversification of these countries' exports to India.

6.3 India's top services imports exclusively from rest of the world

An analysis of the detailed data reported in Annex B tables reveals that India imports positive values of all services from the Rest of the World (ROW), though the magnitudes differ considerably. This said, India's global services imports are generally concentrated in travel, transportation and OBS.

However, at the level of the individual African and Asian country included in this study, there are specific services sectors where the concerned country does not export at all to India. These sectors are reported in Table 6.3 for the recent years 2010-2012 (avg.) country-wise and are likely indicative of supply-side constraints in that particular country and sector. Note that Table 6.3 only reports those countries and sectors where such a supply-side constraint may exist i.e. where India's imports from ROW are significantly positive but those from the concerned country are zero (this was found to be the case for 15 of the 31 African countries and 6 of the 7 Asian countries).

Table 6.3: India imports from ROW but not from the concerned African or Asian country, 2010-2012 (avg.)

Burkina Faso	Central African Republic	Chad	Djibouti
Royalties and license fees	Royalties and license fees; Computer and information services; Communication services; Insurance services; Personal, cultural and recreational services	Royalties and license fees; Computer and information services; Construction; Personal, cultural and recreational services	Computer and information services; Construction; Personal, cultural and recreational services
Gambia	Guinea	Lesotho	Liberia
Royalties and license fees; Personal, cultural and recreational services	Royalties and license fees	Royalties; Computer and information services; Construction; Personal, recreational services	Personal, cultural and recreational services
Malawi	Mauritania	Ghana	Niger
Royalties and license fees; Computer and information services; Construction; Personal, cultural and recreational services	Computer and information services	Insurance services	Royalties and license fees; Construction
Sierra Leone	Sudan	Togo	Bangladesh
Computer and information services	Personal, cultural and recreational services	Royalties and license fees	Personal, cultural and recreational services
Bhutan	Cambodia	Laos	Myanmar
Royalties and license fees; Construction; Personal, cultural and recreational services	Construction: Insurance services	Royalties and license fees; Computer and information services; Personal, cultural and recreational services	Computer and information services; Insurance services
Nepal Royalties and license fee			

Source: OECD-WTO BaTIS; own calculations

CHAPTER 7

Supply constraints in selected African and Asian countries

Limited capacities in LDCs mainly of Africa, and even Asian, for supplying goods and services to India has been regarded as one of the main challenges that restrict harnessing of the full potential of their bilateral trade. From a strategic perspective, India can invest in those sectors/products/services in selected African and Asian countries which do have some growing competitiveness, as also listed in previous chapters, but also urgently need support to overcome supply constraints.

Given this, the analysis in this chapter aims to identify such sectors in the case of both goods and services segments in the concerned African and Asian countries with investment potential for India. The analysis considers those sectors (goods and services) where India's global imports exceed the concerned country's global exports as well as the associated region's (African or Asian) global exports, thus, indicating a supply constraint for that product or service in the concerned country as well as the geographical neighbourhood. Thus, the chapter looks at those goods and services for which the selected African and Asian countries may require supply capacities in order to be able to export the necessary quantities to India. Indian companies can form JVs with local companies in these countries or invest in their export projects, etc. so that the selected countries can enhance their capacity to meet the demand of the Indian market.

One major finding from the analysis undertaken in this chapter is that none of the eight Asian LDCs has a supply constraint in any of the goods and services items in either of the time periods. This is largely due to the significant effect of the Asian region's high global exports (likely due to more linkages in GVCs of South East Asian countries). Thus, although there were several goods and services where India's global imports exceeded global exports of the concerned Asian country, the former could not surpass the exports of the Asian region to the world, alluding to the existence of supply-side capacities in the immediate neighbourhood.

On the other hand, there are a number of merchandise goods followed by some services sectors in African countries where there exist quite significant supply constraints in exporting to India. Accordingly, Table 7.1 (Part A-C) reports such commodities (sectors) at the HS 2-digit level. In this table, 'E' denotes the presence of a supply constraint in the concerned country only during the earlier time-period 2010-2012 (avg.); 'B' denotes the presence of a supply constraint in the concerned country during both time periods 2010-2012 (avg.) and 2015-2017 (avg.); and 'L' denotes the presence of a supply constraint in the concerned country only in the later time period 2015-2017 (avg.). Significantly, a majority of the products with a lack of supply capacity, identified using the methodlogy described in Section 3.1.2 of Chapter 3 i.e. where India's global imports exceed the concerned country's global exports as well as the associated region's global exports, are almost the same across all the concerned African countries.

The products identified with supply constraints and with investment potential for India in both the time periods 2010-2012 (avg.) and 2015-2017 (avg.) include: animal or vegetable fats; organic chemicals; pharmaceutical products; fertilizers; tanning or dyeing extracts; albuminoidal substances; miscellaneous chemical products; photographic or cinematographic goods; plastics and articles thereof; rubber and articles thereof; pulp of wood or of other fibrous cellulosic material; paper and paperboard; articles of paper pulp, of paper or of paperboard;

printed books, newspapers, pictures; silk; other vegetable textile fibers; man-made filaments; man-made staple fibers; wadding, felt and nonwoven; special woven fabric; impregnated, coated, covered or laminated textile fabrics; knitted or crocheted fabrics; umbrellas, sun umbrellas, walking-sticks, seat-sticks, whips, riding-crops and parts thereof; articles of stone, plaster, cement, asbestos, mica or similar materials; ceramic products; glass and glassware; iron and steel; articles of iron or steel; lead and articles thereof; tin and articles thereof; tools, implements, cutlery, spoons; miscellaneous articles of base metal; nuclear reactors, boilers, machinery and mechanical appliances; parts thereof; electrical machinery and equipment; railway or tramway locomotives; aircraft, spacecraft, and parts thereof; ships, boats and floating structures; optical, photographic, cinematographic, measuring, checking, precision, medical or surgical instruments and apparatus; parts and accessories thereof; clocks and watches and parts thereof; musical instruments; toys, games and sports requisite; and miscellaneous manufactured articles. Commodities with supply constraints only in the later time period 2015-17 include vegetable plaiting materials; inorganic chemicals; zinc and articles; furniture, bedding, mattresses etc.; and in the category of commodities not specified anywhere. So, there are number of manufacturing sectors where India can invest.

However, India also needs to re-work some of its import restrictions/prohibitions or lower down its tariff rates and/or reduce number of mandatory certification/registration requirements which can also help African and Asian countries to export their manufactured goods to India in a more feasible manner. For instance, India currently bans import of animal fats, puts high duties on plastic articles and essential drugs, require import licences for imports of chemicals (in fact, pharmaceutical products are mainly permitted to be imported by government trading monopolies), practices price control on medical devices, and requires complex in-country testing of toys and electronic/telecommunication equipment, as well as certification for steel products; etc. There is scope for liberalization of these sectors by India at least on a bilateral basis.

Table 7.1 (Part A): Commodities with supply constraints in select African countries but with potential for Indian investment, 2010-2012 (avg.) and 2015-2017 (avg.)

Code	Product Description	Angola	Benin	Burkina Faso	Burundi	Central African Republic	Chad	Dem. Rep. of Congo	Djibouti	Eritrea	Ethiopia
14	Vegetable plaiting materials	L	L	L	L	L	L	L	L	L	L
15	Animal or vegetable fats/oils and their cleavage products	В	В	В	В	В	В	В	В	В	В
28	Inorganic chemicals; organic or inorganic compounds of precious metals	L	L	L	L	L	L	L	L	L	L
29	Organic chemicals	В	В	В	В	В	В	В	В	В	В
30	Pharmaceutica 1 products	В	В	В	В	В	В	В	В	В	В
31	Fertilizers	В	В	В	В	В	В	В	В	В	В
32	Tanning or dyeing extracts; tannins and their deri. dyes, pigments and other	В	В	В	В	В	В	В	В	В	В

	1										
	colouring										
	matter										
35	Albuminoidal substances; modified starches; glues; enzymes	В	В	В	В	В	В	В	В	В	В
37	Photographic /cinematograp hic goods	В	В	В	В	В	В	В	В	В	В
38	Miscellaneous chemical products	В	В	В	В	В	В	В	В	В	В
39	Plastics and articles thereof	В	В	В	В	В	В	В	В	В	В
40	Rubber and articles thereof	В	В	В	В	В	В	В	В	В	В
47	Pulp of wood or of other fibrous cellulosic material; waste and scrap of paper/paperbo ard	В	В	В	В	В	В	В	В	В	В
48	Paper and paperboard; articles of paper pulp, of paper or of paperboard	В	В	В	В	В	В	В	В	В	В
49	Printed books, newspapers, pictures etc. of	В	В	В	В	В	В	В	В	В	В

	the printing industry; manuscripts, typescripts and plans										
50	Silk	В	В	В	В	В	В	В	В	В	В
53	Other vegetable textile fibres; paper yarn and woven fabrics of paper yarn	В	В	В	В	В	В	В	В	В	В
54	Man-made filaments	В	В	В	В	В	В	В	В	В	В
55	Man-made staple fibres	В	В	В	В	В	В	В	В	В	В
56	Wadding, felt and nonwovens; special yarns; twine, cordage, ropes and cables and articles thereof	В	В	В	В	В	В	В	В	В	В
58	Special woven fabrics	В	В	В	В	В	В	В	В	В	В
59	Impregnated, coated, covered or laminated textile fabrics	В	В	В	В	В	В	В	В	В	В
60	Knitted or crocheted fabrics	В	В	В	В	В	В	В	В	В	В

66	Umbrella, sun umbrellas etc. and parts thereof	В	В	В	В	В	В	В	В	В	В
68	Articles of stone, plaster, cement etc.	В	В	В	В	В	В	В	В	В	В
69	Ceramic products	В	В	В	В	В	В	В	В	В	В
70	Glass and glassware	В	В	В	В	В	В	В	В	В	В
71	Natural or cultured pearls, precious/semi- precious stones etc.	E	E	E	E	E	E	E	E	E	Е
72	Iron and steel	В	В	В	В	В	В	В	В	В	В
73	Articles of iron or steel	В	В	В	В	В	В	В	В	В	В
78	Lead and articles thereof	В	В	В	В	В	В	В	В	В	В
79	Zinc and articles thereof	L	L	L	L	L	L	L	L	L	L
80	Tin and articles thereof	В	В	В	В	В	В	В	В	В	В
82	Tools, implements, cutlery, spoons and forks, of base metal; parts thereof of base metal	В	В	В	В	В	В	В	В	В	В

83	Miscellaneous articles of base metal	В	В	В	В	В	В	В	В	В	В
84	Nuclear reactors, boilers, machinery	В	В	В	В	В	В	В	В	В	В
85	Electrical machinery and equipment	В	В	В	В	В	В	В	В	В	В
86	Railway or tramway locomotives, etc.	В	В	В	В	В	В	В	В	В	В
88	Aircraft, spacecraft, railways or tramway track fixtures and fittings and parts thereof	В	В	В	В	В	В	В	В	В	В
89	Ships, boats and floating structure	В	В	В	В	В	В	В	В	В	В
90	Optical, photographic, cinematograph ic, medical/surgic al instruments etc.	В	В	В	В	В	В	В	В	В	В
91	Clocks and watches and parts thereof	В	В	В	В	В	В	В	В	В	В

92	Musical instruments	В	В	В	В	В	В	В	В	В	В
94	Furniture; bedding, mattresses, etc.	L	L	L	L	L	L	L	L	L	L
95	Toys, games and sports requisites; parts and accessories thereof	В	В	В	В	В	В	В	В	В	В
96	Miscellaneous manufactured articles	В	В	В	В	В	В	В	В	В	В
99	Commodities not specified	L	L	L	L	L	L	L	L	L	L

Table 7.1 (Part B): Commodities with supply constraints in select African countries but with potential for Indian investment, 2010-2012 (avg.) and 2015-2017 (avg.)

Code	Product	Gambia	Ghana	Guinea	Guinea- Bissau	Kenya	Lesotho	Liberia	Madagascar	Malawi	Mali
14	Vegetable plaiting materials	L	L	L	L	L	L	L	L	L	L
15	Animal or vegetable fats/oils and their cleavage products	В	В	В	В	В	В	В	В	В	В
28	Inorganic chemicals; organic or	L	L	L	L	L	L	L	L	L	L

	inorganic										
	compounds of										
	precious										
	metals										
29	Organic chemicals	В	В	В	В	В	В	В	В	В	В
30	Pharmaceutical products	В	В	В	В	В	В	В	В	В	В
31	Fertilizers	В	В	В	В	В	В	В	В	В	В
32	Tanning or dyeing extracts; tannins and their deri. dyes, pigments and other colouring matter	В	В	В	В	В	В	В	В	В	В
35	Albuminoidal substances; modified starches; glues; enzymes	В	В	В	В	В	В	В	В	В	В
37	Photographic /cinematograph ic goods	В	В	В	В	В	В	В	В	В	В
38	Miscellaneous chemical products	В	В	В	В	В	В	В	В	В	В
39	Plastics and articles thereof	В	В	В	В	В	В	В	В	В	В
40	Rubber and articles thereof	В	В	В	В	В	В	В	В	В	В
47	Pulp of wood or of other fibrous	В	В	В	В	В	В	В	В	В	В

	cellulosic										
	material; waste										
	and scrap of										
	paper/paperboa										
	rd										
	Paper and										
	paperboard;										
48	articles of	В	В	В	В	В	В	В	В	В	В
	paper pulp, of										
	paper or of paperboard										
	Printed books,										
	newspapers,										
	pictures etc. of										
	the printing										
49	industry;	В	В	В	В	В	В	В	В	В	В
	manuscripts,										
	typescripts and										
	plans										
50	Silk	В	В	В	В	В	В	В	В	В	В
	Other										
	vegetable										
53	textile fibres;	В	В	В	В	В	В	В	В	В	В
33	paper yarn and	Б	Б	Б	Ь	Ь	Ь	Ъ	Б	Ь	Б
	woven fabrics										
	of paper yarn										
54	Man-made	В	В	В	В	В	В	В	В	В	В
	filaments Man-made										
55	staple fibres	В	В	В	В	В	В	В	В	В	В
	Wadding, felt										
	and										
56	nonwovens;	В	В	В	В	В	В	В	В	В	В
	special yarns;										
	twine, cordage,										

cables and										
articles thereof										
Special woven fabrics	В	В	В	В	В	В	В	В	В	В
Impregnated, coated,	D	D.	D	D.	ъ	D	D	D	ъ	D
laminated textile fabrics	В	В	В	В	В	В	В	В	В	В
Knitted or crocheted fabrics	В	В	В	В	В	В	В	В	В	В
Umbrella, sun umbrellas etc. and parts thereof	В	В	В	В	В	В	В	В	В	В
Articles of stone, plaster, cement etc.	В	В	В	В	В	В	В	В	В	В
Ceramic products	В	В	В	В	В	В	В	В	В	В
Glass and glassware	В	В	В	В	В	В	В	В	В	В
Natural or cultured pearls, precious/semi- precious stones etc.	E	E	E	Е	Е	Е	E	E	E	E
Iron and steel	В	В	В	В	В	В	В	В	В	В
Articles of iron or steel	В	В	В	В	В	В	В	В	В	В
Lead and articles thereof	В	В	В	В	В	В	В	В	В	В
	Special woven fabrics Impregnated, coated, covered or laminated textile fabrics Knitted or crocheted fabrics Umbrella, sun umbrellas etc. and parts thereof Articles of stone, plaster, cement etc. Ceramic products Glass and glassware Natural or cultured pearls, precious/semi-precious stones etc. Iron and steel Articles of iron or steel Lead and	cables and articles thereof Special woven fabrics Impregnated, coated, covered or B laminated textile fabrics Knitted or crocheted B fabrics Umbrella, sun umbrellas etc. and parts thereof Articles of stone, plaster, cement etc. Ceramic products Glass and glassware Natural or cultured pearls, precious/semi-precious stones etc. Iron and steel B Articles of iron or steel Lead and	cables and articles thereof Special woven fabrics Impregnated, coated, covered or laminated textile fabrics Knitted or crocheted B fabrics Umbrella, sun umbrellas etc. and parts thereof Articles of stone, plaster, cement etc. Ceramic products Glass and glassware Natural or cultured pearls, precious/semi-precious stones etc. Iron and steel B B B Articles of iron or steel Lead and B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B Articles of iron or steel Lead and B B B B	cables and articles thereof Special woven fabrics Impregnated, coated, covered or B B B B B B B B B B B B B B B B B B	cables and articles thereof Special woven fabrics Impregnated, coated, covered or B B B B B B B B B B B B B B B B B B	cables and articles thereof Special woven fabrics B	cables and articles thereof Special woven	cables and articles thereof Special woven fabrics B	cables and articles thereof Special woven fabrics B	cables and articles thereof Special woven fabrics B

79	Zinc and articles thereof	L	L	L	L	L	L	L	L	L	L
80	Tin and articles thereof	В	В	В	В	В	В	В	В	В	В
82	Tools, implements, cutlery, spoons and forks, of base metal; parts thereof of base metal	В	В	В	В	В	В	В	В	В	В
83	Miscellaneous articles of base metal	В	В	В	В	В	В	В	В	В	В
84	Nuclear reactors, boilers, machinery	В	В	В	В	В	В	В	В	В	В
85	Electrical machinery and equipment	В	В	В	В	В	В	В	В	В	В
86	Railway or tramway locomotives, etc.	В	В	В	В	В	В	В	В	В	В
88	Aircraft, spacecraft, railways or tramway track fixtures and fittings and parts thereof	В	В	В	В	В	В	В	В	В	В
89	Ships, boats and floating structure	В	В	В	В	В	В	В	В	В	В

90	Optical, photographic, cinematographi c, medical/surgic al instruments etc.	В	В	В	В	В	В	В	В	В	В
91	Clocks and watches and parts thereof	В	В	В	В	В	В	В	В	В	В
92	Musical instruments	В	В	В	В	В	В	В	В	В	В
94	Furniture; bedding, mattresses, etc.	L	L	L	L	L	L	L	L	L	L
95	Toys, games and sports requisites; parts and accessories thereof	В	В	В	В	В	В	В	В	В	В
96	Miscellaneous manufactured articles	В	В	В	В	В	В	В	В	В	В
99	Commodities not specified	L	L	L	L	L	L	L	L	L	L

Table 7.1 (Part C): Commodities with supply constraints in select African countries but with potential for Indian investment, 2010-2012 (avg.) and 2015-2017 (avg.)

14	Vegetable plaiting materials	L	L	L	L	L	L	L	L	L	L	L	L	L
15	Animal or vegetable fats/oils and their cleavage products	В	В	В	В	В	В	В	В	В	В	В	В	В
28	Inorganic chemicals ; organic or inorganic compoun ds of precious metals	L	L	L	L	L	L	L	L	L	L	L	L	L
29	Organic chemicals	В	В	В	В	В	В	В	В	В	В	В	В	В
30	Pharmace utical products	В	В	В	В	В	В	В	В	В	В	В	В	В
31	Fertilizers	В	В	В	В	В	В	В	В	В	В	В	В	В
32	Tanning or dyeing extracts; tannins and their deri. dyes, pigments and other colouring matter	В	В	В	В	В	В	В	В	В	В	В	В	В

	4.11													
35	Albumino idal substance s; modified starches; glues; enzymes	В	В	В	В	В	В	В	В	В	В	В	В	В
37	Photogra phic /cinemato graphic goods	В	В	В	В	В	В	В	В	В	В	В	В	В
38	Miscellan eous chemical products	В	В	В	В	В	В	В	В	В	В	В	В	В
39	Plastics and articles thereof	В	В	В	В	В	В	В	В	В	В	В	В	В
40	Rubber and articles thereof	В	В	В	В	В	В	В	В	В	В	В	В	В
47	Pulp of wood or of other fibrous cellulosic material; waste and scrap of paper/pap erboard	В	В	В	В	В	В	В	В	В	В	В	В	В

48	Paper and paperboar d; articles of paper pulp, of paper or of paperboar d	В	В	В	В	В	В	В	В	В	В	В	В	В
49	Printed books, newspape rs, pictures etc. of the printing industry; manuscri pts, typescript s and plans	В	В	В	В	В	В	В	В	В	В	В	В	В
50	Silk	В	В	В	В	В	В	В	В	В	В	В	В	В
53	Other vegetable textile fibres; paper yarn and woven fabrics of paper yarn	В	В	В	В	В	В	В	В	В	В	В	В	В

54	Man- made filaments	В	В	В	В	В	В	В	В	В	В	В	В	В
55	Man- made staple fibres	В	В	В	В	В	В	В	В	В	В	В	В	В
56	Wadding, felt and nonwove ns; special yarns; twine, cordage, ropes and cables and articles thereof	В	В	В	В	В	В	В	В	В	В	В	В	В
58	Special woven fabrics	В	В	В	В	В	В	В	В	В	В	В	В	В
59	Impregna ted, coated, covered or laminated textile fabrics	В	В	В	В	В	В	В	В	В	В	В	В	В
60	Knitted or crocheted fabrics	В	В	В	В	В	В	В	В	В	В	В	В	В

66	Umbrella, sun umbrellas etc. and parts thereof	В	В	В	В	В	В	В	В	В	В	В	В	В
68	Articles of stone, plaster, cement etc.	В	В	В	В	В	В	В	В	В	В	В	В	В
69	Ceramic products	В	В	В	В	В	В	В	В	В	В	В	В	В
70	Glass and glassware	В	В	В	В	В	В	В	В	В	В	В	В	В
71	Natural or cultured pearls, precious/s emi- precious stones etc.	E	E	E	E	E	E	E	E	E	Е	E	E	E
72	Iron and steel	В	В	В	В	В	В	В	В	В	В	В	В	В
73	Articles of iron or steel	В	В	В	В	В	В	В	В	В	В	В	В	В
78	Lead and articles thereof	В	В	В	В	В	В	В	В	В	В	В	В	В
79	Zinc and articles thereof	L	L	L	L	L	L	L	L	L	L	L	L	L

80	Tin and articles thereof	В	В	В	В	В	В	В	В	В	В	В	В	В
82	Tools, implemen ts, cutlery, spoons and forks, of base metal; parts thereof of base metal	В	В	В	В	В	В	В	В	В	В	В	В	В
83	Miscellan eous articles of base metal	В	В	В	В	В	В	В	В	В	В	В	В	В
84	Nuclear reactors, boilers, machiner	В	В	В	В	В	В	В	В	В	В	В	В	В
85	Electrical machiner y and equipmen t	В	В	В	В	В	В	В	В	В	В	В	В	В
86	Railway or tramway locomotiv es, etc.	В	В	В	В	В	В	В	В	В	В	В	В	В

88	Aircraft, spacecraft , railways or tramway track fixtures and fittings and parts thereof	В	В	В	В	В	В	В	В	В	В	В	В	В
89	Ships, boats and floating structure	В	В	В	В	В	В	В	В	В	В	В	В	В
90	Optical, photograp hic, cinemato graphic, medical/s urgical instrumen ts etc.	В	В	В	В	В	В	В	В	В	В	В	В	В
91	Clocks and watches and parts thereof	В	В	В	В	В	В	В	В	В	В	В	В	В
92	Musical instrumen ts	В	В	В	В	В	В	В	В	В	В	В	В	В
94	Furniture; bedding,	L	L	L	L	L	L	L	L	L	L	L	L	L

	mattresse													
	s, etc.													
	Toys,													
	games													
	and sports													
95	requisites	В	В	В	В	В	В	В	В	В	В	В	В	В
93	; parts	Б	Ъ	Ъ	Ъ	Ъ	Ъ	Ъ	Ъ	ь	Ъ	Ъ	В	Ъ
	and													
	accessori													
	es thereof													
	Miscellan													
	eous													
96	manufact	В	В	В	В	В	В	В	В	В	В	В	В	В
	ured													
	articles													
	Commodi													
99	ties not	L	L	L	L	L	L	L	L	L	L	L	L	L
	specified													

Note: 'E' denotes the presence of a supply constraint in the concerned country only during the earlier time-period 2010-2012 (avg.); 'B' denotes the presence of a supply constraint in the concerned country during both time periods 2010-2012 (avg.) and 2015-2017 (avg.); and 'L' denotes the presence of a supply constraint in the concerned country only in the later time period 2015-2017 (avg.).

The above analysis relates to likely supply constraints in the merchandise export sectors of selected African and Asian economies. This analysis is replicated to identify potential supply constraints for the same countries but in the case of their services exports to India.

Like in the case of goods, none of the eight Asian LDCs were found to have a supply constraint in any of the services sectors in either of the time periods, even though there were several services where India's global imports exceeded global exports of the concerned Asian country. This finding is again due to the effect of the Asian region's higher global exports of services, alluding to the existence of supply-side capacities in the immediate neighbourhood.

Similarly, services found to exhibit a likely lack of supply capacity are also found to be the same across all the selected African countries. Accordingly, Table 7.2 identifies services in the concerned African countries with investment potential for India. Here too, 'E' denotes the presence of a supply constraint in the concerned country only during the earlier time-period 2000-2002 (avg.); 'B' denotes the presence of a supply constraint in the concerned country during both time periods 2000-2002 (avg.) and 2010-2012 (avg.); and 'L' denotes the presence of a supply constraint in the concerned country only in the later time period 2010-2012 (avg.).

The results show that African countries' limited capacity to supply exists for five services sectors - insurance, financial, computer, royalties, and other business services (OBS). The greater prevalence of supply constraints in both time periods in Table 7.2 suggests that the more prominent sectors for India's investment in African countries include financial services and computer and information services. However, it may be noted that India too restricts imports of financial and insurance services and puts many barriers to entry for the private sector in its telecommunications sector. Meanwhile, OBS and insurance services are also integral to building supply-side capacities, but constraints in these sectors were identified only for the period of 2010-2012 (avg.) and 2000-2002 (avg.), respectively.

Table 7.2: Services sectors in African countries with supply constraints but with potential for Indian investment, 2000-2002 (avg.) and 2010-2012 (avg.)

Country / Services	Insurance services	Financial services	Computer and information services	Royalties and license fees	OBS
Angola	Е	В	В	В	L
Benin	E	В	В	В	L
Burkina Faso	E	В	В	В	L
Burundi	E	В	В	В	L
Central African Republic	Е	В	В	В	L
Chad	Е	В	В	В	L
Democratic Republic of Congo	E	В	В	В	L
Djibouti	Е	В	В	В	L
Ethiopia	E	В	В	В	L
Gambia	Е	В	В	В	L
Ghana	Е	В	В	В	L
Guinea	Е	В	В	В	L
Guinea-Bissau	Е	В	В	В	L
Kenya	Е	В	В	В	L

Lesotho	Е	В	В	В	L
Liberia	E	В	В	В	L
Madagascar	E	В	В	В	L
Malawi	Е	В	В	В	L
Mali	Е	В	В	В	L
Mauritania	E	В	В	В	L
Mozambique	Е	В	В	В	L
Niger	E	В	В	В	L
Nigeria	Е	В	В	В	L
Rwanda	Е	В	В	В	L
Senegal	Е	В	В	В	L
Sierra Leone	Е	В	В	В	L
Sudan	Е	В	В	В	L
Tanzania	Е	В	В	В	L
Togo	Е	В	В	В	L
Uganda	Е	В	В	В	L
Zambia	Е	В	В	В	L

Source: OECD-WTO BaTIS; own calculations

Note: 'E' denotes the presence of a supply constraint in the concerned country only during the earlier time-period 2000-2002 (avg.); 'B' denotes the presence of a supply constraint in the concerned country during both time periods 2000-2002 (avg.) and 2010-2012 (avg.); and 'L' denotes the presence of a supply constraint in the concerned country only in the later time period 2010-2012 (avg.).

Chapter 8

India's outward investment in African and Asian countries

Over the past few decades, foreign direct investment (FDI) has grown strongly as a major form of international capital transfer. It is considered as a key element for a country's economic integration and a key source of finance particularly for many LDCs. More importantly, developing and emerging-market economies, such as India, who earlier were mere recipients of FDI have now become important investors abroad, including in a number of African as well as some Asian LDCs.

Against this background, this chapter attempts to provide a deeper understanding of India's investment in the chosen African and Asian countries during the time period from 2010 to 2018 based on data sourced from 'fDi Markets'. Note that information reported in this database only pertains to announced greenfield investment projects. The countries for which data on greenfield investment are available in fDI Markets over 2010-2018 include: Angola, Burkina Faso, Burundi, Chad, Democratic Republic of Congo, Ethiopia, Ghana, Kenya, Liberia, Madagascar, Malawi, Mozambique, Niger, Nigeria, Rwanda, Senegal, Sierra Leone, Sudan, Tanzania, Uganda, Zambia, Bangladesh, Bhutan, Cambodia, Laos, Myanmar, Nepal and Yemen (i.e., 28 countries' data available out of total selected 41).

The chapter reports data on total greenfield investment by India in the 21 African and 7 Asian countries cumulative over the time period from 2010 to 2018 in the Tables 8.1 and 8.2, respectively. Tables 8.3 and 8.4 report respective data on sector-wise distribution of total greenfield investment by India in these countries cumulative over the same time period.

8.1 India's total greenfield investment in selected countries

The key findings from the analysis of India's announced greenfield investment (from Tables 8.1 and 8.2) are as follows:

1. Top 5 African host countries for Indian greenfield investment

Mozambique (USD 6.5 billion), Nigeria (USD 2.8 billion), Ghana (USD 2.0 billion), Ethiopia (USD 1.8 billion), and Kenya (USD 1.2 billion)

2. Top 5 Asian host countries for Indian greenfield investment

Bangladesh (USD 7.5 billion), Myanmar (USD 1.5 billion), Nepal (USD 360.6 million), Bhutan (USD 354.7 million), and Yemen (USD 326.2 million)

Over the cumulative time period 2010-2018, India announced greenfield investment worth US\$19.3 billion in the selected sample of 21 African countries (see Table 8.1). Interestingly, the distribution of investment in these countries was highly skewed. Thus, the top 3 African host countries for Indian greenfield investment during 2010-18 have been Mozambique (with a cumulative value of \$6.5 billion), Nigeria (\$2.8 billion) and Ghana (\$2 billion); together these countries account for 58.9% of total Indian greenfield investment to these African countries. These countries are followed by Ethiopia, Kenya and even Zambia who have also been able to attract a fair amount of Indian greenfield investment, accounting for 9.3%, 6.3% and 4.7%, respectively, of India's total outbound greenfield investment to Africa. Thus, these six countries have together accounted for about 79% of India's total outbound greenfield investment to Africa. In the case of the remaining 15 countries, the value of greenfield

investment received has been much lower, with some countries having shares around just 3% including Liberia, Senegal and Tanzania. For others, the shares have been below 2%. The smallest recipients of Indian greenfield investment have been Burundi, Angola and Sudan with combined shares of just 0.1% (\$20 million received in value terms) in India's total outward greenfield investment in Africa. These countries thus need to urgently improve their weak investment climate by removing restrictions on a number of sectors in the form of local content requirements, and removing bar on investment in petroleum, diamonds and financial services as in the case of Angola.

Table 8.1: Announced greenfield investment by India in 21 African countries

Country	Announced greenfield investment (in US\$ million) over the period 2010-2018 (Cumulative)
Mozambique	6535.1
Nigeria	2775.9
Ghana	2029.9
Ethiopia	1798.2
Kenya	1210.3
Zambia	915.4
Liberia	653.4
Senegal	653.4
Tanzania	649.5
Uganda	508.8
Madagascar	350.3
Rwanda	286.4
Sierra Leone	244.1
Democratic Republic of Congo	197.7
Malawi	129.1
Niger	104.5
Burkina Faso	98.6
Chad	98.6
Sudan	13.8
Angola	3
Burundi	2.9
TOTAL	19258.9

Source: fDI Markets; own calculations

In contrast, the seven Asian countries (see Table 8.2) received \$10.4 billion of cumulative greenfield investment from India during the time period 2010-18. Bangladesh received most of India's greenfield investment at \$7.5 billion, with a share of more than 70% in India's cumulative greenfield investment in these countries. The remaining six countries received 28% and within this group, Myanmar received the most at \$1.5 billion (share of 15%). India invested more than \$300 million in each of Bhutan, Yemen and Nepal (each having share of about 3%). However, Laos received the least greenfield investment from India worth \$92 million, amounting to just 0.88% of total greenfield investment by India in these seven Asian LDCs. It is reported that Lao PDR's investment climate is very challenging with weak judicial system, lack of adequate supporting financial services, etc. Cambodia too has issues related to lack of

proper infrastructure and finance with high logistics costs which hinders investment flows into the country. In fact, Cambodia puts restrictions on FDI in some sectors such as rice milling, gemstone processing, publishing, radio and television, etc. with polices to favour local investors over foreign ones. These constraints may be responsible for the stylized facts presented here.

Table 8.2: Announced greenfield investment by India in 7 Asian countries

Country	Announced greenfield investment (in US\$ million) over the period 2010-2018 (Cumulative)
Bangladesh	7467.7
Myanmar	1532.6
Bhutan	354.7
Yemen	326.2
Nepal	323.4
Cambodia	276.3
Laos	91.7
TOTAL	10372.6

Source: fDI Markets; own calculations

8.2 India's sector-wise greenfield investment in selected countries

While selecting the top 3 sectors for India's greenfield investment in selected 21 African countries, only ten countries could report such information. These include Democratic Republic of Congo, Ethiopia, Ghana, Kenya, Mozambique, Nigeria, Rwanda, Tanzania, Uganda and Zambia (see Table 8.3). This implies that India currently invests only in one or two sectors in the other eleven countries. Thus, there may be untapped potential in these countries to attract more investment from emerging economies like India, and that too in a wider range of sectors. In contrast, except for Laos and Yemen, the top 3 sectors for India's greenfield investment in Asian LDCs could be identified easily (see Table 8.4).

Detailed tables for all 28 countries for which data are available in fDi Markets are enclosed in 'Annex C' where India's greenfield investment is reported sector-wise for each country, along with the name of the major investor company in the sector whose share in total greenfield investment is very high.

The following key results from table 8.3 portray India's sector-wise greenfield investment in the selected African countries:

- (i) Of the ten countries for which the top 3 sectors could be identified, the communication sector is found in the top 3 sectors for seven countries: Congo (\$145.7 million); Ghana (\$145 million); Kenya (\$174.8 million); Nigeria (\$979.9 million); Rwanda (\$103.7 million); Tanzania (\$108.90 million) and Uganda (\$265.6 million). Note that in the fDI Markets database, the communication sector covers two main areas, namely, communications equipment and wireless telecommunication carriers.
- (ii) Apart from the communication sector, the other top 3 sectors in these 10 African countries are: real estate (Democratic Republic of Congo, Kenya); software & IT services (Democratic Republic of Congo); textile (Ethiopia); food & tobacco (Ethiopia); paper,

printing & packaging (Ethiopia); chemical (Ghana, Nigeria, Zambia); automotive OEM (Ghana); warehousing (Kenya, Mozambique); coal, oil & gas (Mozambique, Zambia); beverages (Mozambique); non-automotive transport OEM (Nigeria, Uganda); healthcare (Rwanda, Tanzania, Uganda); pharmaceuticals (Rwanda); industrial equipment (Tanzania); and renewable energy (Zambia).

Besides these ten countries, India has also invested heavily in two main sectors of other African countries, namely, Burkina Faso, Chad, Madagascar, Malawi, Niger and Sierra Leone. Communications is one of these two sectors, followed by business services (in case of Burkina Faso, Chad, Niger); minerals (Madagascar); software & IT services (Malawi) and metals (Sierra Leone).

Interestingly, few Indian firms have solely invested in just one sector in the case of five African LDCs. These include the industrial equipment sector in Angola (\$3 million), in Burundi (healthcare - \$2.92 million), Liberia & Senegal (coal, oil & gas - \$653.4 million each), and Sudan (communications - \$94.1 million). Being resource-rich countries, Liberia and Senegal have received a good amount of capital investment from India in coal, oil & gas sector are natural, liquified & compressed gas; fossil fuel electric power; other petroleum & coal products and coal mining. In contrast, despite being amongst the top recipients of FDI in Africa, Nigeria does not have any greenfield investment in this sector owing to preferential treatment given to Nigerian investors over foreign investors. Kenya too is considering a bill on local content requirements in this sector, which may further deter foreign investment.

In fact, restrictions placed by a number of countries on investment in certain sectors could also be responsible for India's limited sectoral investment in those countries. For instance, Ghana puts investment prohibitions in the sectors of petty trading, automobile rental services, lotteries, beauty salons, printing of recharge scratch cards for telecom services subscribers, production of exercise books and stationery, retail of finished pharmaceutical products, etc. as well as bans small-scale mining license for operations on area less than 25 acres. Ethiopia bans investments in banking, insurance and micro-finance sectors, as well as restricts those involved in retail, wholesale or service enterprises such as in transport services, shipping, printing, etc. FDI in its telecommunications sector is allowed only if done in partnership with its government. This could be one reason why India has not invested in this sector of Ethiopia.

Table 8.3: Top three sectors in African countries attracting greenfield investment from India

Country	Top three sectors	Greenfield investment (in US\$ million) Cumulative 2010-2018
Angola	Industrial equipment	3
Burkina Faso	Communications	94.1
	Business services	4.5
Burundi	Healthcare	2.92
Chad	Communications	94.1
	Business services	4.5
Democratic	Communications	145.7
Republic of	Real estate	24.0
Congo (DRC)	Software & IT services	17.0
Ethiopia	Textiles	1307.2
_	Food & tobacco	194.3

	Paper, printing & packaging	150.0
Ghana	Chemicals	1500.0
	Automotive OEM	180.0
	Communications	145.0
Kenya	Real estate	244.9
	Warehousing	220.3
	Communications	174.8
Liberia	Coal, oil & gas	653.4
Madagascar	Communications	282.3
_	Minerals	67.9
Malawi	Communications	112.1
	Software & IT services	17
Mozambique	Coal, oil & gas	6255.6
-	Warehousing	166.6
	Beverages	38.4
Niger	Communications	100
J	Business services	4.5
Nigeria	Chemicals	1326.5
J	Communications	979.9
	Non-automotive transport OEM	145.2
Rwanda	Healthcare	107.9
	Communications	103.7
	Pharmaceuticals	65.0
Senegal	Coal, oil & gas	653.4
Sierra Leone	Metals	150
	Communications	94.1
Sudan	Software & IT services	13.8
Tanzania	Industrial equipment	119.3
	Communications	108.9
	Healthcare	72.9
Uganda	Communications	265.6
S	Non-automotive transport OEM	72.6
	Healthcare	56.6
Zambia	Chemicals	266.5
	Renewable energy	231.7
	Refle wable effergy	

Source: fDI Markets; own calculations

In the case of the Asian countries (see Table 8.4), the sectors are not similar unlike for the African countries. For instance, the communication sector is not among the top three sectors for a majority of the countries, except in Nepal where the sector attracted investment worth \$79.4 million. In countries like Bangladesh and Laos, in contrast, the telecommunication sector is generally highly regulated with a number of restrictions.

We find that Indian firms have invested heavily in textiles (Bangladesh); automotive OEM (Bangladesh, Cambodia); business services (Bangladesh); coal, oil and gas (Myanmar); renewable energy (Bhutan, Yemen); followed by rubber (Laos), financial services (Myanmar, Bhutan, Nepal) and even in wood products (Laos, Myanmar). India's investment in the textiles sector has only been in Bangladesh (\$2.3 billion) and Bhutan (much lower at \$18 million), which could be surprising given the potential to form regional supply chains in this sector as identified by UNCTAD, Commonwealth Secretariat and CWTO (2010).

Furthermore, India has invested in few rare sectors such as pharmaceuticals and hotels & tourism in the case of Cambodia and rubber in Laos (mainly dedicated to tyre making firms). Also, Nepal's top sector to attract Indian investment (\$103 million) has been the non-automotive transport OEM sector. Meanwhile, Myanmar's coal, oil & gas sector attracted \$1.2 billion during the time period 2010-18.

Table 8.4: Top three sectors in Asian countries attracting greenfield investment from India

Country	Top three sectors	Greenfield investment (in USD million) Cumulative 2010-2018
Bangladesh	Textiles	2289.58
	Automotive OEM	1564.8
	Business services	1000
Bhutan	Renewable energy	267.0
	Financial services	69.6
	Textiles	18.1
Cambodia	Automotive OEM	256.4
	Pharmaceuticals	18.4
	Hotels & tourism	1.5
Lao People's Democratic	Rubber	81.7
Republic	Wood products	9.9
Myanmar	Coal, oil & gas	1238.2
-	Financial services	179.9
	Wood products	64.6
Nepal	Non-automotive transport	102.5
	OEM	
	Communications	79.4
	Financial services	68.6
Yemen	Renewable energy	326.2

Source: fDI Markets; own calculations

8.3 Analysis of trade-investment linkages

Using the values of African and Asian countries' exports of goods and services to India for the latest available time periods - 2015-17 and 2010-12, respectively - (as reported in Annex A and B tables), this section draws their parallel linkages with India's greenfield investment in these countries. The objective is to identify, without attributing any causality, whether the sectors that export more to India are also those that are major recipients of Indian greenfield investment. Overall, significant FDI-trade linkages could be traced between India and almost 6 African and 3 Asian countries (of the 28 countries in which India announced greenfield investment as documented in fDI Markets over 2010-2018). The links are found to be significant in the case of communications services and other business services (OBS). For

¹³ It is also possible, indeed likely, that Indian investment in these countries may generate exports to third countries, but we do not examine such instances in this report as it is focused on the Indian market. This does not however imply that Indian outward investment in these countries is largely to serve demand in the Indian market and is basically resource-seeking. In fact, it would have been useful to compare the observed stylized facts against investment originating from China as well as explore the links between Indian outward investment and India's exports of goods and services to these countries. However, both these aspects are beyond the scope of this report.

commodities, prominent sectors for linkages include coal, metals, food, automotive OEM and chemicals (in the case of Africa), and textiles and wood products (in the case of Asia).

Some specific illustrations of these linkages are as follows:

- Over the period 2010-18, India's cumulative greenfield investment in Zambia in metals and automotive OEM (more than \$95 million in each case) accounted for shares of about 11% in India's total greenfield investment in that country. Zambia's exports to India are concentrated in base metals (where unrefined copper exports worth \$606 million account for 86.9% of India's total merchandise imports from that country).
- Indian firms, namely Karli International, Karuturi Global and Ruchi Group, have invested significant amounts of \$194 million cumulative over 2010-18 in Ethiopia's food & tobacco sector (with a share of 11% in India's total greenfield investment in that country). At the same time, Ethiopia's exports of various types of beans, turmeric and oil seeds to India have been significant \$52 million, amounting to 82% of India's total merchandise imports from that country.
- India's greenfield investment in Mozambique has largely been concentrated in coal, oil & gas sector with a share of 95.7% in India's total greenfield investment in that country cumulative over 2010-18, with Bharat Petroleum investing \$6255 million. Mozambique's \$404 million exports of coal to India with a share of 68.8% in India's total merchandise imports from that country show strong linkages.
- During 2010-2018, India's ABG International invested \$150 million (with a share of 61.5%) in Sierra Leone's metal sector, particularly in aluminum's production and processing. The sector exported base metals (majorly ferrous and aluminum wastes and scrap) worth \$9.7 million to India over 2015-2017 (avg.), with a share of almost 47% in India's total merchandise imports from that country, along with exports of aluminium ores (\$3.6 million with a share of 17.5%).
- India invested \$166 million in Kenya's chemical sector cumulative over 2010-18 representing 13.6% of its total outbound greenfield investment in that country. Kenya has been a significant exporter of inorganic chemicals, i.e., disodium carbonates, to India (of \$20 million) accounting for 19% of India's total merchandise imports from that country over 2015-2017 (avg.).
- India's Bharti Group invested \$145 million in Congo's communication services sector cumulative over 2010-18 with a notable share of 73.7%. Congo exported \$5.5 million of this service to India over 2010-12 (avg.) contributing 10% of the country's total services exports to India.

In the case of Asian LDCs, greater FDI-export linkages are found between India and Bangladesh. For instance, cumulative over 2010-2018, India's Arvind Ltd. invested \$2.3 billion in Bangladesh's textiles sector with a share of 30.6% in India's total greenfield investment in that country. Meanwhile, Bangladesh exported textiles and clothing products (fibres, yarn and woven fabrics of jute, men's/boys' shirts and trousers, made-up textile items such as packaging bags) worth \$229 million to India over 2015-17 (avg.), representing 33% of India's total merchandise imports from that country. India also invested \$1000 million in Bangladesh's business sector (share of 13.4%), while the latter exports \$35 million OBS to India (share of 40.8% in that country's total services exports to India over 2010-12 (avg.)).

Other traceable linkages include (i) India's cumulative greenfield investment in Lao's wood sector (11% share) and Lao's exports to India (worth \$20 million, 9% share). (ii) India's

cumulative greenfield investment in Nepal's communications sector (25% share) and Nepal's export of communication services to India (worth \$25 million, with a share of 13.8%).

CHAPTER 9

Quadrant analysis of goods and services

The current level of investment by India in the selected African and Asian countries measures one set of the opportunities that have opened up between India and these countries. But how much more potential can be created and tapped, and where investment should actually be directed depends on these countries' ability to connect with India. The level of India's imports of various goods and services from countries all over the world provides an estimate of the size of the Indian market that can be penetrated by these countries. However, the extent to which these countries can access the Indian market also depends on their export competitiveness and on their capacity to produce goods and services acceptable in the Indian market. Quadrant analysis is thus used to identify these countries' goods and services exporting opportunities based on the BRCA index and export share in India's imports (as already described in Chapter 3). A higher value of the BRCA index accompanied by above-average import shares classify goods and services under Quadrant A while those with the lowest values of the two indicators are placed under Quadrant D. From a policy perspective, the focus is on Quadrant B which identifies products/services with below-average shares in Indian imports but where the exporting countries are competitive, thereby suggesting the need for proactive trade promotion.

For analysis of merchandise trade, data for all the 41 African and Asian countries are again sourced from UN Comtrade over the two time-periods 2010-12 and 2015-17. The quadrant analysis is undertaken to identify goods at a disaggregated level, i.e. the HS 6-digit level. Note that as the BRCA index and shares in India's imports are calculated using three-year averaged trade values, non-reported data are taken as zero so that shares could lie between 1-100% and BRCA (after normalization) lies between -1 to +1. For analysis of services trade, data are sourced from OECD-WTO BaTIS, which provides data for 38 countries (except Eritrea, Somalia and Timor-Leste) for the time-period 2000-02 and 2010-12.

9.1 Quadrant analysis of merchandise trade

Tables 9.1 and 9.2 report the four quadrants with count of products identified therein during the periods 2010-12 and 2015-17 for selected African and Asian countries, respectively.

9.1.1 Analysis of African goods exports

The analysis reveals that a majority of the goods at the HS 6-digit level during 2010-12 and 2015-17 for the 33 African countries belonged to Quadrants B and D (see Table 9.1). During 2010-12, the number of goods in Quadrant B were relatively higher for several countries and have subsequently declined. During 2015-17, the maximum number of goods were identified under Quadrant D (BRCA<0 and low import shares), while Quadrant C (high import shares but BRCA less than zero) continue to record the least number of goods.

The analysis reveals that few African countries have a high comparative advantage in exporting to India with significant shares in India's imports, and that too in a small number of goods as identified under Quadrant A. Kenya exported the maximum (39 goods) during 2010-12, followed by Ghana, Nigeria and Tanzania (25). However, the number of items declined for each of these during 2015-17, such as up to 16 in the case of Kenya, and from 26 to 7 in the case of Ghana, etc. Even so, during 2015-17, Tanzania recorded the maximum number of items (21) which it has been competitively exporting to India, followed by Kenya, Nigeria, Senegal

(13) and Madagascar (11). Better performing items in this quadrant (based on high shares in India's imports and greater levels of BRCA) country-wise during 2015-17 include: dried leguminous vegetable, waste or scrap of cast iron, certain essential oils, and sawn/chipped wood in case of Tanzania; tin ores and concentrates, mineral item like fluorspar, disodium carbonate (chemical), as well as trees, shrubs and bushes (whether grafted or not) in case of Kenya; iron and non-alloy steel ingots, and turmeric (curcuma) for Nigeria; chemical such as phosphoric and polyphosphoric acid, and raw skins of sheep or lambs (without wool) for Senegal; and mica waste, natural graphite (other than in powdered form), whole cloves, and dried and shelled beans in case of Madagascar. Notably, Madagascar holds over 90% share in India's total imports of mica waste. As these products identified in quadrant A already have a good demand in India, they do not require much change in policies, apart from just preventing a loss in their comparative advantage.

Quadrant B is the major focus area of this study due to identification of commodities in which the selected African and Asian countries have a competitive advantage, but with less than average shares in India's imports thereby necessitating proactive policy measures. Kenya exported the maximum number of goods, as identified under this quadrant, during both the time periods, despite a subsequent fall in the number of goods from about 250 during 2010-12 to 98 during 2015-17. Nigeria and Ghana recorded the second and third highest number of commodities, respectively, during 2010-12. But as the number of goods drastically declined for these two from 245 to 50 and from 233 to 23, respectively, the two positions were taken over by Ethiopia and Uganda during 2015-17. Other competitive countries for exports to India during 2010-12 were Tanzania, Ethiopia and Madagascar; and Nigeria, Madagascar and Tanzania during 2015-17.

For these six prominent countries identified during 2015-17 under this quadrant, majority of their exported products are found in the category of general machinery, mechanical appliances and parts (HS 84) and electric machinery/equipment and their parts (HS 85). These are followed by few common products belonging to the categories of precision machinery (optical, photographic, medical or surgical instruments) and their parts (HS 90); articles of wood (HS 44); and raw hides, skins and leather (HS 41). Further, some sectors are majorly identified across specific top countries, such as, edible vegetables and certain roots and tubers (HS 07) in case of Kenya, Ethiopia and Uganda; plastic and its articles (HS 39) and paper, paperboard and their articles (HS 48) for Kenya and Ethiopia; and copper and its articles (HS 74) in case of Nigeria and Ethiopia (Table 9.1). Interestingly, most African exports in this quadrant can be classified as raw materials and intermediate goods such as iron ores, metal wastes, plastic in primary forms, waste papers, unworked diamonds, leather in raw/semi-finished form, seeds and nuts, parts of machinery, etc.

The detailed Table 1 reported in 'Annex D' shows the complete list of products identified under Quadrant B in both the time-periods 2010-12 and 2015-17 ranging from the lowest share in Indian imports (from 0%) to relatively higher shares (i.e. share up to the average level). These are the products for which the respective governments must design and implement appropriate trade policies to address prevailing NTBs. The change in policies should work in such a way that the identified competitive products for exports are also able to garner significant or improved share in India's imports in the coming years.

Table 9.1: Quadrant-wise count of goods

Country 2010-12	2015-17
-----------------	---------

	Quad. A	Quad. B	Quad. C	Quad. D	Quad. A	Quad. B	Quad. C	Quad. D
Angola	6	34	2	18	3	24	4	29
Benin	15	84	3	22	4	14	2	14
Burkina Faso	4	6	0	2	2	3	2	2
Burundi	1	10	0	1	1	3	0	1
Central African Republic	3	7	0	2	0	7	2	1
Chad	2	9	0	1	1	10	1	6
Democratic Republic of Congo	3	8	0	3	4	20	2	11
Djibouti	3	21	0	3	3	13	0	1
Eritrea	1	4	0	1	2	1	1	2
Ethiopia	9	168	6	55	8	92	3	70
Gambia	2	4	1	5	2	6	0	9
Guinea	5	51	2	51	1	19	1	13
Guinea-Bissau	1	4	0	21	1	4	0	2
Lesotho	4	20	0	7	2	2	0	2
Liberia	8	84	1	16	3	17	0	13
Madagascar	18	137	6	31	11	46	2	41
Malawi	4	48	2	10	4	34	2	24
Mali	8	28	1	8	2	7	0	8
Mauritania	3	16	0	3	4	8	0	3
Mozambique	15	85	3	31	5	31	2	31
Niger	0	23	3	18	2	15	2	27
Rwanda	2	6	0	0	2	10	0	6
Senegal	17	84	6	103	13	26	5	50
Sierra Leone	8	23	2	19	2	22	3	16
Somalia	4	15	0	1	1	10	1	3
Togo	7	43	2	22	6	18	0	11
Uganda	14	82	1	44	8	52	1	52
Tanzania	25	196	8	122	21	46	7	104
Zambia	3	34	2	25	2	38	2	54
Nigeria	25	245	16	277	13	50	13	203
Ghana	26	233	4	103	7	23	4	97
Kenya	39	254	9	243	16	98	7	182
Sudan	3	29	1	4	4	20	1	11

Source: UN Comtrade; own calculations

Quadrant D could also be a focal segment signifying potential future possibilities in terms of building competitiveness and supply capacity in different goods of the selected countries. The main African countries, based on the number of goods identified under this quadrant, remained the same during both the selected time periods, viz. Nigeria, Kenya and Tanzania (followed by Ghana and Senegal in 2010-12; and Ghana and Zambia during 2015-17). However, similar to Quadrant B, the number of products identified under Quadrant D have also significantly

declined over time for almost 50% of the countries, with the maximum fall reported in the case of countries such as Nigeria, Kenya and Senegal.

9.1.2 Analysis of Asian goods exports

According to the quadrant analysis, majority of goods for the selected Asian countries during 2010-12 belonged to Quadrant B, which represents lower export shares despite high competitiveness (see Table 9.2), followed by Quadrants D and A. Quadrants B and D continue to be the dominant ones even in 2015-17, but with few changes in the pattern, viz. number of goods declined in many cases in the former quadrant (except for Bhutan and Cambodia), with a rise in those identified under quadrant D (except in the case of Nepal, Yemen and even Timor-Leste). Quadrant C registers the least number of commodities in both time periods. The number of products also declined under Quadrant A over the two periods (except in the case of Bhutan, Cambodia, Laos and Timor-Leste). Thus, some Asian LDCs may have lost comparative advantage while exporting goods to India.

This sub-section further elaborates quadrant-wise results for different countries and products. Illustratively, in terms of already high competitiveness index and above-average shares in India's imports, the goods that are identified under Quadrant A in 2010-12 were majorly exported by Bangladesh (more than 100 goods at HS 6 digit level). This was followed by Nepal and Myanmar (goods exported in the range of 40-50). Although the number of goods competitively exported by Bangladesh declined during 2015-17 to around 87, it still reports higher BRCA index values in a majority of the items in both the time-periods (followed by Nepal and Cambodia during 2015-2017). On the other hand, products identified for Bangladesh have been rising under quadrants C and D, where BRCA have become less than zero. Thus, the Bangladesh government may need to maintain comparative advantage and higher shares in India in already significant export items. Notably, Bangladesh holds almost 100 percent share in the exports of dried/salted fish in brine, dried fish other than cod, jute and other textile bast fibres and their yarn, binder/baler twine of sisal or other textile fibres, and men's/boy's suits (not knitted/crocheted).

Apart from this, certain products from Nepal also held significantly high shares in India's imports during 2015-17, mainly foodstuffs (fresh/chilled garlic; buckwheat; fresh/chilled cabbage; vegetable material and wastes) and textile and clothing articles (multiple folded or cabled yarn of synthetic fibres; and men's/boy's ensembles, excluding knitted/crocheted); etc.

At the overall level, goods exported by Nepal have also declined, and the fall has occurred in all the quadrants keeping the pattern more or less same. Most of its products in 2015-17 lie in quadrant D (200), followed by B (around 150), implying low demand for its products in the Indian market as indicated by below- average shares less. Meanwhile, the number of goods exported by Cambodia have increased in all the quadrants, such as from 9 to 35 under quadrant A and from 61 to 150 in the case of quadrant D.

In the case of quadrant A, the least number of products have been recorded for East Timor/Timor-Leste (0 in 2010-12, while 2 in 2015-17). This country only has some products identified in quadrant B, so the concerned government may need to improve shares of such goods in India's imports (as listed in Table 2 of Annex D).

As compared to the African countries, the Asian LDCs have not experienced a sharp decline in the number of their products identified under Quadrant B, except in the case of Myanmar (with many of its products identified under Quadrant D during 2015-17). In contrast, as already stated above, Bhutan and Cambodia have recorded a significant rise in the number of goods identified in group B. Note that Annexure D (Table 2) lists Asia's goods identified at the disaggregated HS6-digit level for 2010-12 and 2015-17 periods under Quadrant B [products listed as per shares from 0 up to the average].

To elaborate the case of Quadrant B, where India's BRCA is above zero despite low shares, table 9.2 shows the top countries during 2010-12 as Bangladesh (540) and Myanmar (370). However, during 2015-17, the latter witnessed the greatest decline in number of its products (to 70) under this quadrant, while Nepal rose to the second position (despite a fall in the number of products from 173 to 154 over the two periods). Cambodia also continues to export significant number of items under this quadrant in both the periods. Like some top African countries, the selected Asian countries during 2015-17 are also mostly competitively exporting general machinery and parts (HS 84); electric machinery (HS 85); precision machinery and apparatus (HS 90); articles of paper (HS 48); and articles of wood (HS 44). Besides, some product categories are majorly skewed to certain specific countries, viz., products belonging to edible fruit and nuts; peel of citrus fruit or melons (HS 08); and coffee, tea, mate and spices (HS 09) are identified mainly for Myanmar and Nepal; while, cotton (HS 52); man-made staple fibers (HS 55); and wadding, felt and non-wovens, special yarns, twine, cordage, ropes and cables (HS 56) are identified in case of Bangladesh; and automotive and parts (HS 87) in case of Bangladesh and Cambodia (for detailed product list at HS 6-digit, see Annex D). Interestingly, many intermediate products have been exported by Asian LDCs in this quadrant which include parts and tools of machinery (including handtools), textile yarn and fabrics, semi-finished leather, etc. (the sophistication level of exported products has also risen from 2010-12 to 2015-17). At the same time, a significant number of apparel items have also been exported as final products such as sacks and bags, suits and shirts, etc. as well as processed food items, suit-cases, footwears, home furnishing items, etc.

Table 9.2: Quadrant-wise count of goods

		201	0-12		2015-17				
Country	Quad. A	Quad. B	Quad. C	Quad. D	Quad. A	Quad. B	Quad. C	Quad. D	
Bangladesh	109	541	52	232	87	485	83	304	
Bhutan	2	6	0	17	7	38	1	28	
Cambodia	9	124	16	61	35	131	40	151	
Laos	6	20	1	16	7	19	2	36	
Myanmar	40	373	11	167	25	71	4	245	
Nepal	54	173	18	216	47	154	9	200	
Timor- Leste	0	10	0	1	2	5	0	1	
Yemen	8	38	4	21	5	21	1	13	

Source: UN Comtrade; own calculations

9.2 Quadrant analysis of services sector exports

Tables 9.3 and 9.5 map the services exported by African and Asian countries to India across all the four groups in both the years 2000-02 and 2010-12. Tables 9.4 and 9.6 provide the count of services identified in each quadrant for the selected African and Asian countries,

respectively. It may be noted that the quadrant analysis of goods and services are not comparable, given the huge number of products at the HS 6-digit level and only 11 services sectors as covered in the OECD-WTO BaTIS database.

9.2.1 Analysis of Africa's services exports

The analysis shows very few African countries with their limited supply of services identified in Quadrant A in both time-periods 2000-02 and 2010-12 (see Table 9.4). In the former period, out of total eleven services, only Benin competitively exported maximum of four services which also have above-average shares in India's imports (travel, financial, computer and government services), followed by Tanzania and Uganda (three services each including travel and computers; transportation in case of former and financial services in case of latter). Meanwhile, two services were competitively exported by several countries such as Burkina Faso, Central African Republic, Ethiopia, Congo, Ghana, Kenya, Madagascar, Mozambique, Niger, Rwanda, Sierra Leone and Togo (mainly travel, transportation, financial, government, and even computer, insurance and construction services). In contrast, during 2010-12, the number of services in Quadrant A declined even for top exporting countries, such as, for Tanzania (with focus only on travel), for Benin (where government and computer services moved to Quadrants C and B respectively), for Uganda (lost competiveness in CI services), as well as for Kenya (only focus on transportation), for Congo (lost share in Indian imports for financial services), etc. Thus, in the second time-period, out of 31 countries, Benin, Burkina Faso, Burundi, Central African Republic, Guinea, Guinea-Bissau, Malawi, Mozambique, Nigeria, Rwanda, Sierra Leone, Uganda and Zambia were the major exporters to India supplying two services each (common ones include TN, TL, FS, OB, GS, and also CN).

Table 9.3 shows that for the selected African countries under study, a few services lie in Quadrant A, excluding personal and recreational services and royalties in any of the period. Travel and transportation have been the main services where many African countries are already competitively exporting to India. At the same time, other services mainly communication, computer and information, government, financial and construction have registered greater decline in the number of competitive exporting countries.

Services that lie in Quadrant C already have a good demand in India's imports despite the exporting countries having a comparative disadvantage in those services. Main countries during 2000-02 included Burkina Faso and Madagascar (with supply of three services each). Moreover, Burundi followed by Kenya and Togo have also been identified as major exporters of three-four services in 2010-12 including travel, communication, recreational, government, insurance, etc.

Major findings on the services identified under Quadrant B are as follows:

• Countries with the maximum number of services identified under this quadrant during 2000-02 were Djibouti and Sudan (4 services each with common ones as communications and OB services, along with transportation and insurance in case of former and financial and computer in case of latter), followed by Burkina Faso, Madagascar and Mauritania (three services each, with common ones as financial, insurance, transportation, etc.). Other important ones include Burundi, Ethiopia, Ghana, Guinea, Guinea-Bissau, Kenya, Lesotho, Mali, and Senegal who were competitive in case of two services each, i.e., transportation, computer and information, financial service, etc. (table 9.4).

- On the whole for entire selected African region, top services in 2000-02 for this quadrant included transportation (supplied by 13 countries that time, which have increased to 16 in 2010-12), computer and information services (by 10 countries in 2000-02, which although reduced to 6 later) and financial services (by 7-8 countries in both the periods), followed by other business service (number of supplying countries have increased from 5 to 17). In travel services case, most of the services lie in either A or D.
- During 2010-2012, the major countries in Quadrant B were Mali and Mauritania (supplying 4-5 identified services to India), with export competitiveness in services of transportation, other business, financial services and travel, etc. These are followed by other countries, such as, Angola, Benin, Chad, Guinea, supplying exactly the three services to Africa. These have higher comparative advantages while supplying services to India and their share could be increased in concerned services sector (see table 9.4). In short, top services in Quadrant B in 2010-12 thus include transportation, OB, financial, and computer services. These identified services have potentials and thus even few directed schemes by government could help to increase their shares in Indian imports.

Table 9.3: Mapping of services sectors across countries quadrant-wise (figures indicate number of exporting countries for each service under each quadrant)

Services		2000	-2002			2010-2012			
	Quad.	Quad.	Quad.	Quad.	Quad.	Quad.	Quad.	Quad.	
	A	В	C	D	A	В	C	D	
Communications services (CS)	3	3	7	15	0	0	9	21	
Computer and information services (CI)	6	10	0	9	1	6	1	16	
Construction (CN)	4	2	1	18	2	2	2	20	
Financial services (FS)	6	8	0	14	4	7	1	19	
Government services, n.i.e. (GS)	5	0	19	7	3	0	20	8	
Insurance services (IS)	2	3	1	21	2	1	1	25	
Other business services (OB)	1	5	3	22	3	17	1	10	
Personal, cultural and recreational services (PC)	0	0	5	18	0	0	8	15	
Royalties and license fees (RL)	0	0	0	20	0	0	0	22	
Transportation (TN)	9	13	1	8	11	16	1	3	
Travel (TL)	13	1	2	15	12	3	3	13	

Source: OECD-WTO BaTIS; own calculations

Table 9.4: Country-services quadrant-wise representation for Africa

	2000-2002				2010-2012			
Country	Quad.	Quad.	Quad.	Quad.	Quad.	Quad.	Quad.	Quad.
	A	В	C	D	A	В	C	D

				O (TNI				1
Angola	1 (CN)	1 (FS)	1 (CS)	8 (TN, TL, IS, CI, RL, OB, PC, GS)	1 (OB)	3 (TN, FS, CI)	2 (CS, CN)	5 (TL, IS, RL, PC, GS)
Benin	4 (TL, FS,CI, GS)	0	1 (CS)	6 (TN, CN, IS, RL, OB, PC)	2 (TL, FS)	3 (TN, CI, OB)	2 (CS, GS)	4 (CN, IS, RL, PC)
Burkina Faso	2 (TL, CI)	3 (TN, FS, CN)	3 (OB, PC, GS)	2 (CS, IS)	2 (CI, OB)	1 (TN)	2 (CN, GS)	5 (TL, CS, IS, FS, PC)
Burundi	1 (CI)	2 (TN, FS)	2 (IS, GS)	6 (TL, CS, CN, RL, OB, PC)	2 (FS, OB)	0	4 (IS, CI, PC, GS)	5 (TN, TL, CN, CS, RL)
Central African Republic	2 (TL, CS)	1 (TN)	1 (GS)	3 (CN, FS, OB)	2 (TL, GS)	1 (TN)	0	3 (CN, FS, OB)
Chad	0	1 (TN)	1 (GS)	5 (TL, CS, IS, FS, OB)	0	3 (TN, TL, OB)	1 (GS)	3 (CS, IS, FS)
Democratic Republic of the Congo	2 (IS, FS)	1 (CI)	1 (GS)	7 (TL, TN, CS, CN, RL, OB, PC)	1 (IS)	2 (TN, FS)	1 (GS)	7 (TL, CN, CS, CI, RL, OB, PC)
Djibouti	1 (GS)	4 (TN, CS, IS, OB)	0	2 (TL, FS)	0	2 (TN, OB)	1 (GS)	5 (TL, CS, IS, FS, RL)
Ethiopia	2 (CN, GS)	2 (TN, CI)	1 (CS)	6 (TL, IS, FS, RL, OB, PC)	1 (TN)	2 (FS, OB)	2 (CS, GS)	6 (TL, CN, IS, CI, RL, PC)
Gambia	1 (TL)	1 (CI)	1 (CS)	4 (TN, FS, OB, GS)	1 (TL)	1 (TN)	1 (CS)	6 (CN, IS, FS, CI, OB, GS)
Ghana	2 (TN, TL)	2 (FS, CI)	2 (PC, GS)	5 (CS, CN, IS, RL, OB)	1 (TL)	2 (TN, CI)	2 (PC, GS)	5 (CN, CS, FS, RL, OB)
Guinea	1 (CS)	2 (TN, FS)	1 (GS)	6 (TL, CN, IS, CI, OB, PC)	2 (CN, GS)	3 (TN, FS, OB)	2 (CS, PC)	3 (TL, IS, CI)
Guinea- Bissau	1 (TL)	2 (TN, CI)	1 (GS)	6 (CS, CN, IS, RL,	2 (TN, TL)	1 (OB)	2 (PC, GS)	6 (CN, CS, IS,

				OB, PC)				FS, CI,
Kenya	2 (TN, TL)	2 (CS, CI)	2 (OB, GS)	5 (CN, IS, FS, RL, PC)	1 (TN)	1 (CI)	3 (TL, OB, GS)	6 (CN, CS, IS, FS, RL, PC)
Lesotho	1 (CI)	2 (TN, OB)	1 (TL)	1 (GS)	1 (TL)	2 (TN, OB)	0	4 (CS, IS, FS, GS)
Liberia	1 (TN)	1 (FS)	1 (GS)	7 (TL, CS, CN, IS, CI, RL, OB)	1 (TN)	0	1 (GS)	8 (TL, CN, CS, IS, FS, CI, RL, OB)
Madagascar	2 (TN, OB)	3 (IS, FS, CI)	3 (TL, CN, PC)	2 (CS, GS)	1 (TN)	2 (FS, OB)	2 (TL, PC)	6 (CN, CS, IS, CI, RL, GS)
Malawi	1 (TL)	1 (OB)	1 (GS)	1 (TN)	2 (TN, TL)	1 (OB)	1 (GS)	3 (CS, IS, FS)
Mali	1 (CS)	2 (TN, CI)	1 (GS)	7 (TL, CN, IS, FS, RL, OB, PC)	0	5 (TN, TL, FS, CI, OB)	2 (CS, GS)	4 (CN, IS, RL, PC)
Mauritania	1 (FS)	3 (TN, TL, IS)	1 (GS)	4 (CS, CN, RL, OB)	0	4 (TN, IS, FS, OB)	1 (GS)	5 (TL, CN, CS, RL, PC)
Mozambique	2 (TN, TL)	0	1 (OB)	8 (CS, CN, IS, FS, CI, RL, PC, GS)	2 (TN, TL)	0	1 (GS)	8 (CS, CN, IS, FS, CI, RL, OB, PC)
Niger	2 (TN, FS)	0	1 (GS)	5 (TL, IS, CI, OB, PC)	1 (TN)	0	2 (FS, GS)	6 (TL, CS, IS, CI, OB, PC)
Nigeria	1 (CN)	1 (TN)	0	9 (TL, CS, IS, FS, CI, RL, OB, PC, GS)	2 (TN, CN)	1 (TL)	0	8 (CS, IS, FS, CI, RL, OB, PC, GS)
Rwanda	2 (IS, FS)	0	1 (GS)	8 (TL, TN, CS, CN, CI, RL, OB, PC)	2 (IS, FS)	1 (OB)	0	8 (TN, TL, CN, CS, CI, RL, PC, GS)
Senegal	1 (GS)	2 (CN, CI)	1 (CS)	7 (TN, TL, IS, FS, RL,	1 (GS)	2 (TN, OB)	1 (CS)	7 (TL, CN, IS,

				OB, PC)				FS, CI, RL, PC)
Sierra Leone	2 (TN, TL)	0	2 (PC, GS)	6 (CS, CN, IS, FS, RL, OB)	2 (TN, TL)	0	1 (GS)	7 (CS, CN, IS, FS, RL, OB, PC)
Sudan	1 (TN)	4 (CS, FS, CI, OB)	1 (GS)	5 (TL, CN, IS, RL, PC)	0	2 (TN, OB)	1 (GS)	7 (TL, CS, CN, IS, FS, CI, RL)
Tanzania	3 (TN, TL, CI)	0	1 (GS)	7 (CS, CN, IS, FS, RL, OB, PC)	1 (TL)	2 (CN, CI)	2 (TN, GS)	6 (CS, IS, FS, RL, OB, PC)
Togo	2 (CN, GS)	1 (TN)	1 (CS)	6 (TL, IS, FS, CI, OB, PC)	0	2 (TN, OB)	3 (TL, CS, PC)	5 (CN, IS, FS, CI, GS)
Uganda	3 (TL, FS, CI)	1 (OB)	2 (PC, GS)	5 (TN, CS, CN, IS, RL)	2 (TL, FS)	2 (CN, OB)	2 (PC, GS)	5 (TN, CS, IS, CI, RL)
Zambia	1 (TL)	0	2 (TN, CS)	8 (CN, IS, FS, CI, RL, OB, PC, GS)	2 (TN, TL)	1 (OB)	2 (CS, PC)	6 (CN, IS, FS, CI, RL, GS)

Source: OECD-WTO BaTIS; own calculations

9.2.2 Analysis of Asian services exports

As compared to African countries, a fewer number of services were identified in the case of Asian LDCs in the different quadrants. But here too, for a majority of the Asian countries, the concerned services sector lies in Quadrant D in both the time periods 2000-02 and 2010-12, followed by quadrants B and C; while Quadrant A has the least number of countries (see Tables 9.5 and 9.6). Thus, these countries still lack competiveness in exporting a number of services to India as compared to their goods exports.

As per services and their exporting countries identified under Quadrant A, Nepal and Cambodia were found to be supplying the maximum number of services (two services each including travel; while communications for Nepal and transportation in case of latter) to India during 2000-02, followed by Myanmar, Yemen, Bangladesh and Bhutan (exporting one service each, mostly travel, but construction in case of Bangladesh). In contrast, during 2010-2012, Nepal, Myanmar and Bhutan have competitively exported only travel services. On the other hand, Yemen and Laos did not export any services competitively under this quadrant in 2010-12. These findings from the quadrant analysis are consistent with those from services sector analysis done in Chapter 6.

Quadrant B includes services with BRCA index values greater than 0 but with low shares in Indian imports. The top services in which selected countries hold greater comparative

advantages in 2000-02 included computer and information, transportation, followed by construction and communications. But during 2010-12 (Table 9.5), the number of countries declined for all of these services due to movement into other quadrants (particularly Quadrant D), such as for communication, computers, construction and transportation services. In both the periods, other important services included travel, financial and OBS (supplied only by one country each, except for OBS where the number of countries has increased to two during 2010-12). On the other hand, government, insurance, recreational and royalties did not have any Asian country identified under this quadrant for any time period (these too have countries mostly listed under group D).

Further country-services wise quadrant B analysis (Table 9.6) shows that during 2000-02, Bangladesh and Yemen were competitively exporting the maximum number (4-5) of services (mainly transportation, communication and computer/information services, as well as travel and financial services in case of former and construction in case of latter), followed by Laos (transportation and OBS). In fact, during 2010-12, Yemen exported the maximum number of services, i.e., financial, computer, and OBS, followed by Bangladesh and Laos (with common service of transportation, followed by travel in case of former and OBS in case of latter). In other words, there has not been any change in the position of Laos in its services exports in the quadrant analysis. Other significant exporters in 2010-12 included Nepal which competitively supplied computer services to India. Meanwhile, no services have been identified for Bhutan, Cambodia, and Myanmar under Quadrant B during 2010-12.

There have also been some major changes in services under quadrant B. For instance, Bangladesh is now competitive only in travel and transportation, but has lost BRCA in rest of them (FS, CS and CI); Yemen lost competitiveness in case of TN, CS, CN (except CI); and Cambodia lost in case of CI.

Table 9.5: Mapping of services sectors across countries (quadrant-wise)

Country		200	0-02			201	0-12	
	Quad.							
	A	В	C	D	A	В	C	D
Communications services (CS)	1	2	1	3	0	0	2	5
Computer and information services (CI)	0	4	0	1	0	2	0	3
Construction (CN)	1	2	0	1	0	0	1	3
Financial services (FS)	0	1	0	4	0	1	0	6
Government services, n.i.e. (GS)	0	0	3	4	0	0	5	2
Insurance services (IS)	0	0	0	4	0	0	0	5
Other business services (OB)	0	1	1	5	1	2	0	4
Personal, cultural and recreational services (PC)	0	0	1	3	0	0	0	4

Royalties and	0	0	0	3	0	0	0	4
license fees (RL)								
Transportation	1	3	1	2	1	2	1	3
(TN)								
Travel (TL)	5	1	1	0	3	1	3	0

Source: OECD-WTO BaTIS; own calculations

Over the period 2000-02 to 2010-12, except for government and OBS, the number of countries supplying services under Quadrant D have actually increased (Table 9.5).

Table 9.6: Country-services quadrant-wise representation

		200	0-02			2010-12				
Country	Quad.	Quad.	Quad.	Quad.	Quad.	Quad.	Quad.	Quad.		
	A	В	C	D	A	В	C	D		
Bangladesh	1 (CN)	5 (TN, TL, FS, CS, CI)	2 (OB, GS)	2 (IS, RL)	1 (OB)	2 (TN, TL)	2 (CN, GS)	5 (CS, IS, FS, CI, RL)		
Bhutan	1 (TL)	0	0	6 (TN, CS, IS, CI, OB, GS)	1 (TL)	0	0	7 (TN, CS, IS, FS, CI, OB, GS)		
Cambodia	2 (TN, TL)	1 (CI)	0	4 (CS, OB, PC, GS)	1 (TN)	0	1 (TL)	7 (CS, FS, CI, RL, OB, PC, GS)		
Lao People's Democratic Republic	0	2 (TN, OB)	2 (TL, CS)	3 (CN, FS, GS)	0	2 (TN, OB)	3 (TL, CS, GS)	3 (CN, IS, FS)		
Myanmar	1 (TL)	1 (CN)	2 (TN, GS)	5 (CS, FS, RL, OB, PC)	1 (TL)	0	2 (TN, GS)	6 (CS, CN, FS, RL, OB, PC)		
Nepal	2 (TL, CS)	1 (CI)	1 (PC)	5 (TN, IS, FS, OB, GS)	1 (TL)	1 (CI)	2 (CS, GS)	5 (TN, IS, FS, OB, PC)		
Yemen	1 (TL)	4 (TN, CS, CN, CI)	1 (GS)	5 (IS, FS, RL, OB, PC)	0	3 (FS, CI, OB)	2 (TL, GS)	6 (TN, CS, CN, IS, RL, PC)		

Source: OECD-WTO BaTIS; own calculations

For few services identified under Quadrant C, which already have above-average shares in Indian imports though with low BRCA, Bangladesh, Laos, Nepal, Yemen and Myanmar have been the top exporters in the both the periods (with a rise in the number of services for most of them).

CHAPTER 10

Conclusion and policy recommendations

This study has aimed at examining trade and the investment opportunities between India and selected 41 African and Asian economies mainly LDCs. The focus has been on these countries' export opportunities in the Indian market and on India's investment opportunities in them. A discussion of general regulatory and sector-wise barriers to trade and investment in these countries has also been provided based on secondary data and a review of the existing literature on this subject.

10.1 Summary of key findings

Amongst the African countries studied in this report, Nigeria and Angola have been the top sources of merchandise imports for India in 2010-12 (followed by Tanzania, Ghana, Senegal, and Guinea) and in 2015-17 (Ghana, Zambia, Mozambique, and Guinea). Amongst the sample of Asian LDCs, India imported goods from Yemen, Myanmar, Bangladesh and Nepal in 2010-12, but over time Yemen lost its position while that of Laos improved significantly. In both time periods, India's main imports from Africa include petroleum oils, precious stones and metals like gold and diamonds, mineral ores (manganese, copper, iron), coal, natural gas, and agricultural products covering dried leguminous vegetables, oil seeds, cashew nuts, beans, sesamum seeds, cloves). India's imports from Asia too have been concentrated in few traditional product lines such as jute, fish, gold, petroleum oil, copper ores, ferro-silico, beans, nuts, non-alcoholic beverages, etc. Thus the export baskets of the 41 countries studied in this report are still not diversified enough to meet the rising demand for consumer and capital goods in India. At the same time, this shows the existence of immense untapped potential for enhancing these countries' trade and investment relationship with India. This is corroborated by identifying those commodities in 2015-17 which India imports significantly from the rest of the world but not from the concerned African and Asian countries. Such products include petroleum oils, gold, diamonds, coal, telephone for cellular network, parts of telephones sets, palm oil etc. Ironically, some of these products have been identified as India's top global imports. This implies that only few of the sample countries are significant exporters to India, while there exist supply-side constraints in others, which can be addressed by appropriate policy action.

Secondly, the value of these countries' services exports to India is much lower compared to their merchandise exports. But over the years 2010-12 and 2015-17, India's services imports have increased from all the 38 countries for which services trade data were available, even as India's merchandise imports from five Asian LDCs (excluding Cambodia, Laos and Bangladesh), and from Angola, Benin, Central African Republic, Guinea-Bissau, Kenya, Niger, and Nigeria, have actually declined. This shows these countries' growing services trade linkages with India, which must be enhanced with right polices. But as top merchandise exporters to India (Nigeria, Angola) are also experiencing a decline in their trade, sustaining manufacturing competitiveness of these countries becomes a policy priority. At the same time, some of the lowest exporting countries (Burundi, Central African Republic, Rwanda in Africa for goods and Lesotho for services; and Cambodia, Timor-Leste and Laos in Asia) could be explicitly targeted for removing barriers to trade.

Over the years, Nigeria and Nepal have remained the top African and Asian services exporters to India, respectively. Other major services exporters include Liberia (2000-02, replaced by Angola in 2010-12), Senegal, Kenya, and Tanzania in Africa; and Yemen, Myanmar, Bangladesh and Bhutan in Asia. The major services exports of these countries to India in both time-periods include travel and transportation and other business services (OBS). This also points to the need for enhancing these countries' export capacities in other sectors especially computer and information services, where there is potential for value-chain integration with India.

Limited capacities in LDCs of Africa, and even Asia, for supplying goods and services to India has been regarded as one of the main obstacles in full realization of their bilateral trade. However, India can potentially invest in those sectors in the selected African and Asian countries which have supply constraints (i.e., India's global imports exceed the concerned country's global exports as well as the associated region's global exports). Results show that none of the Asian LDCs have a supply constraint in any of the goods and services, due to the Asian region's huge global exports. In Africa, in contrast, commodities with lack of supply capacity have almost been the same across all the countries. India thus has an investment potential in several commodities namely animal or vegetable fats, organic chemicals, pharmaceuticals, fertilizers, tanning or dyeing extracts, albuminoidal substances, photographic or cinematographic goods, plastics and rubber articles, wood, paper, printed books, textiles, umbrellas, articles of stone, ceramic and glass products, base metals and their articles, general, electrical and precision machinery, transport machinery (excluding automotive), toys, etc. India can also potentially invest in Africa's services sectors including financial services, computer and information services, and OBS.

For the cumulative time period 2010-2018, India's greenfield investment stood at \$19.2 billion and \$10.3 billion in the selected African and Asian countries, respectively. In Africa, the top recipients have been Mozambique, Nigeria and Ghana with a cumulative 58.8% share, while in Asia, Bangladesh alone received around 72% of total greenfield investment. Sector-wise, India has majorly invested in the communications sector in Africa (\$2.7 billion in selected countries). Other sectors such as real estate and software & IT services in case of Democratic Republic of Congo; chemical and automotive OEM in Ghana; chemical and non-automotive transport OEM in Nigeria; etc., have received significant Indian greenfield investment. This is in contrast to the Asian countries where India has invested heavily in sectors like textiles (\$2.3 billion) and automotive OEM (\$1.8 billion), followed by coal, oil and gas, business services, renewable energy, financial services and even wood products. Further, some significant tradeinvestment linkages could be traced between India and six African countries (Zambia, Ethiopia, Mozambique, Sierra Leone, Kenya, and Congo) and three Asian countries (Bangladesh, Laos and Nepal). Such linkages are prominent in the case of communications services and OBS. Prominent manufacturing sectors in terms of linkages are coal, metals, food, automotive OEM and chemicals (in the case of Africa), and textiles and wood products (in the case of Asia).

At the same time, there are a number of non-tariff barriers (NTBs) in the selected African and Asian countries including restrictions/ban of imports in certain sectors such as agricultural goods including poultry, processed meat and vegetables, soaps, chemicals, steel and iron articles, used clothing, footwear, rubber articles, cosmetics, mining/oil and gas, pharmaceuticals, genetically engineered food grains, toys, antiques, narcotic drugs, motor vehicles, wines and spirits (bans/restrictions as reported by Nigeria, Angola, Ethiopia, Kenya, Uganda, Myanmar, Bangladesh, Laos, etc.) and even in lotteries, beauty salons, printing of recharge cards, automobile rentals services, petty trading (like in Ghana). There are restrictions

on foreign entry in banking, financial and insurance services and even in telecommunication. These are followed by non-transparent and complex custom valuation and clearance procedures, along with huge infrastructural bottlenecks resulting in high logistic costs. Many of these countries are LDCs which have huge trade finance gap, less conducive business environment with many unorganized enterprises, and less developed manufacturing sector with weak product diversification. Limited capital availability and supply side capacities further limit their linkages in global value chains (GVCs). In the case of Africa particularly, there is a lack of awareness about India's DFTP scheme which leads to under-utilization of the scheme, despite the high product coverage under the duty-free list. There is also an asymmetry of information between India and Africa, viz. lack of online data to provide information to Indian firms, as well as hurdles in tracing reliable local business partners. Another key barrier is the absence of bilateral investment agreements between India and the selected 33 African countries. On the other hand, Asian LDCs are geographically closer to India especially Bangladesh, Nepal and Myanmar, but continue to face connectivity issues, disturbed border trade or weak border-related infrastructure. India's trade and investment linkages are still more with developing/developed ASEAN (Indonesia, Thailand and Singapore) rather than with LDCs (Laos, Cambodia). Further, Asian LDCs prefer sourcing from other Asian countries (in sectors like textiles) than India thereby resulting in lower imports and investment from India. Restrictions under internal laws and policies of selected Asian LDCs to attract FDI further act as an obstacle, for instance, FDI reforms have been liberalized in Bangladesh but it still has limited scope and coverage under its foreign investment promotion law. On the other hand, India continues to raise its import tariffs on a number of goods including automobiles, coffee, alcoholic beverages, agricultural goods, gemstones, electronics, etc., or bans imports in case of items of animal origin or restricts imports of ICT goods, chemicals, etc. India also has huge certification and in-country inspection and testing as well as registration requirements with national authorities which involve fees and long procedures. Foreign entry is also barred in India's legal services, with restrictions in the case of financial and insurance services and in telecom/media sectors.

Finally, the study has also undertaken a quadrant analysis to identify the products and services where the selected African and Asian countries still have low shares in Indian imports despite a revealed comparative advantage in exporting (identified under Quadrant B). Active trade promotion policies may be required in this case. Across both regions, most of these products belong to the categories of general, electric and precision machinery followed by wood articles, paper articles, as well as fruits and vegetable products, and also some textiles items in the case of Asian LDCs. For services, transportation, OBS, financial, and computer and information services are identified under this quadrant for both the regions, where export shares need to be improved.

10.2 Policy recommendations

On the whole, there exist a large number of trade and investment opportunities between India and the selected African and Asian countries studied in this report. However, many of these opportunities are still untapped due to barriers including supply-side constraints, limited finance, weak infrastructure, and lack of supportive polices to expand these countries' exports to India and India's investment in them. If government policies and the business environment become more conducive mainly in African LDCs, their manufacturing industries can upgrade better and link into India's value chains. At the same time, these countries would also need to improve the cost-competitiveness of their products so that India can import more inputs from them as compared to other existing exporters. Meanwhile, LDCs are also gaining strength in

exporting several services (supply constraints exist only in financial services, computer services and OBS). As India already has a competitive services sector, these countries should also integrate into India's services value-chains especially in computer and information services. Nevertheless, technological upgrading and skill development is needed in all the countries, even in India, so as to develop a strong competitive trade and investment environment. A big push is required on both sides for providing adequate infrastructure and appropriate domestic reforms to produce and supply quality goods and services. There is also a need to fully explore the vertical trade complementarities that exist in the textiles and clothing sector, especially between India and Africa.

Moreover, in the light of China's growing investments in the selected countries particularly in Africa, as well as its ambitious Belt and Road Initiative, India urgently needs to look at the strategic aspect of its relationship with African and Asian LDCs in terms of not only improved economic cooperation, but also closer diplomatic relations. While China is a manufacturing hub, India has an edge in the services sectors; on soft power; in people relations-related investments like health and education; in investment in startups and new technologies suited to poor countries, all of which can be tapped to explore forward-looking possibilities in trade and investment between India and these countries.

Against this background, some important policy suggestions which can help develop and enhance trade and investment linkages between India and the selected African and Asian countries are as follows:

- 1. India can invite more companies from the selected countries to showcase their products and services in international exhibitions held in India. The concerned governments should also incentivize such participation.
- 2. Industry associations of the exporting countries should organize regular meetings at least once a year with their Indian counterparts to gain more awareness about existing schemes such as the DFTP offered by India and about changes that may be required to meet Indian product/sector-specific SPS and TBT requirements. Such sharing of knowledge could be vital, in helping Indian companies better identify sectoral investment opportunities in the selected countries as well as understanding the effects of past and current investment on the host country. Thus, there is a greater need to enhance the role of industry chambers, associations as well as sectoral bodies in promoting trade and investment linkages between India and the selected African and Asian economies.
- 3. African and Asian countries need to diversify their export basket by especially moving beyond trading of agricultural and mineral goods, textiles, etc., to more sophisticated and even capital-intensive items. For this, they first must focus on the items which are currently being imported by India from rest of the world but not competitively from them. Second, they could even integrate into the bottom end of manufacturing chains of Indian brands. Many small players in these countries could be encouraged to connect their manufacturing and services activities with designing and high-tech production activities of major players in India.
- 4. Domestic enterprises in these countries must be given easy loans and finance in order to form such value-chain linkages with India. They could also be allowed duty-free imports of inputs not available in their domestic markets to improve their cost competitiveness.
- 5. Indian investment should be increased in sectors which are competitive but may lack supply capacities in the selected exporting countries (some of these goods and services have been identified in this study lying in quadrant B during the quadrant analysis).

- 6. Such investment should be complemented by pro-active trade promotion policies by the concerned governments of the selected countries especially for those products and services where they have a revealed competitiveness in exporting to India, but which still have lower than average shares in India's total imports.
- 7. These countries must also re-visit their trade policies given the growing servicification of manufacturing GVCs and the spread of digital trade. In this respect, Banga and Velde (2018) also state that "African countries both face a digital divide and reap fewer benefits from digital technologies once installed. African countries should build up industrial capabilities in less-automated sectors (such as food and beverages, garments, metals, and paper) that can help them to move into higher-value-added activities. They should also urgently prepare for a more digital future by stimulating access to broadband and developing technical skills and technology hubs."
- 8. All countries, including India, could reconsider some of the restrictions and prohibitions that they have placed on imports of their goods and services, particularly in those sectors where demand and supply tends to match between India and the selected countries (for instance in agricultural goods, electronics, textiles, IT and financial and business services, etc. based on their competitiveness).
- 9. In the longer term, African governments could also make efforts to upgrade the quality and skillset of their labour force if they desire to compete with other growing economies. LDC firms could also be incentivized to form JVs with firms in other countries including India so as to learn from best practices in designing, manufacturing, branding, etc. Such practices could also be shared by India with better "performing" African countries such as Nigeria, Ghana and Kenya.
- 10. Along with utilising its existing capabilities in certain manufacturing sectors, India can tap its potential in soft power, in people-relations related investments such as health, and education, or even for investments in start-ups and new technologies that are likely to be more suitable for vulnerable economies of selected LDCs.

REFERENCES

- Afreximbank, & Exim India. (2018). *Deepening South-South Collaboration: An Analysis of Africa and India's Trade and Investment*. (Report). Egypt: African Export-Import Bank. Available at https://afreximbank.com/downloads/deepening-south-south-collaboration-an-analysis-ofafrica-and-indias-trade-and-investment-2/
- Assocham. (2016). *India ASEAN Trade and Investment Relations: Opportunities and Challenges*. (Report). New Delhi: The Associated Chambers of Commerce and Industry of India.
- Banga, R., & Saha, A. (2016). *Boosting India's Exports by Linking LDCs into India's Potential Global Value Chains*. (Policy Advice Report). London: Commonwealth Secretariat.
- Banga, K., & Velde, D.W. (2018). *Digitalisation and the future of African manufacturing*. (Briefing Paper).London: Overseas Development Institute. Available at https://set.odi.org/wp-content/uploads/2018/03/SET_Future-of-manufacturing_Brief_Final.pdf
- Bhogal, P. (2018). *India-ASEAN economic relations: Examining future possibilities*. (ORF Issue Brief). New Delhi: Observer Research Foundation.
- Broadman, H. G. (2008). China and India Go to Africa: New Deals in the Developing World. *Foreign Affairs*, 87(2), 95-109.
- Dixit, N. (2015). *CLMV in India's "Act East" Policy*. (News article). Asia Briefing Ltd. Available at https://www.aseanbriefing.com/news/2015/03/13/cmlv-in-indias-act-east-policy.html
- Exim Bank. (2014). Enhancing India's Bilateral Ties with Cambodia, Lao PDR, Myanmar, Vietnam: A Brief Analysis. (Working paper No.34). Mumbai, India: Export-Import Bank of India.
- Exim Bank. (2015). Bangladesh: A Study of India's Trade and Investment Potential. (Occasional paper No.172). Mumbai, India: Export-Import Bank of India.
- Exim Bank. (2018a). *Act East: Enhancing India's Trade with Bangladesh and Myanmar across Border*. (Working paper No.77). Mumbai, India: Export-Import Bank of India.
- Exim Bank. (2018b). *Indian Investments in West Africa: Recent Trends and Prospects*. (Working paper No.82). Mumbai, India: Export-Import Bank of India.
- European Commission. (n.d.). *European Commission Market Access Database: Trade Barriers*. Brussels: European Commission. Available at https://madb.europa.eu/madb/barriers_crossTables.htm?isSps=false
- Gupta, N. (2018). Constraints to linking into global value chains: Do Indian industries lack capacities and skills? *Productivity*, 58(4), 363-379.
- International Trade Centre. (2015). *India's duty-free tariff preference scheme for LDCs: A business guide*. (Technical paper). Geneva: International Trade Centre. Available at http://www.intracen.org/uploadedFiles/intracenorg/IndiaDutyFreeTariff.pdf
- KPMG, & DFID. (2016). Bangladesh Growth & Commercial Opportunities for Foreign Investors Scoping Study: Interim Report. KPMG.
- Kumar, R., & Singh, M. (2009). *India's Role in South Asia Trade and Investment Integration*. (ADB Working Paper Series on Regional Economic Integration No. 32). Manila, Philippines: Asian Development Bank.
- Liu, A., & Tang, B. (2018). US and China Aid to Africa: Impact on the Donor-Recipient Trade Relations. *China Economic Review*, 48, 46-65.
- Lucey, A., Schoeman, M., & Makokera, C.G. (2015). *India–Africa relations: The role of the private sector*. (ISS Paper No.285). Nairobi: Institute for Security Studies.
- Ministry of Commerce. (2017). *India's Duty Free Tariff Preference (DFTP) Scheme for Least Developed Countries (LDCs)*. New Delhi: Ministry of Commerce. Available at

- https://commerce.gov.in/writereaddata/UploadedFile/MOC_636434269763910839_international_tpp_DFTP.pdf
- Mazumdar, R. (2017). From trade to investments, re-orienting India's ASEAN strategy. New Delhi: Observer Research Foundation. Available at https://www.orfonline.org/expert-speak/reorienting-india-asean-strategy-trade-investment/
- McCormick, D. (2008). China & India as Africa's New Donors: The Impact of Aid on Development. *Review of African Political Economy*, *35*(115), 73-92.
- Nayyar, R., & Aggarwal, P. (2014). Africa India's new trade and investment partner. *International Journal of Scientific and Research Publications*, 4(3).
- PHD Research Bureau. (2017). Reshaping India-Africa trade: Dynamics and export potentiality of Indian products in Africa. New Delhi: PHD Chamber Of Commerce and Industry.
- Schiere, R. (2011). China and Africa: An Emerging Partnership for Development? An overview of issues, Series N° 125, African Development Bank, Tunisia.
- UNCTAD, Commonwealth Secretariat, & CWTO. (2010). *Identification of potential supply chains in textiles and clothing sector in South Asia*. Geneva and New Delhi: United Nations Publications.
- UNCTAD. (2013). *Investment Policy Review: Bangladesh*. (Report). Geneva: United Nations Publications.
- UNCTAD. (2018a). Committee for Development Policy Report on the twentieth session (12–16 March 2018). (Economic and Social Council Official Records, 2018, E/2018/33, Supplement No. 13). New York: United Nations Publications. Available at https://undocs.org/en/E/2018/33
- UNCTAD. (2018b). World Investment Report: Investment and new industrial policies. Geneva: United Nations Publications.
- UNCTAD. (2018c). The Least Developed Countries Report 2018: Entrepreneurship for structural transformation: Beyond business as usual. New York: United Nations Publications.
- UNECA, & CII. (2018). *Deepening Africa-India trade and investment Partnership*. (Report). Ethiopia: UNECA.
- USTR. (2019). *United States Trade Representative's National Trade Estimate Report 2019 on Foreign Trade Barriers (NTE)*. (Report). Washington, DC: United States Trade Representative. Available at https://ustr.gov/sites/default/files/2019_National_Trade_Estimate_Report.pdf
- Widodo, T. (2009). Comparative Advantage: Theory, Empirical Measures and Case Studies. *Review of Economic and Business Studies*, 4, 57-82.
- WTO, & CII. (2013). *India-Africa: South-South Trade and Investment for Development*. (Report). Confederation of Indian Industry (CII) and WTO Secretariat.

ANNEX A (Goods)

[Note: The tables report the top ten commodities imported by India from select African (33) and Asian (8) countries averaged over three-year periods from 2010 to 2012 and 2015 to 2017. It also reports the share of each of the top ten commodities in the total merchandise imports of India from these countries along with India's imports from rest of the world (ROW). The top ten commodities imported are highlighted in different colours according to whether they belong to the exclusion list (where no tariff concessions are given), Margin of Preference (MOP) list (some tariff concession is given) or if they are under the duty-free list under the Duty-Free Tariff Preference (DFTP) Scheme. Thus, the commodities are highlighted in orange if they belong to the duty-free list, purple if they are under the MOP list and bold if they are in the exclusion list. If the products of a country are not highlighted in any colour or bold, it means that country has not been a part of the DFTP scheme offered by India in that particular year.]

Angola

Produ ct Code	Product Description	India's imports from Angola (avg. 2010-2012, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)	Product Code	Product Description	India's imports from Angola (avg. 2015- 2017, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)
270900	Petroleum oils & oils obtained from bituminous minerals, crude	6238947.9	99.2	113592248.5	270900	Petroleum oils & oils obtained from bituminous minerals, crude	2622006.0	82.3	69142223.1
720449	Ferrous waste & scrap	36369.4	0.6	2385100.2	710231	Diamonds, non- industrial, un-worked, simply sawn, cleaved or bruted	293789.7	9.2	16191319.3
760200	Aluminium waste & scrap	5140.1	0.1	1110713.4	271111	Natural gas, liquefied	251861.5	7.9	6419352.9
740400	Copper waste & scrap	2361.5	0.0	883064.5	720449	Ferrous waste & scrap (excl. of 720	7011.1	0.2	1466262.9
720410	Waste & scrap of cast iron	1169.3	0.0	63060.1	760200	Aluminium waste & scrap	1760.4	0.1	1531864.2
854810	Waste & scrap of primary cells	1103.3	0.0	11714.6	730210	Rails of iron/steel	1592.6	0.1	47675.6

	Total imports	6290570.8	100.0	427514724.1		Total imports	3187000.0	100.0	394070687.8
	Top 10 Share	6287019.1	99.9			Top 10 Share	3182341.0	99.9	
	rubber (SBR), other					primary cells			
400219	stainless steel Styrene-butadiene	318.2	0.0	529776.9	854810	Waste & scrap of	995.3	0.0	57368.0
720421	Waste & scrap of	328.9	0.0	829965.3	740400	Copper waste & scrap	1062.3	0.0	746858.3
	trimmed					copper of w			
251611	than refined Granite, crude/roughly	627.4	0.0	13464.2	740811	trimmed Copper wire, of refined	1130.1	0.0	459359.9
780199	Unwrought lead other	653.2	0.0	228744.5	251611	Granite, crude/roughly	1131.8	0.0	14930.5

Note- Share (%) is the share of each of the top ten products in the total merchandise imports of India from Angola

Benin

Product Code	Product Description	India's imports from Benin (avg. 2010-2012, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)	Product Code	Product Description	India's imports from Benin (avg. 2015-2017, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)
80131	Cashew nuts, in shell	137386.2	53.6	742771.6	80131	Cashew nuts, in shell	145012.3	61.3	1170065.5
440349	Topical wood, other	17749.5	6.9	635424.1	520100	Cotton, not carded/combed	22962.4	9.7	717570.9
760200	Aluminium waste & scrap	16364.8	6.4	1099488.7	440349	Topical wood, other	19790.9	8.4	415996.1
740400	Copper waste & scrap	13050.7	5.1	872375.3	760200	Aluminium waste & scrap	11557.1	4.9	1522067.6
310530	Di-ammonium hydrogen orthophosphate	11621.9	4.5	3181353.8	120100	Soya beans, whether/not broken	9372.2	4.0	20346.8
720449	Ferrous waste & scrap	11162.3	4.4	2410307.2	720449	Ferrous waste & scrap	9208.0	3.9	1461729.0
151110	Palm oil, crude	8543.4	3.3	5133467.8	740400	Copper waste & scrap	4993.4	2.1	742219.0
260800	Zinc ores & concentrates	7883.6	3.1	68198.7	854810	Waste & scrap of primary cells	3237.7	1.4	55125.6

	Total imports	256416.6	100.0	43357640 7.2		Total imports	236613.9	100.0	396935505.8
	Total imm auta	25(41()	100.0	42257640		Total imm auta	22((12.0	100.0	20/025505 0
	Top 10 Share	232774.8	90.8			Top 10 Share	231209.8	97.7	
2,1110	Dumies, iiqueirea	<i>2,</i> , 0, <u>2</u>	1.0	201033010	11072	lengthwise	2002	1.0	12001111
271113	Butanes, liquefied	3996.2	1.6	2578993.5	440729	Wood sawn/chipped	2384.2	1.0	123811.4
520100	Cotton, not carded/combed	5016.0	2.0	268656.1	120799	Oil seeds & oleaginous fruits (excl	2691.7	1.1	54067.0
500100	0.44	50160	2.0	2606561	120500	01 101	2601.7	1 1	5.40 <i>6</i> 7.0

Note- Share (%) is the share of each of the top ten products in the total merchandise imports of India from Benin.

Burkina Faso

Product Code	Product Description	India's imports from Burkina Faso (avg. 2010-2012, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)	Product Code	Product Description	India's imports from Burkina Faso (avg. 2015-2017, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)
80131	Cashew nuts, in shell	7920.9	50.2	874877.1	710812	Gold	311488.2	85.0	30874929.6
520100	Cotton, not carded/combed	4197.1	26.6	267803.0	80131	Cashew nuts, in shell	31538.3	8.6	1283539.5
260200	Manganese ores & concentrates	2087.9	13.2	423241.6	520100	Cotton, not carded/combed	22064.8	6.0	718468.5
844520	Textile spinning machines	1270.8	8.1	172212.2	120740	Sesamum seeds, whether/not broken	1008.9	0.3	38965.0
410691	Tanned/crust hides & skins	53.3	0.3	17190.4	120799	Oil seeds & oleaginous fruits	472.8	0.1	56285.9
230240	Bran, sharps & other residues, whether or not in the form pellets, derived from sifting, milling or other working of cereals or of leguminous plants	47.2	0.3	16476.7	842121	Filtering/purifying machinery & app.	21.8	0.0	107116.9
830210	Hinges of base metal	46.3	0.3	46613.1	392530	Shutters, blinds	9.8	0.0	3750.1
130120	Gum Arabic	42.0	0.3	14490.9	151590	Fixed vegetable fats & oils	8.7	0.0	5591.2

842139	Filtering/purifying machinery & app	36.5	0.2	180152.0	999999	Commodities not specified according	0.1	0.0	8647656.3
271290	Micro-crystalline petroleum	27.0	0.2	60797.6	732591	Grinding balls & similar cast	0.0	0.0	883.1
	wax Top 10 Share	15729.2	99.7			articles Top 10 Share	366613.3	100.0	
	Total imports	15776.1	100.0	433790749.1		Total imports	366613.3	100.0	396800706.0

Note- Share (%) is the share of each of the top ten products in the total merchandise imports of India from Burkina Faso.

<u>Burundi</u>

Product Code	Product Description	India's imports from Burundi (avg. 2010-2012, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)	Product Code	Product Description	India's imports from Burundi (avg. 2015-2017, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)
120799	Oil seeds & oleaginous fruits	312.9	27.8	16718.8	780110	Unwrought lead, refined	7527.5	67.0	235636.3
410221	Raw skins of sheep/lambs, whether or not with wool on or split, other	179.5	16.0	19438.4	790112	Zinc, not alloyed, unwrought	1312.8	11.7	78042.5
410510	Tanned/crust skins of sheep/lambs	106.2	9.4	58439.7	330129	Essential oils	1100.8	9.8	93779.9
293331	Pyridine & its salts	93.2	8.3	22212.8	780199	Unwrought lead other than refined,	683.3	6.1	233206.0
410190	Raw hides & skins of bovine	88.3	7.8	30765.2	780191	Unwrought lead other than refined	395.4	3.5	9902.2
382490	Chemical products &preparations of t	76.4	6.8	390865.6	410691	Tanned/crust hides & skins	125.0	1.1	15185.8
410419	Tanned/crust hides & skins of bovine	74.5	6.6	126056.6	847170	Storage units	50.1	0.5	808182.3
410621	Tanned/crust hides & skins of goats	49.6	4.4	10261.6	440839	Sheets for veneering	36.2	0.3	63364.6

854810	Waste & scrap of primary	46.7	4.2	12786.8	841290	Parts of the engines & motors of 84	5.7	0.1	84656.1
847490	cells Parts of the machinery of	31.8	2.8	170812.0	850120	Universal AC/DC motors	2.1	0.0	82405.3
	84.74 Top 10 Share	1059.1	94.1			of an output Top 10 Share	11238.8	100.0	
-	Total imports	1125.4	100.0	433802395.2		Total imports	11241.1	100.0	397163178.5

Note- Share (%) is the share of each of the top ten products in the total merchandise imports of India from Burundi.

Central African Republic

Product Code	Product Description	India's imports from Central Afr. Rep. (avg. 2010-2012, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)	Product Code	Product Description	India's imports from Central Afr. Rep. (avg. 2015-2017, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)
440349	Topical wood	1015.3	47.4	652158.3	710231	Diamonds, non- industrial, un-worked, simply sawn, cleaved or bruted	958.6	58.4	16484469.9
440399	Wood, in the rough	228.5	10.7	1013384.0	440399	Wood, in the rough	250.2	15.2	787089.9
720449	Ferrous waste & scrap	198.2	9.2	2421271.0	440349	Topical wood	196.4	12.0	435590.5
440729	Wood sawn/chipped lengthwise, slice	138.1	6.4	45419.4	780199	Unwrought lead other than refined,	84.8	5.2	233405.5
440320	Wood, in the rough	128.7	6.0	6529.2	440320	Wood, in the rough	42.7	2.6	15790.7
740400	Copper waste & scrap	106.5	5.0	885390.4	853669	Plugs & sockets for a voltage not >	33.8	2.1	203581.7
400129	Natural rubber other than latex/smooth	64.8	3.0	53835.4	760200	Aluminium waste & scrap	32.8	2.0	1533613.7
760120	Aluminium alloys, unwrought	52.7	2.5	329141.4	720421	Waste & scrap of stainless steel	24.4	1.5	960782.6
780200	Lead waste & scrap	51.8	2.4	112113.7	720410	Waste & scrap of cast iron	15.4	0.9	24761.9

780199	Unwrought lead other than refined	44.6	2.1	229165.2	560410	Rubber thread & cord, textile cover	2.0	0.1	1934.0
	Top 10 Share	2029.0	94.6			Top 10 Share	1641.0	99.9	
	Total imports 2144.4		100.0	433801073.0		Total imports	1642.5	100.0	397166127.7

Note- Share (%) is the share of each of the top ten products in the total merchandise imports of India from Central African Republic.

Chad

Product Code	Product Description	India's imports from Chad (avg. 2010-2012, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)	Product Code	Product Description	India's imports from Chad (avg. 2015-2017, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)
520100	Cotton, not carded/combed	2583.5	74.5	270277.7	270900	Petroleum oils & oils obtained from bituminous minerals, crude	181264.1	85.1	71582964.7
900669	Other photographic flashlight	139.7	4.0	3119.6	310530 Di-ammonium hydrogen orthophosphate		19765.5	9.3	2081210.1
910119	Other Wrist-watches, electrically operated	132.2	3.8	32105.3	520100	Cotton, not carded/combed	8065.8	3.8	732467.5
130120	Gum Arabic	116.5	3.4	14402.4	130120	Gum Arabic	2479.2	1.2	32928.1
903300	Parts & accessories. in Ch.90	91.1	2.6	297979.3	270119	Coal other than anthracite & bituminous	966.5	0.5	14380862.8
851770	Parts of telephone sets	91.1	2.6	2788448.0	130190	Lac; Natural gums (excl. of 1301.20	112.1	0.1	115049.5
390120	Polyethylene having a sp.gr. of 0.9	69.0	2.0	531981.2	841381	Pumps in 84.13	96.8	0.1	144264.2
400270	Ethylene-propylene- non-conjugated	57.5	1.7	98026.9	80131	Cashew nuts, in shell	59.5	0.0	1315058.0
850131	DC motors (excl. universal AC/DC mo	47.2	1.4	82110.6	320411	Disperse dyes & preparations based	41.8	0.0	38733.5
851712	Telephones for cellular networks/for other wireless	40.1	1.2	5229184.0	410190	Raw hides & skins of bovine	34.6	0.0	25729.3
	Top 10 Share	3367.6	97.1			Top 10 Share	212885.8	100.0	

Total imports 3467.8 100.0 433800755.2 Total import	s 212953.1 100.0 396968398.0
---	------------------------------

Note- Share (%) is the share of each of the top ten products in the total merchandise imports of India from Chad.

Democratic Republic of Congo

Product Code	Product Description	India's imports from Dem. Rep. of Congo(avg . 2010-2012, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)	Product Code	Product Description	India's imports from Dem. Rep. of Congo (avg. 2015-2017, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)
740319	Unwrought products of refined copper	12056.0	52.2	34827.0	260300	Copper ores & concentrates	63397.3	35.6	3442659.2
260500	Cobalt ores & concentrates	6411.2	27.8	19762.3	270900	Petroleum oils & oils obtained from bituminous minerals, crude	37895.1	21.3	71738965.3
740311	Cathodes & sections of cathodes	1490.4	6.5	64243.5	710231	Diamonds, non- industrial, un-worked, simply sawn, cleaved or bruted	30945.4	17.4	16454163.6
720449	Ferrous waste & scrap (excl. of 720	1105.4	4.8	2421101.0	740200	Unrefined copper; copper anodes for electrolytic refining	25100.8	14.1	676580.0
121190	Plants & parts of plants, incl. see	975.1	4.2	39491.0	121190	Plants & parts of plants	4283.5	2.4	62245.7
720410	Waste & scrap of cast iron	409.4	1.8	64092.9	740311	Cathodes & sections of cathodes	3187.0	1.8	164437.7
400122	Technically spec. natural rubber	321.6	1.4	370238.4	810520	Cobalt mattes & other intermediate product of cobalt metallurgy	3169.5	1.8	23985.2
740400	Copper waste & scrap	97.8	0.4	885393.4	180100	Cocoa beans, whole/broken	3142.8	1.8	58552.0

711319	Articles of jewelry & parts there	66.0	0.3	2002883.0	710221	Industrial diamonds, un-worked, simply sawn, cleaved or bruted	1811.1	1.0	9283.5
901850	Ophthalmic instr. & appliances	47.5	0.2	120949.2	260500	Cobalt ores & concentrates	684.0	0.4	497.7
	Top 10 Share	22980.2	99.6			Top 10 Share	173616.7	97.4	
	Total imports	23084.1	100.0	433790233.2		Total imports	178172.4	100.0	397040583.9

Note- Share (%) is the share of each of the top ten products in the total merchandise imports of India from Democratic Republic of Congo.

<u>Djibouti</u>

Product Code	Product Description	India's imports from Djibouti (avg. 2010-2012, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)	s Product Code	Product Description	India's imports from Djibouti (avg. 2015-2017, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)
850433	Electrical transformers	998.2	22.5	11267.6	170310	Cane molasses	1732.0	32.1	3165.6
720449	Ferrous waste & scrap	868.5	19.6	2420601.0	120799	Oil seeds & oleaginous fruits	653.1	12.1	56323.3
410530	Tanned/crust skins of sheep/lambs,	704.4	15.9	10742.2	120100	Soya beans, whether/not broken	490.5	9.1	26104.5
410190	Raw hides & skins of bovine	212.7	4.8	30582.0	410229	Raw skins of sheep/lambs with wool	489.6	9.1	6752.9
740319	Unwrought products of refined copper	150.6	3.4	38795.4	854810	Waste & scrap of primary cells	341.7	6.3	58135.5
71333	Kidney beans, incl. white pea beans	130.0	2.9	71696.3	842290	Parts of the machinery of 8422.11-8	297.0	5.5	67068.8
410221	Raw skins of sheep/lambs	129.3	2.9	19412.0	410190	Raw hides & skins of bovine	252.0	4.7	25488.8
540120	Sewing thread of man-made filaments	113.4	2.6	66213.3	721310	Bars & rods, hot-rolled, in irregularly wound coils, of iron or non-alloy steel	244.7	4.5	4567.4

410210	Raw skins of sheep/lambs with wool	102.5	2.3	5490.8	740721	Bars, rods & profiles, of copper-zinc base alloys	184.3	3.4	45840.7
410229	Raw skins of sheep/lambs with wool	100.9	2.3	7189.0	271019	Petroleum oils & oils obtained from bituminous minerals	119.2	2.2	2471265.0
	Top 10 Share	3510.4	79.0			Top 10 Share	4803.9	89.0	
	Total imports	4442.1	100.0	433800358.6		Total imports	5395.2	100.0	397164583.9

Note- Share (%) is the share of each of the top ten products in the total merchandise imports of India from Djibouti.

Eritrea

Product Code	Product Description	India's imports from Eritrea (avg. 2010-2012, \$ 1000)	Shar e (%)	India's imports from ROW (\$ 1000)	Produc t Code	Product Description	India's imports from Eritrea (avg. 2015-2017, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)
260112	Iron ores & concentrates	9680.2	76.7	186921.4	260300	Copper ores & concentrates	103636.9	88.3	3415832.8
710310	Precious stones and semi- precious stones	1117.5	8.9	207696.0	261690	Precious metal ores & concentrates	5847.4	5.0	186857.3
720449	Ferrous waste & scrap	777.5	6.2	2420951.0	710310	Precious stones and semi- precious stones	3424.5	2.9	589337.4
720421	Waste & scrap of stainless steel	278.6	2.2	830201.3	261610	Silver ores & concentrates	3162.0	2.7	39979.2
470790	Recovered (waste & scrap) paper/paperboard, other	125.9	1.0	550481.6	710399	Precious stones and semi- precious stones	1097.5	0.9	59739.5
740400	Copper waste & scrap	107.8	0.9	885390.0	720449	Ferrous waste & scrap	136.7	0.1	1470891.4
260700	Lead ores & concentrates	105.7	0.8	65458.8	390190	Polymers of ethylene	26.6	0.0	518438.2
999999	Commodities not specified according	83.9	0.7	10834563.0	710391	Rubies, sapphires & emeralds, worked	21.0	0.0	1158381.5
410621	Tanned/crust hides & skins of goats	69.4	0.6	10248.4	853710	Boards, panels, consoles, desks, cabinets and other bases	0.2	0.0	356115.2
410449	Tanned/crust hides & skins of bovine	68.8	0.6	86964.5	847990	Parts of Machines & mechanical appliances	0.2	0.0	319001.3

Top 10 Share	12415.3	98.4	Top 10 Share	117353.0	100.0	
Total imports	12617.2	100.0 433797895.1	Total imports	117353.0	100.0	397088665.4

Note- Share (%) is the share of each of the top ten products in the total merchandise imports of India from Eritrea.

Ethiopia

Product Code	Product Description	India's imports from Ethiopia (avg. 2010-2012, \$ 1000)	Shar e (%)	India's imports from ROW (\$ 1000)	Produc t Code	Product Description	India's import s from Ethiopi a (avg. 2015-2017, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)
71333	Kidney beans, incl. white pea beans	6183.7	14.9	65555.9	71333	Kidney beans, incl. white pea beans	15658.1	24.8	74279.4
410530	Tanned/crust skins of sheep/lambs,	5036.6	12.1	5940.5	120100	Soya beans, whether/not broken	15355.7	24.3	11239.3
71331	Beans of the species Vigna mungo	4916.2	11.8	461206.6	71320	Chickpeas (garbanzos), dried, shell	7965.6	12.6	801970.1
71320	Chickpeas (garbanzos), dried, shell	3140.7	7.6	181255.8	71331	Beans of the species Vigna mungo	7701.4	12.2	583127.4
271113	Butanes , liquefied	1610.8	3.9	2581121.0	91030	Turmeric (curcuma)	3097.7	4.9	21486.9
271112	Propane, liquefied	1312.3	3.2	1151381.0	120799	Oil seeds & oleaginous fruits	2158.8	3.4	54599.9
520100	Cotton, not carded/combed	1254.3	3.0	271163.9	710310	Precious stones and semi- precious stones	1442.8	2.3	591319.1
710310	Precious stones (excl. diamonds) &	1088.0	2.6	207353.0	411200	Leather further prepared after tanning/crust leather of sheep/lamb without wool	1346.1	2.1	3843.6
91010	Ginger	1084.3	2.6	14526.5	850421	Liquid dielectric transformers having power handling capacity not exceeding 650 KVA	802.8	1.3	3741.6

411200	Leather further prepared after tanning/crust leather of sheep/lamb without wool	1020.5	2.5	2647.8	410719	Leather further prepared after tanning/crusting	722.7	1.1	72534.3
	Top 10 Share	26647.5	64.1			Top 10 Share	56251.7	89.0	
	Total imports	41566.0	100.0	433772253.9		Total imports	63231.4	100.0	397108712.5

Note- Share (%) is the share of each of the top ten products in the total merchandise imports of India from Ethiopia.

<u>Gambia</u>

Product Code	Product Description	India's imports from Gambia(a vg. 2010-2012, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)	Product Code	Product Description	India's imports from Gambia (avg. 2015-2017, \$ 1000)	Shar e (%)	India's imports from ROW (\$ 1000)
80131	Cashew nuts, in shell	25324.0	91.0	854833.7	80131	Cashew nuts, in shell	39315.2	89.1	1275762.7
760200	Aluminium waste & scrap	829.7	3.0	1115024.0	720449	Ferrous waste & scrap	2327.8	5.3	1468609.2
330190	Essential oils	271.2	1.0	34702.1	854810	Waste & scrap of primary cells,	1363.5	3.1	56999.8
740400	Copper waste & scrap	261.6	0.9	885164.3	760200	Aluminium waste & scrap	769.2	1.7	1532855.5
854810	Waste & scrap of primary cells	222.2	0.8	12669.8	710310	Precious stones and semi-precious stones	190.4	0.4	592634.9
400231	Isobutene-isoprene (butyl) rubber	202.0	0.7	199165.0	780200	Lead waste & scrap	61.7	0.1	116742.4
722860	Bars & rods of other alloy steel	130.9	0.5	69435.6	720410	Waste & scrap of cast iron	16.8	0.0	24761.4
720449	Ferrous waste & scrap	122.4	0.4	2421347.0	440349	Topical wood	14.0	0.0	435782.3
720429	Waste & scrap of alloy steel other	116.8	0.4	52836.2	740400	Copper waste & scrap	13.6	0.0	747207.9
790200	Zinc waste & scrap	46.7	0.2	75211.3	440399	Wood, in the rough	12.7	0.0	787335.9
	Top 10 Share	27527.5	98.9			Top 10 Share	44084.9	99.9	

Total imports	27837.3	100.0	433775760.0	Total imports	44118.2	100.0	397123339.1		
Note- Share (%) is the share of each of the top ten products in the total merchandise imports of India from Gambia.									

<u>Ghana</u>

Product Code	Product Description	India's imports from Ghana (avg. 2010-2012, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)	Produc t Code	Product Description	India's imports from Ghana (avg. 2015-2017, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)
80131	Cashew nuts, in shell	98158.7	28.6	781999.1	710812	Gold, unwrought or in semi-manufactured forms	2261737.5	90.7	28924680.0
260300	Copper ores & concentrates	48706.4	14.2	4892630.0	80131	Cashew nuts, in shell	78621.2	3.2	1236457.0
440349	Topical wood	47869.6	13.9	605304.1	440349	Topical wood	51186.7	2.1	384600.2
180100	Cocoa beans, whole/broken, raw/roasted	27131.2	7.9	32516.0	710813	Gold, unwrought or in semi-manufactured forms	21691.7	0.9	164926.7
280920	Phosphoric acid & poly- phosphoric acid	17239.1	5.0	1657496.0	760200	Aluminium waste & scrap	14780.9	0.6	1518844.0
720449	Ferrous waste & scrap	16638.1	4.9	2404831.0	180310	Cocoa paste, not defatted	13993.4	0.6	3534.3
760200	Aluminium waste & scrap	10477.4	3.1	1105376.0	120799	Oil seeds & oleaginous fruits	10243.3	0.4	46515.3
120799	Oil seeds & oleaginous fruits	8680.2	2.5	8142.9	440399	Wood, in the rough	6439.2	0.3	780901.0
780199	Unwrought lead other than refined	7935.8	2.3	221244.2	740400	Copper waste & scrap	4587.8	0.2	742624.7
261690	Precious metal ores & concentrates	5466.0	1.6	2029.8	260600	Aluminium ores & concentrates	4004.5	0.2	107061.9
	Top 10 Share	288302.4	84.0			Top 10 Share	2467286.2	99.0	

Total imports	343322.1	100.0	433528726.7	Total Trade	2492764.6	100.0	394689987.1
_							

Note- Share (%) is the share of each of the top ten products in the total merchandise imports of India from Ghana.

Guinea

Product Code	Product Description	India's imports from Guinea (avg. 2010-2012, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)	Produc t Code	Product Description	India's imports from Guinea (avg. 2015-2017, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)
270900	Petroleum oils & oils obtained from bituminous oils, crude	232380.2	69.5	120000000.0	710812	Gold, unwrought or in semi-manufactured forms	243488.6	53.6	30942929.2
260300	Copper ores & concentrates	63289.7	18.9	4866673.0	270900	Petroleum oils & oils obtained from bituminous oils, crude	47799.5	10.5	71716429.3
80131	Cashew nuts, in shell	14884.6	4.5	865273.1	260600	Aluminium ores & concentrates	44972.0	9.9	63424.7
720449	Ferrous waste & scrap	13346.3	4.0	2408123.0	271111	Natural gas, liquefied	33092.7	7.3	6576229.7
440349	Topical wood	1275.2	0.4	651898.5	80131	Cashew nuts, in shell	26286.5	5.8	1288791.4
760200	Aluminium waste & scrap	1191.8	0.4	1114662.0	282540	Nickel oxides & hydroxides	21759.2	4.8	14434.1
90111	Coffee, not roasted, not decaffeinated	953.0	0.3	92257.9	260400	Nickel ores & concentrates	16016.2	3.5	16393.5
721012	Flat-rolled products of iron/non-alloy steel	600.2	0.2	111440.8	720449	Ferrous waste & scrap	8602.7	1.9	1462334.3
847990	Parts of Machines & mechanical appliances	516.5	0.2	371369.3	760200	Aluminium waste & scrap	4457.1	1.0	1529167.6

400220	Butadiene rubber (BR)	405.1	0.1	187100.0	854810	Waste & scrap of primary cells	2955.0	0.7	55408.3
	Top 10 Share	328842.6	98.4			Top 10 Share	449429.4	98.9	
	Total imports	334245.8	100.0	433571357.4		Total imports	454312.0	100.0	396756398.6

Note- Share (%) is the share of each of the top ten products in the total merchandise imports of India from Guinea.

Guinea-Bissau

Product Code	Product Description	India's imports from Guinea-Bissau (avg. 2010-2012, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)	Product Code	Product Description	India's imports from Guinea-Bissau (avg. 2015-2017, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)
80131	Cashew nuts, in shell	131925.8	57.7	748231.9	80131	Cashew nuts, in shell	220725.4	97.4	1094352.5
270900	Petroleum oils & oils obtained from bituminous oils, crude	57711.5	25.3	120000000.0	260600	Aluminium ores & concentrates	5168.0	2.3	106674.0
260300	Copper ores & concentrates	21114.4	9.2	4901828.0	720449	Ferrous waste & scrap	261.3	0.1	1470675.7
270799	Oils & other products of the distillation of high temp. coal	6014.0	2.6	63064.0	120740	Sesamum seeds, whether/not broken	160.5	0.1	39920.4
280920	Phosphoric acid & poly- phosphoric acid	4884.0	2.1	1661615.0	854810	Waste & scrap of primary cells	133.9	0.1	58229.4
400219	Styrene-butadiene rubber (SBR)	907.5	0.4	529278.0	230650	Oil-cake & other solid residues	46.5	0.0	36770.3
720449	Ferrous waste & scrap	883.9	0.4	2420586.0	710310	Precious stones and semi- precious stones	18.1	0.0	592743.8
270119	Coal other than anthracite & bituminous substances	633.2	0.3	12850369.0	80132	Cashew nuts, shelled	11.1	0.0	28923.5

845929	Drilling machines,other	462.8	0.2	73942.0	903180	Measuring/checking instr.,	1.3	0.0	556070.7
845140	Washing/bleaching/dyeing machines	379.2	0.2	57998.3	610590	apparatus Men's/boys' shirts, knitted/crochet	0.1	0.0	442.3
	Top 10 Share	224916.2	98.4			Top 10 Share	226526.2	100.0	
	Total imports	228583.8	100.0	433618246.4		Total imports	226526.2	100.0	396944358.0

Note- Share (%) is the share of each of the top ten products in the total merchandise imports of India from Guinea-Bissau.

Kenya

Product Code	Product Description	India's imports from Kenya (avg. 2010-2012, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)	Product Code	Product Description	India's imports from Kenya (avg. 2015-2017, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)
283620	Disodium carbonate	30136.1	21.7	96085.8	71331	Beans of the species Vigna mungo	21563.8	20.1	566697.8
270119	Coal other than anthracite & bituminous substances	14468.8	10.4	12845968.0	283620	Disodium carbonate	20357.3	19.0	140009.0
71331	Beans of the species Vigna mungo (L	12218.1	8.8	452266.0	90240	Tea, black (fermented)	11491.6	10.7	28309.5
90240	Tea, black (fermented)	11405.6	8.2	29850.8	71390	Dried leguminous vegetables	10344.4	9.7	502818.4
252922	Fluorspar, containing by weight >97	11140.6	8.0	47750.3	252922	Fluorspar, containing by weight >97	5405.2	5.0	32757.1
410510	Tanned/crust skins of sheep/lambs,	5474.1	3.9	53001.0	740400	Copper waste & scrap	5367.7	5.0	741844.7
843810	Bakery machinery & machinery for manufacture of macaroni, spaghetti/similar products	4478.0	3.2	19026.6	410419	Tanned/crust hides & skins of bovine	3057.4	2.9	97419.5

	Total imports	138822.0	100.0	433688560.8		Total imports	107164.9	100.0	397063835.2
	Top 10 Share	96231.0	69.3			Top 10 Share	85311.2	79.6	
410419	Tanned/crust hides & skins of bovine	1650.5	1.2	124480.6	410510	Tanned/crust skins of sheep/lambs	2200.2	2.1	62511.0
71390	Dried leguminous vegetables	1860.9	1.3	333081.6	90111	scrap Coffee, not roasted, not decaffeinated	2682.8	2.5	121978.7
720410	Waste & scrap of cast iron	3398.2	2.5	60831.2	760200	Aluminium waste &	2840.8	2.7	1530783.9

Note- Share (%) is the share of each of the top ten products in the total merchandise imports of India from Kenya.

Lesotho

Product Code	Product Description	India's import s from Lesoth o(avg. 2010-2012, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)	Product Code	Product Description	India's imports from Lesotho (avg. 2015-2017, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)
510119	Wool, not carded/combed, greasy	2272.5	74.9	229757.2	710231	Diamonds, non- industrial, un- worked/simply sawn cleaved/bruted	30336.8	97.8	16454772.1
510129	Wool, not carded/combed, degreased	327.9	10.8	87330.6	510119	Wool, not carded/combed, greasy, in	576.4	1.8	192111.7
853620	Automatic circuit breakers	62.3	2.1	92770.3	903300	Parts & accessories	67.3	0.2	214170.9
540244	Yarn other than high tenacity	43.3	1.4	45584.3	520548	Cotton yarn	21.5	0.0	6018.7
853669	Plugs & sockets	42.1	1.4	168178.5	121190	Plants & parts of plants	7.6	0.0	66526.7
390110	Polyethylene having a sp.gr. of <0.	41.1	1.4	1025047.0	902290	X-ray generators	2.6	0.0	225383.9

382490	Chemical products	40.8	1.3	390877.5	999999	Commodities not specified according	0.0	0.0	8647656.3
392690	Articles of plastics &articles	36.6	1.2	536728.3		_	ducts available		
121190	Plants & parts of plants	32.8	1.1	40444.2					
731819	Threaded articles of iron/steel	27.3	0.9	68570.2					
	Top 10 Share	2926.6	96.5		Top 7 Share		31012.4	100.0	
	Total imports	3033.3	100.0	433800315	Total imports		31012.4	100.0	397136338.9

Note- Share (%) is the share of each of the top ten products in the total merchandise imports of India from Lesotho.

<u>Liberia</u>

Product Code	Product Description	India's imports from Liberia (avg. 2010-2012, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)	Product Code	Product Description	India's imports from Liberia (avg. 2015-2017, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)
271011	Light petroleum oils & preparations	7947.3	28.7	2116893.9	890120	Tankers	43301.7	71.4	389828.5
890400	Tugs & pusher craft	4016.1	14.5	980195.3	720449	Ferrous waste & scrap	4953.6	8.2	1465983.4
180100	Cocoa beans, whole/broken, raw/roasted	3497.3	12.6	58481.5	890110	Cruise ships, excursion boats & similar vessels	3999.5	6.6	132505.2
720449	Ferrous waste & scrap (excl. of 720	2627.4	9.5	2418842.1	440399	Wood, in the rough	3114.5	5.1	785263.8
740400	Copper waste & scrap	1249.5	4.5	884176.4	760200	Aluminium waste & scrap	1721.7	2.8	1531903.0
440349	Topical wood	805.3	2.9	652368.3	854810	Waste & scrap of primary cells	1115.8	1.8	57247.5

760200	Aluminium waste & scrap	752.9	2.7	1115100.6	740400	Copper waste & scrap	749.5	1.2	746463.0
440399	Wood, in the rough	563.2	2.0	1013424.7	400121	Natural rubber	366.8	0.6	132035.9
890110	Cruise ships, excursion boats & similar vessels	560.5	2.0	370152.8	720430	Waste & scrap of tinned iron/steel	280.8	0.5	2189.0
841440	Air compressors mounted on a wheeled chassis for towing	532.0	1.9	172347.2	730210	Rails of iron/steel	193.0	0.3	48142.1
	Top 10 Share	22551.5	81.3			Top 10 Share	59796.8	98.6	
	Total imports	27737.2	100.0	433789415.1		Total imports	60666.1	100.0	397125781.2

Note- Share (%) is the share of each of the top ten products in the total merchandise imports of India from Liberia.

Madagascar

Product Code	Product Description	India's imports from Madagas car (avg. 2010-2012, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)	Produc t Code	Product Description	India's imports from Madaga scar (avg. 2015-2017, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)
90700	Cloves	23403.1	34.4	48745.9	90700	Cloves	78233.3	56.5	71844.8
720449	Ferrous waste & scrap	13183.1	19.4	2408286.4	750210	Nickel, not alloyed, unwrought	14168.9	10.2	591963.9
71339	Beans	5537.3	8.1	101866.8	71339	Beans	8926.8	6.5	94216.7
710310	Precious stones and semi- precious stones	1561.4	2.3	206879.6	710310	Precious stones and semi-precious stones	6802.2	4.9	585959.7
845969	Milling machines	1126.6	1.7	24499.7	330129	Essential oils	4680.6	3.4	89833.1
90411	Pepper (genus Piper), neither crush	926.8	1.4	73847.6	90500	Vanilla	3986.8	2.9	14428.5
750210	Nickel, not alloyed, unwrought	874.6	1.3	585951.2	80131	Cashew nuts, in shell	2983.8	2.2	1312094.1
330129	Essential oils	835.1	1.2	54671.1	71331	Beans of the species Vigna mungo	2816.3	2.0	585445.3

250490	Natural graphite other than in powder	733.8	1.1	894.1	854810	Waste & scrap of primary cells	1803.1	1.3	56560.2
71331	Beans of the species Vigna mungo	699.8	1.0	463784.3	250410	Natural graphite, in powder/flakes	1322.1	1.0	18248.2
	Top 10 Share	48881.7	71.8			Top 10 Share	125723.8	90.9	
	Total imports	68054.0	100.0	433744604.5		Total imports	138377.9	100.0	397035319.6

Note- Share (%) is the share of each of the top ten products in the total merchandise imports of India from Madagascar.

Malawi

Product Code	Product Description	India's imports from Malawi (avg. 2010-2012, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)	Product Code	Product Description	India's imports from Malawi (avg. 2015-2017, \$ 1000)	Shar e (%)	India's imports from ROW (\$ 1000)
71390	Dried leguminous vegetables,	26862.1	71.7	308080.4	71390	Dried leguminous vegetables	38052.8	81.9	475110.1
520100	Cotton, not carded/combed	3310.5	8.8	270896.6	120100	Soya beans, whether/not broken	3540.9	7.6	25414.7
90240	Tea, black (fermented)	1129.1	3.0	40127.3	71339	Beans	1464.8	3.2	102167.0
843920	Machinery for making paper/paperboard	473.2	1.3	43419.2	90240	Tea, black (fermented)	1051.1	2.3	38750.0
720449	Ferrous waste & scrap	432.9	1.2	2421036.6	520100	Cotton, not carded/combed	516.1	1.1	740017.2
71310	Peas (Pisum sativum), dried, shelled	420.7	1.1	652353.3	71333	Kidney beans, incl. white pea beans	284.1	0.6	89842.8
71339	Beans	350.7	0.9	107170.3	71320	Chickpeas (garbanzos), dried, shell	186.7	0.4	809749.1

	Total imports	37448.7	100.0	433770696.4		Total imports	46462.8	100.0	397124446.0
	Top 10 Share	33896.9	90.5			Top 10 Share	45508.0	97.9	
847330	Parts & accessories of the machines	289.3	0.8	1528644.8	90230	Vigna mungo Tea, black (fermented)	133.4	0.3	540.2
410190	networks Raw hides & skins of bovine	308.4	0.8	30691.9	71331	Beans of the species	136.6	0.3	588125.1
851712	Telephones for cellular networks/for other wireless	320.1	0.9	5229090.6	410190	Raw hides & skins of bovine	141.7	0.3	25646.4

Note- Share (%) is the share of each of the top ten products in the total merchandise imports of India from Malawi.

<u>Mali</u>

Product Code	Product Description	India's imports from Mali (avg. 2010-2012, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)	Product Code	Product Description	India's imports from Mali (avg. 2015-2017, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)
520100	Cotton, not carded/combed	10386.6	63.7	265075.7	710812	Gold, unwrought or in semi-manufactured forms	78044.7	51.0	31108373.1
260111	Iron ores & concentrates	1304.5	8.0	59145.2	520100	Cotton, not carded/combed	65179.4	42.6	675353.9
80131	Cashew nuts, in shell	1052.6	6.5	879105.1	130120	Gum Arabic	4951.4	3.2	30455.9
999999	Commodities not specified according	450.6	2.8	10834140.3	80131	Cashew nuts, in shell	2494.1	1.6	1313415.1
854011	Cathode-ray television picture tube	326.1	2.0	197580.1	130190	Lac; Natural gums	1019.8	0.7	114104.4
270119	Coal other than anthracite & bituminous substances	272.8	1.7	12850700.0	410510	Tanned/crust skins of sheep/lambs	366.8	0.2	64588.9
271019	Petroleum oils & oils obtained from	237.7	1.5	3982199.5	120740	Sesamum seeds, whether/not broken	243.4	0.2	39892.8

851770	Parts of telephone sets	222.7	1.4	2788255.3	740819	Copper wire, of refined copper	178.2	0.1	526581.1
410510	Tanned/crust skins of sheep/lambs	219.5	1.4	58328.8	720449	Ferrous waste & scrap	168.0	0.1	1470769.0
130190	Lac; Natural gums	137.8	0.9	73118.3	847130	-Portable automatic data processing machines	137.0	0.1	2557163.5
	Top 10 Share	14610.9	89.6			Top 10 Share	152782.8	99.8	
	Total imports	16305.4	100.0	433791756.5		Total imports	153076.8	100.0	397015828.0

Note- Share (%) is the share of each of the top ten products in the total merchandise imports of India from Mali.

Mauritania

Product Code	Product Description	India's imports from Maurit ania (avg. 2010-2012, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)	Produc t Code	Product Description	India's imports from Maurita nia (avg. 2015-2017, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)
760200	Aluminium waste & scrap	1524.3	31.3	1114329.2	260111	Iron ores & concentrates	9493.5	40.5	428908.7
720449	Ferrous waste & scrap	1423.3	29.2	2420046.2	230120	Flours, meals & pellets of fish/of crustaceans, mollusks or other aquatic invertebrates	3707.0	15.8	13877.1
740400	Copper waste & scrap	831.2	17.1	884871.8	720449	Ferrous waste & scrap	3706.1	15.8	1467230.8
854810	Waste & scrap of primary cells	194.4	4.0	12688.4	760200	Aluminium waste & scrap	2097.5	8.9	1531527.2
30749	Cuttle fish	183.0	3.8	1135.7	740400	Copper waste & scrap	2028.5	8.7	745184.0
847330	Parts & accessories of the machines	125.5	2.6	1528795.8	854810	Waste & scrap of primary cells	2000.2	8.5	56363.2

	Total imports	4866.8	100.0	433798869.8		Total imports	23449.4	100.0	397151582.3
	Top 10 Share	4557.6	93.7			stainless steel Top 10 Share	23326.2	99.5	
410210	goats Raw skins of sheep/lambs	53.1	1.1	5575.6	720421	shavings, chips etc. Waste & scrap of	41.4	0.2	960749.3
410621	unbleached Tanned/crust hides & skins of	73.2	1.5	10270.3	720441	Ferrous turnings,	46.2	0.2	36545.8
470710	sheep/lambs, Recovered (waste & scrap)	74.0	1.5	2945.6	780200	iron Lead waste & scrap	72.0	0.3	116690.9
410510	Tanned/crust skins of	75.6	1.6	58424.7	720410	Waste & scrap of cast	133.9	0.6	24722.4

Note- Share (%) is the share of each of the top ten products in the total merchandise imports of India from Mauritania.

Mozambique

Product Code	Product Description	India's imports from Mozamb ique (avg. 2010-2012, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)	Produ ct Code	Product Description	India's imports from Mozamb ique (avg. 2015-2017, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)
270119	Coal other than anthracite & bituminous substances	87258.9	54.9	12792618.3	270119	Coal other than anthracite & bituminous substances	390889.5	66.6	13990295.4
71390	Dried leguminous vegetables	35801.1	22.5	299141.4	71390	Dried leguminous vegetables	92750.1	15.8	420412.7
80131	Cashew nuts, in shell	27889.8	17.6	852268.0	80131	Cashew nuts, in shell	44822.0	7.6	1270255.9
261400	Titanium ores & concentrates	11860.7	7.5	24034.2	270112	Bituminous coal, whether/not pulverized/ not agglomerated	12805.8	2.2	1103434.7

	Total imports	158949.7	100.0	433643902.3		Total imports	586727.2	100.0	396580565.3
	plastics Top 10 Share	183736.7	115.6			phosphates, natural Top 10 Share	585077.0	99.7	
390421	Poly(vinyl chloride), non-	900.1	0.6	800082.9	251020	Scrap Natural calcium	2833.4	0.5	331783.6
520100	whether/not Cotton, not carded/combed	1587.5	1.0	271471.0	760200	scrap Aluminium waste &	2913.6	0.5	1530711.1
130232	Mucilage & thickeners,	1678.6	1.1	4150.0	720449	Vigna mungo Ferrous waste &	6730.4	1.2	1464206.6
71339	mungo Beans	2128.9	1.3	105275.3	71331	concentrates Beans of the species	8979.2	1.5	579282.4
71331	Beans of the species Vigna	6899.1	4.3	457585.0	261400	semi-precious stones Titanium ores &	10975.4	1.9	14601.3
720449	Ferrous waste & scrap	7732.3	4.9	2413737.2	710310	Precious stones and	11377.6	1.9	581384.2

Note- Share (%) is the share of each of the top ten products in the total merchandise imports of India from Mozambique.

<u>Niger</u>

Product Code	Product Description	India's imports from Niger (avg. 2010-2012, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)	Produc t Code	Product Description	India's imports from Niger (avg. 2015-2017, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)
271111	Natural gas, liquefied	72315.6	94.8	5430374.6	271111	Natural gas, liquefied	19464.1	97.5	6580772.5
999999	Commodities not specified according	3528.1	4.6	10831062.9	780199	Unwrought lead other than refined	124.0	0.6	233392.5
390120	Polyethylene having a sp.gr. of 0.9	86.8	0.1	531975.3	760120	Aluminium alloys, unwrought	102.5	0.5	503052.1
470790	Recovered (waste & scrap) paper/paperboard	60.1	0.1	550545.5	853590	Electrical apparatus for switching	41.2	0.2	59549.2

	Total imports	76248.7	100.0	433775083.0		Total imports	19965.4	100.0	397160636.9
	Top 10 Share	76174.4	99.9			nographic/hydrological, meteorological/geophysical instruments and appliances Top 10 Share	19870.0	99.5	
130190	Lac; Natural gums	12.5	0.0	73206.1	901580	Surveying/hydrographic/ocea	19.5	0.1	123375.2
320810	Paints & varnishes	15.4	0.0	17312.1	842389	Weighing machinery	19.6	0.1	5622.3
901890	Instruments & appliances used in me	21.1	0.0	475200.2	440349	Topical wood	20.0	0.1	435780.3
810420	Magnesium waste & scrap	35.9	0.1	466.4	840999	Parts suit. for use solely/principally with engines of heading	21.3	0.1	568486.0
	•					incorporating bal			
760200	transformers having a power handling capacity exceeding 650 KVA Aluminium waste & scrap	49.5	0.1	1115837.0	848320	other Bearing housings,	23.7	0.1	36169.6
850422	Liquid dielectric	49.5	0.1	33069.2	902139	Artificial parts of the body	34.1	0.2	115866.4

Note- Share (%) is the share of each of the top ten products in the total merchandise imports of India from Niger.

<u>Nigeria</u>

Product Code	Product Description	India's imports from Nigeria (avg. 2010-2012, \$ 1000)	Shar e (%)	India's imports from ROW (\$ 1000)	Product Code	Product Description	India's imports from Nigeria (avg. 2015-2017, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)
270900	Petroleum oils & oils obtained from bituminous minerals, crude	11959176.5	94.6	107872019.8	270900	Petroleum oils & oils obtained from bituminous minerals, crude	7700824.2	88.8	64063404.5
271111	Natural gas, liquefied	504313.8	4.0	4950166.0	271111	Natural gas, liquefied	775695.2	9.0	5811565.4

	Total imports	12645582.7	100.0	421194674.9		Total imports	8671345.4	100.0	388505868.7
	Top 10 Share	12556525.4	99.3			Top 10 Share	8601544.8	99.2	
390120	Polyethylene having a sp.gr. of 0.9	7596.6	0.1	526939.8	780199	Unwrought lead other than refined	4784.3	0.1	228649.5
780199	Unwrought lead other than refined	9722.2	0.1	219457.8	180100	Cocoa beans, whole/broken, raw/roasted	5235.6	0.1	56459.3
440349	Topical wood	10194.0	0.1	642979.6	440349	Topical wood	6402.3	0.1	429384.6
760200	Aluminium waste & scrap	10368.2	0.1	1105485.3	91010	Ginger	6530.3	0.1	6366.8
440399	Wood, in the rough	11669.7	0.1	1001942.7	120740	Sesamum seeds, whether/not broken	6727.4	0.1	33246.5
760120	Aluminium alloys, unwrought	13012.8	0.1	316146.1	760120	Aluminium alloys, unwrought	7481.3	0.1	495605.0
80131	Cashew nuts, in shell	14657.4	0.1	865500.3	760200	Aluminium waste & scrap	31025.9	0.4	1502598.8
841940	Distilling/rectifying plant	15814.1	0.1	58354.9	80131	Cashew nuts, in shell	56838.2	0.7	1258239.6

Note- Share (%) is the share of each of the top ten products in the total merchandise imports of India from Nigeria.

Rwanda

Product Code	Product Description	India's imports from Rwand a (avg. 2010-2012, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)	Product Code	Product Description	India's imports from Rwanda (avg. 2015-2017, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)
90111	Coffee, not roasted, not decaffeinated	168.8	29.8	92519.3	780199	Unwrought lead other than refined,	1212.7	47.8	232221.1
90240	Tea, black (fermented)	161.2	28.5	41202.7	760200	Aluminium waste & scrap	410.5	16.2	1533214.2
90230	Tea, black (fermented)	102.3	18.0	2979.2	261100	Tungsten ores & concentrates	205.8	8.1	358.7
720410	Waste & scrap of cast iron	77.1	13.6	64203.7	90240	Tea, black (fermented)	139.4	5.5	39708.2

	Total imports	566.8	100.0	433802653.8		Total imports	2536.1	100.0	397165133.4
	Top 10 Share	566.8	100.0			scrap Top 10 Share	2453.3	96.7	
10690	natural sands Live animals	0.1	0.0	2826.8	720449	alloys (brass) Ferrous waste &	30.4	1.2	1470926.8
250620	Quartzite, other than	0.2	0.0	74.8	740321	concentrates Copper-zinc base	43.5	1.7	6291.8
261790	according Ores & concentrates	0.8	0.1	473.6	260700	semi-precious stones Lead ores &	48.1	1.9	8234.9
999999	precious stones Commodities not specified	2.8	0.5	10834589.1	710310	(fermented) Precious stones and	108.9	4.3	592653.0
710310	Capsicum/Pimen Precious stones and semi-	12.6	2.2	208428.4	90230	scrap Tea, black	118.1	4.7	634.2
90420	Fruits of the genera	40.9	7.2	3662.5	740400	Copper waste &	136.1	5.4	747076.4

Note- Share (%) is the share of each of the top ten products in the total merchandise imports of India from Rwanda.

Senegal

Product Code	Product Description	India's imports from Senegal (avg. 2010-2012, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)	Product Code	Product Description	India's imports from Senegal (avg. 2015-2017, \$ 1000)	Shar e (%)	India's imports from ROW (\$ 1000)
280920	Phosphoric acid & poly-phosphoric acid	264750.5	77.9	1398492.2	280920	Phosphoric acid & poly-phosphoric acid	286341.6	77.0	1335943.0
720449	Ferrous waste & scrap	18687.0	5.5	2402782.5	80131	Cashew nuts, in shell	38808.9	10.4	1276269.0
80131	Cashew nuts, in shell	14993.7	4.4	865164.0	720449	Ferrous waste & scrap	18343.7	4.9	1452593.2
251020	Natural calcium phosphates, natural	6330.3	1.9	506663.6	760200	Aluminium waste & scrap	6652.4	1.8	1526972.3
151190	Palm oil, other than crude	5877.0	1.7	1238484.2	251020	Natural calcium phosphates, natural	4194.4	1.1	329931.8
760200	Aluminium waste & scrap	5459.4	1.6	1110394.2	740400	Copper waste & scrap	3339.3	0.9	743873.2

	Total imports	339823.4	100.0	433481929.6		Total imports	371935.6	100.0	396798420.3
	iron/non-alloy steel Top 10 Share	325918.1	95.9			primary cells Top 10 Share	363762.4	97.8	
720851	Flat-rolled products of	997.8	0.3	497711.4	854810	sheep/lambs Waste & scrap of	1312.0	0.4	57051.3
740400	of the distil Copper waste & scrap	1281.1	0.4	884144.8	410510	concentrates Tanned/crust skins of	1416.7	0.4	63294.4
270799	concentrates Oils & other products	2555.6	0.8	64216.8	261510	Zirconium ores &	1419.2	0.4	65884.2
260111	Iron ores &	4985.8	1.5	57125.8	130120	Gum Arabic	1934.3	0.5	33473.0

Note- Share (%) is the share of each of the top ten products in the total merchandise imports of India from Senegal.

Sierra Leone

Product Code	Product Description	India's imports from Sierra Leone (avg. 2010-2012, \$ 1000)	Shar e (%)	India's imports from ROW (\$ 1000)	Product Code	Product Description	India's imports from Sierra Leone (avg. 2015-2017, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)
720449	Ferrous waste & scrap	2867.6	39.5	2418601.9	720449	Ferrous waste & scrap	6820.3	32.7	1464116.6
760200	Aluminium waste & scrap	1161.4	16.0	1114692.1	710231	Diamonds, non-industrial, un- worked/simply sawn cleaved/bruted	4641.6	22.3	16480467.4
740400	Copper waste & scrap	616.8	8.5	884809.2	260600	Aluminium ores & concentrates	3643.1	17.5	107182.3
710231	Diamonds, non-industrial, un- worked/simply sawn cleaved/bruted	331.8	4.6	12310402.2	760200	Aluminium waste & scrap	2057.4	9.9	1531567.3

851770	Parts of telephone sets	305.5	4.2	2788376.2	854810	Waste & scrap of primary cells	1297.4	6.2	57065.9
720410	Waste & scrap of cast iron	303.8	4.2	64128.1	261400	Titanium ores & concentrates	1005.9	4.8	24570.9
741129	Copper tubes & pipes, of copper all	226.5	3.1	75945.7	740400	Copper waste & scrap	722.0	3.5	746490.4
854810	Waste & scrap of primary cells	189.8	2.6	12691.4	854231	Electronic integrated circuits	224.6	1.1	816436.5
261400	Titanium ores & concentrates	189.2	2.6	35705.7	780200	Lead waste & scrap	143.6	0.7	116619.4
720429	Waste & scrap of alloy steel other	153.4	2.1	52863.0	261590	Niobium/tantalum/vanadium ores & concentrates	60.9	0.3	4271.3
	Top 10 Share	6345.6	87.3			Top 10 Share	20616.9	99.0	
	Total imports	7266.3	100.0	433797020.0		Total imports	20829.7	100.0	1588596655.0

Note- Share (%) is the share of each of the top ten products in the total merchandise imports of India from Sierra Leone.

Somalia

Product Code	Product Description	India's imports from Somalia (avg. 2010-2012, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)	Product Code	Product Description	India's imports from Somalia (avg. 2015-2017, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)
120740	Sesamum seeds, whether/not broken	3878.2	45.0	3358.4	120740	Sesamum seeds, whether/not broken	10617.4	81.4	29356.5
844790	Knitting machines	953.5	11.1	223326.0	130190	Lac; Natural gums	793.1	6.1	114331.1
410190	Raw hides & skins of bovine	635.3	7.4	30159.4	410190	Raw hides & skins of bovine	385.5	3.0	25355.3
410229	Raw skins of sheep/lambs	560.8	6.5	6729.2	780200	Lead waste & scrap	248.7	1.9	116680.1
410692	Tanned/crust hides & skins	348.0	4.0	1513.8	271490	Bitumen & asphalt, natural; asphalt	169.6	1.3	8217.8

71331	Beans of the species Vigna mungo	281.1	3.3	464390.4	410229	Raw skins of sheep/lambs	124.5	1.0	6833.1
410530	Tanned/crust skins of sheep/lambs	250.8	2.9	10893.4	410419	Tanned/crust hides & skins of bovine	112.7	0.9	100439.4
410419	Tanned/crust hides & skins of bovine	214.2	2.5	125988.3	410510	Tanned/crust skins of sheep/lambs	92.8	0.7	64649.3
410510	Tanned/crust skins of sheep/lambs	206.7	2.4	58268.4	720410	Waste & scrap of cast iron	91.7	0.7	24736.5
410691	Tanned/crust hides & skins	157.9	1.8	17102.9	760110	Aluminium, not alloyed, unwrought	77.9	0.6	290111.4
	Top 10 Share	7486.5	86.8			Top 10 Share	12713.8	97.5	
	Total imports	8626.6	100.0	433796089.6		Total imports	13044.6	100.0	397154768.5

Note- Share (%) is the share of each of the top ten products in the total merchandise imports of India from Somalia.

Sudan

Product Code	Product Description	India's imports from Sudan (avg. 2010-2012, \$ 1000)	Share (%)	Product Code	Product Description	India's imports from Sudan (avg. 2015-2017, \$ 1000)	Share (%)	India's imports from ROW
270900	Petroleum oils & oils obtained from bituminous minerals, crude	134680.0	75.1	270900	Petroleum oils & oils obtained from bituminous minerals, crude	135497.2	48.7	71628731.5
440349	Topical wood	10488.7	5.9	71390	Dried leguminous vegetables	33870.0	12.2	479292.9
720449	Ferrous waste & scrap	9030.7	5.0	120799	Oil seeds & oleaginous fruits	31107.8	11.2	25650.9

130120	Gum Arabic	4772.5	2.7	120740	Sesamum seeds, whether/not broken	18536.4	6.7	21437.6
760200	Aluminium waste & scrap	3272.8	1.8	130120	Gum Arabic	17780.5	6.4	17626.8
120799	Oil seeds & oleaginous fruits	2371.6	1.3	440349	Topical wood	15608.0	5.6	420179.0
740400	Copper waste & scrap	2091.8	1.2	710812	Gold, unwrought or in semi-manufactured forms	9167.6	3.3	31183361.9
520100	Cotton, not carded/combed	1326.8	0.7	71320	Chickpeas (garbanzos), dried, shell	8821.1	3.2	801114.6
120740	Sesamum seeds, whether/not broken	983.1	0.6	760200	Aluminium waste & scrap	3307.3	1.2	1530317.4
410190	Raw hides & skins of bovine	871.1	0.5	440729	Wood sawn/chipped lengthwise, slice	2631.9	1.0	123563.7
	Top 10 Share	169889.2	94.7		Top 10 Share	276327.7	99.3	
	Total imports	179313.1	100.0		Total imports	278171.2	100.0	396889121.2

Note- Share (%) is the share of each of the top ten products in the total merchandise imports of India from Sudan.

Togo

Produc t Code	Product Description	India's imports from Togo (avg. 2010-2012, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)	Product Code	Product Description	India's imports from Togo (avg. 2015-2017, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)
251020	Natural calcium phosphates, natural	70767.2	46.7	438006.5	270900	Petroleum oils & oils obtained from bituminous minerals, crude	65478.4	29.4	71742402.6
251010	Natural calcium phosphates, natural	31046.5	20.5	729837.7	251020	Natural calcium phosphates, natural	42505.9	19.1	290222.2
720449	Ferrous waste & scrap	11096.5	7.3	2410373.0	251010	Natural calcium phosphates, natural	27790.7	12.5	477827.7

	Total imports	151607.1	100.0	433663810.2		Total imports	223088.0	100.0	443907719.6
	Top 10 Share	144605.2	95.4			primary cells Top 10 Share	217787.0	97.6	
291612	Esters of acrylic acid	1709.6	1.1	267131.6	854810	Waste & scrap of	3553.6	1.6	54809.7
720839	Flat-rolled products of iron/non-alloy steel	1797.8	1.2	269354.6	520100	Cotton, not carded/combed	6209.6	2.8	734323.7
760200	Aluminium waste & scrap	2019.2	1.3	1113834.3	740400	scrap Copper waste & scrap	6586.9	3.0	740625.6
80131	whole/broken, raw/roasted Cashew nuts, in shell	3296.4	2.2	876861.4	760200	Aluminium waste &	8891.1	4.0	1524733.6
180100	Cocoa beans,	6554.3	4.3	55277.8	80131	Cashew nuts, in shell	14419.4	6.5	1300658.4
440349	Topical wood	8010.6	5.3	645163.0	440349	Topical wood	15773.5	7.1	420013.4
520100	Cotton, not carded/combed	8307.1	5.5	269231.1	720449	Ferrous waste & scrap	26578.0	11.9	1444359.0

Note- Share (%) is the share of each of the top ten products in the total merchandise imports of India from Togo.

Tanzania

Produc t Code	Product Description	India's imports from Tanzania (avg. 2010-2012, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)	Product Code	Product Description	India's imports from Tanzania (avg. 2015-2017, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)
80131	Cashew nuts, in shell	131224.2	30.3	748933.5	710812	Gold, unwrought or in semi-manufactured forms	477460.8	48.4	30708957.0
710812	Gold, unwrought or in semi-manufactured forms	95665.5	22.1	45397602.1	80131	Cashew nuts, in shell	235999.0	23.9	1079078.9

71390	Dried leguminous vegetables	45956.1	10.6	288986.4	71390	Dried leguminous vegetables	85409.4	8.7	427753.4
71331	Beans of the species Vigna mungo	20302.6	4.7	444181.5	71320	Chickpeas (garbanzos), dried, shell	31447.7	3.2	778488.1
71320	Chickpeas (garbanzos), dried, shell	15344.2	3.6	168005.4	71331	Beans of the species Vigna mungo	31374.6	3.2	556887.1
520100	Cotton, not carded/combed	14859.0	3.4	257141.1	90700	Cloves	27103.3	2.7	122974.9
90700	Cloves	11454.5	2.7	60694.6	440729	Wood sawn/chipped lengthwise, slice	17429.8	1.8	108765.8
720449	Ferrous waste & scrap	8889.3	2.1	2412580.2	710310	Precious stones and semi-precious stones	10925.7	1.1	581836.2
710310	Precious stones and semi- precious stones	8280.1	1.9	200160.9	330129	Essential oils	8763.6	0.9	85750.1
290243	p-Xylene	7852.3	1.8	613758.9	520100	Cotton, not carded/combed	8510.7	0.9	732022.5
	Top 10 Share	359827.8	83.2			Top 10 Share	934424.5	94.6	
	Total imports	432529.3	100.0	433459266.6		Total imports	987578.8	100.0	396184931.0

Note- Share (%) is the share of each of the top ten products in the total merchandise imports of India from Tanzania.

<u>Uganda</u>

Product Code	Product Description	India's imports from Uganda (avg. 2010-2012, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)	Produc t Code	Product Description	India's imports from Uganda (avg. 2015-2017, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)
90111	Coffee, not roasted, not decaffeinated	9166.2	36.1	83409.3	90111	Coffee, not roasted, not decaffeinated	17851.4	31.2	106810.0
520100	Cotton, not carded/combed	3604.3	14.2	269597.3	71390	Dried leguminous vegetables	7374.3	12.9	505788.6

	Total imports	25409.6	100.0	433783529.6		Total imports	57276.3	100.0	397112218.3
	Top 10 Share	20505.0	80.7			Top 10 Share	51677.4	90.2	
200400	concentrates	303.0	1.4	1/940.2	440729	Wood sawn/chipped lengthwise, slice	713.3	1.0	123260.3
260400	in irregularly wound coils, of iron or non-alloy steel Nickel ores &	363.8	1.4	17948.2	440729	skins of bovine	915.3	1.6	125280.3
721399	Bars & rods, hot-rolled,	371.0	1.5	84735.6	410419	mungo Tanned/crust hides &	957.5	1.7	99519.4
410691	Tanned/crust hides & skins	501.0	2.0	16707.2	71331	Beans of the species Vigna	1368.0	2.4	586893.6
730619	Other line pipe of a kind used for	513.1	2.0	11129.3	40590	Fats & oils derived from milk, other	2681.6	4.7	5776.1
480100	mechanical appliances Newsprint, in rolls/sheets	566.0	2.2	872193.1	330129	whole/broken, raw/roasted Essential oils	3957.3	6.9	90556.5
847989	residues Other machines &	781.9	3.1	837580.3	180100	Cocoa beans,	4436.3	7.8	57258.6
230630	raw/roasted Oil-cake & other solid	1564.1	6.2	4243.9	440349	Topical wood	5358.7	9.4	430428.2
180100	Cocoa beans, whole/broken,	3073.7	12.1	56573.6	520100	Cotton, not carded/combed	6777.1	11.8	733756.2

Note-Share (%) is the share of each of the top ten products in the total merchandise imports of India from Uganda.

Zambia

Product Code	Product Description	India's imports from Zambia (avg. 2010-2012, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)	Product Code	Product Description	India's imports from Zambia (avg. 2015-2017, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)
-----------------	---------------------	---	-----------	------------------------------------	-----------------	---------------------	---	-----------	---

740200	Unrefined copper	94534.4	54.6	26804.9	740200	Unrefined copper	606468.5	86.9	78478.5
740311	Cathodes & sections of cathodes	20289.5	11.7	44450.7	710310	Precious stones and semi-precious stones	67894.5	9.7	524867.4
260300	Copper ores & concentrates	20055.5	11.6	4902180.7	740311	Cathodes & sections of cathodes	8659.2	1.2	158965.6
710310	Precious stones and semi- precious stones	16523.1	9.5	191917.9	260200	Manganese ores & concentrates	4042.6	0.6	442723.8
740319	Unwrought products of refined copper	4352.6	2.5	37394.7	520100	Cotton, not carded/combed	3563.9	0.5	736969.3
260200	Manganese ores & concentrates	4228.8	2.4	421100.6	780199	Unwrought lead other than refined	3246.4	0.5	231269.5
520100	Cotton, not carded/combed	4181.5	2.4	267818.6	710391	Rubies, sapphires & emeralds	959.9	0.1	1157428.6
740400	Copper waste & scrap	3535.5	2.0	881890.4	261690	Precious metal ores & concentrates	386.9	0.1	190626.6
810520	Cobalt mattes & other intermediate	2224.3	1.3	26669.4	410419	Tanned/crust hides & skins of bovine	355.8	0.1	100121.2
740710	Bars, rods & profiles, of refined copper	487.5	0.3	272860.2	810520	Cobalt mattes & other intermediate	330.0	0.1	27044.7
	Top 10 Share	170412.8	98.4			Top 10 Share	695907.6	99.7	
	Total imports	173243.0	100.0	433647656.6		Total imports	698053.5	100.0	396473398.9

Note- Share (%) is the share of each of the top ten products in the total merchandise imports of India from Zambia.

Bangladesh

Product Code	Product Description	India's imports from Banglade sh (avg. 2010-2012, \$ 1000)	Shar e (%)	India's imports from ROW (\$ 1000)	Produc t Code	Product Description	India's imports from Banglade sh (avg. 2015-2017, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)
-----------------	---------------------	--	------------------	---	------------------	---------------------	--	-----------	------------------------------------

	Total imports	533962.9	100.0	433301409.1		Total imports	690270.6	100.0	396531101.7
	Top 10 Share	336119.0	63.0			& both persons and goods Top 10 Share	311572.6	45.1	
251710	Pebbles/gravel/broken/crushed stone	9674.3	1.8	7199.5	890190	Vessels for the transport of goods	16376.9	2.4	381450.6
720421	Waste & scrap of stainless steel	12103.0	2.3	818191.1	620520	Men's/boys' shirts	17004.5	2.5	23035.7
740400	Copper waste & scrap	15936.8	3.0	869489.2	890120	Tankers	18642.2	2.7	412482.3
252329	Portland cement	19007.7	3.6	34888.1	271011	Light petroleum oils & preparations	22218.7	3.2	1340686.9
271011	Light petroleum oils & preparations	26368.4	4.9	2093174.6	531010	Woven fabrics of jute/other textile	23895.1	3.5	9667.3
30710	Yarn of jute/of other textile bast	32170.2	6.0	240.8	780199	Unwrought lead other than refined	25331.5	3.7	208102.3
80290	Nuts, n.e.s. in 08.01 & 08.02, fres	40633.8	7.6	19773.8	630510	Sacks & bags, of a kind used for th	32575.0	4.7	14190.6
80269	Fish,n.e.s. in 03.02, fresh/chilled	52459.2	9.8	959.8	530710	brace o Yarn of jute/of other textile bast	44738.1	6.5	20.4
530510	Sacks & bags, of a kind used for packing of goods	56716.6	10.6	17439.0	620342	roasted Men's/boys' trousers, bib &	49661.8	7.2	24319.8
30310	Jute & other textile bast fibres, raw or roasted	71049.1	13.3	13.1	530310	Jute & other textile bast fibres, raw or	61128.9	8.9	83.4

Note- Share (%) is the share of each of the top ten products in the total merchandise imports of India from Bangladesh.

Bhutan

Product Code	Product Description	India's imports from Bhutan (avg. 2010-2012, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)	Product Code	Product Description	India's imports from Bhutan (avg. 2015-2017, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)
720221	Ferro-silicon, containing by weight	86719.9	43.7	73802.0	720221	Ferro-silicon, containing by weight	92994.0	47.5	67812.8
740819	Copper wire, of refined copper	20698.0	10.4	36037.7	271600	Electrical energy	38505.7	19.7	0.0
284910	Carbides, of calcium, whether/not chemically defined	16668.1	8.4	25918.3	284910	Carbides, of calcium, whether/not chemically defined	13988.0	7.1	25146.1
740319	Unwrought products of refined copper	10860.7	5.5	31605.2	284920	Carbides, of silicon, whether/not chemically defined	9495.1	4.9	26606.9
721430	Bars & rods of iron/non-alloy steel	10056.0	5.1	1448.0	392099	Plates, sheets, film, foil & strip etc. of other plastics	4345.0	2.2	135042.3
722720	Bars & rods, hot-rolled, in irregularly wound coils, of iron or non-alloy steel	8787.1	4.4	517.4	251810	Dolomite, not calcined/sintered	3656.1	1.9	38939.4
720610	Iron & non-alloy steel in ingots	6028.6	3.0	4798.2	740819	Copper wire, of refined copper	2625.0	1.3	524015.5
720229	Ferro-silicon, other than that cont	5890.8	3.0	23192.2	441011	Particle board of wood	1952.5	1.0	28152.6
284920	Carbides, of silicon, whether/not chemically defined	4271.1	2.2	27050.5	720719	Semi-finished products of iron/non-alloy steel	1920.4	1.0	15300.6
392099	Plates, sheets, film, foil & strip etc. of other plastics	4034.1	2.0	92236.8	220720	Ethyl alcohol & other spirits	1859.5	1.0	195369.7
	Top 10 Share	174014.2	87.7			Top 10 Share	171341.3	87.5	
	Total imports	198449.6	100.0	433615972		Total imports	195895.3	100.0	396990844.6

Note- Share (%) is the share of each of the top ten products in the total merchandise imports of India from Bhutan.

Cambodia

Product Code	Product Description	India's imports from Cambo dia (avg. 2010-2012, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)	Product Code	Product Description	India's imports from Cambod ia (avg. 2015-2017, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)
151110	Palm oil, crude	3845.6	34.6	5132470.0	440349	Topical wood	7359.4	14.4	430880.7
400122	Technically spec. natural rubber	1399.8	12.6	368945.8	400121	Natural rubber (excl. latex)	7195.8	14.1	124962.3
400121	Natural rubber (excluding latex)	892.9	8.0	341017.5	440399	Wood, in the rough	6584.0	12.9	780756.1
620342	Men's/boys' trousers, etc., not knit, cotton	366.4	3.3	31582.6	250590	Natural sands, other	2751.7	5.4	2212.5
80131	Cashew nuts, in shell	322.7	2.9	879835.1	110814	Manioc (cassava) starch	2639.5	5.2	5505.5
640399	Other footwear without outer soles	282.8	2.6	24576.0	871200	Bicycles & other cycles	2041.4	4.0	36999.1
120220	Ground-nuts, not roasted/othw. cook	246.2	2.2	88.0	611030	Jerseys, pullovers, cardigans, etc. of man-made fibres	1315.4	2.6	18263.8
852851	Other monitors, of a kind solely/pr	231.4	2.1	430400.4	210690	Food preparations	1172.4	2.3	58103.4
610469	Women's/girls' trousers, bib & brace overalls, breeches and shorts of other textile materials	178.8	1.6	632.7	640399	Other footwear without outer soles	1021.7	2.0	42688.3
611030	Jerseys, pullovers, cardigans, etc. of man-made fibres	170.5	1.5	4840.6	610990	T-shirts, singlet's & other vests, etc. of other textile materials	942.0	1.8	19791.5
	Top 10 Share	7937.0	71.5			Top 10 Share	33023.5	64.6	
	Total imports	11105.4	100.0	433794127.3		Total imports	51152.8	100.0	397122597.8

Note- Share (%) is the share of each of the top ten products in the total merchandise imports of India from Cambodia.

Laos

Product Code	Product Description	India's imports from Laos (avg. 2010-2012, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)	Produc t Code	Product Description	India's imports from Laos (avg. 2015-2017, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)
260300	Copper ores & concentrates	77397.8	98.2	4831468.0	710812	Gold	85240.1	39.1	31101177.6
848190	Parts of the appliances of 84.81	287.2	0.4	223230.7	260300	Copper ores & concentrates	58495.6	26.8	3426428.5
846210	Forging/die-stamping machines	137.7	0.2	109007.8	440399	Wood, in the rough	13268.2	6.1	774071.9
720449	Ferrous waste & scrap	119.6	0.2	2421429.7	710813	Gold	11071.0	5.1	175697.5
740939	Plates, sheets & strip, of copper-tin base alloys not in coils	78.8	0.1	7328.7	710691	Silver	8469.7	3.9	2786038.6
130190	Lac; Natural gums	77.4	0.1	73132.9	780199	Unwrought lead other than refined	8137.9	3.7	228008.5
721119	Flat-rolled products of iron/non-alloy steel	62.7	0.1	5106.2	261610	Silver ores & concentrates	7781.2	3.6	38439.4
440349	Topical wood	58.7	0.1	653114.9	441299	Plywood, veneered panels & similar laminated wood	3647.2	1.7	35230.5
848320	Bearing housings, incorporating ball/roller bearings	53.8	0.1	35687.7	440349	Topical wood	3062.2	1.4	433745.5
240220	Cigarettes containing tobacco	40.2	0.1	16130.5	841950	Heat exchange units	3039.0	1.4	180557.0
	Top 10 Share	78314.0	99.4			Top 10 Share	202212.1	92.8	
	Total imports	78829.8	100.0	433724847.2		Total imports	217961.7	100.0	396978884.7

Note- Share (%) is the share of each of the top ten products in the total merchandise imports of India from Laos.

Myanmar

Product Code	Product Description	India's imports from Myanmar (avg. 2010-2012, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)	Produ ct Code	Product Description	India's imports from Myanmar (avg. 2015-2017, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)
71331	Beans of the species Vigna mungo	326038.0	25.2	138446.1	71331	Beans of the species Vigna mungo	431465.0	43.3	156796.9
440349	Topical wood	289519.2	22.4	363654.4	71390	Dried leguminous vegetables	225965.0	22.7	287198.2
440399	Wood, in the rough	268167.6	20.7	745444.8	71339	Beans	56261.8	5.6	46881.7
71390	Dried leguminous vegetables	196338.5	15.2	138604.0	880240	Airplanes & other aircraft, of an un- laden weight exceeding 15000 kg	50396.2	5.1	1199009.8
71339	Beans	65487.2	5.1	41916.9	440839	Sheets for veneering	40986.4	4.1	22390.3
71320	Chickpeas (garbanzos), dried, shell	15876.2	1.2	167473.4	440890	Sheets for veneering	33816.6	3.4	65277.3
71333	Kidney beans, incl. white pea beans	7828.8	0.6	63910.8	440729	Wood sawn/chipped lengthwise, slice	25122.5	2.5	101073.1
281410	Anhydrous ammonia	7370.2	0.6	958883.7	71320	Chickpeas (garbanzos), dried, shell	15512.1	1.6	794423.7
50100	Human hair, un-worked, waste of human hair	6417.6	0.5	15715.3	720260	Ferro-nickel, in granular/powder form	13242.2	1.3	274994.1
271019	Petroleum oils & oils obtained from bituminous minerals	5323.8	0.4	3978729.5	80290	Nuts, n.e.s. in 08.01 & 08.02, fres	12577.3	1.3	105237.2
	Top 10 Share	1188367.2	91.9			Top 10 Share	905344.0	90.8	

Total imports	1293421.4 100	00.0 432559394.2	Total imports	996775.0 100	.0 396220867.6
N	2.1				

Note- Share (%) is the share of each of the top ten products in the total merchandise imports of India from Myanmar.

Nepal

Product Code	Product Description	India's imports from Nepal (avg. 2010-2012, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)	Product Code	Product Description	India's imports from Nepal (avg. 2015-2017, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)
392690	Other articles of plastics	41107.6	8.8	495632.9	220290	Non-alcoholic beverages	75846.2	16.5	33351.0
220290	Non-alcoholic beverages	30691.8	6.6	15255.3	90830	Cardamoms	41086.8	8.9	11331.3
721049	Flat-rolled products of iron/non-alloy steel	30152.7	6.4	287995.0	392690	Other articles of plastics	34319.6	7.5	743042.6
721041	Flat-rolled products of iron/non-alloy steel	21520.4	4.6	146.4	640419	Footwear (excl. waterproof) with outer soles of rubber/plastics and with upper of textile materials	21517.8	4.7	74195.9
550921	Yarn other than sewing thread	20450.4	4.4	16798.8	630510	Sacks & bags	17227.0	3.7	35280.9
630510	Sacks & bags, of a kind used for packing of goods	16913.5	3.6	57242.1	721720	Wire of iron/non-alloy steel, plate	15936.8	3.5	24100.6
90240	Tea, black (fermented)	15265.2	3.3	25991.2	531010	Woven fabrics of jute/other textile	14410.4	3.1	23955.4
90830	Cardamoms	14822.7	3.2	1685.3	380610	Rosin & resin acids	12673.2	2.8	56541.8
730690	Tubes, pipes & hollow profiles of iron or steel	14339.6	3.1	85442.7	999999	Commodities not specified according	11232.7	2.4	8636423.6
721720	Wire of iron/non-alloy steel, plate	13571.5	2.9	18891.5	90240	Tea, black (fermented)	10278.0	2.2	29523.1

Top 10 Share	218835.3	46.7		Top 10 Share	254528.5	55.3	
Total imports	468243.5	100.0	433362427	Total imports	460673.8	100.0	396737817.5

Note- Share (%) is the share of each of the top ten products in the total merchandise imports of India from Nepal.

Timor-Leste

Product Code	Product Description	India's imports from Timor-Leste (avg. 2010-2012, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)	Product Code	Product Description	India's imports from Timor-Leste (avg. 2015-2017, \$ 1000)	Shar e (%)	India's imports from ROW (\$ 1000)
999999	Commodities not specified according	10466.1	82.0	10827613.5	80290	Nuts, n.e.s. in 08.01 & 08.02, fres	96.8	43.4	117750.0
851770	Parts of telephone sets	899.4	7.1	2788178.2	90700	Cloves	55.0	24.7	150059.8
382490	Chemical products	767.0	6.0	390635.4	847990	Parts of Machines & mechanical appliances	36.4	16.3	318989.3
720449	Ferrous waste & scrap	341.3	2.7	2421355.7	760200	Aluminium waste & scrap	21.5	9.6	1533617.5
760200	Aluminium waste & scrap	121.8	1.0	1115772.3	940180	Seats (excl. of 64.02), n.e.s. in 9	7.2	3.2	38491.8
853641	Relays, for a voltage not >60V	51.2	0.4	35327.6	940389	Furniture of other materials, incl.	2.8	1.3	107184.0
290919	Acyclic ethers other than diethyl e	37.3	0.3	36181.2	732429	Baths of iron other than cast iron/	1.2	0.5	8677.1
960831	Indian ink drawing pens	35.0	0.3	2249.6	721790	Wire of iron/non-alloy steel	0.7	0.3	41593.3
844610	Weaving machines (looms) for weaving fabrics of width less than/equal to 30cm	10.1	0.1	23719.8	853710	Boards, panels, consoles, desks, cabinets and other bases	0.6	0.3	356115.1
820790	Interchangeable tools for hand tool	6.3	0.1	162154.0	90611	Cinnamon	0.4	0.2	4618.9
	Top 10 Share	12735.4	99.8			Top 10 Share	222.6	99.8	

Total imports 12	2761.3 100	0.0	433795069.0	Total imports	223.0	100.0	397167185.9
------------------	------------	-----	-------------	---------------	-------	-------	-------------

Note- Share (%) is the share of each of the top ten products in the total merchandise imports of India from Timor-Leste.

Yemen

Product Code	Product Description	India's imports from Yemen (avg. 2010-2012, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)	Product Code	Product Description	India's imports from Yemen (avg. 2015-2017, \$ 1000)	Share (%)	India's imports from ROW (\$ 1000)
270900	Petroleum oils & oils obtained from bituminous minerals, crude	1165316.6	87.0	118665879.8	271111	Natural gas, liquefied	118743.5	62.0	6547679.4
271111	Natural gas, liquefied	104490.1	7.8	5349989.7	270900	Petroleum oils & oils obtained from bituminous minerals, crude	59874.8	31.3	71724312.2
271019	Petroleum oils & oils obtained from bituminous minerals	46900.0	3.5	3935378.7	780199	Unwrought lead other than refined,	2947.9	1.5	230485.9
780199	Unwrought lead	7275.1	0.5	221904.9	890190	Vessels for the transport of goods & persons	1874.6	1.0	386284.7
740811	Copper wire, of refined copper of w	2552.8	0.2	314783.3	780200	Lead waste & scrap	1305.2	0.7	116327.9
410221	Raw skins of sheep/lambs, whether or not with wool on or split, other	1794.0	0.1	17704.2	854810	Waste & scrap of primary cells	1290.0	0.7	57933.3
470790	Recovered (waste & scrap) paper/paperboard	1656.7	0.1	548908.8	410190	Raw hides & skins of bovine	1151.8	0.6	24589.0
410190	Raw hides & skins of bovine	1362.0	0.1	29432.6	410221	Raw skins of sheep/lambs, whether or not with	752.0	0.4	6780.3

30741	Cuttle fish	1137.2	0.1	4022.3	760200	wool on or split, other Aluminium waste &	416.7	0.2	1533346.9
760200	Aluminium waste & scrap	718.5	0.1	1115135.0	470790	scrap Recovered (waste & scrap)	408.5	0.2	703951.1
	Top 10 Share	1333203.1	99.5			paper/paperboard Top 10 Share	188765.0	98.6	
	Total imports	1339376.0	100.0	432468031.9		Total imports	191551.8	100.0	397079624.4

Note- Share (%) is the share of each of the top ten products in the total merchandise imports of India from Yemen.

Source: Own Calculations, UN Comtrade

ANNEX B (Services)

[Note: The annex tables report the top ten services exported by select African (31) and Asian (7) countries to India averaged over three-year periods from 2000 to 2002 and 2010 to 2012. These tables also mention the share of each of the top ten commodities in the total services exported by these same countries to India along with India's services imports from rest of the world (ROW).]

Angola

S.No.	Product Description	Angola's exports to India (avg. 2000-2002, \$ millions)	Share (%)	India's imports from ROW(\$ millions)	Angola's exports to India (avg. 2010-2012, \$ millions)	Share (%)	India's imports from ROW(\$ millions)
1	Transportation	0.4	11.1	3995.4	80.8	22.2	23370.1
2	Travel	0.4	13.2	4360.4	21.4	5.9	21737.2
3	Communications services	0.1	3.5	198.4	9.5	2.6	1058.3
4	Construction	1.6	50.5	342.4	102.2	28.0	2087.4
5	Insurance services	0.0	0.0	310.0	0.0	0.0	1032.2
6	Financial services	0.1	3.7	602.9	8.9	2.4	3429.7
7	Computer and information services	0.0	0.2	386.5	1.8	0.5	2303.4
8	Royalties and license fees	0.0	0.2	291.1	1.9	0.5	3006.8
9	Other business services	0.5	15.9	2369.1	134.2	36.8	17893.0
10	Personal, cultural and recreational services	0.0	0.0	31.5	0.1	0.0	404.2
11	Government services, n.i.e.	0.1	1.8	393.7	3.7	1.0	1519.4
12	Total services	3.2	100.0	13281.3	364.4	100.0	77841.8

Note- Share (%) is the share of each of the top ten services in the total exports of services of Angola to India.

Benin

S.No.	Product Description	Benin's exports to India (avg. 2000- 2002, \$ millions)	Share (%)	India's imports from ROW(\$ millions)	Benin's exports to India (avg. 2010-2012, \$ millions)	Share (%)	India's imports from ROW(\$ millions)
1	Transportation	1.1	7.3	3994.7	7.3	11.8	23443.5
2	Travel	7.1	47.7	4353.8	24.4	39.2	21734.2
3	Communications services	0.4	2.9	198.1	2.7	4.3	1065.2
4	Construction	0.0	0.1	344.0	0.1	0.1	2189.5
5	Insurance services	0.1	0.3	310.0	0.2	0.3	1032.1
6	Financial services	1.0	6.7	602.0	5.1	8.2	3433.6
7	Computer and information services	0.9	6.0	385.6	2.2	3.5	2303.1
8	Royalties and license fees	0.0	0.1	291.0	0.1	0.2	3008.5
9	Other business services	2.6	17.2	2367.1	16.0	25.8	18011.2
10	Personal, cultural and recreational services	0.0	0.1	31.5	0.3	0.5	404.0
11	Government services, n.i.e.	1.7	11.7	392.1	3.9	6.3	1519.1
12	Total services	14.9	100.0	13269.7	62.2	100.0	78144.0

Note- Share (%) is the share of each of the top ten services in the total exports of services of Benin to India.

Burkina Faso

S.No.	Product Description	Burkina Faso's exports to India (avg. 2000- 2002, \$ millions)	Share (%)	India's imports from ROW(\$ millions)	Burkina Faso's exports to India (avg. 2010- 2012, \$ millions)	Share (%)	India's imports from ROW(\$ millions)
1	Transportation	0.3	4.9	3995.5	1.4	8.2	23449.4
2	Travel	3.5	54.8	4357.3	6.0	35.3	21752.6
3	Communications services	0.0	0.1	198.5	0.0	0.2	1067.8
4	Construction	0.1	2.1	343.9	1.0	5.6	2188.6
5	Insurance services	0.0	0.5	310.0	0.2	1.3	1032.0
6	Financial services	0.0	0.5	602.9	0.3	1.6	3438.4
7	Computer and information services	0.4	6.5	386.1	0.9	5.5	2304.3
8	Royalties and license fees	0.0	0.0	291.1	0.0	0.0	3008.7
9	Other business services	1.4	21.2	2368.2	5.8	33.7	18021.4
10	Personal, cultural and recreational services	0.0	0.3	31.5	0.0	0.2	404.3
11	Government services, n.i.e.	0.6	9.1	393.2	1.4	8.4	1521.6
12	Total services	6.4	100.0	13278.2	17.1	100.0	78189.1

Note- Share (%) is the share of each of the top ten services in the total exports of services of Burkina Faso to India.

<u>Burundi</u>

S.No.	Product Description	Burundi's exports to India (avg. 2000- 2002, \$ millions)	Share (%)	India's imports from ROW(\$ millions)	Burundi's exports to India (avg. 2010-2012, \$ millions)	Share (%)	India's imports from ROW(\$ millions)
1	Transportation	0.0	19.4	3995.8	0.1	11.8	23450.7
2	Travel	0.0	8.5	4360.8	0.0	3.4	21758.6
3	Communications services	0.0	0.1	198.5	0.0	0.1	1067.8
4	Construction	0.0	0.3	344.0	0.0	0.1	2189.6
5	Insurance services	0.0	6.5	310.0	0.0	2.6	1032.2
6	Financial services	0.0	7.3	603.0	0.1	9.1	3438.6
7	Computer and information services	0.0	22.4	386.4	0.1	6.0	2305.2
8	Royalties and license fees	0.0	0.1	291.1	0.0	0.3	3008.7
9	Other business services	0.0	18.0	2369.6	0.5	45.8	18026.7
10	Personal, cultural and recreational services	0.0	0.1	31.5	0.0	1.0	404.3
11	Government services, n.i.e.	0.0	17.4	393.8	0.2	19.7	1522.9
12	Total services	0.1	100.0	13284.5	1.1	100.0	78205.1

Note- Share (%) is the share of each of the top ten services in the total exports of services of Burundi to India.

Central African Republic

S.No.	Product Description	Central Afr. Rep's exports to India (avg. 2000- 2002, \$ millions)	Share (%)	India's imports from ROW(\$ millions)	Central Afr. Rep's exports to India (avg. 2010- 2012, \$ millions)	Share (%)	India's imports from ROW(\$ millions)
1	Transportation	0.0	0.5	3995.8	0.1	1.1	23450.8
2	Travel	1.4	85.2	4359.5	3.8	81.0	21754.8
3	Communications services	0.0	0.0	198.5	0.0	0.0	1067.9
4	Construction	0.0	0.1	344.1	0.0	0.0	2189.6
5	Insurance services	0.0	0.0	310.0	0.0	0.0	1032.3
6	Financial services	0.0	0.3	603.0	0.0	0.7	3438.6
7	Computer and information services	0.0	0.0	386.5	0.0	0.0	2305.2
8	Royalties and license fees	0.0	0.0	291.1	0.0	0.0	3008.7
9	Other business services	0.0	0.3	2369.6	0.0	0.5	18027.2
10	Personal, cultural and recreational services	0.0	0.0	31.5	0.0	0.0	404.3
11	Government services, n.i.e.	0.2	13.6	393.6	0.8	16.8	1522.3
12	Total services	1.6	100.0	13283.0	4.7	100.0	78201.5

Note- Share (%) is the share of each of the top ten services in the total exports of services of Central African Republic to India.

Chad

S.No.	Product Description	Chad's exports to India (avg. 2000-2002, \$ millions)	Share (%)	India's imports from ROW(\$ millions)	Chad's exports to India (avg. 2010-2012, \$ millions)	Share (%)	India's imports from ROW(\$ millions)
1	Transportation	0.8	47.0	3994.9	1.5	45.3	23449.3
2	Travel	0.4	21.8	4360.5	1.0	29.7	21757.6
3	Communications services	0.0	0.1	198.5	0.0	0.1	1067.8
4	Construction	0.0	0.0	344.0	0.0	0.0	2189.6
5	Insurance services	0.0	0.3	310.0	0.0	0.5	1032.2
6	Financial services	0.0	0.0	603.0	0.0	0.0	3438.6
7	Computer and information services	0.0	0.0	386.5	0.0	0.0	2305.2
8	Royalties and license fees	0.0	0.0	291.1	0.0	0.0	3008.7
9	Other business services	0.0	2.0	2369.6	0.1	2.4	18027.1
10	Personal, cultural and recreational services	0.0	0.0	31.5	0.0	0.0	404.3
11	Government services, n.i.e.	0.5	28.9	393.3	0.7	21.9	1522.3
12	Total services	1.8	100.0	13282.8	3.4	100.0	78202.9

Note- Share (%) is the share of each of the top ten services in the total exports of services of Chad to India.

Democratic Republic of Congo

S.No.	Product Description	Angola's exports to India (avg. 2000-2002, \$ millions)	Share (%)	India's imports from ROW(\$ millions)	Angola's exports to India (avg. 2010-2012, \$ millions)	Share (%)	India's imports from ROW(\$ millions)
1	Transportation	0.2	6.6	3995.7	4.4	8.2	23447.9
2	Travel	0.3	9.8	4360.8	6.0	11.2	21758.2
3	Communications services	0.2	7.9	198.5	5.5	10.2	1067.8
4	Construction	0.8	30.8	344.0	8.7	16.1	2189.5
5	Insurance services	0.1	2.7	309.8	5.9	10.9	1029.2
6	Financial services	0.1	3.0	602.9	1.1	1.9	3437.4
7	Computer and information services	0.2	8.3	386.5	6.5	12.0	2305.2
8	Royalties and license fees	0.0	0.0	291.1	0.0	0.0	3008.6
9	Other business services	0.6	22.7	2369.5	11.9	22.0	18025.9
10	Personal, cultural and recreational services	0.1	4.2	31.5	1.5	2.8	404.3
11	Government services, n.i.e.	0.1	3.9	393.7	2.5	4.7	1520.6
12	Total services	2.7	100.0	13284.0	54.2	100.0	78194.6

Note- Share (%) is the share of each of the top ten services in the total exports of services of Dem. Rep. of Congo to India.

Djibouti

S.No.	Product Description	Djibouti's exports to India (avg. 2000- 2002, \$ millions)	Share (%)	India's imports from ROW(\$ millions)	Djibouti's exports to India (avg. 2010-2012, \$ millions)	Share (%)	India's imports from ROW(\$ millions)
1	Transportation	1.3	24.2	3994.5	7.9	32.6	23443.0
2	Travel	2.1	39.2	4358.8	8.4	35.0	21750.1
3	Communications services	0.1	1.5	198.4	0.3	1.3	1067.5
4	Construction	0.0	0.0	344.0	0.0	0.0	2189.6
5	Insurance services	0.0	0.1	310.0	0.0	0.1	1032.2
6	Financial services	0.0	0.0	603.0	0.0	0.0	3438.6
7	Computer and information services	0.0	0.0	386.5	0.0	0.0	2305.2
8	Royalties and license fees	0.0	0.0	291.1	0.0	0.0	3008.7
9	Other business services	0.4	7.3	2369.2	3.1	12.7	18024.1
10	Personal, cultural and recreational services	0.0	0.0	31.5	0.0	0.0	404.3
11	Government services, n.i.e.	1.5	27.8	392.3	4.4	18.3	1518.6
12	Total services	5.2	100.0	13279.3	24.1	100.0	78182.1

Note- Share (%) is the share of each of the top ten services in the total exports of services of Djibouti to India.

Ethiopia

S.No.	Product Description	Ethiopia's exports to India (avg. 2000- 2002, \$ millions)	Share (%)	India's imports from ROW(\$ millions)	Ethiopia's exports to India (avg. 2010-2012, \$ millions)	Share (%)	India's imports from ROW(\$ millions)
1	Transportation	3.3	33.4	3992.5	18.4	45.4	23432.4
2	Travel	1.9	19.2	4358.9	10.7	26.4	21747.9
3	Communications services	0.3	2.5	198.3	0.9	2.2	1067.0
4	Construction	0.5	4.6	343.6	0.4	1.1	2189.1
5	Insurance services	0.0	0.1	310.0	0.0	0.1	1032.2
6	Financial services	0.1	0.7	602.9	0.9	2.3	3437.7
7	Computer and information services	0.0	0.1	386.5	0.0	0.1	2305.2
8	Royalties and license fees	0.0	0.1	291.0	0.2	0.4	3008.5
9	Other business services	0.9	9.4	2368.7	5.2	12.9	18022.0
10	Personal, cultural and recreational services	0.0	0.1	31.5	0.0	0.1	404.3
11	Government services, n.i.e.	3.0	29.9	390.8	3.7	9.2	1519.3
12	Total services	9.9	100.0	13274.7	40.5	100.0	78165.7

Note- Share (%) is the share of each of the top ten services in the total exports of services of Ethiopia to India.

<u>Gambia</u>

S.No.	Product Description	Gambia's exports to India (avg. 2000- 2002, \$ millions)	Share (%)	India's imports from ROW(\$ millions)	Gambia's exports to India (avg. 2010-2012, \$ millions)	Share (%)	India's imports from ROW(\$ millions)
1	Transportation	0.1	2.8	3995.7	1.4	4.8	23449.4
2	Travel	2.7	86.0	4358.2	24.5	81.0	21734.1
3	Communications services	0.2	6.9	198.3	3.2	10.6	1064.7
4	Construction	0.0	0.0	344.0	0.0	0.0	2189.6
5	Insurance services	0.0	0.0	310.0	0.1	0.4	1032.1
6	Financial services	0.0	0.3	603.0	0.1	0.3	3438.6
7	Computer and information services	0.1	1.9	386.4	0.2	0.6	2305.1
8	Royalties and license fees	0.0	0.0	291.1	0.0	0.0	3008.7
9	Other business services	0.0	1.0	2369.6	0.5	1.7	18026.7
10	Personal, cultural and recreational services	0.0	0.0	31.5	0.0	0.0	404.3
11	Government services, n.i.e.	0.0	1.0	393.8	0.2	0.7	1522.9
12	Total services	3.1	100.0	13281.5	30.2	100.0	78176.0

Note- Share (%) is the share of each of the top ten services in the total exports of services of Gambia to India.

Ghana

S.No.	Product Description	Ghana's exports to India (avg. 2000- 2002, \$ millions)	Share (%)	India's imports from ROW(\$ millions)	Ghana's exports to India (avg. 2010-2012, \$ millions)	Share (%)	India's imports from ROW(\$ millions)
1	Transportation	1.6	19.4	3994.2	21.2	27.9	23429.7
2	Travel	6.0	71.8	4354.9	40.1	53.0	21718.5
3	Communications services	0.0	0.2	198.5	0.3	0.4	1067.6
4	Construction	0.0	0.1	344.0	0.0	0.0	2189.5
5	Insurance services	0.0	0.0	310.0	0.0	0.0	1032.3
6	Financial services	0.0	0.5	602.9	0.4	0.5	3438.3
7	Computer and information services	0.0	0.5	386.4	0.6	0.7	2304.7
8	Royalties and license fees	0.0	0.0	291.1	0.0	0.0	3008.7
9	Other business services	0.4	4.8	2369.2	8.0	10.5	18019.2
10	Personal, cultural and recreational services	0.0	0.3	31.5	1.0	1.3	403.3
11	Government services, n.i.e.	0.2	2.4	393.6	4.3	5.6	1518.8
12	Total services	8.3	100.0	13276.3	75.8	100.0	78130.4

Note- Share (%) is the share of each of the top ten services in the total exports of services of Ghana to India.

Guinea

S.No.	Product Description	Guinea's exports to India (avg. 2000- 2002, \$ millions)	Share (%)	India's imports from ROW(\$ millions)	Guinea's exports to India (avg. 2010-2012, \$ millions)	Share (%)	India's imports from ROW(\$ millions)
1	Transportation	0.4	29.4	3995.4	6.0	28.0	23444.8
2	Travel	0.2	15.8	4360.6	2.8	12.8	21755.8
3	Communications services	0.2	17.0	198.3	4.3	19.8	1063.6
4	Construction	0.0	2.5	344.0	2.7	12.5	2186.9
5	Insurance services	0.0	0.0	310.0	0.4	1.7	1031.9
6	Financial services	0.1	7.8	602.9	1.7	7.8	3437.0
7	Computer and information services	0.0	0.0	386.5	0.0	0.0	2305.2
8	Royalties and license fees	0.0	0.0	291.1	0.0	0.0	3008.7
9	Other business services	0.1	4.6	2369.5	1.6	7.4	18025.6
10	Personal, cultural and recreational services	0.0	0.4	31.5	0.6	2.7	403.7
11	Government services, n.i.e.	0.3	22.4	393.5	1.6	7.4	1521.5
12	Total services	1.4	100.0	13283.2	21.5	100.0	78184.7

Note- Share (%) is the share of each of the top ten services in the total exports of services of Guinea to India.

Guinea-Bissau

S.No.	Product Description	Guinea- Bissau's exports to India (avg. 2000- 2002, \$ millions)	Share (%)	India's imports from ROW(\$ millions)	Guinea- Bissau's exports to India (avg. 2010- 2012, \$ millions)	Share (%)	India's imports from ROW(\$ millions)
1	Transportation	0.9	21.7	3994.8	7.6	29.9	23443.3
2	Travel	2.1	48.0	4358.8	11.0	43.6	21747.6
3	Communications services	0.0	0.1	198.5	0.1	0.3	1067.8
4	Construction	0.0	0.0	344.0	0.0	0.0	2189.6
5	Insurance services	0.0	0.0	310.0	0.0	0.0	1032.3
6	Financial services	0.0	0.0	603.0	0.0	0.0	3438.6
7	Computer and information services	0.1	2.4	386.4	0.5	1.9	2304.7
8	Royalties and license fees	0.0	0.5	291.0	0.4	1.4	3008.3
9	Other business services	0.3	7.4	2369.3	3.9	15.5	18023.3
10	Personal, cultural and recreational services	0.0	0.1	31.5	0.1	0.6	404.2
11	Government services, n.i.e.	0.8	19.8	393.0	1.7	6.9	1521.3
12	Total services	4.3	100.0	13280.3	25.3	100.0	78180.9

Note- Share (%) is the share of each of the top ten services in the total exports of services of Guinea-Bissau to India.

Kenya

S.No.	Product Description	Kenya's exports to India (avg. 2000- 2002, \$ millions)	Share (%)	India's imports from ROW(\$ millions)	Kenya's exports to India (avg. 2010- 2012, \$ millions)	Share (%)	India's imports from ROW(\$ millions)
1	Transportation	12.2	40.6	3983.6	60.8	46.9	23390.0
2	Travel	13.4	44.7	4347.5	47.5	36.6	21711.1
3	Communications services	0.1	0.3	198.4	0.3	0.2	1067.5
4	Construction	0.0	0.1	344.0	0.1	0.1	2189.4
5	Insurance services	0.1	0.3	309.9	0.5	0.4	1031.8
6	Financial services	0.2	0.5	602.8	0.9	0.7	3437.8
7	Computer and information services	0.0	0.1	386.4	0.2	0.1	2305.1
8	Royalties and license fees	0.0	0.0	291.1	0.0	0.0	3008.7
9	Other business services	3.4	11.3	2366.2	17.6	13.5	18009.6
10	Personal, cultural and recreational services	0.0	0.0	31.5	0.2	0.2	404.1
11	Government services, n.i.e.	0.6	2.0	393.2	1.7	1.3	1521.3
12	Total services	29.9	100.0	13254.7	129.8	100.0	78076.4

Note- Share (%) is the share of each of the top ten services in the total exports of services of Kenya to India.

Lesotho

S.No.	Product Description	Lesotho's exports to India (avg. 2000- 2002, \$ millions)	Share (%)	India's imports from ROW(\$ millions)	Lesotho's exports to India (avg. 2010-2012, \$ millions)	Share (%)	India's imports from ROW(\$ millions)
1	Transportation	0.0	9.0	3995.8	0.2	12.4	23450.6
2	Travel	0.1	86.1	4360.8	1.4	83.9	21757.2
3	Communications services	0.0	0.0	198.5	0.0	0.3	1067.8
4	Construction	0.0	0.0	344.0	0.0	0.0	2189.6
5	Insurance services	0.0	0.0	310.0	0.0	0.2	1032.3
6	Financial services	0.0	0.0	603.0	0.0	0.0	3438.6
7	Computer and information services	0.0	2.6	386.5	0.0	0.0	2305.2
8	Royalties and license fees	0.0	0.0	291.1	0.0	0.0	3008.7
9	Other business services	0.0	2.2	2369.6	0.1	3.0	18027.1
10	Personal, cultural and recreational services	0.0	0.0	31.5	0.0	0.0	404.3
11	Government services, n.i.e.	0.0	0.1	393.8	0.0	0.1	1523.1
12	Total services	0.1	100.0	13284.5	1.6	100.0	78204.6

Note- Share (%) is the share of each of the top ten services in the total exports of services of Lesotho to India.

Liberia

S.No.	Product Description	Liberia's exports to India (avg. 2000- 2002, \$ millions)	Share (%)	India's imports from ROW(\$ millions)	Liberia's exports to India (avg. 2010-2012, \$ millions)	Share (%)	India's imports from ROW(\$ millions)
1	Transportation	32.6	85.1	3963.2	75.0	82.6	23375.9
2	Travel	2.9	7.5	4358.0	4.3	4.7	21754.3
3	Communications services	0.2	0.6	198.3	0.3	0.3	1067.5
4	Construction	0.0	0.1	344.0	0.0	0.0	2189.6
5	Insurance services	0.0	0.0	310.0	0.0	0.0	1032.2
6	Financial services	0.1	0.3	602.8	0.4	0.5	3438.2
7	Computer and information services	0.1	0.1	386.4	0.1	0.1	2305.1
8	Royalties and license fees	0.0	0.0	291.1	0.0	0.0	3008.7
9	Other business services	0.5	1.4	2369.1	2.4	2.7	18024.8
10	Personal, cultural and recreational services	0.0	0.0	31.5	0.0	0.0	404.3
11	Government services, n.i.e.	1.9	4.9	391.9	8.3	9.1	1514.8
12	Total services	38.3	100.0	13246.3	90.8	100.0	78115.4

Note- Share (%) is the share of each of the top ten services in the total exports of services of Liberia to India.

Madagascar

S.No.	Product Description	Madagascar's exports to India (avg. 2000-2002, \$ millions)	Share (%)	India's imports from ROW(\$ millions)	Madagascar's exports to India (avg. 2010-2012, \$ millions)	Share (%)	India's imports from ROW(\$ millions)
1	Transportation	0.6	24.4	3995.1	11.3	31.9	23439.5
2	Travel	1.5	57.7	4359.3	18.3	51.4	21740.3
3	Communications services	0.0	0.1	198.5	0.1	0.2	1067.8
4	Construction	0.1	3.2	344.0	0.6	1.6	2189.0
5	Insurance services	0.0	0.8	310.0	0.1	0.4	1032.1
6	Financial services	0.0	0.8	602.9	0.3	0.8	3438.4
7	Computer and information services	0.0	0.2	386.5	0.1	0.1	2305.2
8	Royalties and license fees	0.0	0.0	291.1	0.0	0.0	3008.7
9	Other business services	0.3	12.5	2369.3	4.3	12.1	18022.9
10	Personal, cultural and recreational services	0.0	0.3	31.5	0.5	1.5	403.8
11	Government services, n.i.e.	0.0	0.0	393.8	0.0	0.0	1523.1
12	Total services	2.6	100.0	13282.0	35.5	100.0	78170.7

Note- Share (%) is the share of each of the top ten services in the total exports of services of Madagascar to India.

<u>Malawi</u>

S.No.	Product Description	Malawi's exports to India (avg. 2000- 2002, \$ millions)	Share (%)	India's imports from ROW(\$ millions)	Malawi's exports to India (avg. 2010-2012, \$ millions)	Share (%)	India's imports from ROW(\$ millions)
1	Transportation	0.2	28.5	3995.6	1.7	31.2	23449.1
2	Travel	0.4	58.8	4360.5	3.0	53.7	21755.6
3	Communications services	0.0	0.0	198.5	0.0	0.5	1067.8
4	Construction	0.0	0.0	344.0	0.0	0.0	2189.6
5	Insurance services	0.0	0.0	310.0	0.0	0.3	1032.2
6	Financial services	0.0	0.0	603.0	0.0	0.2	3438.6
7	Computer and information services	0.0	0.0	386.5	0.0	0.0	2305.2
8	Royalties and license fees	0.0	0.0	291.1	0.0	0.0	3008.7
9	Other business services	0.1	8.7	2369.5	0.6	10.7	18026.6
10	Personal, cultural and recreational services	0.0	0.0	31.5	0.0	0.0	404.3
11	Government services, n.i.e.	0.0	4.0	393.8	0.2	3.5	1522.9
12	Total services	0.6	100.0	13283.9	5.5	100.0	78200.7

Note- Share (%) is the share of each of the top ten services in the total exports of services of Malawi to India.

<u>Mali</u>

S.No.	Product Description	Mali's exports to India (avg. 2000- 2002, \$ millions)	Share (%)	India's imports from ROW(\$ millions)	Mali's exports to India (avg. 2010-2012, \$ millions)	Share (%)	India's imports from ROW(\$ millions)
1	Transportation	1.1	14.4	3994.7	1.5	9.5	23449.3
2	Travel	1.6	21.6	4359.3	4.4	27.6	21754.2
3	Communications services	2.0	27.5	196.5	2.9	18.1	1065.0
4	Construction	0.0	0.0	344.0	0.0	0.0	2189.6
5	Insurance services	0.0	0.0	310.0	0.0	0.0	1032.3
6	Financial services	0.3	4.3	602.7	1.2	7.5	3437.5
7	Computer and information services	0.1	1.1	386.4	0.1	0.8	2305.1
8	Royalties and license fees	0.0	0.2	291.0	0.1	0.6	3008.6
9	Other business services	1.6	22.5	2368.0	4.9	31.0	18022.3
10	Personal, cultural and recreational services	0.0	0.0	31.5	0.0	0.1	404.3
11	Government services, n.i.e.	0.6	8.4	393.2	0.8	4.7	1522.3
12	Total services	7.3	100.0	13277.3	15.8	100.0	78190.4

Note- Share (%) is the share of each of the top ten services in the total exports of services of Mali to India.

Mauritania

S.No.	Product Description	Mauritania's exports to India (avg. 2000-2002, \$ millions)	Share (%)	India's imports from ROW(\$ millions)	Angola's exports to India (avg. 2010-2012, \$ millions)	Share (%)	India's imports from ROW(\$ millions)
1	Transportation	0.0	12.0	3995.8	0.6	13.8	23450.3
2	Travel	0.1	43.6	4360.8	1.4	34.9	21757.2
3	Communications services	0.0	0.7	198.5	0.0	0.7	1067.8
4	Construction	0.0	0.1	344.0	0.0	0.0	2189.6
5	Insurance services	0.0	1.8	310.0	0.1	1.7	1032.2
6	Financial services	0.0	8.0	603.0	0.3	6.8	3438.4
7	Computer and information services	0.0	0.0	386.5	0.0	0.0	2305.2
8	Royalties and license fees	0.0	2.2	291.0	0.2	5.2	3008.5
9	Other business services	0.0	7.1	2369.6	0.4	10.5	18026.8
10	Personal, cultural and recreational services	0.0	0.0	31.5	0.0	0.0	404.3
11	Government services, n.i.e.	0.0	24.5	393.8	1.1	26.3	1522.0
12	Total services	0.2	100.0	13284.4	4.0	100.0	78202.2

Note- Share (%) is the share of each of the top ten services in the total exports of services of Mauritania to India.

Mozambique

S.No.	Product Description	Mozambique's exports to India (avg. 2000-2002, \$ millions)	Share (%)	India's imports from ROW(\$ millions)	Mozambique's exports to India (avg. 2010-2012, \$ millions)	Share (%)	India's imports from ROW(\$ millions)
1	Transportation	4.6	69.1	3991.1	22.7	68.7	23428.2
2	Travel	1.4	20.6	4359.5	6.1	18.5	21752.5
3	Communications services	0.0	0.5	198.5	0.1	0.3	1067.8
4	Construction	0.0	0.6	344.0	0.1	0.3	2189.5
5	Insurance services	0.0	0.1	310.0	0.0	0.1	1032.2
6	Financial services	0.0	0.2	603.0	0.0	0.0	3438.6
7	Computer and information services	0.0	0.0	386.5	0.0	0.0	2305.2
8	Royalties and license fees	0.0	0.0	291.1	0.0	0.0	3008.7
9	Other business services	0.6	8.7	2369.0	3.4	10.2	18023.8
10	Personal, cultural and recreational services	0.0	0.0	31.5	0.0	0.1	404.3
11	Government services, n.i.e.	0.0	0.3	393.8	0.6	1.8	1522.5
12	Total services	6.7	100.0	13277.9	33.0	100.0	78173.2

Note- Share (%) is the share of each of the top ten services in the total exports of services of Mozambique to India.

<u>Niger</u>

S.No.	Product Description	Niger's exports to India (avg. 2000- 2002, \$ millions)	Share (%)	India's imports from ROW(\$ millions)	Niger's exports to India (avg. 2010-2012, \$ millions)	Share (%)	India's imports from ROW(\$ millions)
1	Transportation	0.4	83.4	3995.4	6.0	90.1	23444.9
2	Travel	0.0	7.3	4360.8	0.2	3.4	21758.4
3	Communications services	0.0	0.0	198.5	0.0	0.0	1067.9
4	Construction	0.0	0.0	344.0	0.0	0.0	2189.6
5	Insurance services	0.0	0.1	310.0	0.0	0.1	1032.3
6	Financial services	0.0	3.4	603.0	0.2	3.4	3438.4
7	Computer and information services	0.0	0.2	386.5	0.0	0.2	2305.2
8	Royalties and license fees	0.0	0.0	291.1	0.0	0.0	3008.7
9	Other business services	0.0	2.0	2369.6	0.1	1.7	18027.1
10	Personal, cultural and recreational services	0.0	0.0	31.5	0.0	0.0	404.3
11	Government services, n.i.e.	0.0	3.7	393.8	0.1	1.1	1523.0
12	Total services	0.5	100.0	13284.1	6.6	100.0	78199.6

Note- Share (%) is the share of each of the top ten services in the total exports of services of Niger to India.

<u>Nigeria</u>

S.No.	Product Description	Nigeria's exports to India (avg. 2000- 2002, \$ millions)	Share (%)	India's imports from ROW(\$ millions)	Nigeria's exports to India (avg. 2010-2012, \$ millions)	Share (%)	India's imports from ROW(\$ millions)
1	Transportation	35.5	43.0	3960.3	261.6	48.2	23189.3
2	Travel	5.9	7.1	4355.0	98.4	18.2	21660.2
3	Communications services	0.6	0.8	197.9	3.6	0.7	1064.3
4	Construction	28.3	34.2	315.8	104.6	19.3	2085.0
5	Insurance services	0.2	0.3	309.8	5.0	0.9	1027.2
6	Financial services	0.6	0.7	602.4	1.9	0.4	3436.8
7	Computer and information services	0.1	0.1	386.4	0.8	0.1	2304.5
8	Royalties and license fees	0.1	0.1	291.0	0.6	0.1	3008.1
9	Other business services	9.8	11.8	2359.8	61.5	11.3	17965.7
10	Personal, cultural and recreational services	0.0	0.0	31.5	0.2	0.0	404.1
11	Government services, n.i.e.	1.7	2.0	392.1	4.3	0.8	1518.8
12	Total services	82.6	100.0	13202.0	542.3	100.0	77663.8

Note- Share (%) is the share of each of the top ten services in the total exports of services of Nigeria to India.

Rwanda

S.No.	Product Description	Rwanda's exports to India (avg. 2000- 2002, \$ millions)	Share (%)	India's imports from ROW(\$ millions)	Rwanda's exports to India (avg. 2010-2012, \$ millions)	Share (%)	India's imports from ROW(\$ millions)
1	Transportation	0.0	5.4	3995.8	0.1	6.2	23450.7
2	Travel	0.1	28.8	4360.8	0.4	22.4	21758.2
3	Communications services	0.0	0.0	198.5	0.0	0.3	1067.8
4	Construction	0.0	0.1	344.0	0.0	0.1	2189.6
5	Insurance services	0.0	14.2	310.0	0.3	12.9	1032.0
6	Financial services	0.1	28.8	602.9	0.7	36.2	3437.9
7	Computer and information services	0.0	0.3	386.5	0.0	0.2	2305.2
8	Royalties and license fees	0.0	0.3	291.1	0.0	0.8	3008.7
9	Other business services	0.0	13.0	2369.6	0.3	17.0	18026.9
10	Personal, cultural and recreational services	0.0	0.1	31.5	0.0	0.5	404.3
11	Government services, n.i.e.	0.0	9.1	393.8	0.1	3.5	1523.0
12	Total services	0.2	100.0	13284.4	2.0	100.0	78204.3

Note- Share (%) is the share of each of the top ten services in the total exports of services of Rwanda to India.

Senegal

S.No.	Product Description	Senegal's exports to India (avg. 2000- 2002, \$ millions)	Share (%)	India's imports from ROW(\$ millions)	Senegal's exports to India (avg. 2010-2012, \$ millions)	Share (%)	India's imports from ROW(\$ millions)
1	Transportation	2.5	6.8	3993.3	11.5	8.1	23439.4
2	Travel	15.8	43.1	4345.1	58.0	41.0	21700.6
3	Communications services	3.8	10.5	194.7	13.3	9.4	1054.6
4	Construction	0.7	1.8	343.4	1.9	1.3	2187.7
5	Insurance services	0.0	0.1	310.0	0.3	0.2	1032.0
6	Financial services	0.2	0.4	602.8	1.1	0.8	3437.6
7	Computer and information services	0.1	0.4	386.3	0.5	0.3	2304.7
8	Royalties and license fees	0.0	0.0	291.0	0.1	0.1	3008.6
9	Other business services	5.7	15.5	2363.9	33.9	24.0	17993.3
10	Personal, cultural and recreational services	0.0	0.1	31.5	0.8	0.6	403.5
11	Government services, n.i.e.	7.8	21.3	386.0	20.4	14.4	1502.7
12	Total services	36.6	100.0	13247.9	141.6	100.0	78064.6

Note- Share (%) is the share of each of the top ten services in the total exports of services of Senegal to India.

Sierra Leone

S.No.	Product Description	Sierra Leone's exports to India (avg. 2000- 2002, \$ millions)	Share (%)	India's imports from ROW(\$ millions)	Sierra Leone's exports to India (avg. 2010- 2012, \$ millions)	Share (%)	India's imports from ROW(\$ millions)
1	Transportation	0.4	39.7	3995.4	2.4	42.8	23448.5
2	Travel	0.4	42.4	4360.4	2.5	44.5	21756.1
3	Communications services	0.0	0.0	198.5	0.0	0.0	1067.8
4	Construction	0.0	0.3	344.0	0.0	0.2	2189.6
5	Insurance services	0.0	0.4	310.0	0.0	0.3	1032.2
6	Financial services	0.0	0.0	603.0	0.0	0.0	3438.6
7	Computer and information services	0.0	0.0	386.5	0.0	0.0	2305.2
8	Royalties and license fees	0.0	0.1	291.1	0.0	0.2	3008.7
9	Other business services	0.0	4.2	2369.6	0.3	5.2	18026.9
10	Personal, cultural and recreational services	0.0	0.7	31.5	0.0	0.3	404.3
11	Government services, n.i.e.	0.1	12.4	393.7	0.4	6.6	1522.7
12	Total services	1.0	100.0	13283.6	5.5	100.0	78200.7

Note- Share (%) is the share of each of the top ten services in the total exports of services of Sierra Leone to India.

Sudan

S.No.	Product Description	Sudan's exports to India (avg. 2000- 2002, \$ millions)	Share (%)	India's imports from ROW(\$ millions)	Sudan's exports to India (avg. 2010-2012, \$ millions)	Share (%)	India's imports from ROW(\$ millions)
1	Transportation	1.9	33.7	3993.9	9.2	39.6	23441.7
2	Travel	1.7	29.5	4359.2	5.2	22.3	21753.4
3	Communications services	0.0	0.4	198.5	0.1	0.3	1067.8
4	Construction	0.0	0.4	344.0	0.4	1.6	2189.2
5	Insurance services	0.0	0.2	310.0	0.0	0.0	1032.3
6	Financial services	0.0	0.2	603.0	0.1	0.4	3438.6
7	Computer and information services	0.0	0.1	386.5	0.0	0.0	2305.2
8	Royalties and license fees	0.0	0.0	291.1	0.0	0.0	3008.7
9	Other business services	0.6	10.2	2369.0	3.2	13.7	18024.0
10	Personal, cultural and recreational services	0.0	0.0	31.5	0.0	0.0	404.3
11	Government services, n.i.e.	1.4	25.3	392.4	5.2	22.2	1517.9
12	Total services	5.7	100.0	13278.9	23.2	100.0	78183.0

Note- Share (%) is the share of each of the top ten services in the total exports of services of Sudan to India.

Togo

S.No.	Product Description	Togo's exports to India (avg. 2000-2002, \$ millions)	Share (%)	India's imports from ROW(\$ millions)	Togo's exports to India (avg. 2010-2012, \$ millions)	Share (%)	India's imports from ROW(\$ millions)
1	Transportation	0.4	10.4	3995.3	4.8	18.4	23446.0
2	Travel	2.2	53.4	4358.6	14.6	55.9	21743.9
3	Communications services	0.3	7.9	198.2	2.7	10.1	1065.2
4	Construction	0.7	16.6	343.4	0.7	2.6	2188.9
5	Insurance services	0.0	0.0	310.0	0.0	0.0	1032.3
6	Financial services	0.0	0.5	602.9	0.2	0.7	3438.5
7	Computer and information services	0.0	0.0	386.5	0.0	0.0	2305.2
8	Royalties and license fees	0.0	0.0	291.1	0.0	0.0	3008.7
9	Other business services	0.2	4.7	2369.4	2.0	7.4	18025.2
10	Personal, cultural and recreational services	0.0	0.3	31.5	0.7	2.6	403.6
11	Government services, n.i.e.	0.3	6.2	393.5	0.6	2.3	1522.5
12	Total services	4.1	100.0	13280.5	26.2	100.0	78180.0

Note- Share (%) is the share of each of the top ten services in the total exports of services of Togo to India.

Tanzania

S.No.	Product Description	Tanzania's exports to India (avg. 2000-2002, \$ millions)	Share (%)	India's imports from ROW(\$ millions)	Tanzania's exports to India (avg. 2010-2012, \$ millions)	Share (%)	India's imports from ROW(\$ millions)
1	Transportation	11.8	34.4	3983.9	39.5	30.9	23411.3
2	Travel	19.1	55.6	4341.7	78.3	61.2	21680.3
3	Communications services	0.2	0.6	198.3	0.6	0.5	1067.3
4	Construction	0.1	0.3	344.0	1.8	1.4	2187.7
5	Insurance services	0.0	0.1	310.0	0.2	0.2	1032.0
6	Financial services	0.0	0.0	603.0	0.0	0.0	3438.6
7	Computer and information services	0.7	2.0	385.8	1.7	1.4	2303.5
8	Royalties and license fees	0.0	0.0	291.1	0.0	0.0	3008.7
9	Other business services	0.7	2.0	2368.9	3.9	3.0	18023.3
10	Personal, cultural and recreational services	0.0	0.0	31.5	0.1	0.1	404.2
11	Government services, n.i.e.	1.8	5.1	392.0	1.8	1.4	1521.3
12	Total services	34.4	100.0	13250.2	128.1	100.0	78078.2

Note- Share (%) is the share of each of the top ten services in the total exports of services of Tanzania to India.

<u>Uganda</u>

S.No.	Product Description	Uganda's exports to India (avg. 2000- 2002, \$ millions)	Share (%)	India's imports from ROW(\$ millions)	Uganda's exports to India (avg. 2010-2012, \$ millions)	Share (%)	India's imports from ROW(\$ millions)
1	Transportation	0.3	9.4	3995.5	2.5	8.1	23448.4
2	Travel	1.7	57.3	4359.1	15.1	49.7	21743.5
3	Communications services	0.0	1.4	198.5	0.3	1.0	1067.5
4	Construction	0.0	0.4	344.0	0.9	2.8	2188.7
5	Insurance services	0.0	0.3	310.0	0.1	0.3	1032.2
6	Financial services	0.3	8.3	602.7	2.9	9.4	3435.8
7	Computer and information services	0.1	3.2	386.4	0.8	2.6	2304.4
8	Royalties and license fees	0.0	0.4	291.0	0.4	1.2	3008.3
9	Other business services	0.4	13.7	2369.2	4.8	15.7	18022.4
10	Personal, cultural and recreational services	0.0	0.5	31.5	1.0	3.2	403.3
11	Government services, n.i.e.	0.2	5.1	393.6	1.8	6.0	1521.2
12	Total services	3.0	100.0	13281.6	30.5	100.0	78175.8

Note- Share (%) is the share of each of the top ten services in the total exports of services of Uganda to India.

Zambia

S.No.	Product Description	Zambia's exports to India (avg. 2000- 2002, \$ millions)	Share (%)	India's imports from ROW(\$ millions)	Zambia's exports to India (avg. 2010- 2012, \$ millions)	Share (%)	India's imports from ROW(\$ millions)
1	Transportation	1.7	30.7	3994.1	22.2	45.5	23428.7
2	Travel	3.3	58.5	4357.6	21.7	44.7	21736.8
3	Communications services	0.5	9.0	198.0	3.0	6.1	1064.9
4	Construction	0.0	0.1	344.0	0.0	0.1	2189.5
5	Insurance services	0.0	0.0	310.0	0.0	0.1	1032.2
6	Financial services	0.0	0.0	603.0	0.0	0.0	3438.6
7	Computer and information services	0.0	0.1	386.5	0.1	0.1	2305.2
8	Royalties and license fees	0.0	0.0	291.1	0.0	0.0	3008.7
9	Other business services	0.1	1.1	2369.5	0.9	1.9	18026.3
10	Personal, cultural and recreational services	0.0	0.0	31.5	0.7	1.4	403.7
11	Government services, n.i.e.	0.0	0.4	393.8	0.1	0.2	1523.0
12	Total services	5.6	100.0	13279.0	48.7	100.0	78157.5

Note- Share (%) is the share of each of the top ten services in the total exports of services of Zambia to India.

Bangladesh

S.No.	Product Description	Bangladesh's exports to India (avg. 2000-2002, \$ millions)	Share (%)	India's imports from ROW(\$ millions)	Bangladesh's exports to India (avg. 2010-2012, \$ millions)	Share (%)	India's imports from ROW(\$ millions)
1	Transportation	3.7	27.4	3992.1	26.4	30.8	23424.4
2	Travel	2.3	16.9	4358.6	9.4	10.9	21749.2
3	Communications services	0.2	1.3	198.3	0.4	0.5	1067.4
4	Construction	2.6	19.2	341.5	3.0	3.5	2186.5
5	Insurance services	0.0	0.2	310.0	0.2	0.2	1032.1
6	Financial services	0.1	0.9	602.9	1.0	1.1	3437.7
7	Computer and information services	0.1	0.5	386.4	0.3	0.4	2304.9
8	Royalties and license fees	0.0	0.0	291.1	0.0	0.0	3008.7
9	Other business services	3.5	26.3	2366.1	35.1	40.9	17992.1
10	Personal, cultural and recreational services	0.0	0.0	31.5	0.0	0.0	404.3
11	Government services, n.i.e.	1.0	7.5	392.8	10.1	11.7	1513.0
12	Total services	13.4	100.0	13271.2	85.9	100.0	78120.3

Note- Share (%) is the share of each of the top ten services in the total exports of services of Bangladesh to India.

Bhutan

S.No.	Product Description	Bhutan's exports to India (avg. 2000- 2002, \$ millions)	Share (%)	India's imports from ROW(\$ millions)	Bhutan's exports to India (avg. 2010-2012, \$ millions)	Share (%)	India's imports from ROW(\$ millions)
1	Transportation	0.1	0.8	3995.7	0.9	1.6	23449.9
2	Travel	8.6	96.9	4352.3	56.9	96.7	21701.7
3	Communications services	0.0	0.3	198.5	0.2	0.3	1067.7
4	Construction	0.0	0.0	344.0	0.0	0.0	2189.6
5	Insurance services	0.0	0.1	310.0	0.1	0.1	1032.2
6	Financial services	0.0	0.0	603.0	0.0	0.0	3438.6
7	Computer and information services	0.0	0.3	386.4	0.2	0.3	2305.0
8	Royalties and license fees	0.0	0.0	291.1	0.0	0.0	3008.7
9	Other business services	0.0	0.3	2369.6	0.3	0.4	18027.0
10	Personal, cultural and recreational services	0.0	0.0	31.5	0.0	0.0	404.3
11	Government services, n.i.e.	0.1	1.3	393.7	0.3	0.5	1522.8
12	Total services	8.8	100.0	13275.7	58.8	100.0	78147.4

Note- Share (%) is the share of each of the top ten services in the total exports of services of Bhutan to India.

Cambodia

S.No.	Product Description	Cambodia's exports to India (avg. 2000-2002, \$ millions)	Share (%)	India's imports from ROW(\$ millions)	Cambodia's exports to India (avg. 2010-2012, \$ millions)	Share (%)	India's imports from ROW(\$ millions)
1	Transportation	0.7	48.6	3995.1	7.0	50.3	23443.8
2	Travel	0.7	49.0	4360.2	6.4	45.8	21752.2
3	Communications services	0.0	0.3	198.5	0.0	0.1	1067.8
4	Construction	0.0	0.0	344.1	0.0	0.0	2189.6
5	Insurance services	0.0	0.0	310.0	0.0	0.0	1032.3
6	Financial services	0.0	0.0	603.0	0.0	0.1	3438.6
7	Computer and information services	0.0	0.6	386.5	0.1	0.5	2305.2
8	Royalties and license fees	0.0	0.0	291.1	0.0	0.0	3008.7
9	Other business services	0.0	1.5	2369.6	0.4	3.0	18026.8
10	Personal, cultural and recreational services	0.0	0.1	31.5	0.0	0.1	404.3
11	Government services, n.i.e.	0.0	0.2	393.8	0.0	0.3	1523.0
12	Total services	1.3	100.3	13283.2	13.9	100.0	78192.3

Note- Share (%) is the share of each of the top ten services in the total exports of services of Cambodia to India.

Laos

S.No.	Product Description	Laos's exports to India (avg. 2000- 2002, \$ millions)	Share (%)	India's imports from ROW(\$ millions)	Laos's exports to India (avg. 2010- 2012, \$ millions)	Share (%)	India's imports from ROW(\$ millions)
1	Transportation	0.1	13.5	3995.7	3.2	9.2	23447.7
2	Travel	0.5	78.1	4360.4	27.8	81.2	21730.8
3	Communications services	0.0	1.5	198.5	0.4	1.2	1067.4
4	Construction	0.0	0.3	344.0	0.1	0.3	2189.5
5	Insurance services	0.0	0.0	310.0	0.1	0.2	1032.2
6	Financial services	0.0	0.0	603.0	0.0	0.0	3438.6
7	Computer and information services	0.0	0.0	386.5	0.0	0.0	2305.2
8	Royalties and license fees	0.0	0.0	291.1	0.0	0.0	3008.7
9	Other business services	0.0	5.4	2369.6	2.2	6.6	18025.0
10	Personal, cultural and recreational services	0.0	0.0	31.5	0.0	0.0	404.3
11	Government services, n.i.e.	0.0	1.1	393.8	0.5	1.3	1522.6
12	Total services	0.6	100.0	13284.0	34.2	100.0	78172.0

Note- Share (%) is the share of each of the top ten services in the total exports of services of Laos to India.

Myanmar

S.No.	Product Description	Myanmar's exports to India (avg. 2000-2002, \$ millions)	Share (%)	India's imports from ROW(\$ millions)	Myanmar's exports to India (avg. 2010-2012, \$ millions)	Share (%)	India's imports from ROW(\$ millions)
1	Transportation	11.3	49.4	3984.5	47.4	45.7	23403.5
2	Travel	10.1	44.4	4350.7	46.4	44.7	21712.2
3	Communications services	0.0	0.0	198.5	0.1	0.1	1067.8
4	Construction	0.2	1.0	343.8	0.8	0.7	2188.8
5	Insurance services	0.0	0.0	310.0	0.0	0.0	1032.3
6	Financial services	0.0	0.0	603.0	0.0	0.0	3438.6
7	Computer and information services	0.0	0.0	386.5	0.0	0.0	2305.2
8	Royalties and license fees	0.0	0.0	291.1	0.0	0.0	3008.7
9	Other business services	0.8	3.6	2368.8	7.5	7.3	18019.7
10	Personal, cultural and recreational services	0.0	0.0	31.5	0.0	0.0	404.3
11	Government services, n.i.e.	0.4	1.7	393.4	1.7	1.6	1521.4
12	Total services	22.8	100.0	13261.7	103.8	100.0	78102.4

Note- Share (%) is the share of each of the top ten services in the total exports of services of Myanmar to India.

Nepal

S.No.	Product Description	Nepal's exports to India (avg. 2000- 2002, \$ millions)	Share (%)	India's imports from ROW(\$ millions)	Nepal's exports to India (avg. 2010-2012, \$ millions)	Share (%)	India's imports from ROW(\$ millions)
1	Transportation	2.5	4.1	3993.3	7.3	4.0	23443.5
2	Travel	44.0	71.9	4316.8	134.0	73.0	21624.6
3	Communications services	10.4	17.0	188.1	25.4	13.9	1042.4
4	Construction	0.0	0.0	344.0	0.0	0.0	2189.6
5	Insurance services	0.0	0.0	310.0	0.1	0.0	1032.2
6	Financial services	0.1	0.2	602.9	0.9	0.5	3437.8
7	Computer and information services	0.3	0.4	386.2	0.8	0.5	2304.4
8	Royalties and license fees	0.0	0.0	291.1	0.0	0.0	3008.7
9	Other business services	0.9	1.4	2368.7	5.3	2.9	18021.9
10	Personal, cultural and recreational services	0.3	0.5	31.2	1.0	0.5	403.3
11	Government services, n.i.e.	2.7	4.4	391.1	8.8	4.8	1514.3
12	Total services	61.2	100.0	13223.4	183.5	100.0	78022.7

Note- Share (%) is the share of each of the top ten services in the total exports of services of Nepal to India.

Yemen

S.No.	Product Description	Yemen's exports to India (avg. 2000- 2002, \$ millions)	Share (%)	India's imports from ROW(\$ millions)	Yemen's exports to India (avg. 2010-2012, \$ millions)	Share (%)	India's imports from ROW(\$ millions)
1	Transportation	2.0	12.6	3993.8	14.2	10.8	23436.7
2	Travel	8.1	52.7	4352.7	76.8	58.2	21681.8
3	Communications services	0.1	0.5	198.4	0.5	0.4	1067.3
4	Construction	0.2	1.1	343.9	0.5	0.4	2189.1
5	Insurance services	0.0	0.0	310.0	0.1	0.1	1032.2
6	Financial services	0.3	1.6	602.7	1.6	1.2	3437.0
7	Computer and information services	0.1	0.4	386.4	0.6	0.5	2304.6
8	Royalties and license fees	0.1	0.3	291.0	1.0	0.8	3007.7
9	Other business services	1.6	10.1	2368.0	22.7	17.2	18004.5
10	Personal, cultural and recreational services	0.0	0.0	31.5	0.1	0.1	404.2
11	Government services, n.i.e.	3.2	20.7	390.6	13.8	10.5	1509.3
12	Total services	15.4	100.0	13269.1	131.8	100.0	78074.4

Note- Share (%) is the share of each of the top ten services in the total exports of services of Yemen to India.

Source: Own Calculations, OECD-WTO BaTIS

ANNEX C

[India's greenfield investment in selected African and Asian countries during cumulative 2010-18 (all values in USD million)]

I) India's investment in 21 African countries (top investing company is given at the end of each table in that particular sector where India's investment is the highest)

1. Angola

Sector	Value (cum. 2010-2018)	Share (%)
Industrial equipment	3	100 ^a
Total	3	100.00

^aThe share is due solely to investment by Karam Industries

2. Burkina Faso

Sector	Value (cum. 2010-2018)	Share (%)
Communications	<mark>94.1</mark>	95.44 ^a
Business services	4.5	4.56
Total	98.6	100.00

^aThe share is due solely to investment by Bharti Group

3. Burundi

Sector	Value (cum. 2010-2018)	Share (%)
Healthcare	2.92	100 ^a
Total	2.92	100.00

^aThe share is due solely to investment by Healthcare Global Enterprises

4. Chad

Sector	Value (cum. 2010-2018)	Share (%)
Communications	<mark>94.10</mark>	<mark>95.44^a</mark>
Business services	4.50	4.56
Total	98.60	100.00

^aThe share is due solely to investment by Bharti Group

5. Democratic Republic of Congo (DRC)

Sector	Value (cum. 2010-2018)	Share (%)
Communications	145.70	73.70 ^a
Financial services	11.00	5.56
Real estate	24.00	12.14
Software & IT services	17.00	<mark>8.60</mark>
Total	197.70	100.00

^aThe share is due solely to investment by Bharti Group

6. Ethiopia

Sector	Value (cum. 2010-2018)	Share (%)
Automotive OEM	94.50	5.25
Building materials	0.60	0.03

Financial services	22.00	1.22
Food & tobacco	194.36	10.81
Healthcare	2.40	0.13
Industrial equipment	2.80	0.16
Paper, printing & packaging	150.00	8.34
Pharmaceuticals	24.60	1.37
Textiles	1307.20	72.68 ^a
Total	1798.46	100.00

^aThe share is due solely to investment by Tata Group

7. Ghana

Sector	Value (cum. 2010-2018)	Share (%)
Automotive OEM	180.00	8.87
Business services	69.60	3.43
Chemicals	1500.00	73.89 ^a
Communications	145.00	7.14
Financial services	9.50	0.47
Healthcare	5.32	0.26
Metals	98.30	4.84
Software & IT services	22.20	1.09
Total	2029.92	100.00

^aThe share is due solely to investment by Rastriya Chemicals & Fertilizers

8. Kenya

Sector	Value (cum. 2010-2018)	Share (%)
Automotive OEM	121.30	10.02
Business services	57.70	4.77
Chemicals	165.70	13.69
Communications	174.85	14.45
Consumer products	16.30	1.35
Financial services	31.50	2.60
Healthcare	2.70	0.22
Leisure & entertainment	42.70	3.53
Non-automotive transport OEM	72.60	6.00
Pharmaceuticals	24.60	2.03
Plastics	2.50	0.21
Real estate	244.90	20.24^{a}
Software & IT services	30.90	2.55
Textiles	1.70	0.14
Warehousing	220.30	18.20
Total	1210.25	100.00

^aThe share is due solely to investment by Delta Corporation

9. Liberia

Sector	Value (cum. 2010-2018)	Share (%)
Coal, oil & gas	653.4	100 ^a

Total 653.4 100

^aThe share is due solely to investment by Jindal Steel & Power

10. Madagascar

Sector	Value (cum. 2010-2018)	Share (%)
Communications	282.3	80.59 ^a
Minerals	67.99	19.41
Total	350.29	100.00

^aThe share is due solely to investment by Bharti Group

11. Malawi

Sector	Value (cum. 2010-2018)	Share (%)
Communications	112.1	86.83 ^a
Software & IT services	17	13.17
Total	129.1	100.00

^aThe share is due solely to investment by Bharti Group

12. Mozambique

Sector	Value (cum. 2010-2018)	Share (%)
Beverages	38.4	0.59
Coal, oil & gas	<mark>6255.6</mark>	95.72 ^a
Consumer products	11	0.17
Financial services	11	0.17
Food & tobacco	36.824	0.56
Metals	15.6	0.24
Warehousing	166.66	2.55
Total	6535.08	100.00

^aThe share is due solely to investment by Bharat Petroleum

13. Niger

Sector	Value (cum. 2010-2018)	Share (%)
Business services	4.5	4.31
Communications	100	95.69 ^a
Total	104.5	100.00

^aThe share is due solely to investment by Bharti Group

14. Nigeria

Sector	Value (cum. 2010-2018)	Share (%)
Automotive OEM	92.78	3.34
Business services	97.3	3.51
Chemicals	1326.5	47.79 ^a
Communications	979.9	35.30
Consumer products	3.3	0.12
Electronic components	11.8	0.43
Financial services	9.5	0.34
Healthcare	10.72	0.39
Industrial equipment	2.8	0.10

Total	2775.86	100.00
Software & IT services	37.1	1.34
Pharmaceuticals	51.3	1.85
Paper, printing & packaging	7.66	0.28
Non-automotive transport OEM	145.2	5.23

^aThe share is due solely to investment by Nagarjuna Group

15. Rwanda

Sector	Value (cum. 2010-2018)	Share (%)
Communications	103.70	36.21 ^b
Consumer products	0.25	0.09
Financial services	9.50	3.32
Healthcare	107.92	37.69 ^a
Pharmaceuticals	65.00	22.70
Total	286.37	100.00

The share is due solely to investment by Healthcare Global Enterprises (HCG)

bThe share is due solely to investment by Bharti Group

16. Senegal

Sector	Value (cum. 2010-2018)	Share (%)
Coal, oil & gas	653.4	100 ^a
Total	653.4	100

^aThe share is due solely to investment by Jindal Organization

17. Sierra Leone

Sector	Value (cum. 2010-2018)	Share (%)
Communications	94.1	38.55^{a}
Metals	150	61.45 ^b
Total	244.1	100.00

18. Sudan

Sector	Value (cum. 2010-2018)	Share (%)
Software & IT services	13.8	100 ^a
Total	13.8	100

^aThe share is due solely to investment by Ramco Systems

19. Tanzania

Sector	Value (cum. 2010-2018)	Share (%)
Building materials	64.70	9.96
Business services	27.70	4.27
Communications	108.90	16.77
Financial services	69.50	10.70
Healthcare	<mark>72.92</mark>	11.23
Industrial equipment	119.35	18.38 ^a
Leisure & entertainment	34.50	5.31

^aThe share is due solely to investment by Bharti Group

^bThe share is due solely to investment by ABG International

Metals	50.00	7.70
Non-automotive transport OEM	60.60	9.33
Pharmaceuticals	31.20	4.80
Software & IT services	10.10	1.56
Total	649.47	100.00

^aThe share is due solely to investment by Sonalika Group & Mahindra Group

20. Uganda

Sector	Value (cum. 2010-2018)	Share (%)
Communications	<mark>265.60</mark>	52.20 ^a
Financial services	31.50	6.19
Healthcare	<mark>56.62</mark>	11.13
Industrial equipment	2.40	0.47
Leisure & entertainment	34.50	6.78
Non-automotive transport OEM	72.60	14.27
Pharmaceuticals	45.60	8.96
Total	508.82	100.00

^aThe share is due solely to investment by Bharti Group

21. Zambia

Sector	Value (cum. 2010-2018)	Share (%)
Automotive OEM	99.5	10.87
Chemicals	<mark>266.5</mark>	29.11 ^a
Coal, oil & gas	108	11.80
Communications	94.1	10.28
Healthcare	2.4	0.26
Metals	96.2	10.51
Renewable energy	231.7	$25.31^{\rm b}$
Software & IT services	17	1.86
Total	915.4	100

^aThe share is due solely to investment by Premier Explosives

II) India's investment in 7 Asian countries (top investing company is given at the end of each table in that particular sector where India's investment is the highest)

1. Bangladesh

Sector	Value (cum. 2010-2018)	Share (%)
Automotive OEM	1564.8	20.95
Business services	1000	13.39
Chemicals	133.3	1.79
Coal, oil & gas	611.4	8.19
Communications	75.7	1.01
Consumer products	39.6	0.53
Financial services	612.2	8.20
Food & tobacco	8.2	0.11
Healthcare	1.8	0.02
Leisure & entertainment	33.8	0.45

^bThe share is due solely to investment by Nava Bharat Ventures

Non-automotive transport OEM	67	0.90
Pharmaceuticals	40.9	0.55
Plastics	939.8	12.58
Rubber	40	0.54
Software & IT services	9.6	0.13
Textiles	<mark>2289.58</mark>	30.66 ^a
Total	7467.68	100.00

^aThe share is due solely to investment by Arvind Ltd (Arvind Mills)

2. Bhutan

Sector	Value (cum. 2010-2018)	Share (%)
Financial services	69.60	19.62
Textiles	18.10	5.10
Renewable energy	267.00	75.27^{a}
Total	354.70	100.00

^aThe share is due solely to investment by Tata Group

3. Cambodia

Sector	Value (cum. 2010-2018)	Share (%)
Automotive OEM	256.40	92.80 ^a
Hotels & tourism	1.50	0.54
Pharmaceuticals	18.40	6.66
Total	276.30	100.00

^aThe share is due solely to investment by Tata Group

4. Lao People's Democratic Republic

Sector	Value (cum. 2010-2018)	Share (%)
Wood products	9.99	10.90
Rubber	81.70	89.10^{a}
Total	91.69	100.00

^aThe share is due solely to investment by Apollo Tyres

5. Myanmar

Sector	Value (cum. 2010-2018)	Share (%)
Automotive OEM	1.35	0.09
Coal, oil & gas	1238.2	80.79 ^a
Consumer products	8.6	0.56
Financial services	179.9	11.74
Industrial equipment	3.5	0.23
Non-automotive transport OEM	36.4	2.38
Wood products	64.6	4.22
Total	1532.55	100.00

^aThe share is due solely to investment by Tata Group & Indian Oil

6. Nepal

Sector	Value (cum. 2010-2018)	Share (%)
Communications	<mark>79.4</mark>	24.55

Consumer products	21.83	6.75
Financial services	68.6	21.21
Food & tobacco	20.33	6.29
Non-automotive transport OEM	102.5	31.70^{a}
Pharmaceuticals	6.4	1.98
Software & IT services	6.2	1.92
Textiles	18.1	5.60
Total	323.36	100.00

^aThe share is due solely to investment by Indian Railways

7. Yemen

Sector	Value (cum. 2010-2018)	Share (%)
Renewable energy	326.2	100^{a}
Total	326.2	100

^aThe share is due solely to investment by Bharat Heavy Electricals (BHEL)

Source: fDI Markets; own calculations

ANNEX D

[Products at HS 6-digit level identified under Quadrant B during both the time-periods 2010-12 and 2015-17 for African and Asian countries: these products require proactive trade policies on part of the concerned countries' government due to their low shares in Indian imports in spite of having BRCA>0]

Table 1: Products identified under Quadrant B for selected African countries

Country	2010-2012	2015-2017
Country Angola	Machine-tools; parts and accessories, dividing heads and other special attachments for machine-tools (846630); Iron ores & concentrates (excl. Roasted iron pyrites) (260112); Glazed ceramic flags & paving/hearth/wall tiles (690890); Iron or steel tubes, pipes and hollow profiles, seamless, n.e.s. (730490); Plates, sheets, film, foil & strip, of polymers of propylene, non-cellular and not reinforced, laminated, supported or similarly combined with other materials (392020); Plates, sheets, film, foil & strip, non-cellular and not reinforced, laminated, supported or similarly combined with other materials (392049); Compound optical microscopes (excl. stereoscopic and microscopes for photomicrography, cinephotomicrography or microprojection) (901180); Aluminium structures (like plates, rods, profiles, tubes and the like) and their parts, n.e.s. (761090); Unglazed ceramic flags & paving/hearth/wall tiles (690790); Parts of the air-conditioning machines, with motor driven fan and elements for temperature control (841590); Nails, tacks, drawing pins, corrugated nails, staples (of iron and steel) and the like, with heads of other material or not, but excluding articles with heads of copper (731700); Wood, in the rough, n.e.s., whether or not stripped of bark or sapwood, or roughly squared, untreated, (440399); Iron or steel (excluding cast iron or stainless steel); other seamless, casing and tubing, of a kind used in drilling for oil or gas (730429); Pumps and compressors; parts, of air or vacuum pumps, air or other gas compressors and fans, ventilating or recycling hoods incorporating a fan (841490);	Plates, sheets, film, foil & strip, of plastics n.e.s., non-cellular and not reinforced, laminated, supported or similarly combined with other materials (392099); Plates, sheets, film, foil & strip, other than cellular (392190); Ethylene polymers; in primary forms, polyethylene having a specific gravity of less than 0.94 (390110); Polyethylene having a sp.gr. Of 0.94 or more (390120); Titanium & articles thereof, n.e.s (810890); Spring washers & other lock washers of iron and steel, non-threaded (731821); Waste & scrap of stainless steel (720421); Recovered (waste & scrap) paper/paperboard, n.e.s., and of unsorted waste and scrap (470790); Polymers of ethylene, in primary forms, n.e.s. (390190); Aluminium oxide (excl. artificial corundum) (281820); Polypropylene, in primary forms (390210); Aluminium, not alloyed, unwrought (760110); Unsaturated acyclic hydrocarbons, n.e.s (290129); Ceramic wares; for laboratory, chemical or other technical uses, articles having a hardness equivalent to 9 or more on the Mohs scale (690912); Lead waste & scrap (780200); Copper wire, of refined copper, of which the maximum cross-sectional dimension exceeds 6mm (740811); Aluminium waste & scrap (760200); Tubes, pipes & hoses, of vulcanised rubber (other than hard rubber), reinforced or otherwise combined with materials other than metal or textiles, with fittings (400942);
	Marble and travertine, merely cut, by sawing or otherwise, into blocks or slabs of a rectangular (including square) shape, having a specific gravity of 2.5 or more (251512); Aluminium alloys, bars, rods and profiles, other than hollow (760429);	Waste & scrap of cast iron (720410); Ferrous waste & scrap, n.e.s. (720449); Vinyl chloride, other halogenated olefin polymers; vinylidene chloride polymers, in primary forms (390450);

Engines; parts, suitable for use solely or principally with spark-ignition internal combustion piston engines (for other than aircraft) (840991); Human hair, unworked, whether or not washed or scoured, and waste of human hair (50100); Packing/wrapping machinery (842240); Interchangeable tools for hand tool, whether or not power-operated, for screw-driving or uses n.e.c (820790);

Other drinking glasses, other than of lead crystal or glass-ceramics (701337);

Waste & scrap of stainless steel (720421); Flat-rolled products of stainless steel, width less than 600mm, nes (722090):

Bearing housings, incorporating ball or roller bearings (848320);

Railway/tramway track construction material of iron and steel, n.e.s. (730290);

Bicycles and other cycles; including delivery tricycles, not motorised (871200);

Other tubes, pipes and hollow profiles (for example, open seam or welded, riveted or similarly closed), of iron or steel (730690); Bentonite (250810);

Other tubes and pipes (for example, welded, riveted or similarly closed), having circular cross-sections, the external diameter of which exceeds 406.4 mm, of iron or steel (730590);

Hand-/foot-operated air pumps (841420); Worked monumental/building stone of granite; articles thereof, simply cut or sawn, with a flat or

Unwrought lead other than refined, not containing by weight antimony as the principal other element (780199):

Rails of iron/steel (730210);

even surface (680223);

Copper waste & scrap (740400)

Worked monumental/building stone of granite; articles thereof, simply cut or sawn, with a flat or even surface (680223); Waste and scrap of primary cells, primary batteries and electric accumulators; spent primary cells, spent primary batteries and spent electric accumulators (854810); Diamonds; non-industrial, unworked or simply sawn, cleaved or bruted, but not mounted or set (710231)

Benin

Interchangeable tools for hand tool, whether or not power-operated, for screw-driving or uses n.e.c (820790);

Plates, sheets, film, foil & strip, other than cellular (392190);

Ball bearings (848210);

Articles of vulcanised rubber other than hard rubber, n.e.s. (401699);

Carboys, bottles, flasks, jars, pot and other containers of glass, (not ampoules), used for the conveyance or packing of goods (701090); Photosensitive semiconductor device (854140); Pulleys and flywheels, including pulley blocks (848350);

Self-adhesive plates, sheets, film, foil, tape, strip, other flat shapes thereof, other than in rolls of a width not exceeding 20cm (391990);

Fixed electrical resistors for a power handling capacity <= 20 W (excl. Heating resistors) (853321);

Other parts & accessories for the motor vehicles (870899);

Lac; natural gums, gum resins, resins and balsams, n.e.s. (130190);

Compound optical microscopes (excl. Stereoscopic and microscopes for photomicrography, cinephotomicrography or microprojection) (901180);

Waste & scrap of stainless steel (720421); Recovered (waste & scrap)

paper/paperboard, n.e.s., and of unsorted

waste and scrap (470790); Preparations of a kind used in animal

feeding, put up for retail sale (except dog or cat food) (230990);

Dried leguminous vegetables, n.e.c., shelled, whether or not skinned or split (71390);

Wood sawn/chipped length wise, sliced or peeled, whether or not planed, sanded or end-jointed, of a thickness exceeding 6mm (440710):

Waste & scrap of cast iron (720410); Ferrous waste & scrap, n.e.s. (720449); Aluminium waste & scrap (760200); Taps, cocks, valves and similar appliances; for pipes, boiler shells, tanks, vats or the like, including thermostatically controlled valves (848180);

Mechanical appliances, for projecting, dispersing or spraying liquids or powders, for other than agricultural or horticultural use, whether or not hand-operated (842489);

Printing ink (excl. Black) (321519);

Tapered roller bearings, incl. Cone and tapered roller assemblies (848220);

Organic comps. N.e.s. (294200);

Iron or steel threaded articles except screw, nut, bolts (731819);

Electrical apparatus; n.e.c., for switching or protecting electrical circuits, for a voltage not exceeding 1000 volts (853690);

Relays for a voltage > 60 V but $\leq 1.000 \text{ V}$ (853649);

Articles of plastics, n.e.s. (392690);

Parts and accessories of data processing equipment (847330);

Machines for used in the milling industry or for the working of cereals or dried leguminous vegetables, other than farm-type machinery (843780);

Transmission belts/belting, n.e.s., of vulcanized rubber (401039);

Machine-tools for wood, cork or hard plastic, etc. N.e.s. (846599);

Other Parts suitable for use solely or principally with the apparatus of headings 8525 to 8528 (reception and transmission apparatus, monitors, etc.) (852990);

Air conditioning machines comprising a motordriven fan and elements for changing the temperature and humidity, window or wall types, self-contained or split-system (841510); Static converters (850440);

Electric motors of an output not exceeding 37.5W (850110);

Woven fabrics of synthetic staple fibres, containing 85% or more by weight of polyester staple fibres, other than unbleached or bleached (551219);

Parts of cranes, work trucks, shovels, constr machine (843149);

Perfuming or deodorizing preparations n.e.s., Depilatories & other perfumery (330790); Sewing machines of the household type (845210); Ceramic sinks, wash basins, wash basin pedestals, baths, bidets, water closet pans, flushing cisterns, urinals and similar sanitary fixtures; of other than porcelain or china (691090);

Dried leguminous vegetables, n.e.c., shelled, whether or not skinned or split (71390); Electric lamps & lighting fittings, nec (940540); Palm oil, other than crude, & its fractions, whether or not refined, but not chemically modified (151190);

Waste & scrap of tinned iron/steel (720430);

Gum Arabic (130120);

Wood, n.e.s., sawn/chipped lengthwise/sliced lengthwise/sliced/peeled, whether or not planed, sanded or endjointed, of a thickness exceeding 6mm (440729);

Topical wood, n.e.s., in the rough, whether or not stripped of bark or sapwood, or roughly squared, untreated (440349)

Zinc alloys, unwrought (790120);

Parts of electric filament or discharge lamps, sealed beam lamp units, ultra-violet or infra-red lamps and arc-lamps, n.e.s. (853990);

Electrical apparatus like switches n.e.c. In heading no. 8536, for a voltage not exceeding 1000 volts (853650);

Sheets n.e.s. For veneer or plywood, other wood sawn lengthwise, sliced or peeled, whether or not planed, sanded or end-jointed, not thicker than 6mm (440890);

Parts suitable for use solely or principally with the apparatus of heading 8535, 8536 or 8537, n.e.s. (excl. boards, panels, consoles, desks, cabinets and other bases for the goods of heading 8537, not equipped with their apparatus) (853890);

Projectors, not of a kind solely principally used in an automatic data processing system of heading 84.71 (852869);

Filtering/purifying machinery & apparatus, other than intake air filters for internal combustion engines (842139);

Men's/boys' shirts (excl. knittedd/crocheted), of textile materials n.e.s. (620590);

Plywood; consisting only of sheets of wood (not bamboo), each ply 6mm or thinner, with at least one outer ply of tropical wood (441231);

Hotographic goods; chemical preparations other than sensitised emulsions, put up in measured portions or put up for retail sale in a form ready for use (370790):

Tubes, pipes & hose fittings (e.g. Joints, elbows, flanges) (391740);

Other printers, copying machines, single-function printing, not capable of connecting to an automatic data processing machine or to a network (844339); Quicklime (252210);

Waste & scrap of stainless steel (720421); Permanent magnets and articles intended to become permanent magnets after magnetization, of materials other than metal (850519);

Newsprint, in rolls/sheets (480100);

Paper and paperboard; coated with kaolin or other inorganic substances, for printing/writing/graphics, having 10% or less by weight of total fibres got by mechanical/chemi-mechanical process, sides exceeding 435mm and 297mm, unfolded, sheets (481019);

Parts of the air-conditioning machines, with motor driven fan and elements for temperature control (841590);

Other colour reception apparatus for television, whether not incorporating radio broadcast receivers sound video recording reproducing apparatus, (852872);

Pears & quinces, fresh (80820);

Tubes, pipes & hollow profiles of stainless steel (excluding cold-drawn or cold-rolled) (730449);

Mountings, fittings and similar articles; suitable for other than buildings or furniture, of base metal (830249):

Carbon; carbon blacks and other forms of carbon n.e.s. (280300);

Sunglasses (900410);

Aluminium alloys, unwrought (760120);

Articles & equipment for sports, for outdoor games and recreation n.e.c. (950699);

Paper & paperboard, coated/impregnated or covered with plastics (excluding adhesives), other than bleached and weighing more than 150g/m2 (481159):

Electric conductors, for a voltage > 1.000 V, insulated, n.e.s. (854460);

Clays (excluding expanded clays of heading no. 6806); n.e.s. In heading no. 2508, whether or not calcined (250840);

Peas (Pisum sativum), dried, shelled, whether or not skinned or split (71310);

Wooden doors & their frames & thresholds (441820);

Plates, sheets, strip, rods and profile shapes, of vulcanised rubber other than hard rubber (400829);

Limestone flux; limestone and other calcareous stone, of a kind used for the manufacture of lime or cement (252100);

Palm oil and its fractions, crude, not chemically modified (151110);

Floor coverings of plastics (other than polymers of vinyl chloride), whether or not self-adhesive, in rolls or in the form of tiles (391890);

Iron/steel wool; pot scourers & scouring or polishing pads, gloves and the like, of iron/steel (732310);

Bran, sharps and other residues; of other cereals, whether or not in the form of pellets, derived from the sifting, milling or other workings thereof (230240);

Vegetable seeds, of a kind used for sowing (120991);

Sacks & bags (excl. those having a base of a width of 40cm or more) (481940);

Washing/bleaching/dyeing machines (845140); Wooden furniture of a kind used in office (940330):

Unglazed ceramic flags & paving/hearth/wall tiles (690790);

Butanes, liquefied (271113);

Diammonium hydrogenorthophosphate (diammonium phosphate) (310530);

Recovered waste and scrap paper or paperboard, incl. Unsorted waste and scraperboard, Mechanical Pulp (470730);

Mattress supports (940410);

Machinery for the preparation of fruits, nuts or vegetables (843860);

Ginger (91010);

Ferrous waste & scrap, n.e.s. (720449)

Burkina	Ferrous waste & scrap, n.e.s. (720449);	Filtering/purifying machinery & apparatus
Faso	Mixtures & articles of heat-insulating, sound-insulating or sound-absorbing mineral materials, n.e.s. (680690); Reciprocating positive displacement pumps, n.e.s. In 8413, for liquids (841350); Lubricating preparations (other than for the treatment of textile and similar materials) containing less than 70% (by weight) of petroleum oils or oils obtained from bituminous minerals (340319); Micro-crystalline petroleum wax, slack wax, lignite wax, ozokerite, peatwax other mineral waxes and similar products obtained by synthesis or by other processes, whether or not coloured (271290); Bran, sharps and other residues; of other cereals, whether or not in the form of pellets, derived from the sifting prilling or other wordings thereof	(842121); Shutters, blinds (incl. Venetian blinds) and similar articles and parts thereof (392530); Oil seeds & oleaginous fruits, n.e.s., whether or not broken (120799)
	the sifting, milling or other workings thereof (230240)	
Burundi	Ferrous waste & scrap, n.e.s. (720449); Phosphoric acid and polyphosphoric acids (280920); Fuel/lubricating/cooling medium pumps for internal combustion piston engines (841330); Winding wire, other than of copper (854419); Parts of machinery for sorting, screening, separating, washing, crushing etc mineral substances, for agglomerating, shaping or moulding solid fuels, ceramic pastes etc, for forming foundry moulds of sand (847490); Chemical products & preparations of the chemical or allied industries, n.e.s. (382490); DC motors and generators of an output exceeding 375kw (850134); Tanned/crust hides & skins of bovine or equine animals, in the wet state (including wet blue), without hair on, excluding full grains, unsplit; grain splits (410419); Tanned/crust skins of sheep/lambs, without wool on, whether or not split, but not further prepared, in the wet state (including wet-blue) (410510); Raw hides & skins of bovine (including buffalo) and equine animals, fresh, salted or preserved, but not tanned, parchment dressed or further prepared, whether or not dehaired or split (410190)	Universal AC/DC motors of an output exceeding 37.5W (850120); Storage units of ADP (847170); Parts of the engines & motors of 8412 (841290)
Central African Republic	Articles of iron/steel, n.e.s. (732690); Taps, cocks, valves and similar appliances; for pipes, boiler shells, tanks, vats or the like, including thermostatically controlled valves (848180); Fuel/lubricating/cooling medium pumps for internal combustion piston engines (841330); Aluminium waste & scrap (760200); Copper waste & scrap (740400); Ferrous waste & scrap, n.e.s. (720449); Wood, in the rough, n.e.s., whether or not stripped of bark or sapwood, or roughly squared, untreated, (440399)	Photosensitive semiconductor device (854140); Monofilament of which any cross-sectional dimension exceeds 1mm, rods, sticks and profile shapes, whether or not surfaceworked but not otherwise worked (391690); Aluminium waste & scrap (760200); Vacuum cleaners, with self-contained electric motor, n.e.c. (850819); Diamonds; non-industrial, unworked or simply sawn, cleaved or bruted, but not mounted or set (710231); Plugs and sockets for a voltage <= 1000 V (excl. Lamp-holders) (853669);

	T	Unwrought lead other than refined, not
		containing by weight antimony as the
		principal other element (780199);
		Winding wire for electrical purposes, of
		copper, insulated (854411);
		Paper/paperboard labels of all kind, printed (482110);
		Handbags, whether/not with shoulder strap
		and including those without handle, with
		outer surface of plastic sheeting or of textile materials (420222);
		Other parts and accessories for printing
		machinery excl. 844391 (844399);
		Semi conductor media, Smart cards for the
		recording of sound of other phenomena, but excl. Products of Ch. 37. (852352);
		Pumps n.e.s. In 84.13 (841381);
		Disperse dyes & preparations based thereon
		(320411);
		Raw hides & skins of bovine (including buffalo) and equine animals, fresh, salted or
		preserved, but not tanned, parchment
		dressed or further prepared, whether or not
		dehaired or split (410190);
		Lac; natural gums, gum resins, resins and
		balsams, n.e.s. (130190);
		Petroleum oils and oils obtained from
		bituminous minerals, crude (270900)
Chad	Parts and accessories of office machines, nes	Winding wire for electrical purposes, of
	(847340);	copper, insulated (854411);
	Other parts and accessories for printing machinery excl. 844391 (844399);	Paper/paperboard labels of all kind, printed (482110);
	Watch straps, watch bands, watch bracelets, and	Handbags, whether/not with shoulder strap
	parts thereof; of base metal, whether or not gold-	and including those without handle, with
	or silver-plated (911320);	outer surface of plastic sheeting or of textile
	Telephones for cellular networks or for other wireless networks (851712);	materials (420222); Other parts and accessories for printing
	Storage units of ADP (847170);	machinery excl. 844391 (844399);
	Articles of plastics, n.e.s. (392690);	Semi conductor media, smart cards for the
	Parts of telephone sets, incl. Telephones for	recording of sound of other phenomena, but
	cellular networks for other wireless networks other	excl. Products of Ch. 37. (852352);
	apparatus for the transmission reception of voice,	Pumps n.e.s. In 84.13 (841381);
	images other data, incl. Apparatus for	Disperse dyes & preparations based thereon
	communication in a wired wireless network	(320411);
	(851770);	Raw hides & skins of bovine (including
	Parts & accessories n.e.s. In Ch.90 (903300);	buffalo) and equine animals, fresh, salted or
	Imitation jewellery other than of base metal, whether or not plated with precious metal	preserved, but not tanned, parchment dressed or further prepared, whether or not
	(711790)	dehaired or split (410190);
	(,)	Lac; natural gums, gum resins, resins and
		balsams, n.e.s. (130190);
		Petroleum oils and oils obtained from
		bituminous minerals, crude (270900)
Democratic	Instruments, appliances and machines; for	Non-threaded articles, n.e.s., of iron/steel
Republic of	measuring or checking n.e.c. (903180);	(731829);
Congo	Articles of jewellery & parts thereof, of precious	Microtomes and parts & accessories thereof
	metal (excluding silver) whether or not plated or	(902790);
	clad with precious metal (711319);	Aluminium alloys, unwrought (760120);
	Copper waste & scrap (740400);	

Anti-oxidising preparations and other compound stabilisers; for rubber or plastics, n.e.s. Or included (381230);

Machines for arc welding of metals, incl. Plasma arc welding, neither fully nor partly automatic (851539);

Ferrous waste & scrap, n.e.s. (720449); Technically spec. Natural rubber (TSNR), in primary forms or in plates, sheets or strip (excluding latex and smoked sheets) (400122); Waste & scrap of cast iron (720410) Machines and mechanical appliances; having individual functions, n.e.c. (847989);

Precious and semi-precious stones (excl. Diamonds, rubies, sapphires and emeralds), worked other than simply sawn or roughly shaped, not strung, mounted or set (710399);

Precious (other than diamonds) and semiprecious stones, unworked or simply sawn or roughly shaped, not strung, mounted or set (710310);

Aluminium waste & scrap (760200); Ferrous waste & scrap, n.e.s. (720449); Stationery n.e.s., made of paper or paperboard (482090);

Coniferous wood, in the rough, whether/not stripped of bark or sapwood, or roughly squared, untreated (440320);

Petroleum oils and oils obtained from bituminous minerals, crude (270900); Sheets n.e.s. For veneer or plywood, other wood sawn lengthwise, sliced or peeled, whether or not planed, sanded or endjointed, not thicker than 6mm (440890); Copper waste & scrap (740400); Diamonds; non-industrial, unworked or

Diamonds; non-industrial, unworked or simply sawn, cleaved or bruted, but not mounted or set (710231);

Diamonds, unsorted (710210);

Waste and scrap of primary cells, primary batteries and electric accumulators; spent primary cells, spent primary batteries and spent electric accumulators (854810); Copper ores & concentrates (260300); Unrefined copper; copper anodes for electrolytic refining (740200);

Waste & scrap of cast iron (720410); Vegetable alkaloids, natural/reproduced by synthesis, and their salts, ethers, esters and other derivatives, n.e.s. (293999)

Djibouti

Articles of iron/steel, n.e.s. (732690);

Petroleum oils and oils from bituminous minerals, not containing biodiesel, not crude, not waste oils; preparations n.e.s. (271019);

Parts of the filtering/purifying machinery for filtering or purifying liquids or gases, n.e.s. (842199);

Pumps n.e.s. In 84.13 (841381);

Articles of vulcanised rubber other than hard rubber, n.e.s. (401699);

Copper waste & scrap (740400);

Marble and travertine, merely cut, by sawing or otherwise, into blocks or slabs of a rectangular (including square) shape, having a specific gravity of 2.5 or more (251512);

Aluminium waste & scrap (760200);

Precious (other than diamonds) and semi-precious stones, unworked or simply sawn or roughly shaped, not strung, mounted or set (710310);

Aluminium waste & scrap (760200); Ferrous waste & scrap, n.e.s. (720449); Petroleum oils and oils from bituminous minerals, not containing biodiesel, not crude, not waste oils; preparations n.e.s. (271019);

Copper waste & scrap (740400); Chickpeas (garbanzos), dried, shelled, whether or not skinned or split (71320); Marble and travertine, merely cut, by sawing or otherwise, into blocks or slabs of a rectangular (including square) shape, having a specific gravity of 2.5 or more (251512);

Precious (other than diamonds) and semiprecious stones, unworked or simply sawn or roughly shaped, not strung, mounted or set (710310);

Chickpeas (garbanzos), dried, shelled, whether or Glass mirrors other than rear-view not skinned or split (71320); (700991): Tubes, pipes & hoses, of vulcanised rubber (other Waste oils; of petroleum or obtained from than hard rubber), reinforced or otherwise bituminous minerals, not crude and combined with materials other than metal or preparations n.e.c., weight 70% or textiles, with fittings (400942); preparations of the same, not containing Beans of the species vigna mungo (L.) Hepper or polychlorinated biphenyls (pcbs), vigna radiata (L.) Wilczek, shelled, whether or not polychorinated terphenyls (pcts) or skinned or split, dried (71331); polybrominated biphenyls (pbbs) (271099); Tanned/crust hides & skins of bovine or equine Ash & residues (excl. From the manufacture animals, in the dry state (crust), without hair on, of iron or steel), containing mainly zinc, (other than full grains, unsplit; grain splits) hard zinc spelter (262019); (410449);Parts of the machinery of heading 8422 Tubes, pipes & hollow profiles of cast iron (842290); Waste and scrap of primary cells, primary (730300);Ferrous waste & scrap, n.e.s. (720449); batteries and electric accumulators; spent Kidney beans, incl. White pea beans (phaseolus primary cells, spent primary batteries and vulgaris), dried, shelled, whether or not skinned or spent electric accumulators (854810); split (71333); Oil seeds & oleaginous fruits, n.e.s., Waste & scrap of cast iron (720410); whether or not broken (120799) Ginger (91010); Raw skins of sheep/lambs, pickled (but not tanned, parchment-dressed or further preserved), without wool on (410221); Turmeric (curcuma) (91030); Waste and scrap of primary cells, primary batteries and electric accumulators; spent primary cells, spent primary batteries and spent electric accumulators (854810) Eritrea Commodities not specified (999999); Polymers of ethylene, in primary forms, Ferrous waste & scrap, n.e.s. (720449); n.e.s. (390190) Tanned/crust hides & skins of bovine or equine animals, in the wet state (including wet blue), without hair on, excluding full grains, unsplit; grain splits (410419); Vegetable saps and extracts; n.e.s. In item no. 1302.1 (130219) Discs, tapes, solid state non-volatile storage Ethiopia Iron or steel tubes, pipes and hollow devices, smart cards and other media for the profiles, seamless, n.e.s. (730490); recording of sound of other phenomena, whether Vinyl acetate copolymers other than Poly not recorded, incl. Matrices and masters for the (vinyl alcohol), in primary forms (390529); production of discs, but excl. Products of Ch.37., Polyesters, n.e.s., unsaturated, in primary other n.e.s. (852380); forms (390791); Other printers, copying machines, single-function Sacks & bags, of a kind used for the printing, capable of connecting to an automatic packing of goods, of jute or of other textile bast fibres (630510); data processing machine or to a network (844332); Transmission shafts (including cam shafts and Paper & paperboard, coated with inorganic crank shafts) and cranks (848310); substances only, whether or not surface-Parts suitable for use solely or principally with coloured, surface-decorated or printed, electric motors and generators, electric generating n.e.s., in rolls or sheets (481099); sets and rotary converters, n.e.s. (850300): Brooms, brushes, mops, feather dusters, squeegees, hand operated mechanical floor "Therapeutic respiration apparatus such as ozone, oxygen, aerosol therapy apparatus; artificial sweepers; (not motorised), prepared knots respiration or other therapeutic respiration and tufts for brooms or brush making n.e.c. apparatus (960390); (901920);" Iron ores and concentrates; non-Hooks, eyes & eyelets of a kind used for clothing, agglomerated (260111); footwear, awnings, handbags, travel goods or other Tricycles, scooters, pedal cars and similar made up articles, of base metal (830810); wheeled toys; dolls' carriages; dolls; other toys; reduced-size (scale) models and

Sewing needles, knitting needles, bodkins, crochet hooks and similar articles, for use in the hand, n.e.s., of iron/steel (731990);

Threaded screws & bolts of iron/steel, n.e.s., whether or not with their nuts or washers (731815);

Pens and pencils n.e.s.; duplicating stylos, penholders, pencil-holders, and similar holders, parts of the articles (960899);

Iron or steel threaded articles except screw, nut, bolts (731819);

Ball point pens (960810);

Gears & gearing (not toothed wheels, chain sprockets and other transmission elements presented separately); ball or roller screws; gear boxes and other speed changers, including torque converters (848340);

Filtering/purifying machinery & apparatus (842121);

Gaskets, washers & other seals of vulcanised rubber, of non-cellular rubber (401693); Lubricating preparations (excl. Those for the treatment of textile and similar materials), not containing petroleum oils or oils obtained from bituminous minerals (340399);

Engines parts for internal combustion piston engines (excluding spark-ignition) (840999); Springs & leaves for springs, n.e.s., of iron/steel (732090);

Input or output units for automatic data processing machines, whether or not containing storage units in the same housing (excl. Keyboards) (847160); Articles of aluminium n.e.s. (761699);

Lenses, prisms, mirrors & other optical elements, mounted, being parts or fittings for instruments or apparatus, of any material (excluding elements of glass not optically worked) (900290);

Speed indicators & tachometers; stroboscopes (902920);

Bearings; n.e.c., including combined ball/roller (848280);

Clasps, frames with clasps, buckles, buckle clasps; of base metal, of a kind used for clothing, footwear, awnings, handbags, travel goods or other made up articles, including parts (830890); Electronic integrated circuits; processors and controllers, whether or not combined with memories, converters, logic circuits, amplifiers, clock and timing circuits, or other circuits (854231);

Parts of boring or sinking machinery (843143); Articles of plastics, n.e.s. (392690);

Electrical apparatus; n.e.c., for switching or protecting electrical circuits, for a voltage not exceeding 1000 volts (853690);

Hydrometers and similar floating instruments, barometers, hygrometers, psychrometers, thermometers, pyrometers; recording or not, any combination of these instruments (excluding similar recreational models, working or not; puzzles of all kinds (950300);

Electric machines and apparatus for laser or other light or photon beam, ultrasonic, electron beam, magnetic pulse or plasma arc welding; electric machines and apparatus for hot spraying of metals, metal carbides or cermets (excl. Metal spray guns specified elsewhere) (851580);

Carpets and other textile floor coverings; tufted, of nylon or other polyamides, whether or not made up (570320); Fittings for office articles (eg. Letter clips, letter corners, paper clips, indexing tags and similar articles), including parts, of base metal (830590):

Filter paper & paperboard, cut to shape (482320);

Static converters (850440);

Plates, separators and other parts of electric accumulators, n.e.s. (850790);

Input or output units for automatic data processing machines, whether or not containing storage units in the same housing (excl. Keyboards) (847160);

Instruments For Measuring Length, Nesoi (901780);

Ties, bow ties & cravats, of silk or silk waste (not knittedd or crocheted) (621510); Road wheels & parts & accessories (870870);

Ball point pens (960810);

Dental instruments and appliances; other than dental drill engines (901849); Paper knives, letter openers, erasing knives, pencil sharpeners and blades therefor (821410);

Paints & varnishes (incl. Enamels & lacquers), based on polyesters, dispersed or dissolved in a non-aqueous medium (320810);

Projectors, not of a kind solely principally used in an automatic data processing system of heading 84.71 (852869);

Parts of the machinery for treating textile fabrics (845190);

Table/kitchen/other h-hold. Articles and their parts, of Iron or steel, n.e.s. (732399); Articles of aluminium n.e.s. (761699); Parts of Machinery for working rubber or plastics (847790);

Alkyd resins, in primary forms (390750); Men's/boys' jackets & blazers, of textile materials n.e.s. (not knittedd or crocheted) (620339);

Sign-plates, name-plates, address-plates and similar plates, numbers, letters and other symbols; of base metal, excluding those of heading no. 9405 (831000);

thermometers and barometers not combined with other instruments) (902580);

Parts suitable for use solely or principally with the apparatus of heading 8535, 8536 or 8537, n.e.s. (excl. Boards, panels, consoles, desks, cabinets and other bases for the goods of heading 8537, not equipped with their apparatus) (853890); Headphones, earphones and combined microphone/speaker sets (excl. Telephone sets, hearing aids and helmets with built-in headphones, whether or not incorporating a microphone) (851830):

Other stemware drinking glasses, other than of lead crystal or glass-ceramics (701328); Processing units for automatic data processing machines, whether or not containing in the same housing one or two of the following types of unit storage units, input units, output units (847150); Parts of Machines & mechanical appliances of those having individual functions (847990); Clays (excluding expanded clays of heading no. 6806); n.e.s. In heading no. 2508, whether or not calcined (250840);

Television cameras, digital cameras and video camera recorders (852580);

DC Electric motors and generators of an output not exceeding 750W (850131);

Fixed electrical capacitors (excl. Tantalum, aluminium electrolytic, ceramic, paper, plastic and power capacitors) (853229);

Ties, bow ties & cravats, of textile materials other than silk or man-made fibres (not knittedd or crocheted) (621590);

Ethylene polymers; in primary forms, polyethylene having a specific gravity of less than 0.94 (390110);

Engines; parts, suitable for use solely or principally with spark-ignition internal combustion piston engines (for other than aircraft) (840991); Diagnostic/laboratory reagents on a on a backing and prepared diagnostic or laboratory reagents whether or not on a backing; certified reference material (382200);

Relays for a voltage > 60 V but $\leq 1.000 \text{ V}$ (853649);

Ultrasonic scanning apparatus (901812); Articles of leather/of composition leather, n.e.s. (420500);

Aluminium foil, not backed, not rolled, of a thickness not exceeding 0.2mm (760719); Ball bearings (848210);

Silica sands & quartz sands, whether or not coloured (250510);

Centrifugal pumps, n.e.s. In 8413, for liquids (841370);

Poly(methyl methacrylate), in primary forms (390610);

Boxes, cases, crates and similar articles for the conveyance or packing of goods (392310); Articles of iron/steel, n.e.s. (732690); Trade advertising material, commercial catalogue and the like (491110); Stoppers, lids, caps and other closures, for the conveyance or packing of goods (392350);

Weighing machinery, having a maximum weighing capacity not exceeding 30kg (excluding balances of a sensitivity of 5cg or better) (842381);

Garden/similar umbrellas (660110); Printed/illustrated postcards; printed cards bearing personal greetings, messages or announcements, whether or not illustrated, with or without envelopes or trimmings (490900):

Wind musical instruments, other than brass, (eg. Clarinet, bagpipes, pipe organs), other than fairground organs and mechanical street organs (920590);

Weighing machine weights of all kinds; parts of weighing machinery (842390); Worn clothing & other worn articles (630900):

Tailors' dummies and other lay figures; automata and other animated displays used for shop window dressing (961800);
Zinc waste & scrap (790200);
Raw skins of sheep/lambs
(fresh/salted/dried/limed/otherwise,
preserved other than by pickling, (but not tanned, parchment-dressed or further preserved), without wool on (410229);
Instruments and apparatus using optical radiations (UV, visible, IR), (other than spectrometers, spectrophotometers and

Paper/paperboard labels of all kind, printed (482110);

spectrographs) (902750);

Men's/boys', anoraks (incl. Ski-jackets), wind-cheaters, wind-jackets and similar articles; men's or boys', of textile materials (other than wool or fine animal hair, cotton or man-made fibres), not knittedd or crocheted (620199);

Boring/sinking machinery, self-propelled, n.e.s. (843041);

Hydraulic presses for working metal (846291);

Articles of apparel & clothing accessories of paper, cellulose wadding or fibres (481850);

Table/kitchen/other h-hold. Articles and their parts, of aluminium (761519); Household articles & toilet articles, of plastic (392490);

Ties, bow ties & cravats, of man-made fibres (not knittedd or crocheted) (621520); Paper/paperboard labels of all kind, unprinted (482190); Bamboos (140110);

Men's/boys', anoraks (incl. Ski-jackets), wind-cheaters, wind-jackets and similar articles; men's or boys', of textile materials (other than wool or fine animal hair, cotton or man-made fibres), not knittedd or crocheted (620199);

Ferrous waste & scrap, n.e.s. (720449);

Food preparations, n.e.s. (210690);

Dental instruments and appliances; other than dental drill engines (901849);

Other articles of glass, n.e.s. (702000);

Woven fabrics obt. From high tenacity yarn, of nylon, other polyamides or of polyesters (540710); Non-threaded articles, n.e.s., of iron/steel (731829);

Machines for the reception, conversion and transmission regeneration of voice, images other data, incl. Switching and routing apparatus (851762);

Gaskets and similar joints of metal sheeting combined with other material or two or more layers of metal (848410);

Essential oils, n.e.s., (terpeneless or not), including concretes and absolutes (330129);

Boards, panels, consoles, desks, cabinets and other bases for the goods of heading 8537, not equipped with their apparatus (853810);

Pumps n.e.s. In 84.13 (841381);

Parts & accessories n.e.s. In Ch.90 (903300); Articles of glass such as statuettes and other ornaments of lamp worked glass, other than imitation jewellery (701890);

Telephones for cellular networks or for other wireless networks (851712);

Connectors for optical fibres, optical fibre bundles cables (853670);

Parts & accessories of the of spinning, combing, drawing machines, n.e.s. (844839);

Plates, sticks, tips & the like for tools, unmounted, of sintered metal carbides or cermets (820900); Paper/paperboard labels of all kind, printed (482110);

Steering wheels, steering columns and steering boxes & parts thereof (870894);

Handbags, whether/not with shoulder strap and including those without handle, with outer surface of plastic sheeting or of textile materials (420222); Made up clothing accessories, n.e.s., (not knittedd or crocheted) (621710);

Insulated electric conductors for a voltage not exceeding 1000 volts, not fitted with connectors (854449):

Portable automatic data processing (ADP) machines, weighing not more than 10kg, consisting of at least a central processing unit, a keyboard and a display (847130);

Envelopes, of paper/paperboard (481710); Tableware & kitchenware, of porcelain or china (691110); Dried leguminous vegetables, n.e.c., shelled, whether or not skinned or split (71390);

Tanned/crust skins of sheep/lambs, without wool on, whether or not split, but not further prepared, in the wet state (including wet-blue) (410510);

Fresh orchids (60313);

Pepper (genus Piper), neither crushed nor ground (90411);

Floor coverings of coconut fibres (coir), woven, (not tufted or flocked), whether or not made up (570220);

Ash & residues (excl. From the manufacture of iron or steel), containing mainly zinc, hard zinc spelter (262019);

Marble and travertine, merely cut, by sawing or otherwise, into blocks or slabs of a rectangular (including square) shape, having a specific gravity of 2.5 or more (251512);

Envelopes, of paper/paperboard (481710); Removable insoles, heel cushions & and similar articles, of materials n.e.s. (640699); Bars, rods & profiles, of copper-zinc base alloys (brass) (740721);

Other articles of copper, n.e.s. (741999); Cardamoms (90830);

Raw skins of sheep/lambs, fresh, or salted, dried, limed, pickled or otherwise preserved, (but not tanned, parchment-dressed or further prepared), with wool on (410210);

Articles of paper, cellulose wadding or fibres n.e.s. (481890);

Raw hides & skins of bovine (including buffalo) and equine animals, fresh, salted or preserved, but not tanned, parchment dressed or further prepared, whether or not dehaired or split (410190);

Parts of the non-electric domestic appliances, of iron or steel (732190); Stationery n.e.s., made of paper or paperboard (482090);

Calendars of any kind, printed, including calendar blocks (491000);

Human blood; animal blood prepared for therapeutic, prophylactic or diagnostic uses; cultures of micro-organisms (excluding yeast) and similar products (300290); Sanitary ware & parts thereof, of copper (741820):

Leather further prepared after tanning or crusting, including parchment-dressed leather, of animals (other than sheep and lambs, goats and kids, swine and reptiles), without wool or hair on, whether or not split, n.e.s. (411390);

Precious (other than diamonds) and semiprecious stones, unworked or simply sawn Paints & varnishes (incl. Enamels & lacquers), based on polyesters, dispersed or dissolved in a non-aqueous medium (320810);

Parts of the sewing machines n.e.s. (845290); Transistors with a dissipation rate >= 1 W (excl. Photosensitive transistors) (854129);

Parts and accessories of data processing equipment (847330);

Copper waste & scrap (740400);

Non-ionic surface-active agents, whether or not put up for retail sale (340213);

Belts & bandoliers, of leather/composition leather (420330);

Parts of the h-hold./laundry-type washing machine (845090);

Static converters (850440);

Parts & accessories of regulating or controlling instruments and apparatus; automatic (903290); Polyethylene having a sp.gr. Of 0.94 or more (390120);

Parts and accessories for machine tools, for laser operation, metalworking machining centers, lathes and drilling machines, etc., others (846693);

Filtering/purifying machinery & apparatus, other than intake air filters for internal combustion engines (842139);

Parts and accessories for instruments, appliances and machines measuring or checking (903190); Pumps and compressors; parts, of air or vacuum pumps, air or other gas compressors and fans, ventilating or recycling hoods incorporating a fan (841490);

Spectrometers, spectrophotometers and spectrographs using optical radiations (UV, visible, IR) (902730);

Other parts and accessories of looms, auxiliary machinery, n.e.s. (844849);

Ties, bow ties & cravats, of man-made fibres (not knittedd or crocheted) (621520);

Sunglasses (900410);

Clothing accessories, of leather/composition leather, n.e.s. (420340);

Diodes (excl. Photosensitive or light emitting diodes) (854110);

Thermometers and pyrometers; (other than liquid filled, for direct reading), not combined with other instruments (902519);

Refrigerating/freezing equip. N.e.s (841869); Machines and mechanical appliances; having individual functions, n.e.c. (847989); Calendars of any kind, printed, including calendars

Calendars of any kind, printed, including calendar blocks (491000);

Combined refrigerator-freezers, fitted with separate external doors, electric or other (841810); Furniture for medical, surgical, veterinary use (e.g. Operating tables, examination tables, hospital beds with mechanical fittings) and parts thereof (940290);

Marble, travertine and alabaster; simply cut or sawn, with a flat or even surface (680221);

or roughly shaped, not strung, mounted or set (710310);

Tanned/dressed furskins, whole, with or without head, tail or paws, not assembled, n.e.s. (430219);

Seeds of cumin (90930);

Manifold business forms & interleaved carbon sets, of paper/paperboard (482040); Beans n.e.s., dried/shelled, including seed (71339);

Tanned/crust hides & skins of bovine or equine animals, in the dry state (crust), without hair on, (other than full grains, unsplit; grain splits) (410449);

Leather further prepared after tanning or crusting, including parchment-dressed leather, of bovine (including buffalo) or equine animals, without hair on, whole hides and skins, full grain, unsplit (410711); Tanned/crust hides & skins of bovine or equine animals, in the wet state (including wet blue), without hair on, full grains, unsplit; grain splits (410411);

Lead ores & concentrates (260700); Precious and semi precious stones (excl. Diamonds, rubies, sapphires and emeralds), worked other than simply sawn or roughly shaped, not strung, mounted or set (710399);

Marble & travertine, crude/roughly (251511);

Leather further prepared after tanning incl. Parchment-dressed, of bovine (including buffalo) or equine animals, no hair, and whole hides and skins, and sides, (full grains, unsplit and grain splits) (410799); Gum Arabic (130120);

Seeds of coriander (90920);

Beans of the species vigna mungo (L.) Hepper or vigna radiata (L.) Wilczek, shelled, whether or not skinned or split, dried (71331);

Chickpeas (garbanzos), dried, shelled, whether or not skinned or split (71320); Leather further prepared after tanning or crusting, including parchment-dressed, of bovine (including buffalo) or equine animals, without hair on, split or not, (other than grain splits and full grains, unsplit) (410719);

Linseed, whether/not broken (120400); Sunflower seeds, whether/not broken (120600) Other parts & accessories for the motor vehicles (870899);

Non-refractory surfacing preparations, mastics n.e.s. (321490);

Other parts and accessories for printing machinery excl. 844391 (844399);

Marble and travertine, merely cut, by sawing or otherwise, into blocks or slabs of a rectangular (including square) shape, having a specific gravity of 2.5 or more (251512);

Synthetic/reconstructed precious/semi precious stones, worked or graded or not, (but not strung, mounted or set), temporarily strung for transport convenience (710490);

Parts of the air-conditioning machines, with motor driven fan and elements for temperature control (841590);

Knittedd/crocheted fabrics, n.e.s., of synthetic fibres, dyed (600632);

Pistachios (80250);

Fruit, edible; guavas, mangoes and mangosteens, fresh or dried (80450);

Floor coverings of plastics (other than polymers of vinyl chloride), whether or not self-adhesive, in rolls or in the form of tiles (391890);

Fixed electrical resistors for a power handling capacity > 20 W (excl. Heating resistors) (853329);

Dried leguminous vegetables, n.e.c., shelled, whether or not skinned or split (71390);

Articles of paper, cellulose wadding or fibres n.e.s. (481890);

Rodenticides & other biocides, n.e.s. (380899); Other printers, copying machines, single-function printing, not capable of connecting to an automatic data processing machine or to a network (844339); Stationery n.e.s., made of paper or paperboard (482090);

Cinnamon and cinnamon-tree flowers, other than cinnamon (Cinnamonum zeylanicum Blume), neither crushed nor ground (90619);

Newspapers, journals & periodicals, whether or not illustrated or containing advertising material, appearing less frequently than four times a week (490290);

Articles of vulcanised rubber (other than hard rubber), moulded rubber mats and mats of non-rectangular shape made by cutting from the piece, of cellular rubber (401610);

Aluminium waste & scrap (760200);

Newspapers, journals & periodicals, whether or not illustrated or containing advertising material, appearing at least four times a week (490210); Plates, sheets, film, foil & strip, of polymers of propylene, non-cellular and not reinforced, laminated, supported or similarly combined with other materials (392020);

Women's/girls' overcoats, raincoats, car-coats, capes, cloaks and similar articles, of cotton, not knittedd or crocheted (620212);

Machines and appliances for testing the hardness, strength, compressibility, elasticity or other mechanical properties of materials other than metals (902480);

Revolution counters, production counters, taximeters, mileometers, pedometers and the like (902910);

Women's/girls', trousers, bib & brace overalls, of cotton (not knittedd or crocheted) (620462); Manifold business forms & interleaved carbon sets, of paper/paperboard (482040);

Women's/girls' jackets & blazers (excl.

knittedd/crocheted), of textile materials (620439); Beans n.e.s., dried/shelled, including seed (71339); Glass beads, imitation pearls, imitation precious or semi-precious stones and similar glass smallwares (701810);

Rolled/flaked grains of oats (110412);

Cinematographic film, exposed & developed, whether or not incorporating sound track or consisting only of sound track, of a width of 35cm or more (370610);

Precious and semi precious stones (excl. Diamonds, rubies, sapphires and emeralds), worked other than simply sawn or roughly shaped, not strung, mounted or set (710399);

Zinc waste & scrap (790200);

Women's/girls', trousers, bib & brace overalls, of textile materials (other than wool, fine animal hair, cotton or synthetic fibres), (not knittedd or crocheted) (620469);

Women's/girls' dresses, of cotton (not knittedd or crocheted) (620442);

Women's/girls' skirts & divided skirts, of cotton (not knittedd or crocheted) (620452);

Unwrought lead, refined (780110);

Seeds of coriander (90920);

Pepper (genus Piper), neither crushed nor ground (90411);

Raw hides & skins (fresh,/salted, dried, limed, pickled or otherwise preserved), but not tanned, parchment-dressed or further prepared, whether or not dehaired or split (410390);

Women's/girls' skirts & divided skirts, of synthetic fibres (not knittedd or crocheted) (620453);

Tanned/crust skins of sheep/lambs, without wool on, whether or not split, but not further prepared, in the wet state (including wet-blue) (410510); Maize (corn) flour (110220);

Leather further prepared after tanning incl. Parchment-dressed, of bovine (including buffalo) or equine animals, no hair, and whole hides and skins, and sides, (full grains, unsplit and grain splits) (410799);

Tanned/crust hides & skins of bovine or equine animals, in the wet state (including wet blue), without hair on, excluding full grains, unsplit; grain splits (410419);

Women's/girls' jackets & blazers, of cotton (not knittedd or crocheted) (620432);

Cotton, not carded/combed (520100); Unwrought lead other than refined, not containing by weight antimony as the principal other element (780199): Lead ores & concentrates (260700); Tanned/crust hides & skins of bovine or equine animals, in the wet state (including wet blue), without hair on, full grains, unsplit; grain splits (410411);Tanned/crust hides & skins of goats or kids, without wool or hair on, whether or not split, but not further prepared, in the wet state (including wet blue) (410621); Cardamoms (90830): Tanned/crust hides & skins, n.e.s., of animals other than equine, ovine, bovine, goats or kids, swine and reptiles; without wool or hair on, whether or not split, but not further prepared, in the dry state (crust) (410692); Women's/girls' blouses, shirts & shirt blouses, of textile materials (other than cotton or man-made fibres), knittedd or crocheted (610690); Precious (other than diamonds) and semi-precious stones, unworked or simply sawn or roughly shaped, not strung, mounted or set (710310); Leather further prepared after tanning or crusting, including parchment-dressed, of bovine (including buffalo) or equine animals, without hair on, split or not, (other than grain splits and full grains, unsplit) (410719); Tanned/crust hides & skins of bovine or equine animals, in the dry state (crust), without hair on, (other than full grains, unsplit; grain splits) (410449); Beans of the species vigna mungo (L.) Hepper or vigna radiata (L.) Wilczek, shelled, whether or not skinned or split, dried (71331); Raw skins of sheep/lambs (fresh/salted/dried/limed/otherwise, preserved other than by pickling, (but not tanned, parchmentdressed or further preserved), without wool on (410229);Turmeric (curcuma) (91030); Chickpeas (garbanzos), dried, shelled, whether or not skinned or split (71320); Spices, n.e.s. In Ch. 9 (91099) Gambia Wooden furniture (other than for office, kitchen or Instruments and apparatus for measuring or bedroom use), nesoi (940360); checking the flow or level of liquids Fixed carbon resistors, composition or film types (902610); (excl. Heating resistors) (853310); Fixed electrical capacitors (excl. Tantalum, Aluminium waste & scrap (760200); aluminium electrolytic, ceramic, paper, Waste & scrap of alloy steel other than stainless plastic and power capacitors) (853229); steel (720429) Parts of Machines & mechanical appliances of those having individual functions (847990); Precious (other than diamonds) and semiprecious stones, unworked or simply sawn or roughly shaped, not strung, mounted or set (710310); Aluminium waste & scrap (760200);

		Ferrous waste & scrap, n.e.s. (720449)
Ghana	Fixed electrical capacitors (excl. Tantalum, aluminium electrolytic, ceramic, paper, plastic and power capacitors) (853229); Parts of electrical ignition or starting equipment, generators, etc. Of heading 8511, n.e.s. (851190); Other apparatus for transmission reception of voice, images other data, incl. Apparatus for communication in a wired wireless network (such as a local wide area network) (851769); Sunglasses (900410); Limestone flux; limestone and other calcareous stone, of a kind used for the manufacture of lime or cement (252100); Dentists', barbers' or similar chairs having rotating as well as both reclining and elevating movements, and parts thereof (940210); Other parts & accessories for the motor vehicles (870899); Liquid dielectric transformers, having a power handling capacity > 650 kva but <= 10000 kva (850422); Curtains (incl. Drapes) & interior blinds, curtain or bed valances; of textile materials other than cotton or synthetic fibres, not knittedd or crocheted (630399); Fixed electrical capacitors, ceramic dielectric, multilayer (excl. Power capacitors) (853224); Air compressors mounted on a wheeled chassis for towing (841440); Aluminium oxide (excl. Artificial corundum) (281820); Electrical apparatus like switches n.e.c. In heading no. 8536, for a voltage not exceeding 1000 volts (853650); Ethylene polymers; in primary forms, polyethylene having a specific gravity of less than 0.94 (390110); Sound signalling equipment; electrical, used on cycles or motor vehicles (851230); Electric space heating and soil heating apparatus (excl. Storage heating radiators) (851629); Wool, not carded/combed, greasy, (including fleece-washed wool), not carded or combed (510119); Other flat-rolled products of other steel, flat-rolled, width 600mm or more, hot-rolled, not in coils (722540); Parts of mechanical appliances (parts of machines projecting, dispersing or spraying liquids or powders, whether or not hand-operated) (842490); Millstones, grindstones, grindstones, grindstones or of ceramics (680422);	Ferrous waste & scrap, n.e.s. (720449) Poly(vinyl chloride), not mixed with any other substances, in primary forms (390410); Parts of Machinery for working rubber or plastics (847790); Diamonds; non-industrial, unworked or simply sawn, cleaved or bruted, but not mounted or set (710231); Rolls, sheets & dials, printed for self-recording apparatus (482340); Hard rubber (e.g., ebonite) in all forms, including waste and scrap, and articles of hard rubber (401700); Carbonates, n.e.s. (283699); Recovered (waste & scrap) paper/paperboard, n.e.s., and of unsorted waste and scrap (470790); Casks, barrels, vats, tubs & other coopers' products and parts thereof, of wood, including staves (441600); Ash & residues (excl. From the manufacture of iron or steel), containing mainly zinc, hard zinc spelter (262019); Waste & scrap of stainless steel (720421); Cotton, not carded/combed (520100); Vegetable saps and extracts; n.e.s. In item no. 1302.1 (130219); Ferrous waste & scrap, n.e.s. (720449); Coniferous wood, in the rough, whether/not stripped of bark or sapwood, or roughly squared, untreated (440320); Gum Arabic (130120); Waste & scrap of cast iron (720410); Waste & scrap of tinned iron/steel (720430); Edible mixtures or preparations of animal or vegetable fats or oils or of fractions of different fats or oils of this chapter, other than edible fats or oils of heading no. 1516 (151790); Roundabouts, swings, shooting galleries, other fairground amusements; travelling theatres (950890); Sesamum seeds, whether/not broken (120740); Unwrought lead other than refined, not containing by weight antimony as the principal other element (780199); Aluminium waste & scrap (760200); Lead waste & scrap (780200)

Tricycles, scooters, pedal cars and similar wheeled toys; dolls' carriages; dolls; other toys; reduced-size (scale) models and similar recreational models, working or not; puzzles of all kinds (950300);

Video recording or reproducing apparatus, whether or not incorporating a video tuner (excl. Magnetic tape-type and video camera recorders) (852190); Air conditioning machines comprising a motor-driven fan and elements for changing the temperature and humidity, window or wall types, self-contained or split-system (841510); Textured yarn other than sewing thread, of polyesters, not for retail sale (540233); Catheters, cannulae & the like (901839); Parts and accessories nesoi of bicycles etc. (871499);

Helical springs of iron/steel (732020);

Valves for oleohydraulic/pneumatic transmissions (848120);

Glass beads, imitation pearls, imitation precious or semi-precious stones and similar glass smallwares (701810);

Plates, sheets, film, foil & strip, other than cellular (392190);

Brakes & servo-brakes; parts thereof (vehicles) (870830);

Paper and paperboard; coated with kaolin or other inorganic substances, for printing/writing/graphics, having 10% or less by weight of total fibres got by mechanical/chemi-mechanical process, sides exceeding 435mm and 297mm, unfolded, sheets (481019);

Other Parts suitable for use solely or principally with the apparatus of headings 8525 to 8528 (reception and transmission apparatus, monitors, etc.) (852990);

Coal; (other than anthracite and bituminous), whether or not pulverised but not agglomerated (270119);

Printed matter, n.e.s. (491199);

Saw blades, n.e.s. (820299);

Articles of iron/steel, n.e.s., forged/stamped, but not further worked (732619);

Fittings for furniture/coachwork/the like, of plastic (392630);

Polymers of ethylene, in primary forms, n.e.s. (390190);

Lubricating preparations (excl. Those for the treatment of textile and similar materials), not containing petroleum oils or oils obtained from bituminous minerals (340399);

Woven fabrics containing 85%/more by weight of silk or of silk waste other than noil silk (500720); Moulds for metal or metal carbides, other than injection or compression types (848049); Gummed/adhesive paper,paperboard, pressuresensitive (481141);

Machines for the reception, conversion and transmission regeneration of voice, images other

data, incl. Switching and routing apparatus (851762);

Parts & accessories n.e.s. In Ch.90 (903300); Stranded wire, ropes & cables of iron/steel, not electrically insulated (731210);

Fans, other than

table/floor/wall/window/ceiling/roof (841459); Mountings, fittings & similar articles, for motor vehicles, of base metal (830230);

Parts of the air-conditioning machines, with motor driven fan and elements for temperature control (841590);

Refrigerating/freezing equip. n.e.s (841869); Beans of the species vigna mungo (L.) Hepper or vigna radiata (L.) Wilczek, shelled, whether or not skinned or split, dried (71331);

Textile fabrics; impregnated, coated, covered or laminated with plastics, (excluding polyvinyl chloride, polyurethane, polyamides polyesters) (590390);

Used/new rags, scrap twine, cordage, rope and cables and worn out articles of twine, cordage, rope or cables, of textile materials; other than sorted (631090);

Parts of other agricultural, horticultural or forestry use n.e.s. (843699);

Parts and accessories for machine tools, for laser operation, metalworking machining centers, lathes and drilling machines, etc., others (846693);

Pigments & preparations based on titanium dioxide, containing 80% or more by weight of titanium dioxide calculated on the dry weight (320611);

Sewing machines (excl. H-hold. Type), automatic units (845221);

Polypropylene, in primary forms (390210); Clutches & parts thereof for the motor vehicles (870893);

Plates, sheets, film, foil & strip, of plastics n.e.s., non-cellular and not reinforced, laminated, supported or similarly combined with other materials (392099);

Refractory bricks, blocks, tiles & similar refractory ceramic constructional goods, n.e.s. (690290); Knitting Machines, Others, And Machines For Making Gimped Yarn, Tulle, Lace, Embroidery, Trimmings, Braid Or Net And Machines For Tufting (844790);

Electric motors of an output not exceeding 37.5W (850110);

Tableware & kitchenware, of porcelain or china (691110):

Wooden furniture of a kind used in bedroom (940350);

Sewing machines, not household or automatic unit type (845229);

Parts of the industrial/laboratory furnaces and ovens (including incinerators) (841790);

Unglazed ceramic tiles, cubes and similar articles, whether or not rectangular, the largest surface area

of which is capable of being enclosed in a square with side<7 cm (690710);

Silicones, in primary forms (391000);

Paper, paperboard, cellulose wadding or webs of cellulose fibres (482390);

Apples, fresh (80810);

Sanitary towels & tampons, napkins (481840); Bidets, lavatory pans, flushing cisterns and similar

sanitary ware, n.e.s. (392290);

Glass fibres (incl. Glass wool) & articles thereof, n.e.s. (701990);

Multiple loudspeakers, mounted in the same enclosure (851822);

Unglazed ceramic flags & paving/hearth/wall tiles (690790);

Flat-rolled products of iron/non-alloy steel, width 600mm or more, electrolytically plated or coated with zinc (721030);

Electrical lighting or visual signalling equipment for motor vehicles (excl. Lamps of heading 8539) (851220);

Dried leguminous vegetables, n.e.c., shelled, whether or not skinned or split (71390);

Lamps and light fittings; parts thereof, of materials other than glass or plastics (940599);

Furniture of cane, osier, or similar materials (other than bamboo or rattan) (940389);

Drive-axles with differential, whether or not provided with other transmission components, and non-driving axles; parts thereof (870850);

Parts & accessories of motorcycles (mopeds), other than saddles (871419);

Thermometers and pyrometers; (other than liquid filled, for direct reading), not combined with other instruments (902519);

Isocyanates (292910);

Boring machines n.e.s. (845940);

Spark-ignition reciprocating/rotary internal combustion engines, outboard motors for marine propulsion, (840721);

Tanned/crust hides & skins of bovine or equine animals, in the dry state (crust), without hair on, (other than full grains, unsplit; grain splits) (410449);

Amino-naphthols &other amino-phenols; their ethers and esters, (other than those containing more than one kind of oxygen function), n.e.s. (292229);

Articles of vulcanised rubber other than hard rubber, n.e.s. (401699);

Artificial waxes & prepared waxes (other than of chemically modified lignite or polyethylene glycol) (340490);

Warp knit fabrics. (including those made on galloon knitting machines), n.e.s., of synthetic fibres, dved (600532);

Marble, travertine and alabaster; simply cut or sawn, with a flat or even surface (680221); Lip make-up preparations (330410);

Pneumatic elevators & conveyors (842820);

Hydraulic power engines & motors other than linear acting (cylinders) (841229);

Vegetable oils; soya-bean oil and its fractions, crude, whether or not degummed, not chemically modified (150710);

Diamonds; non-industrial, unworked or simply sawn, cleaved or bruted, but not mounted or set (710231);

Nucleic acids & their salts, whether or not chemically defined, n.e.s. (293499);

Pistachios (80250);

Synthetic staple fibres, not carded, combed or otherwise processed for spinning (550320); Bars, rods & profiles, of refined copper (740710); Polyethylene having a sp.gr. Of 0.94 or more (390120):

Anti-oxidising preparations and other compound stabilisers; for rubber or plastics, n.e.s. Or included (381230);

Hollow drill bars & rods of alloy/non-alloy steel (722880);

Paper, paperboard, cellulose wadding and webs of soft cellulose fibres; coated, impregnated, covered, surface-decorated or surface-coloured, n.e.s. (481190);

Refrigerating or freezing equipment; parts thereof, other than furniture (841899);

Vehicles with only compression-ignition internal combustion piston engine (diesel or semi-diesel), cylinder capacity over 1500 but not over 2500cc (870332);

Internal combustion piston engines (diesel or semidiesel engines), of a kind used for other than marine propulsion or the vehicles of chapter 87 (840890):

Furniture for medical, surgical, veterinary use (e.g. Operating tables, examination tables, hospital beds with mechanical fittings) and parts thereof (940290);

Umbrella frames, incl. Frames mounted on shafts (sticks) (660320);

Woven fabrics, containing 85% or more by weight of filaments of nylon or other polyamides, dyed (540742);

Anchors, grapnels & parts thereof, of iron/steel (731600);

Wheel rims & spokes for bicycles (871492); Paving blocks, slabs, bricks and other articles of pressed or moulded glass (701690);

Slide fasteners, other than those fitted with chain scoops of base metal (960719);

Other made-up textile articles, (including dress patterns), n.e.s. (630790);

Welded angles, shapes & sections of Iron or steel (730120);

Weighing machine weights of all kinds; parts of weighing machinery (842390);

Starches (excl. Of 1108.11-1108.14) (110819);

Glass in rods, unworked (700220);

Calcium carbonate (283650);

Inflatable balls (950662);

Natural/artificial abrasive powder/grain, on a base of paper or paperboard only, whether or not cut to shape or sewn or otherwise made up (680520); Bedspreads other than knittedd/crocheted (630419);

Methanol (methyl alcohol) (290511);

Articles of plaster/of compositions based on plaster; n.e.s. (680990);

Newsprint, in rolls/sheets (480100);

Knives & cutting blades, for machines or mechanical appliances, n.e.s. (820890);

Synthetic organic products of a kind used as fluorescent brightening agents (320420);

Perfumes & toilet waters (330300);

Handbags, whether/not with shoulder strap (including those without handle), with outer surface of leather, of composition leather or of patent leather (420221);

Hinges of base metal (830210);

Drilling/threading/tapping tools (820510);

T-shirts, singlets & other vests, of cotton, knittedd or crocheted (610910);

Particle board of wood, whether/not agglomerated with resins or other organic binding substances (441011);

Artificial flowers, foliage & fruit, artificial, and parts thereof; articles made of artificial flowers, foliage or fruit, of materials other than plastics (670290);

Tea, black; (fermented) and partly fermented tea, in immediate packings of a content exceeding 3kg (90240):

Hairpins, curling pins, curling grips, hair curlers and the like and parts thereof (961590);

Refractory ceramic goods, containing by weight more than 50% of graphite or other forms of carbon or of a mixture of these products, excluding those of siliceous fossil meals or similar earths (690310);

Washing/bleaching/dyeing machines (845140); Essential oils, n.e.s., (terpeneless or not), including concretes and absolutes (330129);

Wire of iron/non-alloy steel, n.e.s (721790); Flat-rolled products of iron/non-alloy steel, in coils, width 600mm or more, cold-rolled, of a thickness of 0.5mm or more but not exceeding 1mm (720917);

AC motors, multi-phase, of an output > 750 W but <= 75 kw (850152);

Umbrellas & sun umbrellas (excl. Garden and similar umbrellas) (660199):

Festive, carnival or other entertainment articles including novelty jokes and conjuring tricks; other than Christmas festivity articles (950590);

Polymers of propylene/of other olefin polymers, n.e.s. (390290);

Woven fabrics, containing 85% or more by weight of textured polyester filaments, dyed (540752);

Threaded Self-tapping screws of iron/steel (731814);

Portland cement (excl. White cement), whether or not artificially coloured (252329);

Kaolin and other kaolinic clays; whether or not calcined (250700);

Bran, sharps and other residues; of other cereals, whether or not in the form of pellets, derived from the sifting, milling or other workings thereof (230240):

Paper & paperboard, coated/impregnated or covered with plastics (excluding adhesives), other than bleached and weighing more than 150g/m2 (481159):

Articulated link chain (other than roller), of iron or steel (731512);

Builders' ware of plastics, n.e.s. (392590);

Uppers & parts thereof (excl. Stiffeners) (640610); Agarbatti & other odoriferous preparations, which operate by burning (including those used during religious rites) (330741);

Hat-shapes, plaited or made by assembling strips of any material, neither blocked to shape, nor with made brims, nor lined, nor trimmed (650200); Sewing thread of artificial filaments, whether or not put up for retail sale (540120);

Wooden doors & their frames & thresholds (441820);

Waste & scrap of stainless steel (720421); Articles of gut (other than silk-worm gut), of goldbeater's skin, of bladders or of tendons (420600);

Rear-view mirrors for vehicles (700910); Candles, tapers&the like (340600); Multi-station transfer machines for working metal (845730);

Tubes, pipes & hoses, of vulcanised rubber, reinforced or otherwise combined only with metal, with fittings (400922);

Composite paper & paperboard (made by sticking layers together with an adhesive), not surface-coated or impregnated, whether or not internally reinforced, in rolls or sheets (480700);

Mattresses of other materials, not cellular rubber or plastics (940429);

Boards, sheets, panels, tiles & similar articles, (other than faced or reinforced with paper or paperboard only), not ornamented (680919); Aluminium nails, tacks, staples, screws, bolts, nuts, screw hooks, rivets, cotters, cotter-pins, washers and similar articles (761610); Other n.e.s., flat-rolled, width 600mm or more, n.e.s., (other than plated or coated with zinc) (722599).

Recovered (waste & scrap) paper/paperboard, n.e.s., and of unsorted waste and scrap (470790); Auminium plates, sheets & strip, rectangular (including square), of thickness exceeding 0.2mm, (not alloyed) (760611);

Peptones&their derivatives; other protein substances and their derivatives n.e.s. Or included, hide powder, whether or not chromed (350400); Cyclanes, cyclenes & cycloterpenes (290219); Mechano-therapy appliances, massage apparatus and psychological aptitude-testing apparatus (901910);

Cans to be closed by soldering/crimping, for any material (excluding compressed or liquefied gas), less than 50l capacity, of iron or steel (731021); Cyclanic/cyclenic/cycloterpenic ketoneswithout other oxygen function, n.e.s. (291429); Reaction initiators, reaction accelerators and catalytic preparations, unsupported, n.e.s. Or included (381590);

Harvesting/threshing machinery, n.e.s. (843359); Articles of asphalt/similar material in rolls (680710);

Glazed ceramic tiles, cubes and similar articles, whether or not rectangular or on a backing, with the largest surface area<7cm square (690810); Woven fabrics of cotton, containing 85% or more by weight of cotton, printed, of weaves n.e.s., weighing more than 200g/m2 (520959); Wood (including strips & friezes for parquet flooring), continuously shaped along any edges, ends or faces, whether or not planed, sanded or end-jointed (440910);

Ferrous turnings, shavings, chips, milling waste,

sawdust, fillings, trimmings and stampings, whether or not in bundles (720441); Beech (Fagus spp.), sawn or chipped lengthwise, sliced or peeled, whether or not planed, sanded or finger-jointed, thicker than 6mm (440792); Dyed Woven fabrics of synthetic, polyester staple fibres (less than 85% by weight), mixed mainly or solely with cotton, not exceeding 170g/m2

Flat-rolled products of iron/non-alloy steel, in coils, without patterns in relief, of a width 600mm or more, hot-rolled, of a thickness exceeding 10mm (720836);

(551323);

Table/kitchen/other h-hold. Articles and their parts, of Iron or steel, n.e.s. (732399); Machinery for the extraction or preparation of animal or fixed vegetable fats or oils (847920); Iron or steel (excluding cast iron or stainless steel); other seamless, casing and tubing, of a kind used in drilling for oil or gas (730429);

Shaving, hair, nail, eyelash and other toilet brushes for use on the person, including brushes as parts of appliances (960329);

Spools, cops, bobbins & similar supports, for the conveyance or packing of goods (392340); Quilted textile products in the piece, composed of one or more layers of textile materials assembled with padding by stitching or otherwise (excluding embroidery in strips or motifs) (581100); Adhesive dressings & other articles having an adhesive layer, packed for retail sale for medical, surgical, dental or veterinary purposes (300510);

Vulcanised rubber thread & cord (400700);

Cotton, not carded/combed (520100);

Waste & scrap of tinned iron/steel (720430);

Combined refrigerator-freezers, fitted with separate external doors, electric or other (841810);

Zinc waste & scrap (790200); Sodium sulphates other than disodium sulphate

Sodium sulphates other than disodium sulphate (283319);

Golf equipment, other than clubs and balls (950639);

Particle board & similar board of ligneous material, n.e.s. (441090);

Wood, in the rough, n.e.s., whether or not stripped of bark or sapwood, or roughly squared, untreated, (440399):

Gummed/adhesive paper,paperboard (excluding self-adhesive), n.e.s. (481149);

Blankets (excl. Electric) & travelling rugs, n.e.s. (630190);

Plaits&similar products of plaiting (460192);

Planes, chisels, gouges & similar cutting tools for working wood (820530);

Machinery for wringing, dressing, finishing, coating or impregnating textile yarns, fabrics or made up textile articles; for applying paste to base fabric used in manufacture of floor coverings (845180);

Machines for used in the milling industry or for the working of cereals or dried leguminous vegetables, other than farm-type machinery (843780);

Other glassware of a kind used for table or kitchen purposes (not drinking glasses or of glass-ceramics) (701349);

Copper-tin base alloys (bronze), unwrought (740322);

Oak (Quercus spp.), in the rough, whether or not stripped of bark or sapwood, or roughly squared, untreated (440391);

Other casing, tubing, of a kind used in drilling for oil or gas, seamless, of stainless steel (730424); Copper waste & scrap (740400);

Natural steatite, whether/not roughly trimmed or merely cut, by sawing or otherwise, into blocks or slabs of a rectangular (including square) shape; talc (252620);

Saturated acyclic hydrocarbons (290110); Bamboos (140110);

Copper ores & concentrates (260300);

Acetone (291411);

Phosphoric acid and polyphosphoric acids (280920);

Unwrought lead, refined (780110);

Frozen shrimps & prawns, whether/not in shell, , whether/not cooked by steaming or by boiling in water (30613);

Paper & paperboard, not containing fibre greater than 10%, weight 40-150g/m2, in sheets 435mm or less by 297mm or less (unfolded) (480256); Electrical signalling, safety or traffic control equipment (excl. That for railways or tramways and mechanical or electromechanical equipment of heading 8608) (853080);

Wood sawn/chipped length wise, sliced or peeled, whether or not planed, sanded or end-jointed, of a thickness exceeding 6mm (440710);

Apparatus based on the use of alpha, beta or gamma radiations, including radiography or radiotherapy apparatus; (for other than medical, surgical, dental or veterinary uses) (902229); Waste & scrap of alloy steel other than stainless steel (720429):

Household and toilet articles, n.e.s., in of porcelain or china (691190);

Ferrous waste & scrap, n.e.s. (720449)

Guinea

Diamonds, non-industrial, (other than unworked or simply sawn, cleaved or bruted), but not mounted or set (710239);

Textile fabrics; impregnated, coated, covered or laminated with plastics, (excluding polyvinyl chloride, polyurethane, polyamides polyesters) (590390);

Waste & scrap of stainless steel (720421); Other drinking glasses, other than of lead crystal or glass-ceramics (701337);

Parts of the converters, ladles, ingot moulds and casting machines, of a kind used in metallurgy or in metal foundries (845490);

Knitting Machines, Others, And Machines For Making Gimped Yarn, Tulle, Lace, Embroidery, Trimmings, Braid Or Net And Machines For Tufting (844790);

Palm oil and its fractions, crude, not chemically modified (151110);

Essential oils, n.e.s., (terpeneless or not), including concretes and absolutes (330129);

Jacks & hoists, (other than hydraulic), of a kind used for raising vehicles (842549);

Ethylene polymers; in primary forms, polyethylene having a specific gravity of less than 0.94 (390110);

Sewing machines, not household or automatic unit type (845229);

Beans of the species vigna mungo (L.) Hepper or vigna radiata (L.) Wilczek, shelled, whether or not skinned or split, dried (71331);

Electric table, desk, bedside/floor (940520); Paints & varnishes (incl. Enamels & lacquers), based on polymers n.e.s., dispersed or dissolved in a non-aqueous medium (320890);

Carpets & other textile floor coverings, woven, (not tufted or flocked), of wool or fine animal hair, of pile construction, made up, n.e.s. (570241); Beauty/make-up preparations & preparations for skin care, n.e.s. (330499);

Electronic integrated circuits; processors and controllers, whether or not combined with memories, converters, logic circuits, amplifiers, clock and timing circuits, or other circuits (854231);

Dental instruments and appliances; other than dental drill engines (901849); Pumps and compressors; parts, of air or vacuum pumps, air or other gas compressors and fans, ventilating or recycling hoods incorporating a fan (841490);

Folding cartons, boxes & cases, of non-corrugated paper or paperboard (481920); Monofilament of which any cross-sectional dimension exceeds 1mm, rods, sticks and profile shapes, whether or not surface-worked but not otherwise worked (391620); Precious (other than diamonds) and semi-precious stones, unworked or simply sawn or roughly shaped, not strung, mounted or set (710310);

Medicaments containing hormones/adrenal cortex hormones/antibiotics, for therapeutic or prophylactic uses, packaged for retail sale (300439);

Petroleum oils and oils obtained from bituminous minerals, crude (270900); Copper waste & scrap (740400); Other electrical parts of machinery (854890);

Waste oils; of petroleum or obtained from bituminous minerals, not crude and preparations n.e.c., weight 70% or preparations of the same, not containing polychlorinated biphenyls (pcbs), polychorinated terphenyls (pcts) or polybrominated biphenyls (pbbs) (271099); Raw hides & skins of bovine (including buffalo) and equine animals, fresh, salted or preserved, but not tanned, parchment

Pigments & preparations based on titanium dioxide, containing 80% or more by weight of titanium dioxide calculated on the dry weight (320611);

Outer soles & heels, of rubber/plastics (640620); Tableware & kitchenware, of wood (441900); Dried leguminous vegetables, n.e.c., shelled, whether or not skinned or split (71390); Air-conditioning machines, containing a motor driven fan, other than window or wall types, not incorporating a refrigerating unit (841583); Mechano-therapy appliances, massage apparatus and psychological aptitude-testing apparatus (901910);

Mixtures of odoriferous substance and mixtures, used as raw materials in industries other than the food or drink industries (330290);

Granite, crude/roughly trimmed (251611); Polishes & creams, scouring pastes, n.e.s. (340590);

Portland cement (excl. White cement), whether or not artificially coloured (252329);

Apples, fresh (80810);

Articles & equipment for sports, for outdoor games and recreation n.e.c. (950699);
Men's/boys' trousers, bib & brace overalls, of cotton (not knittedd or crocheted) (620342);
Furniture of cane, osier, or similar materials (other

furniture of cane, osier, or similar materials (other than bamboo or rattan) (940389);

Acrylic polymers (other than polymethyl methacrylate), in primary forms (390690); Ball bearings (848210);

Precious and semi-precious stones (excl. Diamonds, rubies, sapphires and emeralds), worked other than simply sawn or roughly shaped, not strung, mounted or set (710399);

Surface-active preparations, Washing and cleaning preparations, whether or not containing soap, put up for retail sale (340220);

Medium density of fibreboard of wood, of a thickness exceeding 9 mm (441114);

Unwrought lead other than refined, not containing by weight antimony as the principal other element (780199);

Artificial flowers, foliage & fruit, artificial, parts thereof and their plastic articles (670210); Parts of Machines & mechanical appliances of those having individual functions (847990); Metal furniture of a kind used in office (940310); Plaited bands, slings & the like, of iron/steel, not electrically insulated (731290);

Threaded Screw hooks & screw rings, of iron/steel (731813);

Wood sawn/chipped length wise, sliced or peeled, whether or not planed, sanded or end-jointed, of a thickness exceeding 6mm (440710);

Butadiene rubber (BR), in primary forms or in plates, sheets or strip (400220);

Personal deodorants & antiperspirants (330720); Aluminium waste & scrap (760200); dressed or further prepared, whether or not dehaired or split (410190);
Aluminium waste & scrap (760200);
Waste & scrap of cast iron (720410);
Diamonds, unsorted (710210);
Ferrous waste & scrap, n.e.s. (720449);
Gold (incl. Gold plated with platinum, unwrought, for non-monetary purposes (excl. Gold in powder form) (710812);
Cashew nuts, in shell (80131);
Niobium/tantalum/vanadium ores &

concentrates (261590)

Precious (other than diamonds) and semi-precious stones, unworked or simply sawn or roughly shaped, not strung, mounted or set (710310); Petroleum oils and oils obtained from bituminous minerals, crude (270900); Flat-rolled products of iron/non-alloy steel, width 600mm or more, plated or coated with tin, thickness of less than 0.5mm (721012); Topical wood, n.e.s., in the rough, whether or not stripped of bark or sapwood, or roughly squared, untreated (440349); Waste & scrap of tinned iron/steel (720430); Vegetable fats and oils and their fractions; partly or wholly hydrogenated, inter-esterified, reesterified or elaidinised, whether or not refined, but not further prepared (151620) Precious (other than diamonds) and semi-precious Instruments, appliances and machines; for Guinea-Bissau stones, unworked or simply sawn or roughly measuring or checking n.e.c. (903180); shaped, not strung, mounted or set (710310); Ferrous waste & scrap, n.e.s. (720449); Waste and scrap of primary cells, primary Tableware & kitchenware, of plastic (392410); Other machines and apparatus for electrical batteries and electric accumulators; spent machines and apparatus, other than than machines primary cells, spent primary batteries and and apparatus for electroplating electrolysis spent electric accumulators (854810); Aluminium ores & concentrates (260600) electrophoresis signal generators particle accelerators. (854370); Non-threaded washers, of iron/steel, excluding spring and lock (731822) Kenya Heterocyclic comps. Containing a pyrimidine ring Parts & accessories of the machines or (whether or not hydrogenated) or piperazine ring auxiliary machinery of heading no. 8444 in the structure, nucleic acids and their salts. (844820); excluding malonylurea (293359): Saw blades, n.e.s. (820299): Copper ores & concentrates (260300); Spherical roller bearings (848230); Nucleic acids&their salts, whether or not Parts of railway/tramway rolling stock n.e.c. chemically defined, n.e.s. (293499); (860799): Parts of the slide fasteners (960720); Parts and accessories for instruments, Parts and accessories nesoi of bicycles etc. appliances and machines measuring or (871499); checking (903190); Apparatus based on the use of alpha, beta or Drive-axles with differential, whether or not provided with other transmission components, and gamma radiations, including radiography or non-driving axles; parts thereof (870850); radiotherapy apparatus; (for other than Salicylic acid & its salts (291821); medical, surgical, dental or veterinary uses) Other parts and accessories of looms, auxiliary (902229): machinery, n.e.s. (844849); Injection-moulding machines (847710); Aromatic monoamines & their derivatives; salts Propylene copolymers, in primary forms thereof; n.e.s. (292149); (390230);Reaction initiators, reaction accelerators and Hearing aids (excl. Parts & accessories) (902140); catalytic preparations, unsupported, n.e.s. Or included (381590); Microtomes and parts & accessories thereof Heterocyclic comps. With nitrogen hetero-atom (902790);only, n.e.s. (293399); Wood, in the rough, n.e.s., whether or not Kraft paper & paperboard, n.e.s., coated with stripped of bark or sapwood, or roughly kaolin or other inorganic substances only, for nonsquared, untreated, (440399); graphic purposes, in rolls or sheets (481039); Parts of the steam turbines & other vapour Parts of the seats of 94.01 (seats in vehicles, turbines (840690); aircrafts, convertible to bed, etc.) (940190); Toothed wheels, chain sprockets and other Parts of the sewing machines n.e.s. (845290); transmission elements presented separately; Headphones, earphones and combined parts of transmission components (848390); microphone/speaker sets (excl. Telephone sets, Poly(vinyl chloride), non-plasticised, in primary forms, mixed with other substances hearing aids and helmets with built-in headphones, (390421);

whether or not incorporating a microphone) (851830);

Millstones, grindstones, grinding wheels and the like; of other agglomerated abrasives or of ceramics (680422);

Apparatus based on use of x-rays and similar; parts and accessories (x-ray generators, tubes, high tension generators, control panels and desks, screens, examination or treatment tables, chairs and like (902290);

Clutches and shaft couplings (including universal joints) (848360);

Sanitary ware & parts thereof, of copper (741820);

Amino-resins, n.e.s., in primary forms (390930); Textile fabrics; impregnated, coated, covered or laminated with plastics, (excluding polyvinyl chloride, polyurethane, polyamides polyesters) (590390);

Non-threaded washers, of iron/steel, excluding spring and lock (731822);

Other parts & accessories for the motor vehicles (870899);

Polyethylene having a sp.gr. Of 0.94 or more (390120);

Aluminium alloys, unwrought (760120);

Newsprint, in rolls/sheets (480100);

Steering wheels, steering columns and steering boxes & parts thereof (870894);

Rolls for rolling mills (845530);

Bearing housings, incorporating ball or roller bearings (848330);

Slide fasteners, other than those fitted with chain scoops of base metal (960719);

Esters of acetic acid, n.e.s. (291539);

Parts of machines and apparatus for soldering or welding or for hot spraying of metals, metal carbides or cermets, n.e.s. (851590);

Pens and pencils n.e.s.; duplicating stylos, penholders, pencil-holders, and similar holders, parts of the articles (960899);

Dates, fresh/dried (80410);

Parts of machinery, plant/laboratory equipment, for treating materials by a process involving a change of temperature (841990);

Parts & accessories of bodies (other than safety seat belts) (870829);

Ultrasonic scanning apparatus (901812); Dental cements & other dental fillings, bone reconstruction cements (300640);

Electrical apparatus; n.e.c., for switching or protecting electrical circuits, for a voltage not exceeding 1000 volts (853690);

Other Insulated electric conductors; for a voltage not exceeding 1000 volts, fitted with connectors (854442):

Extruders for working rubber or plastics or otherwise (847720);

Other parts and accessories of looms, auxiliary machinery, n.e.s. (844849);

Warp knit fabrics, including those made on galloon knitting machines, n.e.s., excluding those made from cotton or synthetic or artificial fibres (600590);

Parts & accessories for stone, ceramic machine tools (846691);

Aluminium oxide (excl. Artificial corundum) (281820);

Machinery for working rubber/plastic (847780);

Polymers of ethylene, in primary forms, n.e.s. (390190);

Pepper (genus Piper), neither crushed nor ground (90411);

Leather further prepared after tanning or crusting, including parchment-dressed, of bovine (including buffalo) or equine animals, without hair on, split or not, (other than grain splits and full grains, unsplit) (410719);

Sugars; n.e.s., including invert sugar and other sugar and sugar syrup blends containing, in the dry state, 50% by weight of fructose (170290);

Offset printing machinery, used for printing by means of plates, cylinders and other printing components of heading 84.42, n.e.s. (844319);

Ferrous waste & scrap, n.e.s. (720449); Other reception apparatus for radio broadcasting, excl. 8527.91 and 8527.92 (852799);

Spectrometers, spectrophotometers and spectrographs using optical radiations (UV, visible, IR) (902730);

Zinc, not alloyed, unwrought, containing by weight 99.99% or more of zinc (790111); Manganese ores and concentrates, including manganiferous iron ores and concentrates with a manganese content of 20% or more, calculated on the dry weight (260200); Articles of apparel & clothing accessories (including gloves, mittens and mitts) (392620):

Tools for milling, for hand tools, whether or not power-operated (820770);

Copper foil, whether/not printed, backed with paper, paperboard, plastics or similar backing material, of a thickness (excluding any backing) not exceeding 0.15mm, of refined copper (741021);

Flat-rolled products of silicon-electrical steel, flat-rolled, width 600mm or more, grain-oriented, (722511);

Lead ores & concentrates (260700); Unwrought lead, refined (780110); Amino-alcohol-phenols, amino-acid-phenols and other amino-compounds with oxygen function (292250);

Paper and paperboard; coated with kaolin or other inorganic substances, for printing/writing/graphics, having 10% or less by weight of total fibres got by mechanical/chemi-mechanical process, sides exceeding 435mm and 297mm, unfolded, sheets (481019);

Warp knit fabrics, including those made on galloon knitting machines, n.e.s., excluding those made from cotton or synthetic or artificial fibres

Taps, cocks, valves and similar appliances: for pipes, boiler shells, tanks, vats or the like, including thermostatically controlled valves (848180):

Pigments & preparations based on titanium dioxide, containing 80% or more by weight of titanium dioxide calculated on the dry weight (320611);

Springs & leaves for springs, n.e.s., of iron/steel (732090);

Plates, sheets, film, foil & strip, of plastics n.e.s., non-cellular and not reinforced, laminated, supported or similarly combined with other materials (392099);

Peas (Pisum sativum), dried, shelled, whether or not skinned or split (71310);

Bearings parts, (other than balls, needles and rollers) (848299):

Gaskets, washers & other seals of vulcanised rubber, of non-cellular rubber (401693);

Parts & accessories of motorcycles (mopeds), other than saddles (871419):

Polypropylene, in primary forms (390210); Toothed wheels, chain sprockets and other transmission elements presented separately; parts of transmission components (848390);

Plaited bands, slings & the like, of iron/steel, not electrically insulated (731290);

Moulds for rubber or plastics, other than injection or compression types (848079);

Ball bearings (848210);

Acetone (291411);

Chemical wood pulp, soda/sulphate, (other than dissolving grades), semi-bleached or bleached, of non-coniferous wood (470329);

Brakes & servo-brakes; parts thereof (vehicles)

Used/new rags, scrap twine, cordage, rope and cables and worn out articles of twine, cordage, rope or cables, of textile materials; other than sorted (631090);

Parts of lifting handling machinery n.e.s. (843139):

Vitamin E & its derivatives (293628);

Internal combustion piston engines (diesel or semidiesel engines), of a kind used for other than

Poly(ethylene terephthalate), in primary forms, having a viscosity of 78ml/g or higher (390760);

Hydraulic presses for working metal (846291);

Parts of the machinery of 84.41, for making up paper pulp, paper or paperboard, including cutting machines of all kinds (844190);

Parts of centrifuges, including centrifugal dryers (842191);

Rolls for rolling mills (845530);

Plates, sheets, film, foil & strip, of polyethylene terephthalate, non-cellular and not reinforced, laminated, supported or similarly combined with other materials (392062):

Sanitary towels & tampons, napkins (481840):

Organic comps. N.e.s. (294200); Lead waste & scrap (780200);

Chickpeas (garbanzos), dried, shelled, whether or not skinned or split (71320);

Zinc alloys, unwrought (790120);

or light, n.e.c. (902780);

Paper, paperboard, cellulose wadding and webs of soft cellulose fibres; coated, impregnated, covered, surface-decorated or surface-coloured, n.e.s. (481190); Instruments and apparatus for physical or chemical analysis, for measuring or checking viscosity, porosity, expansion, surface tension or quantities of heat, sound

Oil-cake and other solid residues: whether or not ground or in the form of pellets, resulting from the extraction of oils, n.e.s. (230690);

Machines and mechanical appliances for treating metal, including electric wire coilwinders (847981);

Flat-rolled products of iron/non-alloy steel, width 600mm or more, cold-rolled, n.e.s. (720990):

Woven fabrics of synthetic filament yarn, dyed, n.e.s. (540792);

Woven fabrics of jute/other textile bast fibres, n.e.s., unbleached (531010); Lac; natural gums, gum resins, resins and balsams, n.e.s. (130190);

Non-electrical lamps & light fittings (940550);

Wood, n.e.s., sawn/chipped lengthwise/sliced lengthwise/sliced/peeled, whether or not planed, sanded or endjointed, of a thickness exceeding 6mm (440729):

Parts of the machinery, used in the industrial preparation or manufacture of food or drink (843890);

Vitamin E & its derivatives (293628);

marine propulsion or the vehicles of chapter 87 (840890);

Pulleys and flywheels, including pulley blocks (848350);

Apples, fresh (80810);

Paper & paperboard of a kind used for printing & writing, graphics, containing no, or not more than 10% by weight of total fibres obtained by mechanical or chemi-mechanical process, in rolls (481013):

Parts of the machinery of 84.41, for making up paper pulp, paper or paperboard, including cutting machines of all kinds (844190);

Electrical ignition or starting equipment, incl. Cutouts, of a kind used for spark-ignition or compression-ignition internal combustion engines (excl. Generators, starter motors, distributors, ignition coils, ignition magnetos, magnetic flywheels and sparking plugs) (851180); Textile spinning machines (844520);

Other apparatus for transmission reception of voice, images other data, incl. Apparatus for communication in a wired wireless network (such as a local wide area network) (851769);

Semi conductor media, solid state non-volatile storage devices, for the recording of sound of other phenomena, but excl. Products of Ch. 37. (852351);

Methylamine, di- or trimethylamine and their salts (292111);

Machinery for wringing, dressing, finishing, coating or impregnating textile yarns, fabrics or made up textile articles; for applying paste to base fabric used in manufacture of floor coverings (845180):

Parts of the steam turbines & other vapour turbines (840690);

Machines for used in the milling industry or for the working of cereals or dried leguminous vegetables, other than farm-type machinery (843780);

Thermostats, automatic types (903210); Textile fabrics; impregnated, coated, covered or

laminated with poly(vinyl chloride) (590310); Cashew nuts, in shell (80131);

Other flat-rolled products of other steel, flat-rolled, width 600mm or more, hot-rolled, in coils

(722530); Wood, in the rough, n.e.s., whether or not stripped

of bark or sapwood, or roughly squared, untreated, (440399);

Lentils, dried, shelled, whether/not skinned or split (71340);

Needle roller bearings (848240);

Iron or steel tubes, pipes and hollow profiles, seamless, n.e.s. (730490);

Machines for mixing, kneading, crushing, grinding, screening, sifting, homogenising, emulsifying or stirring (847982);

Zinc dust (790310);

Topical wood, n.e.s., in the rough, whether or not stripped of bark or sapwood, or roughly squared, untreated (440349); Waste & scrap of stainless steel (720421);

Sacks & bags (incl. Cones), of plastics other than ethylene polymers, for the conveyance or packing of goods (392329);

Electric generating sets with compressionignition internal combustion piston engines (diesel or semi-diesel engines), of an output exceeding 375kva (850213);

Machinery for sugar manufacture (843830); Machine-tools for deburring, polishing or otherwise finishing metal, sintered metal carbides or cermets by means of grinding stones, abrasives or polishing products, n.e.s. (846090);

Stationery n.e.s., made of paper or paperboard (482090);

Beans n.e.s., dried/shelled, including seed (71339);

Sodium hydroxide (caustic soda), solid (281511);

Zinc, not alloyed, unwrought, containing by weight less than 99.99% of zinc (790112); Chassis fitted with engines, for the motor vehicles (870600);

Recovered (waste & scrap) unbleached kraft paper or paperboard or of corrugated paper or paperboard (470710);

Recovered (waste & scrap)

paper/paperboard, n.e.s., and of unsorted waste and scrap (470790);

Tanned/crust hides & skins of bovine or equine animals, in the dry state (crust), without hair on, (other than full grains, unsplit; grain splits) (410449);

Aluminium waste & scrap (760200); Liquid dielectric transformers, having a power handling capacity <= 650 kva (850421);

Precious (other than diamonds) and semiprecious stones, unworked or simply sawn or roughly shaped, not strung, mounted or set (710310);

Carding machines (844511);

Liquid dielectric transformers, having a power handling capacity > 650 kva but <= 10000 kva (850422);

Wattle extract (320120);

Unwrought lead other than refined, not containing by weight antimony as the principal other element (780199);

Articles for the conveyance/packing of goods, n.e.s. (392390);

Coral & similar materials, unworked (50800);

Parts of electric filament or discharge lamps, sealed beam lamp units, ultra-violet or infra-red lamps and arc-lamps, n.e.s. (853990);

Parts and accessories of office machines, nes (847340);

Synthetic organic colouring matter, dyes n.e.s. (320419);

Natural rubber (excl. Latex), in smoked sheets (400121);

Refractory cements, mortars, concretes and similar composition (381600);

Parts of the engines & motors of 8412 (841290); Parts of aircraft engines (spark-ignition reciprocating or rotary internal combustion piston engines) (840910);

Rubies, sapphires & emeralds, worked (other than simply sawn or roughly shaped), not strung, mounted or set (710391);

Mountings, fittings & similar articles, of base metal, suitable for buildings (830241);

Paraffin wax; containing by weight<0.75% of oil, obtained by synthesis or by other processes, whether or not coloured (271220);

Parts of the air-conditioning machines, with motor driven fan and elements for temperature control (841590);

Textile fabrics & felts, endless/ fitted with linking devices, of a kind used in paper-making or similar machines, weighing 650g/m2 or more (591132); Silicones, in primary forms (391000);

Auminium plates, sheets & strip, rectangular (including square), of thickness exceeding 0.2mm, alloys (760612);

Electric ovens, cookers, cooking plates and boiling rings, electric grillers and roasters, for domestic use (excl. Space heating stoves and microwave ovens) (851660);

Polycarbonates, in primary forms (390740); Clocks (excluding those with watch movements and instrument panel clocks), other than alarm or wall clocks, other than electrically operated (910599);

Tailors' dummies and other lay figures; automata and other animated displays used for shop window dressing (961800);

Plates, sheets, film, foil & strip, non-cellular and not reinforced, laminated, supported or similarly combined with other materials (392051);

Acrylic polymers (other than polymethyl methacrylate), in primary forms (390690); Activated carbon (380210);

Statuettes & other ornamental articles (392640); Ceramic tableware, kitchenware, household articles and toilet articles, other than of porcelain or china (691200):

Sewing machines, not household or automatic unit type (845229);

Erythromycin & its derivatives; salts thereof (294150);

Kidney beans, incl. White pea beans (phaseolus vulgaris), dried, shelled, whether or not skinned or split (71333);

Containers for compressed/liquefied gas, of iron or steel (731100);

Parings & other waste of

leather/composition leather; not suitable for the manufacture of leather articles; leather dust, powder and flour (411520);

Rum & tafia (220840);

Mineral substance, n.e.s. In Ch.25 (253090);

Tanned/crust hides & skins, n.e.s., (410691);

Copper waste & scrap (740400);

Fats & oils derived from milk (other than butter or dairy spreads) (40590);

Alkaloid of cinchona and their derivatives and salts thereof (293920);

Waste and scrap of primary cells, primary batteries and electric accumulators; spent primary cells, spent primary batteries and spent electric accumulators (854810); Milk in powder/granules/other solid form,

Milk in powder/granules/other solid form of a fat content not exceeding 1.5% (by weight) (40210);

Zinc waste & scrap (790200);

Milk in powder/granules/other solid form, of a fat content not exceeding 1.5% (by weight) (40229);

Wool, not carded/combed, greasy, (including fleece-washed wool), not carded or combed (510119);

Zirconium ores & concentrates (261510); "New pneumatic tyres, of rubber, Of a kind used on agricultural or forestry vehicles And machines (401192)

Pumps and compressor for air/vaccum/other gas (841480);

Machine-tools for working metal/cermets/sintered metal carbide, without removing material, n.e.s. In heading no. 8463 (846390);

Lead waste & scrap (780200);

Lubricating preparations (excl. Those for the treatment of textile and similar materials), not containing petroleum oils or oils obtained from bituminous minerals (340399);

Washing/bleaching/dyeing machines (845140); Tube/pipe fittings of stainless steel, n.e.s (730729); Carboxymethylcellulose & its salts, in primary forms (391231);

Parts of liquid elevators (841392);

Silicon dioxide (281122);

Heterocyclic compounds with oxygen (293299); Unglazed ceramic flags & paving/hearth/wall tiles (690790);

Other colour reception apparatus for television, whether not incorporating radio broadcast receivers sound video recording reproducing apparatus, (852872);

Articles of plastics, n.e.s. (392690);

Refractory ceramic goods, composition of which n.e.s., other than those of siliceous fossil meals or similar earths (690390);

Shearing machines for working metal (846239); Other parts and accessories for printing machinery excl. 844391 (844399);

Portable electric lamps, designed to function by their own source of energy (851310);

Phthalic anhydride (291735);

Chromatographs and electrophoresis instruments (902720);

Filter paper & paperboard, uncoated (480540); Glass in tubes, unworked, other than of glass having a linear coefficient of expansion not exceeding 5 x 10-6 per kelvin within a temperature range of 0-300 degrees C, not fused quartz or other fused silica (700239);

Heat exchange units, whether/not electrically heated, for domestic purposes (841950); Machine-tools for wood, cork or hard plastic, etc. N.e.s. (846599);

Saws for working in the hand, with self-contained electric motor (846722);

Machinery for the manufacture of confectionery, cocoa or chocolate (843820);

Locks (key/combination/electrically operated), other than those for motor vehicles or furniture, of base metal (830140);

Line pipe of a kind used for oil/gas, of stainless steel, seamless (730411);

Waste (incl. Noils, yarn waste and garnetted stock), of synthetic fibres (550510);

Parts of the machinery for treating textile fabrics (845190);

Vehicles with only spark-ignition internal combustion reciprocating piston engine, cylinder capacity over 3000cc (870324);

Synthetic rubber & factice derived, in primary forms or in plates, sheets or strip (400299); Machinery, n.e.s. In heading no. 8441, for making up paper pulp, paper or paperboard (844180); Organic comps. N.e.s. (294200);

Wood, n.e.s., sawn or chipped lengthwise, sliced or peeled, whether or not planed, sanded or finger-jointed, thicker than 6mm (440799);

Other Machine-tools for working stone, ceramics, concrete, asbestos-cement or like mineral materials or for cold working glass, (other than sawing, grinding or polishing machines) (846490); Flat-rolled products of iron/non-alloy steel, width less than 600mm, plated or coated, (excluding tin, zinc or plastic) (721250);

1-Cyanoguanidine (dicyandiamide) (292620); Wooden frames for paintings, photographs, mirrors or similar objects (441400); Fixed electrical resistors for a power handling capacity <= 20 W (excl. Heating resistors) (853321);

Iron or steel threaded articles except screw, nut, bolts (731819);

Sewing thread of artificial filaments, whether or not put up for retail sale (540120);

Coal; (other than anthracite and bituminous), whether or not pulverised but not agglomerated (270119):

Table/floor/wall/window/ceiling/roof fans, with a self-contained electric motor of an output not exceeding 125W (841451);

Printing type, blocks, plates, cylinders and other printing components, lithographic stones prepared for printing purposes (for example, planed, grained or polished) (844250);

Steam turbines & other vapour turbines, (for other than marine propulsion), of an output not exceeding 40MW (840682);

Machinery for filling, closing, sealing, capsuling or labelling bottles, cans, bags or other containers (842230):

Radio-broadcast receivers not capable of operating without an external source of power, of a kind used in motor vehicles, incl. Apparatus capable of also receiving radio-telephony or radio-telegraphy, combined with sound recording or reproducing apparatus (852721);

Bumpers & parts thereof of the motor vehicles (870810):

Mixtures of odoriferous substance and mixtures, used as raw materials in industries other than the food or drink industries (330290);

Butanoic acids, pentanoic acids, their salts and esters (291560);

Halogenated derivatives of aromatic (290369);

Gas-operated machinery & apparatus, n.e.s., for soldering, brazing or welding, whether or not capable of cutting (846820);

Wire, rods, tubes, plates, electrodes of base metal or metal carbides; of a kind used for soldering, brazing, welding; wires and rods for metal spraying, n.e.s. (831190);

Diols other than ethylene glycol (ethanediol) and Propylene glycol (propane-1,2-diol) (290539); Chopped strands, of a length of not more than 50mm (701911);

Fats & oils derived from milk (other than butter or dairy spreads) (40590):

Lac; natural gums, gum resins, resins and balsams, n.e.s. (130190);

Millstones & grindstones for milling, grinding or pulping (680410);

Rails of iron/steel (730210);

Sewing machines of the household type (845210); Hard rubber (e.g., ebonite) in all forms, including waste and scrap, and articles of hard rubber (401700);

Manganese ores and concentrates, including manganiferous iron ores and concentrates with a manganese content of 20% or more, calculated on the dry weight (260200);

Ferrous waste & scrap, n.e.s. (720449);

Sheets for veneering (including those obtained by slicing laminated wood), of other wood sawn length wise, sliced or peeled, whether or not planed, sanded or end-jointed, not thicker than 6mm (440839);

Oil seeds & oleaginous fruits, n.e.s., whether or not broken (120799):

Pigments & preparations based thereon (320417); Waste & scrap of stainless steel (720421); Ferrous turnings, shavings, chips, milling waste, sawdust, fillings, trimmings and stampings, whether or not in bundles (720441);

Flat-rolled products of iron/non-alloy steel, width 600mm or more, cold-rolled, n.e.s. (720990); Office machines; automatic banknote dispensers, coin-sorting machines, coin-counting or wrapping machines, pencil-sharpening machines, perforating

Parts of electrical ignition or starting equipment, generators, etc. Of heading 8511, n.e.s. (851190); Punching/notching machines for work (846241); Topical wood, n.e.s., in the rough, whether or not stripped of bark or sapwood, or roughly squared,

or stapling machines (847290);

untreated (440349);

Electrodes of graphite or other carbon, for electrical purposes (excl. Those used for furnaces) (854519);

Motorcycles (incl. Mopeds) & cycles (871120); Boards, sheets, panels, tiles & similar articles, faced or reinforced with paper or paperboard only, not ornamented (680911);

Chemical products & preparations of the chemical or allied industries, n.e.s. (382490);

Carbon brushes for electrical purposes (854520); Ginger (91010);

Aluminium waste & scrap (760200);

Wood, n.e.s., sawn/chipped lengthwise/sliced lengthwise/sliced/peeled, whether or not planed, sanded or end-jointed, of a thickness exceeding 6mm (440729);

Finishing agents, dye carriers to accelerate dyeing or fixing of dye-stuffs, other products and preparations, used in the textile industry (excluding those with a basis of amylaceous substances) (380991);

Lighting sets of a kind used for Christmas trees (940530);

Flat-rolled products of iron/non-alloy steel, width less than 600mm, plated or coated with tin (721210);

Zinc, not alloyed, unwrought, containing by weight less than 99.99% of zinc (790112); Gears & gearing (not toothed wheels, chain sprockets and other transmission elements presented separately); ball or roller screws; gear boxes and other speed changers, including torque converters (848340);

Alcohol peroxides, ether peroxides, ketone peroxides and their halogenated, sulphonated, nitrated or nitrosated derivatives (290960); Weighing machine weights of all kinds; parts of weighing machinery (842390);

Waste & scrap of alloy steel other than stainless steel (720429);

Artificial waxes & prepared waxes (other than of chemically modified lignite or polyethylene glycol) (340490);

Parts of the non-electric domestic appliances, of iron or steel (732190);

Mineral substance, n.e.s. In Ch.25 (253090); Cinematographic film, exposed & developed, whether or not incorporating sound track or consisting only of sound track, of a width of 35cm or more (370690);

Zinc dust (790310);

Precious and semi-precious stones (excl. Diamonds, rubies, sapphires and emeralds), worked other than simply sawn or roughly shaped, not strung, mounted or set (710399);

Writing/drawing ink & other inks, n.e.s., whether or not concentrated or solid (321590);

Printing ink (excl. Black) (321519);

Leather further prepared after tanning or crusting, including parchment-dressed, of bovine (including buffalo) or equine animals, without hair on, split or not, (other than grain splits and full grains, unsplit) (410719):

Kidney beans, incl. White pea beans (phaseolus vulgaris), dried, shelled, whether or not skinned or split (71333);

Lactones, other lactones, n.e.s. (293229); Hydrides, nitrides, azides, silicides and borides, whether or not chemically defined, other than compounds which are also carbides and already specified (285000);

Thiosulphates (283230);

Labels, badges & similar articles; (other than woven), of textile materials, in the piece, in strips or cut to shape or size, not embroidered (580790); Chickpeas (garbanzos), dried, shelled, whether or not skinned or split (71320);

Cloves (whole fruit, cloves & stems) (90700); Woven fabrics of cotton, containing 85% or more by weight of cotton, of yarns of different colours, plain weave, weighing not more than 100g/m2 (520841);

Gum Arabic (130120);

Cotton sewing thread, put up for RS, containing less than 85% by weight of cotton (520420); Sewing thread of synthetic staple fibres, whether or not put up for retail sale (550810);

Gravure printing machinery (844317);

Copper waste & scrap (740400);

Paving blocks, slabs, bricks and other articles of pressed or moulded glass (701690);

Filter paper & paperboard, cut to shape (482320); Plywood, veneered panels and similar laminated wood; of bamboo (441210);

Coral & similar materials, unworked (50800);

Sack kraft paper, uncoated, n.e.s. (480429);

Sodium dichromate (284130);

Beans n.e.s., dried/shelled, including seed (71339); Tanned/crust skins of sheep/lambs, without wool on, whether or not split, but not further prepared, in the dry state (410530);

Parts & accessories of card clothing (844831); Leather further prepared after tanning incl. Parchment-dressed, of bovine (including buffalo) or equine animals, no hair, and whole hides and skins, and sides, (full grains, unsplit and grain splits) (410799);

Stationery n.e.s., made of paper or paperboard (482090);

Powders of non-lamellar structure, of copper (740610):

Recovered (waste & scrap) paper/paperboard, n.e.s., and of unsorted waste and scrap (470790); Paints & varnishes (incl. Enamels & lacquers), based on polymers n.e.s., dispersed or dissolved in a non-aqueous medium (320890);

Unwrought lead other than refined, containing by weight antimony as the principal other element (780191);

Vegetable alkaloids, natural/reproduced by synthesis, and their salts, ethers, esters and other derivatives, n.e.s. (293999);

Tanned/crust hides & skins of bovine or equine animals, in the dry state (crust), without hair on, (other than full grains, unsplit; grain splits) (410449);

Articles for the conveyance/packing of goods, n.e.s. (392390);

Precious (other than diamonds) and semi-precious stones, unworked or simply sawn or roughly shaped, not strung, mounted or set (710310); Containers for compressed/liquefied gas, of iron or steel (731100);

Unwrought lead other than refined, not containing by weight antimony as the principal other element (780199);

Sewing thread of synthetic filaments, whether or not put up for retail sale (540110);

Kraftliner, uncoated, unbleached, in rolls or sheets, n.e.s. (480411);

Unwrought lead, refined (780110);

Other articles of tin, n.e.s. (800700);

Zinc waste & scrap (790200);

Raw skins of sheep/lambs, fresh, or salted, dried, limed, pickled or otherwise preserved, (but not tanned, parchment-dressed or further prepared), with wool on (410210);

Other sheets, panels, tiles & similar articles (other than corrugated sheets), not containing asbestos (681182);

Carding machines (844511);

Dried leguminous vegetables, n.e.c., shelled, whether or not skinned or split (71390);

Paper & paperboard, other than corrugated (with or without glued flat surface sheets) or kraft paper, creped or crinkled, whether or not embossed or perforated, in rolls or sheets, other than paper already specified (480890);

Prepared driers (321100);

Wool, not carded/combed, greasy, (including fleece-washed wool), not carded or combed (510119)

Lesotho

Threaded screws & bolts of iron/steel, n.e.s., whether or not with their nuts or washers (731815);

Articles of aluminium n.e.s. (761699); Insulated electric conductors for a voltage not exceeding 1000 volts, not fitted with connectors (854449);

Transformers having a power handling capacity <= 1 kva (excl. Liquid dielectric transformers) (850431);

Electrical apparatus; n.e.c., for switching or protecting electrical circuits, for a voltage not exceeding 1000 volts (853690);

Discharge lamps (excl. Flourescent, hot cathode lamps, mercury or sodium vapour lamps, metal halide lamps and ultra-violet lamps) (853939); Locks (key/combination/electrically operated), other than those for motor vehicles or furniture, of base metal (830140):

Non-refractory surfacing preparations, mastics n.e.s. (321490);

Aluminium oxide (excl. Artificial corundum) (281820);

Ethylene polymers; in primary forms, polyethylene having a specific gravity of less than 0.94 (390110):

Apparatus based on use of x-rays and similar; parts and accessories (x-ray generators, tubes, high tension generators, control panels and desks, screens, examination or treatment tables, chairs and like (902290);

Parts & accessories n.e.s. In Ch.90 (903300)

Chandeliers and other electric ceiling or wall light fittings; excluding those used for lighting public open spaces or thoroughfares (940510); Lightning arresterrs, voltage limiters and surge suppressors, for a voltage > 1.000 V (853540); Articles of plastics, n.e.s. (392690); Burglar or fire alarms and similar apparatus (853110); Chemical products&preparations of the chemical or allied industries, n.e.s. (382490); Boards, panels, consoles, desks, cabinets and other bases for the goods of heading 8537, not equipped with their apparatus (853810); Plugs and sockets for a voltage <= 1000 V (excl. Lamp-holders) (853669); Iron or steel threaded articles except screw, nut, bolts (731819); Automatic circuit breakers for a voltage <= 1.000 V (853620): Shawls, scarves, mufflers, mantillas, veils and the like, of textile materials n.e.s., (not knittedd or crocheted) (621490) Parts & accessories of bodies (other than safety Liberia Electric conductors, for a voltage > 1.000 seat belts) (870829); V, insulated, n.e.s. (854460); Parts of the seats of 94.01 (seats in vehicles, Microtomes and parts & accessories thereof aircrafts, convertible to bed, etc.) (940190); (902790); Waste & scrap of stainless steel (720421); Tubes, pipes & hose fittings (e.g. Joints, elbows, flanges) (391740); Printed circuits (853400); Parts of the air-conditioning machines, with motor Food preparations, n.e.s. (210690); driven fan and elements for temperature control Instruments, appliances and machines; for measuring or checking n.e.c. (903180); Threaded screws & bolts of iron/steel, n.e.s.. Precious (other than diamonds) and semiwhether or not with their nuts or washers precious stones, unworked or simply sawn (731815): or roughly shaped, not strung, mounted or Articles of vulcanised rubber other than hard set (710310): rubber, n.e.s. (401699); Other apparatus for transmission reception Stoppers, caps & lids, of base metal (830990); of voice, images other data, incl. Apparatus Clutches & parts thereof for the motor vehicles for communication in a wired wireless network (such as a local wide area network) (870893); Electrical apparatus like switches n.e.c. In heading (851769);no. 8536, for a voltage not exceeding 1000 volts Articles of cutlery (e.g., hair clippers and mincing knives) (821490); Pulleys and flywheels, including pulley blocks Machines and mechanical appliances; (848350): having individual functions, n.e.c. Electrical apparatus; n.e.c., for switching or (847989); protecting electrical circuits, for a voltage not Wood, n.e.s., sawn/chipped exceeding 1000 volts (853690); lengthwise/sliced lengthwise/sliced/peeled, Electric motors of an output not exceeding 37.5W whether or not planed, sanded or end-(850110); jointed, of a thickness exceeding 6mm Insulated electric conductors for a voltage not (440729); exceeding 1000 volts, not fitted with connectors Wood, n.e.s., sawn or chipped lengthwise, (854449);sliced or peeled, whether or not planed, Hinges of base metal (830210); sanded or finger-jointed, thicker than 6mm Parts and accessories for instruments, appliances (440799): Natural rubber (excl. Latex), in smoked and machines measuring or checking (903190); Ceramic articles n.e.s., other than of porcelain or sheets (400121); china (691490); Copper waste & scrap (740400); Jacks & hoists, (other than hydraulic), of a kind Aluminium waste & scrap (760200);

used for raising vehicles (842549);

Transmission shafts (including cam shafts and crank shafts) and cranks (848310);

Parts and accessories for instruments/apparatus measuring or checking the flow, level, pressure or other variables of liquids or gases (902690); Parts & accessories of regulating or controlling instruments and apparatus; automatic (903290); Fuel/lubricating/cooling medium pumps for internal combustion piston engines (841330); Trade advertising material, commercial catalogue and the like (491110);

Non-ionic surface-active agents, whether or not put up for retail sale (340213);

Air-conditioning machines, comprising a motor driven fan and elements for changing the temperature and humidity, of a kind used for persons, in motor vehicles (841520);

Articles of iron/steel, n.e.s., forged/stamped, but not further worked (732619);

Other Parts suitable for use solely or principally with the apparatus of headings 8525 to 8528 (reception and transmission apparatus, monitors, etc.) (852990);

Regulating or controlling instruments and apparatus; automatic, other than hydraulic or pneumatic (903289);

Food preparations, n.e.s. (210690);

Starter motors and dual purpose starter-generators, of a kind used for spark or compression-ignition internal combustion engines (851140);

Parts of microphones, loudspeakers, headphones, earphones or audio-frequency electric amplifiers, n.e.s. (851890);

Portable automatic data processing (ADP) machines, weighing not more than 10kg, consisting of at least a central processing unit, a keyboard and a display (847130);

Brakes & servo-brakes; parts thereof (vehicles) (870830);

Sound signalling equipment; electrical, used on cycles or motor vehicles (851230);

Other parts & accessories for the motor vehicles (870899);

Waste & scrap of stainless steel (720421); Articles of apparel & clothing accessories (including gloves, mittens and mitts) (392620); Suspension systems & parts thereof (vehicles) (870880);

Pumps and compressors; parts, of air or vacuum pumps, air or other gas compressors and fans, ventilating or recycling hoods incorporating a fan (841490);

Radiators & parts thereof for the motor vehicles (870891);

Gears & gearing (not toothed wheels, chain sprockets and other transmission elements presented separately); ball or roller screws; gear boxes and other speed changers, including torque converters (848340);

Isocyanates (292910);

Wood, in the rough, n.e.s., whether or not stripped of bark or sapwood, or roughly squared, untreated, (440399);

Ferrous waste & scrap, n.e.s. (720449)

Tubes, pipes & hoses, of vulcanised rubber (other than hard rubber), reinforced or otherwise combined only with textile materials, with fittings (400932):

Ceramic tableware, kitchenware, household articles and toilet articles, other than of porcelain or china (691200);

Telephones for cellular networks or for other wireless networks (851712);

"Tubes, pipes & hoses, of vulcanised rubber, not reinforced or otherwise combined with other materials, with fittings (400912);"

Rear-view mirrors for vehicles (700910); Ethylene polymers; in primary forms, polyethylene having a specific gravity of less than 0.94 (390110);

Tubes, pipes & hoses, of vulcanised rubber, reinforced or otherwise combined only with metal, with fittings (400922);

Parts and accessories for machine tools, for laser operation, metalworking machining centers, lathes and drilling machines, etc., others (846693); Safety seat belts of the motor vehicles and parts (870821);

Wooden furniture of a kind used in bedroom (940350):

Ignition wiring sets and other wiring sets for vehicles, aircraft or ships (854430);

Sound recording or reproducing apparatus; n.e.c. (851989);

Wooden furniture of a kind used in office (940330);

Boxes, cases, crates and similar articles for the conveyance or packing of goods (392310);

Chemical products&preparations of the chemical or allied industries, n.e.s. (382490);

Floor coverings & mats of vulcanises rubber, non-cellular rubber (401691);

Centrifugal pumps, n.e.s. In 8413, for liquids (841370);

Silencers (mufflers) & exhaust pipe and parts (vehicle) (870892);

Lubricating preparations (excl. Those for the treatment of textile and similar materials), not containing petroleum oils or oils obtained from bituminous minerals (340399);

Lamps and light fittings; parts thereof, of materials other than glass or plastics (940599);

Plates, sheets, film, foil & strip, other than cellular (392190);

Loudspeakers; not mounted in their enclosures (851829);

Paper, paperboard, cellulose wadding or webs of cellulose fibres (482390);

Metal furniture (other than for office use) (940320);

Storage units of ADP (847170);

Waste & scrap of cast iron (720410);

Glazed ceramic flags & paving/hearth/wall tiles (690890);

Air conditioning machines comprising a motordriven fan and elements for changing the temperature and humidity, window or wall types, self-contained or split-system (841510); Inorganic bases, metal oxides, hydroxides and peroxides; n.e.s. (282590);

Wood, in the rough, n.e.s., whether or not stripped of bark or sapwood, or roughly squared, untreated, (440399);

Organic composite solvents & thinners, n.e.s. Or included, prepared paint or varnish removers (381400):

Cruise ships, excursion boats and similar vessels, principally designed for the transport of persons, ferry boats of all kinds (890110);

Prepared rubber accelerators (381210);

Slide fasteners, fitted with chain scoops of base metal (960711);

Aluminium waste & scrap (760200);

Video recording or reproducing apparatus; magnetic tape-type (852110);

Other electrical parts of machinery (854890); Gloves (excl. Surgical), mittens & mits, of

vulcanized rubber (401519); Air compressors mounted on a wheeled chassis for towing (841440);

Ferrous waste & scrap, n.e.s. (720449);

Topical wood, n.e.s., in the rough, whether or not stripped of bark or sapwood, or roughly squared, untreated (440349);

Light petroleum oils & preparations (not crude), made from petroleum and bitum oils, etc. (271011):

Waste & scrap of alloy steel other than stainless steel (720429)

Madagascar

Non-threaded articles, n.e.s., of iron/steel (731829);

Other parts & accessories for the motor vehicles (870899);

Diamonds, non-industrial, (other than unworked or simply sawn, cleaved or bruted), but not mounted or set (710239);

Parts of aircraft engines (spark-ignition reciprocating or rotary internal combustion piston engines) (840910);

Parts of Machinery for working rubber or plastics (847790);

Whiskies (220830):

Self-adhesive plates, sheets, film, foil, tape, strip, other flat shapes thereof, other than in rolls of a width not exceeding 20cm (391990);

Threaded nuts of iron/steel (731816);

Esters of acetic acid, n.e.s. (291539);

Chandeliers and other electric ceiling or wall light fittings; excluding those used for lighting public open spaces or thoroughfares (940510);

Parts of Electrical transformers, static converters and inductors (850490);

Brooms, brushes, mops, feather dusters, squeegees, hand operated mechanical floor sweepers; (not motorised), prepared knots and tufts for brooms or brush making n.e.c. (960390);

Paper/paperboard labels of all kind, unprinted (482190);

Transformers having a power handling capacity <= 1 kva (excl. Liquid dielectric transformers) (850431);

Parts of Electrical transformers, static converters and inductors (850490); Antibiotics & their derivatives, n.e.s. (294190);

Refractory bricks, blocks, tiles & similar refractory ceramic constructional goods, n.e.s. (690290);

Interchangeable tools for hand tool, whether or not power-operated, for screw-driving or uses n.e.c (820790);

Parts of Machines & mechanical appliances of those having individual functions (847990);

Other articles of zinc, n.e.s. (790700);

Other monitors, of a kind solely principally used in an automatic data processing system (852851); Other Insulated electric conductors; for a voltage not exceeding 1000 volts, fitted with connectors (854442);

Bearings parts, (other than balls, needles and rollers) (848299);

Plates, sheets, film, foil & strip, of plastics n.e.s., non-cellular and not reinforced, laminated, supported or similarly combined with other materials (392099);

Articles of vulcanised rubber other than hard rubber, n.e.s. (401699);

Glass mirrors other than rear-view (700991); Petroleum oils and oils from bituminous minerals, not containing biodiesel, not crude, not waste oils; preparations n.e.s. (271019);

Threaded screws & bolts of iron/steel, n.e.s., whether or not with their nuts or washers (731815);

Static converters (850440);

Parts and accessories for instruments, appliances and machines measuring or checking (903190); Tubes, pipes & hoses of plastics, n.e.s. (391739); Electrical apparatus; n.e.c., for switching or protecting electrical circuits, for a voltage not exceeding 1000 volts (853690);

Instruments, appliances and machines; for measuring or checking n.e.c. (903180); Other tubes, pipes and hollow profiles (for example, open seam or welded, riveted or similarly closed), of iron or steel (730690); Engines parts for internal combustion piston engines (excluding spark-ignition) (840999); Regulating or controlling instruments and apparatus; automatic, other than hydraulic or pneumatic (903289);

Copper wire, of refined copper, of which the maximum cross-sectional dimension is 6mm or less (740819);

Gears & gearing (not toothed wheels, chain sprockets and other transmission elements presented separately); ball or roller screws; gear boxes and other speed changers, including torque converters (848340);

Gin & Geneva (220850);

Ethylene polymers; in primary forms, polyethylene having a specific gravity of less than 0.94 (390110);

Filtering/purifying machinery & apparatus, other than intake air filters for internal combustion engines (842139);

Weighing machine weights of all kinds; parts of weighing machinery (842390);

Parts suitable for use solely or principally with electric motors and generators, electric generating sets and rotary converters, n.e.s. (850300);

Wood, in the rough, n.e.s., whether or not stripped of bark or sapwood, or roughly squared, untreated, (440399);

Polypropylene, in primary forms (390210); Copper ores & concentrates (260300); Parts of Machinery for working rubber or plastics (847790);

Portable electric lamps, designed to function by their own source of energy (851310);

Aluminium nails, tacks, staples, screws, bolts, nuts, screw hooks, rivets, cotters, cotter-pins, washers and similar articles (761610):

Parts of the slide fasteners (960720); Sacks & bags, of a kind used for the packing of goods, of jute or of other textile bast fibres (630510);

Ferrous waste & scrap, n.e.s. (720449); Dried leguminous vegetables, n.e.c., shelled, whether or not skinned or split (71390);

Poly(ethylene terephthalate), in primary forms, having a viscosity of 78ml/g or higher (390760);

Tools for working in the hand, pneumatic, rotary type (including combined rotary-percussion) (846711);

Articles & equipment for sports, for outdoor games and recreation n.e.c. (950699); Organic comps. N.e.s. (294200);

Phosphates of potassium (283524);

Optical instruments and appliances; for measuring or checking, n.e.c. (903149); Essential oils (terpeneless/not), including concretes and absolutes, n.e.s. (330190); Recovered (waste & scrap)

paper/paperboard, n.e.s., and of unsorted waste and scrap (470790);

Nuts, n.e.s., fresh or dried, whether or not shelled or peeled (80290);

Seeds, n.e.s., of a kind used for sowing (120999);

Waste & scrap of cast iron (720410); Worked monumental/building stone of granite; articles thereof, simply cut or sawn, with a flat or even surface (680223);

Woven fabrics of cotton, containing 85% or more by weight of cotton, of yarns of different colours, 3-thread or 4-thread twill, including cross twill, weighing not more than 200g/m2 (520843);

Copper waste & scrap (740400); Coral & similar materials, unworked (50800):

Heterocyclic compounds with oxygen (293299);

Cashew nuts, in shell (80131);

Aeroplanes and other aircraft of an unladen weight not exceeding 2000kg (880220);

Parts and accessories for machine tools, for laser operation, metalworking machining centers, lathes and drilling machines, etc., others (846693);

Parts of hoists and winches (843110);

Inductors (excl. Inductors for discharge lamps or tubes) (850450);

Articles of iron/steel, n.e.s., forged/stamped, but not further worked (732619);

Printed circuits (853400);

Electric sound or visual signalling apparatus (excl. Indicator panels with liquid crystal devices or light emitting diodes, burglar or fire alarms and similar apparatus and apparatus for cycles, motor vehicles and traffic signalling) (853180);

Polyethers other than polyacetals, excluding polyacetals (390720);

Bearing housings, incorporating ball or roller bearings (848330);

Boxes, cases, crates and similar articles for the conveyance or packing of goods (392310); Aluminium waste & scrap (760200);

Narrow woven fabrics n.e.s., of textile materials n.e.c., (excl. Labels, badges and similar articles) (580639);

Boring or sinking machinery n.e.s., not self propelled (843049);

Winding wire for electrical purposes, of copper, insulated (854411);

Video recording or reproducing apparatus, whether or not incorporating a video tuner (excl. Magnetic tape-type and video camera recorders) (852190); Parts of taps, cocks, valves and similar appliances (848190);

Input or output units for automatic data processing machines, whether or not containing storage units in the same housing (excl. Keyboards) (847160); Electric sound amplifier sets (851850);

Parts suitable for use solely or principally with the apparatus of heading 8535, 8536 or 8537, n.e.s. (excl. Boards, panels, consoles, desks, cabinets and other bases for the goods of heading 8537, not equipped with their apparatus) (853890);

Parts of machinery for sorting, screening, separating, washing, crushing etc mineral substances, for agglomerating, shaping or moulding solid fuels, ceramic pastes etc, for forming foundry moulds of sand (847490); Radiators & parts thereof for the motor vehicles (870891):

Peas (Pisum sativum), dried, shelled, whether or not skinned or split (71310);

Articles of plastics, n.e.s. (392690);

Ceramic sinks, wash basins, wash basin pedestals, baths, bidets, water closet pans, flushing cisterns, urinals and similar sanitary fixtures; of other than porcelain or china (691090);

Tanned/crust hides & skins of bovine or equine animals, in the wet state (including wet blue), without hair on, excluding full grains, unsplit; grain splits (410419);

Beans of the species vigna mungo (L.) Hepper or vigna radiata (L.) Wilczek, shelled, whether or not skinned or split, dried (71331):

Kidney beans, incl. White pea beans (phaseolus vulgaris), dried, shelled, whether or not skinned or split (71333);

Vegetable materials of a kind used primarily for plaiting, n.e.s. (140490); Raw hides & skins of bovine (including buffalo) and equine animals, fresh, salted or preserved, but not tanned, parchment dressed or further prepared, whether or not dehaired or split (410190);

Pepper (genus Piper), neither crushed nor ground (90411);

Precious (other than diamonds) and semiprecious stones, unworked or simply sawn or roughly shaped, not strung, mounted or set (710310);

Plants & parts of plants, including seeds and fruits, n.e.s., used primarily in perfumery, pharmacy or for insecticidal, fungicidal purposes; fresh or dried, whether or not cut, crushed or powdered (121190); Granite, crude/roughly trimmed (251611); Granite; merely cut, by sawing or otherwise, into blocks or slabs of a rectangular (including square) shape (251612);

Waste and scrap of primary cells, primary batteries and electric accumulators; spent primary cells, spent primary batteries and spent electric accumulators (854810)

Topical wood, n.e.s., in the rough, whether or not stripped of bark or sapwood, or roughly squared, untreated (440349):

Photosensitive semiconductor device (854140); Ball bearings (848210);

Chemical products&preparations of the chemical or allied industries, n.e.s. (382490);

Machines which perform two or more of the functions of printing, copying or facsimile transmission, capable of connecting to an automatic data processing machine or to a network (844331):

Polymers of ethylene, in primary forms, n.e.s. (390190);

Dates, fresh/dried (80410);

Sewing machines, not household or automatic unit type (845229);

Ferrous turnings, shavings, chips, milling waste, sawdust, fillings, trimmings and stampings, whether or not in bundles (720441);

Paints & varnishes (incl. Enamels & lacquers), based on polyesters, dispersed or dissolved in a non-aqueous medium (320810);

Ornamental trimmings in the piece, without embroidery, other than knittedd or crocheted (580890);

Apples, fresh (80810);

Permanent magnets and articles intended to become permanent magnets after magnetization, of materials other than metal (850519);

Storage units of ADP (847170);

Statuettes & other ornamental articles (392640); Manifold business forms & interleaved carbon

sets, of paper/paperboard (482040);

Articles of iron/steel, n.e.s. (732690);

Textile furnishing articles, of textile materials other than cotton or synthetic fibres, not knittedd or crocheted (excluding bedspreads and other articles of heading no. 9404) (630499);

Lentils, dried, shelled, whether/not skinned or split (71340);

Check (nonreturn) valves, for pipes, boiler shells, tanks, vats or the like (848130);

Printing ink, black (321511);

Undenatured ethyl alcohol of an alc (220890); Mountings, fittings and similar articles; suitable for other than buildings or furniture, of base metal (830249);

Lead waste & scrap (780200);

Woven fabrics containing 85%/more by weight of silk or of silk waste other than noil silk (500720); Recovered (waste & scrap) paper/paperboard,

n.e.s., and of unsorted waste and scrap (470790); Nonwovens, whether/not

impregnated/coated/covered/laminated, of manmade filaments, (weighing not more than 25g/m2) (560311);

Auminium plates, sheets & strip, rectangular (including square), of thickness exceeding 0.2mm, alloys (760612);

Auminium plates, sheets & strip, (not rectangular or square), of thickness exceeding 0.2mm, alloys (760692);

Bumpers & parts thereof of the motor vehicles (870810);

Lubricating preparations (excl. Those for the treatment of textile and similar materials), not containing petroleum oils or oils obtained from bituminous minerals (340399);

Pistachios (80250):

Centrifugal pumps, n.e.s. In 8413, for liquids (841370);

Marble and travertine, merely cut, by sawing or otherwise, into blocks or slabs of a rectangular (including square) shape, having a specific gravity of 2.5 or more (251512):

Oranges, fresh/dried (80510);

Parts of Machines & mechanical appliances of those having individual functions (847990);

Parts and accessories for the musical instruments n.e.s. In heading no. 9209 and for other mechanical noise makers (920999);

Narrow woven fabrics n.e.s., of cotton (excl.

Labels, badges and similar articles) (580631);

Tube/pipe fittings of iron/steel, n.e.s,, other than stainless steel (730799);

Nickel, not alloyed, unwrought (750210);

Toilet/facial tissue stock, towel/napkin/similar articles, for household or sanitary use, cellulose, wadding webs of cellulose fibres (480300);

Machines and appliances for testing the hardness, strength, compressibility, elasticity or other mechanical properties of materials other than metals (902480);

Tableware & kitchenware, of plastic (392410); Slide fasteners, other than those fitted with chain scoops of base metal (960719);

Chickpeas (garbanzos), dried, shelled, whether or not skinned or split (71320);

Vermiculite, perlite and chlorites; unexpanded (253010);

Buttons of plastics, not covered with textile material (960621);

Copper waste & scrap (740400);

Synthetic/reconstructed precious/semi-precious stones, worked or graded or not, (but not strung, mounted or set), temporarily strung for transport convenience (710490);

Other articles of copper, n.e.s. (741999);

Knittedd/crocheted fabrics, n.e.s., of synthetic fibres, dyed (600632);

Cashew nuts, in shell (80131);

Dried leguminous vegetables, n.e.c., shelled, whether or not skinned or split (71390);

Synthetic organic colouring matter, dyes n.e.s. (320419);

Waste & scrap of cast iron (720410);

Other offset printing machinery (844313);

Padlocks, of base metal (830110);

Musical instruments (other than keyboard), the sound of which is produced or must be amplified electrically (920790); Artificial flowers, foliage & fruit, artificial, and parts thereof; articles made of artificial flowers, foliage or fruit, of materials other than plastics (670290); Beans of the species vigna mungo (L.) Hepper or vigna radiata (L.) Wilczek, shelled, whether or not skinned or split, dried (71331); Furniture of banboo/rattan (940381); Precious and semi-precious stones (excl. Diamonds, rubies, sapphires and emeralds), worked other than simply sawn or roughly shaped. not strung, mounted or set (710399); Castors of base metal (830220); Twine, cordage, ropes, cables; of polyethylene or polypropylene (excluding binder or baler twine), whether or not plaited, braided or rubber or plastic impregnated, coated, covered or sheathed (560749);Plants & parts of plants, including seeds and fruits, n.e.s., used primarily in perfumery, pharmacy or for insecticidal, fungicidal purposes; fresh or dried, whether or not cut, crushed or powdered (121190); Parts of lifts, skip hoist or escalators (843131); Coral & similar materials, unworked (50800); Woven fabrics of cotton, containing 85% or more by weight of cotton, printed, of weaves n.e.s., weighing not more than 200g/m2 (520859); Urea resins, in primary forms; thiourea resins, (390910): Cocoa paste, not defatted (180310); Raw hides & skins of bovine (including buffalo) and equine animals, fresh, salted or preserved, but not tanned, parchment dressed or further prepared, whether or not dehaired or split (410190); Containers for compressed/liquefied gas, of iron or steel (731100); Kidney beans, incl. White pea beans (phaseolus vulgaris), dried, shelled, whether or not skinned or split (71333); Active yeasts (210210); Ferrous waste & scrap, n.e.s. (720449); Small red (Adzuki) beans (Phaseolus (71332); Yarn of vegetable textile fibres, n.e.s. (530890); Precious (other than diamonds) and semi-precious stones, unworked or simply sawn or roughly shaped, not strung, mounted or set (710310) Other parts and accessories for printing machinery Other Parts suitable for use solely or

Malawi

excl. 844391 (844399);
Instruments and appliances used in medical or veterinary sciences, nes (901890);
Transistors with a dissipation rate >= 1 W (excl. Photosensitive transistors) (854129);
Parts of telephone sets, incl. Telephones for cellular networks for other wireless networks other apparatus for the transmission reception of voice, images other data, incl. Apparatus for

Other Parts suitable for use solely or principally with the apparatus of headings 8525 to 8528 (reception and transmission apparatus, monitors, etc.) (852990); Storage units of ADP (847170); Parts and accessories (excl. Pick-up cartridges) suitable for use solely or principally with sound reproducing and recording apparatus and with video equipment for recording and reproducing

communication in a wired wireless network (851770);

Electric accumulators n.e.s., including separators, whether or not rectangular (including square) (850780);

Aluminium waste & scrap (760200); Parts & accessories n.e.s. In Ch.90 (903300); Mixtures of odoriferous substance & and mixtures; of a kind used in the food or drink industries (330210):

Input or output units for automatic data processing machines, whether or not containing storage units in the same housing (excl. Keyboards) (847160); Springs & leaves for springs, n.e.s., of iron/steel (732090);

Microtomes and parts & accessories thereof (902790);

Portable automatic data processing (ADP) machines, weighing not more than 10kg, consisting of at least a central processing unit, a keyboard and a display (847130);

Processing units for automatic data processing machines, whether or not containing in the same housing one or two of the following types of unit storage units, input units, output units (847150); Other Electronic integrated circuits, other than Amplifiers Memories Processors and controllers (854239);

Polyethylene having a sp.gr. Of 0.94 or more (390120);

Refrigerating or freezing equipment; parts thereof, other than furniture (841899);

Waste & scrap of stainless steel (720421); Telephones for cellular networks or for other wireless networks (851712);

Chemical products&preparations of the chemical or allied industries, n.e.s. (382490);

Narrow woven fabrics (excluding pile fabrics), containing by weight 5% or more of elastomeric yarn or rubber thread (580620);

Sign-plates, name-plates, address-plates and similar plates, numbers, letters and other symbols; of base metal, excluding those of heading no. 9405 (831000);

Self-adhesive plates, sheets, film, foil, tape, strip, other flat shapes thereof, other than in rolls of a width not exceeding 20cm (391990);

Polypropylene, in primary forms (390210); Rubies, sapphires & emeralds, worked (other than simply sawn or roughly shaped), not strung, mounted or set (710391);

Recovered (waste & scrap) paper/paperboard, n.e.s., and of unsorted waste and scrap (470790); Refrigerating/freezing chests, cabinets, display counters, show-cases and the like, for storage and display, n.e.s. (841850);

Pens and pencils n.e.s.; duplicating stylos, penholders, pencil-holders, and similar holders, parts of the articles (960899);

pictures and sound (excl. Pick-up devices for grooved recording media) (852290); Static converters (850440);

Parts of Electrical transformers, static converters and inductors (850490); Parts of boring or sinking machinery (843143);

Portable automatic data processing (ADP) machines, weighing not more than 10kg, consisting of at least a central processing unit, a keyboard and a display (847130); Machines for the reception, conversion and transmission regeneration of voice, images other data, incl. Switching and routing apparatus (851762);

Plugs and sockets for a voltage <= 1000 V (excl. Lamp-holders) (853669); Knives having other than fixed blades, serrated or not, excluding knives and blades of heading no. 8208 (821193); Chemical products&preparations of the chemical or allied industries, n.e.s.

Wooden furniture (other than for office, kitchen or bedroom use), nesoi (940360); Electronic integrated circuits; processors and controllers, whether or not combined with memories, converters, logic circuits, amplifiers, clock and timing circuits, or other circuits (854231);

(382490):

Precious and semi-precious stones (excl. Diamonds, rubies, sapphires and emeralds), worked other than simply sawn or roughly shaped, not strung, mounted or set (710399);

Live animals, n.e.s. (10690); Aluminium alloys, unwrought (760120); Cashew nuts, in shell (80131); Digital Automatic Data Processing Machines And Units Thereof Presented In The Form Of Systems, N.e.s.o.i. (847149); Diodes (excl. Photosensitive or light emitting diodes) (854110); Other monitors, of a kind solely principally

Other monitors, of a kind solely principally used in an automatic data processing system (852851);

Other printers, copying machines, single-function printing, capable of connecting to an automatic data processing machine or to a network (844332);

Precious (other than diamonds) and semiprecious stones, unworked or simply sawn or roughly shaped, not strung, mounted or set (710310);

Projectors other than of a kind solely or principally used in an automatic data processing system (852861); Poly(ethylene terephthalate), in primary

Poly(ethylene terephthalate), in primary forms, having a viscosity of 78ml/g or higher (390760);

Electric ovens, cookers, cooking plates and boiling rings, electric grillers and roasters, for domestic use (excl. Space heating stoves and microwave ovens) (851660):

Parts and accessories of data processing equipment (847330);

Plants & parts of plants, including seeds and fruits, n.e.s., used primarily in perfumery, pharmacy or for insecticidal, fungicidal purposes; fresh or dried, whether or not cut, crushed or powdered (121190); Heterocyclic comps. Containing an unfused pyridine ring (whether or not hydrogenated) in the structure, other than pyridine and its salts (293339);

Ferrous waste & scrap, n.e.s. (720449);

Parts of the air-conditioning machines, with motor driven fan and elements for temperature control (841590);

Plates, sheets, film, foil&strip, not self-adhesive, of polymers of ethylene, non-cellular and not reinforced, laminated, supported or similarly combined with other materials (392010);

Check (nonreturn) valves, for pipes, boiler shells, tanks, vats or the like (848130);

Peas (Pisum sativum), dried, shelled, whether or not skinned or split (71310);

Precious (other than diamonds) and semi-precious stones, unworked or simply sawn or roughly shaped, not strung, mounted or set (710310); Oil seeds & oleaginous fruits, n.e.s., whether or not broken (120799):

Beans of the species vigna mungo (L.) Hepper or vigna radiata (L.) Wilczek, shelled, whether or not skinned or split, dried (71331):

Lentils, dried, shelled, whether/not skinned or split (71340);

Colouring matter&other preparations, n.e.s. (320649);

Kidney beans, incl. White pea beans (phaseolus vulgaris), dried, shelled, whether or not skinned or split (71333);

Illuminated signs, name plates and the like (940560);

Chickpeas (garbanzos), dried, shelled, whether or not skinned or split (71320);

Pictures, designs & photographs, n.e.s. (491191); Electric generating sets with spark-ignition internal combustion piston engines (850220); Motorcycles (including mopeds) and cycles, fitted with auxiliary motor, reciprocating internal combustion piston engine of cylinder capacity exceeding 500cc but not exceeding 800cc, with or without sidecars; side-cars (871140);

Beans n.e.s., dried/shelled, including seed (71339)

Transformers having a power handling capacity <= 1 kva (excl. Liquid dielectric transformers) (850431);

Pumps and compressors; parts, of air or vacuum pumps, air or other gas compressors and fans,

Other printers, copying machines, singlefunction printing, not capable of connecting to an automatic data processing machine or to a network (844339);

Radio navigational aid apparatus (852691); Wooden furniture of a kind used in bedroom (940350);

Chickpeas (garbanzos), dried, shelled, whether or not skinned or split (71320); Beans of the species vigna mungo (L.) Hepper or vigna radiata (L.) Wilczek, shelled, whether or not skinned or split, dried (71331);

Food preparations, n.e.s. (210690): Cotton, not carded/combed (520100); Kidney beans, incl. White pea beans (phaseolus vulgaris), dried, shelled, whether or not skinned or split (71333); Line telephone sets with cordless handsets

(851711);Statuettes & other ornaments, of wood (442010)

Parts of machinery, plant/laboratory equipment, for treating materials by a process involving a change of temperature

(841990);

Plugs and sockets for a voltage <= 1000 V (excl. Lamp-holders) (853669);

Mali

236 | Page

ventilating or recycling hoods incorporating a fan (841490);

Insulated electric conductors for a voltage not exceeding 1000 volts, not fitted with connectors (854449);

Lamp-holders, for a voltage not >10 (853661); Ferrous waste & scrap, n.e.s. (720449);

Gaskets, washers & other seals of vulcanised rubber, of non-cellular rubber (401693); Static converters (850440);

Electrical apparatus; n.e.c., for switching or protecting electrical circuits, for a voltage not exceeding 1000 volts (853690);

Portable automatic data processing (ADP) machines, weighing not more than 10kg, consisting of at least a central processing unit, a keyboard and a display (847130);

Machines for the reception, conversion and transmission regeneration of voice, images other data, incl. Switching and routing apparatus (851762);

New pneumatic tyres, of rubber, of a kind used on motor cars (including station wagons and racing cars) (401110);

Ethylene polymers; in primary forms, polyethylene having a specific gravity of less than 0.94 (390110);

Commodities not specified (999999); Articles for the conveyance/packing of goods, n.e.s. (392390);

Parts of the pumps of 8413.11-8413. (841391); Table/floor/wall/window/ceiling/roof fans, with a self-contained electric motor of an output not exceeding 125W (841451);

Parts of the machinery, used in the industrial preparation or manufacture of food or drink (843890);

Parts of telephone sets, incl. Telephones for cellular networks for other wireless networks other apparatus for the transmission reception of voice, images other data, incl. Apparatus for communication in a wired wireless network (851770);

Fixed electrical capacitors (excl. Tantalum, aluminium electrolytic, ceramic, paper, plastic and power capacitors) (853229);

Electric accumulators, i.e. Lead-acid, (other than for starting piston engines), including separators, whether or not rectangular (including square) (850720);

Hotographic goods; chemical preparations other than sensitised emulsions, put up in measured portions or put up for retail sale in a form ready for use (370790);

Titanium oxides (282300):

Cathode-ray television picture tubes, including video monitor cathode-ray tubes, colour (854011); Other footwear, other than with uppers of lether/composition leather/textile materils (640590);

Portable automatic data processing (ADP) machines, weighing not more than 10kg, consisting of at least a central processing unit, a keyboard and a display (847130); Ferrous waste & scrap, n.e.s. (720449); Waste and scrap of primary cells, primary batteries and electric accumulators; spent primary cells, spent primary batteries and spent electric accumulators (854810); Cashew nuts, in shell (80131);

Lac; Natural gums, Gum Resins, Resins And Balsams, n.e.s. (130190)

Textile fabrics coated with gum/amylaceous substances, of a kind used for the outer covers of books or the like (590110); Soap in other forms, n.e.s. (340120); Cashew nuts, in shell (80131); Lac; natural gums, gum resins, resins and balsams, n.e.s. (130190) Mauritania Electrical apparatus for switching or protecting Interchangeable tools for hand tool, whether electrical circuits, or for making connections to or or not power-operated, for screw-driving or in electrical circuits, for a voltage > 1.000 V (excl. uses n.e.c (820790); Fuses, automatic circuit breakers, isolating Parts and accessories for instruments. switches, make-and-break switches, lightning appliances and machines measuring or checking (903190); arresters, voltage limiters, surge suppressors and Parts of taps, cocks, valves and similar control desks, cabinets, panels etc. Of heading 8537) (853590); appliances (848190); Drilling machines for drilling by removing metal, Raw hides & skins of bovine (including other than numerically controlled (845929); buffalo) and equine animals, fresh, salted or Chemical products&preparations of the chemical preserved, but not tanned, parchment or allied industries, n.e.s. (382490); dressed or further prepared, whether or not Waste & scrap of stainless steel (720421); dehaired or split (410190); Parts and accessories of data processing equipment Ferrous turnings, shavings, chips, milling (847330); waste, sawdust, fillings, trimmings and Catheters, cannulae & the like (901839); stampings, whether or not in bundles (720441);Bars & rods, hot-rolled, in irregularly wound coils, n.e.s. (722790); Aluminium waste & scrap (760200); Lead waste & scrap (780200); Ferrous waste & scrap, n.e.s. (720449); Ferrous waste & scrap, n.e.s. (720449); Copper waste & scrap (740400) Copper waste & scrap (740400); Tanned/crust skins of sheep/lambs, without wool on, whether or not split, but not further prepared, in the wet state (including wet-blue) (410510); Aluminium waste & scrap (760200): Whole bovine (incl. Buffalo)/equine animals, of a weight per skin not exceeding 8kg when simply dried, 10kg when dry-salted or 16kg when fresh, wet-salted or otherwise preserved (410120); Tanned/crust hides & skins of goats or kids, without wool or hair on, whether or not split, but not further prepared, in the wet state (including wet blue) (410621); Raw skins of sheep/lambs, fresh, or salted, dried, limed, pickled or otherwise preserved, (but not tanned, parchment-dressed or further prepared), with wool on (410210); Raw skins of sheep/lambs (fresh/salted/dried/limed/otherwise, preserved other than by pickling, (but not tanned, parchmentdressed or further preserved), without wool on (410229)Limestone flux; limestone and other calcareous Mozambique Plates, sheets, film, foil & strip, of stone, of a kind used for the manufacture of lime polyesters n.e.s., non-cellular and not reinforced, laminated, supported or or cement (252100): Articles of vulcanised rubber other than hard similarly combined with other materials rubber, n.e.s. (401699); (392069): Parts of the furniture of 94.03 (metal/wooden Waste & scrap of stainless steel (720421); furniture etc. For office, kitchen uses, etc.) Iron or steel tubes, pipes and hollow (940390); profiles, seamless, n.e.s. (730490); Organic surface-active agents, surface-active, Lubricating preparations (excl. those for the treatment of textile and similar materials), whether or not containing soap, including auxiliary washing preparations, not for retail sale (340290); not containing petroleum oils or oils

Parts of machines and apparatus for soldering or welding or for hot spraying of metals, metal carbides or cermets, n.e.s. (851590);

Other parts & accessories for the motor vehicles (870899);

Optical devices, appliances and instruments; n.e.c.(including liquid crystal devices) (901380); Parts suitable for use solely or principally with electric motors and generators, electric generating sets and rotary converters, n.e.s. (850300); Pumps and compressor for air/vaccum/other gas (841480);

Parts and accessories nesoi of bicycles etc. (871499);

Fans, other than

table/floor/wall/window/ceiling/roof (841459); Static converters (850440);

Grinding machines other than flat-surface), in which positioning in any one axis can be set up to at least an accuracy of 0.01mm, other than numerically controlled (846029);

Gear cutting, gear grinding or gear finishing machines, working by removing metal, sintered metal carbides or cermets (846140);

Parts of Machinery for working rubber or plastics (847790);

Silica sands & quartz sands, whether or not coloured (250510);

Plates, sticks, tips & the like for tools, unmounted, of sintered metal carbides or cermets (820900); Insulated electric conductors for a voltage not exceeding 1000 volts, not fitted with connectors (854449):

Wooden furniture (other than for office, kitchen or bedroom use), nesoi (940360);

Flat-surface grinding machines, in which positioning in any one axis can be set up to an accuracy of 0.01mm or better, other than numerically controlled (846019);

Wooden furniture of a kind used in office (940330);

Parts of the machinery used for cleaning, sorting or grading seed, grain or dried leguminous vegetables or for milling or working cereals or dried leguminous vegetables (not farm-type machinery) (843790);

Calcium carbonate (283650);

Rivets of iron/steel (731823);

Permanent magnets of metal and articles intended to become permanent magnets after magnetization (excl. Chucks, clamps and similar holding devices) (850511);

Recovered (waste & scrap) paper/paperboard, n.e.s., and of unsorted waste and scrap (470790); Millstones, grindstones, grinding wheels and the like; of other agglomerated abrasives or of ceramics (680422);

Tube/pipe fittings of iron/steel, n.e.s,, other than stainless steel (730799);

Parts & accessories n.e.s. in Ch.90 (903300);

obtained from bituminous minerals (340399);

Parts of metal-rolling mills, other rolls (845590);

Lentils, dried, shelled, whether/not skinned or split (71340);

Zinc waste & scrap (790200);

Kidney beans, incl. White pea beans (phaseolus vulgaris), dried, shelled, whether or not skinned or split (71333);

Activated carbon (380210);

Oil-cake and other solid residues; whether or not ground or in the form of pellets, resulting from the extraction of coconut or copra seed oils (230650);

Boring/sinking machinery, self-propelled, n.e.s. (843041);

Flat-rolled products of silicon-electrical steel, flat-rolled, width less than 600mm, grain-oriented, (722611);

Vegetable saps and extracts; n.e.s. In item no. 1302.1 (130219);

Parts of boring or sinking machinery (843143);

Lead waste & scrap (780200);

Aluminium waste & scrap (760200);

Worked monumental/building stone of granite; articles thereof, simply cut or sawn, with a flat or even surface (680223);

Natural calcium phosphates, natural (251020);

Beans n.e.s., dried/shelled, including seed (71339);

Wood charcoal (including shell or nut charcoal), whether or not agglomerated, n.e.s. (440290);

Ferrous waste & scrap, n.e.s. (720449); Coral & similar materials, unworked (50800);

Ferro-alloys n.e.s. (720299);

Waste & scrap of cast iron (720410); Sodium hexafluoroaluminate (synthet (282630);

Sunflower seeds, whether/not broken (120600);

Beans of the species vigna mungo (L.) Hepper or vigna radiata (L.) Wilczek, shelled, whether or not skinned or split, dried (71331):

Precious (other than diamonds) and semiprecious stones, unworked or simply sawn or roughly shaped, not strung, mounted or set (710310):

Crude mica & mica rifted into sheets or splittings (252510);

Waste and scrap of primary cells, primary batteries and electric accumulators; spent primary cells, spent primary batteries and spent electric accumulators (854810); Soya beans, whether/not broken (120100)

Organo-sulphur compounds, n.e.s. (293090); Tricycles, scooters, pedal cars and similar wheeled toys; dolls' carriages; dolls; other toys; reducedsize (scale) models and similar recreational models, working or not; puzzles of all kinds

(950300);

Polyethylene having a sp.gr. Of 0.94 or more (390120);

Silicones, in primary forms (391000);

Parts suitable for use solely or principally with the apparatus of heading 8535, 8536 or 8537, n.e.s. (excl. Boards, panels, consoles, desks, cabinets and other bases for the goods of heading 8537, not equipped with their apparatus) (853890);

Statuettes & other ornamental articles (392640); Furniture of cane, osier, or similar materials (other than bamboo or rattan) (940389);

Other friction material & articles thereof (e.g. Sheets, rolls, strips, segments, discs, washers, pads) not mounted; for clutches or the like (not brake linings and pads), with a basis of mineral substances or cellulose (other than asbestos) (681389);

Aluminium waste & scrap (760200);

Clutches and shaft couplings (including universal joints) (848360);

Twine, cordage, ropes & cables of materials n.e.s., whether or not plaited, braided or impregnated, coated, covered, or sheathed with rubber or plastics (560790);

Iron or steel threaded articles except screw, nut, bolts (731819);

Weighing machine weights of all kinds; parts of weighing machinery (842390);

Unwrought lead, refined (780110);

Pepper (genus Piper), neither crushed nor ground (90411);

Dates, fresh/dried (80410);

Wood sawn/chipped length wise, sliced or peeled, whether or not planed, sanded or end-jointed, of a thickness exceeding 6mm (440710);

Iron or steel (excluding cast iron or stainless steel); other seamless, casing and tubing, of a kind used in drilling for oil or gas (730429);

Printed books, brochures, leaflets, in single sheets, whether or not folded (490110);

Articles for funfair, table or parlour games, including pintables, special tables for casino games, automatic bowling alley equipment, n.e.c. (950490);

Chickpeas (garbanzos), dried, shelled, whether or not skinned or split (71320);

Ferrous turnings, shavings, chips, milling waste, sawdust, fillings, trimmings and stampings, whether or not in bundles (720441):

Knives & cutting blades, for machines or mechanical appliances, n.e.s. (820890);

Methanol (methyl alcohol) (290511);

Flat-rolled products of iron/non-alloy steel, in coils, without patterns in relief, of a width 600mm

or more, hot-rolled, pickled, of a thickness of 4.75mm or more (720825);

Topical wood, n.e.s., in the rough, whether or not stripped of bark or sapwood, or roughly squared, untreated (440349);

Clutches & parts thereof for the motor vehicles (870893);

Pulley tackle & hoists other than skip hoists or hoists of a kind used for raising vehicles, powered by an electric motor (842511);

Peas (Pisum sativum), dried, shelled, whether or not skinned or split (71310);

Handbags, whether/not with shoulder strap and including those without handle, with outer surface of plastic sheeting or of textile materials (420222); Yarn other than high tenacity/textured, filament, monofilament (less than 67 decitex), elastomeric, single, untwisted or twisted 50 turns or less per metre, not for retail sale, not sewing thread (540244);

Parts of Electrical transformers, static converters and inductors (850490);

Mechanical appliances, for projecting, dispersing or spraying liquids or powders, for agricultural or horticultural use, whether or not hand-operated (842481);

Orthopaedic/fracture appliances (902110); Boring/sinking machinery, self-propelled, n.e.s. (843041);

Floor coverings & mats of vulcanises rubber, non-cellular rubber (401691);

Bicycles and other cycles; including delivery tricycles, not motorised (871200);

Builders' ware of plastics, n.e.s. (392590); Carbonaceous pastes for electrodes and similar pastes for furnace linings (380130);

Winding wire, other than of copper (854419); Bending, folding, straightening or flattening machines (including presses), (other than numerically controlled), for working metal (846229);

Other tubes, pipes and hollow profiles (for example, open seam or welded, riveted or similarly closed), of iron or steel (730690); Cored wire of base metal, for electric arc-welding (831120);

Raw hides & skins of bovine (including buffalo) and equine animals, fresh, salted or preserved, but not tanned, parchment dressed or further prepared, whether or not dehaired or split (410190); Tubular metal needles & needles for sutures (901832):

Lentils, dried, shelled, whether/not skinned or split (71340);

Nails, tacks, drawing pins, corrugated nails, staples (of iron and steel) and the like, with heads of other material or not, but excluding articles with heads of copper (731700);

Precious (other than diamonds) and semi-precious stones, unworked or simply sawn or roughly shaped, not strung, mounted or set (710310); Kidney beans, incl. White pea beans (phaseolus vulgaris), dried, shelled, whether or not skinned or split (71333);

Bars & rods of iron/non-alloy steel, n.e.s., cold-formed or cold-finished, (not free-cutting) (721550);

Unwrought lead other than refined, not containing by weight antimony as the principal other element (780199);

Ferrous waste & scrap, n.e.s. (720449); Rails of iron/steel (730210):

Wood, n.e.s., sawn/chipped lengthwise/sliced lengthwise/sliced/peeled, whether or not planed, sanded or end-jointed, of a thickness exceeding 6mm (440729);

Coal; (other than anthracite and bituminous), whether or not pulverised but not agglomerated (270119):

Waste & scrap of cast iron (720410)

Niger

Mixtures of odoriferous substance & and mixtures; of a kind used in the food or drink industries (330210);

Electrical apparatus like switches n.e.c. In heading no. 8536, for a voltage not exceeding 1000 volts (853650);

Parts of aeroplanes/helicopters, others (880330); Static converters (850440);

Other Parts suitable for use solely or principally with the apparatus of headings 8525 to 8528 (reception and transmission apparatus, monitors, etc.) (852990);

Machines for the reception, conversion and transmission regeneration of voice, images other data, incl. Switching and routing apparatus (851762);

Parts of the pumps of 8413.11-8413. (841391); Parts of Machines & mechanical appliances of those having individual functions (847990); Electrical measurement instruments nes., without a recording device (903089);

Fuel/lubricating/cooling medium pumps for internal combustion piston engines (841330); Automatic circuit breakers for a voltage <= 1.000 V (853620);

Articles of plastics, n.e.s. (392690); Other articles of copper, n.e.s. (741999); Fixed electrical capacitors (excl. Tantalum, aluminium electrolytic, ceramic, paper, plastic and power capacitors) (853229);

Inductors (excl. Inductors for discharge lamps or tubes) (850450);

Parts of the machinery, used in the industrial preparation or manufacture of food or drink (843890);

Plates, sheets, film, foil&strip, not self-adhesive, of polymers of ethylene, non-cellular and not

Self-adhesive plates, sheets, film, foil, tape, strip, other flat shapes thereof, other than in rolls of a width not exceeding 20cm (391990);

Parts suitable for use solely or principally with electric motors and generators, electric generating sets and rotary converters, n.e.s. (850300);

Iron or steel threaded articles except screw, nut,bolts (731819);

Articles of plastics, n.e.s. (392690); Electronic integrated circuits; processors and controllers, whether or not combined with memories, converters, logic circuits, amplifiers, clock and timing circuits, or other circuits (854231);

Cartons, boxes & cases, of corrugated paper or paperboard (481910);

Gears & gearing (not toothed wheels, chain sprockets and other transmission elements presented separately); ball or roller screws; gear boxes and other speed changers, including torque converters (848340); Gaskets and similar joints of metal sheeting combined with other material or two or more layers of metal (848410); Apparatus based on use of x-rays and similar parts and accessories (x-ray)

similar; parts and accessories (x-ray generators, tubes, high tension generators, control panels and desks, screens, examination or treatment tables, chairs and like (902290);

Electric lamps & lighting fittings, nec (940540);

Engines parts for internal combustion piston engines (excluding spark-ignition) (840999);

reinforced, laminated, supported or similarly Copper tube/pipe fittings (e.g. Couplings, combined with other materials (392010); elbows, sleeves) of copper alloys (741220); Speed indicators & tachometers; stroboscopes Paper/paperboard labels of all kind, printed (902920): (482110): Instruments and appliances used in medical or Paper/paperboard labels of all kind, veterinary sciences, nes (901890); unprinted (482190); Aluminium waste & scrap (760200); Surveying equipment; articles n.e.s., Paper, paperboard, cellulose wadding or webs of including hydrographic, oceanographic, cellulose fibres (482390); hydrological, meteorological or geophysical Polyethylene having a sp.gr. Of 0.94 or more instruments and appliances (excluding compasses) (901580) (390120): Commodities not specified (999999) Aluminium foil, not backed, rolled (but not further Halogenated derivatives or adrenal cortical Nigeria hormones (293722); worked), of a thickness not exceeding 0.2mm (760711): Iron ores and concentrates: non-Parts of the h-hold./laundry-type washing machine agglomerated (260111); Clock movements, whether or not complete, (845090); incomplete, assembled or unassembled Sanitary ware & parts thereof, of copper (911090);(741820);Antisera & other blood fractions & human and Parts & accessories of the machines or animal blood (300210); auxiliary machinery of heading no. 8444 Knives having other than fixed blades, serrated or (844820);not, excluding knives and blades of heading no. Drilling machines for drilling by removing metal, other than numerically controlled 8208 (821193); Parts & accessories of the machines or auxiliary (845929); machinery for preparing textile fibres, other than Cans to be closed by soldering/crimping, for card clothing (844832); any material (excluding compressed or Pigments & preparations based thereon (320417); liquefied gas), less than 50l capacity, of iron Parts of locks, of base metal (830160); or steel (731021): Dish washing machines other than household type Leather further prepared after tanning or crusting, including parchment-dressed, of Anti-oxidising preparations and other compound bovine (including buffalo) or equine stabilisers; for rubber or plastics, n.e.s. Or included animals, without hair on, split or not, (other than grain splits and full grains, unsplit) (381230): Textile fabrics; impregnated, coated, covered or (410719): laminated with polyurethane (590320); Machines for agglomerating, shaping or Coated rods & cored wire, of base metal, for moulding solid mineral fuels, ceramic paste, unhardened cements, plastering materials in soldering, brazing or welding by flame (831130); Grinding machines other than flat-surface), in powder or paste form, machines for forming which positioning in any one axis can be set up to foundry moulds of sand (847480); at least an accuracy of 0.01mm, other than Yarn other than sewing thread, of artificial numerically controlled (846029); staple fibres, mixed mainly or solely with Ice skates & roller skates, including skating boots fibres (other than wool, fine animal hair or with skates attached (950670); cotton), not put up for retail sale (551090); Loudspeakers; not mounted in their enclosures Ferrous waste & scrap, n.e.s. (720449); (851829); Polymers of propylene/of other olefin Seats of a kind used for motor vehicles (940120); polymers, n.e.s. (390290); Isocyanates (292910); Sanitary ware & parts thereof, of copper Jams, fruit jellies, marmalades, purees and pastes; (741820);of fruit or nuts, n.e.s. Cooked preparations Rolls for rolling mills (845530); (excluding homogenised), whether or not Other printers, copying machines, singlecontaining added sugar or other sweetening matter function printing, not capable of connecting (200799): to an automatic data processing machine or Wheel rims & spokes for bicycles (871492); to a network (844339); Copper wire, of refined copper, of which Lighting sets of a kind used for Christmas trees (940530): the maximum cross-sectional dimension

Plates, sheets, film, foil & strip, of plastics n.e.s.,

non-cellular and not reinforced, laminated,

supported or similarly combined with other

materials (392099);

exceeds 6mm (740811);

Tanned/crust hides & skins of bovine or

equine animals, in the wet state (including

Non-threaded articles, n.e.s., of iron/steel (731829);

Parts of metal-rolling mills, other rolls (845590); Articulated link chain (other than roller), of iron or steel (731512);

Electric table, desk, bedside/floor (940520); Sewing needles, knitting needles, bodkins, crochet hooks and similar articles, for use in the hand, n.e.s., of iron/steel (731990);

Outer soles & heels, of rubber/plastics (640620); Tricycles, scooters, pedal cars and similar wheeled toys; dolls' carriages; dolls; other toys; reduced-size (scale) models and similar recreational models, working or not; puzzles of all kinds (950300):

Furniture of cane, osier, or similar materials (other than bamboo or rattan) (940389);

Industrial monocarboxylic fatty acids; acid oils from refining; (other than stearic acid, oleic acid or tall oil fatty acids) (382319);

Hat-forms, hat bodies & hoods of felt, neither blocked to shape nor with made brims; plateaux and manchons (including slit manchons) of felt (650100);

Artificial flowers, foliage & fruit, artificial, parts thereof and their plastic articles (670210);

Articles of apparel & clothing accessories (other than gloves, mittens and mitts), of vulcanized rubber (401590);

Parts of microphones, loudspeakers, headphones, earphones or audio-frequency electric amplifiers, n.e.s. (851890);

Drive-axles with differential, whether or not provided with other transmission components, and non-driving axles; parts thereof (870850);

Glass fibres (incl. Glass wool) & articles thereof, n.e.s. (701990);

Machine-tools for deburring, polishing or otherwise finishing metal, sintered metal carbides or cermets by means of grinding stones, abrasives or polishing products, n.e.s. (846090);

Sawing/cutting-off machines working by removing metal, sintered metal carbides or cermets (846150);

Peas (Pisum sativum), dried, shelled, whether or not skinned or split (71310);

Apples, fresh (80810);

Newsprint, in rolls/sheets (480100);

Electric discharge lamps (excluding ultra-violet), fluorescent, hot cathode (853931);

Permanent magnets and articles intended to

become permanent magnets after magnetization, of materials other than metal (850519);

Ceramic tableware, kitchenware, household articles and toilet articles, other than of porcelain or china (691200);

Plywood, veneered panels and similar laminated wood (other than blockboard, laminboard and battenboard, other than of bamboo, and other than

wet blue), without hair on, excluding full grains, unsplit; grain splits (410419); Parts of calendering/other rolling machines, for other than metal or glass (842099); Wood sawn/chipped length wise, sliced or peeled, whether or not planed, sanded or end-jointed, of a thickness exceeding 6mm (440710);

Tools for working in the hand, pneumatic, other than rotary type (846719); Aluminium oxide (excl. artificial

corundum) (281820); Offset printing machinery, sheet-fed, office type (sheet size not exceeding 22 x 36cm in the unfolded state) (844312);

Cathodes & sections of cathodes, of refined copper, unwrought (740311);

Parts of the machinery of 84.41, for making up paper pulp, paper or paperboard, including cutting machines of all kinds (844190);

Preparations based on graphite/other carbon based preparations, in the form of pastes, blocks, plates or other semi-manufactures (380190);

Parts of the vehicles of 87.16 (Trailers, semi-trailers and other vehicles not mechanically propelled) (871690); Polymers of ethylene, in primary forms, n.e.s. (390190);

Bars, rods & profiles, of refined copper (740710):

Flexographic printing machinery (844316); Ethylene polymers; in primary forms, polyethylene having a specific gravity of less than 0.94 (390110);

Refractory ceramic goods, containing by weight more than 50% of graphite or other forms of carbon or of a mixture of these products, excluding those of siliceous fossil meals or similar earths (690310);

Powders of non-lamellar structure, of aluminum (760310);

Parts of the book-binding machinery, including book-sewing machines, parts thereof (844090);

Natural rubber (excl. Latex), in smoked sheets (400121);

Zinc oxide; zinc peroxide (281700);

Machinery for the preparation of fruits, nuts or vegetables (843860);

Machines for mixing or kneading mineral substances, excluding concrete mixers and machines for mixing mineral substances with bitumen (847439):

Vegetable saps and extracts; n.e.s. In item no. 1302.1 (130219);

Cut flowers & flower buds of a kind suitable for bouquets or ornamental

plywood consisting only of sheets of wood each ply 6mm or thinner) (441299);

Chemical products & preparations of the chemical or allied industries, n.e.s. (382490);

Pen nibs & nib points (960891);

Hand-/foot-operated air pumps (841420);

Fixed electrical resistors for a power handling capacity > 20 W (excl. Heating resistors) (853329);

Almonds, in shell (80211);

Springs & leaves for springs, n.e.s., of iron/steel (732090);

Sports footwear; tennis shoes, basketball shoes, gym shoes, training shoes and the like, with outer soles of rubber or plastics and uppers of textile materials (640411):

Parts of railway/tramway rolling stock n.e.c. (860799);

Rear-view mirrors for vehicles (700910);

Statuettes & other ornamental articles (392640); Other centrifuges, incl. Centrifugal dryers (but not clothes-dryers) (842119);

Knitting Machines, Others, And Machines For Making Gimped Yarn, Tulle, Lace, Embroidery, Trimmings, Braid Or Net And Machines For Tufting (844790);

Hydraulic brake fluids & other prepared liquids for hydraulic transmission, not containing or containing less than 70% by weight of petroleum oils or oils obtained from bituminous minerals (381900):

Bedspreads other than knittedd/crocheted (630419);

Parts & accessories of bodies (other than safety seat belts) (870829):

Organic surface-active agents, surface-active, whether or not containing soap, including auxiliary washing preparations, not for retail sale (340290); Plasters (consisting of calcined gypsum or calcium sulphate), whether or not coloured, with or without small quantities of accelerators or retarders (252020);

Buttons, other than of base metal or plastics, not covered with textile material (960629);

Parts of the air-conditioning machines, with motor driven fan and elements for temperature control (841590);

Polyurethanes, in primary forms (390950); Bars & rods, hot-rolled, in irregularly wound coils, containing indentations, ribs, grooves or other deformations produced during the rolling process

Laminated safety glass, of size & shape suitable for incorporation in vehicles, aircraft, spacecraft or vessels (700721):

Wooden furniture of a kind used in office (940330):

Threaded nuts of iron/steel (731816);

Machines which perform two or more of the functions of printing, copying or facsimile

purposes, fresh, other than roses, carnations, orchids, chrysanthemums or lillies (60319); Precious (other than diamonds) and semiprecious stones, unworked or simply sawn or roughly shaped, not strung, mounted or set (710310);

Cotton, not carded/combed (520100); Tanned/crust hides & skins, n.e.s., (410691);

Machines for working wire, without removing material (846330);

Dried leguminous vegetables, n.e.c., shelled, whether or not skinned or split (71390):

Aluminium, not alloyed, unwrought (760110):

Unwrought products of refined copper, n.e.s. (740319);

Semi-finished products of iron/non-alloy steel, containing by weight less than 0.25% of carbon, other than rectangular or square cross-section (720719);

Tanned/crust hides & skins of bovine or equine animals, in the dry state (crust), without hair on, (other than full grains, unsplit; grain splits) (410449); Aluminium waste & scrap (760200);

Unwrought lead other than refined, not containing by weight antimony as the principal other element (780199);

Zinc waste & scrap (790200)

transmission, capable of connecting to an automatic data processing machine or to a network (844331);

Parts of Electrical transformers, static converters and inductors (850490);

Hard rubber (e.g., ebonite) in all forms, including waste and scrap, and articles of hard rubber (401700);

Dichloromethane (methylene chloride) (290312); Other printers, copying machines, single-function printing, not capable of connecting to an automatic data processing machine or to a network (844339); Vinyl acetate copolymers other than Poly (vinyl alcohol), in primary forms (390529);

Other articles of copper, n.e.s. (741999);

Poly(vinyl chloride), plasticised, in primary forms, mixed with other substances (390422);

Casing & tubing of a kind used in drilling for oil and gas (other than stainless steel or seamless), n.e.c., (730629);

Other Machine-tools for working stone, ceramics, concrete, asbestos-cement or like mineral materials or for cold working glass, (other than sawing, grinding or polishing machines) (846490); Textile winding (incl. Weft-winding), or reeling

machines (844540); Natural/artificial abrasive powder/grain, on a base

Natural/artificial abrasive powder/grain, on a base of paper or paperboard only, whether or not cut to shape or sewn or otherwise made up (680520); Glazed ceramic flags & paving/hearth/wall tiles (690890);

Acrylic polymers (other than polymethyl methacrylate), in primary forms (390690); Knives & cutting blades, for machines or for mechanical appliances, for metal working (820810);

Hand saws (820210);

Instruments and apparatus for physical or chemical analysis, for measuring or checking viscosity, porosity, expansion, surface tension or quantities of heat, sound or light, n.e.c. (902780); Clasps, frames with clasps, buckles, buckle clasps; of base metal, of a kind used for clothing, footwear, awnings, handbags, travel goods or other made up articles, including parts (830890); Unglazed ceramic flags & paving/hearth/wall tiles (690790);

Wooden furniture of a kind used in bedroom (940350);

Handkerchiefs, of textile materials other than cotton (not knittedd or crocheted) (621390); Recovered (waste & scrap) paper/paperboard, n.e.s., and of unsorted waste and scrap (470790); Parts of the filtering/purifying machinery for filtering or purifying liquids or gases, n.e.s. (842199);

"Time switches with clock, watch movement or synchronous motor (910700);"

Flat-rolled products of stainless steel, width 600mm or more, n.e.s. (721990);

Plates, sheets, film, foil & strip, of polymers of vinyl chloride, cellular (392112);

Safety pins of iron/steel (731920);

Compounded rubber (other than with carbon black or silica), vulcanized, in plates, sheets and strip (400591);

Parts of the furniture of 94.03 (metal/wooden furniture etc. For office, kitchen uses, etc.) (940390);

Liquid dielectric transformers, having a power handling capacity > 650 kva but <= 10000 kva (850422):

Microwave ovens (851650);

Doors, windows & their frames & thresholds for doors (392520);

Textile fabrics; impregnated, coated, covered or laminated with plastics, (excluding polyvinyl chloride, polyurethane, polyamides polyesters) (590390);

Other parts, trimmings & accessories of umbrellas and walking sticks n.e.c. (660390);

Carbon; carbon blacks and other forms of carbon n.e.s. (280300);

Millstones, grindstones, grinding wheels and the like; of other agglomerated abrasives or of ceramics (680422);

Copper wire, of refined copper, of which the maximum cross-sectional dimension exceeds 6mm (740811);

Wool, not carded/combed, degreased, (not carbonised), (other than shorn) (510129);

Bidets, lavatory pans, flushing cisterns and similar sanitary ware, n.e.s. (392290);

Parts of machinery for making/finishing paper or paperboard (843999);

Parts of the seats of 94.01 (seats in vehicles, aircrafts, convertible to bed, etc.) (940190);

Finishing agents, dye carriers to accelerate dyeing or fixing of dye-stuffs, other products and preparations, used in the textile industry (excluding those with a basis of amylaceous substances) (380991);

Refractory cements, mortars, concretes and similar composition (381600);

Plaited bands, slings & the like, of iron/steel, not electrically insulated (731290);

Carpets and other textile floor coverings; tufted, of man-made textile materials (other than polyamides), whether or not made up (570330); Unglazed ceramic tiles, cubes and similar articles, whether or not rectangular, the largest surface area of which is capable of being enclosed in a square with side<7 cm (690710):

Lubricating preparations (other than for the treatment of textile and similar materials) containing containing less than 70% (by weight) of petroleum oils or oils obtained from bituminous minerals (340319);

Styrene-butadiene rubber (SBR), and carboxylated styrene-butadiene rubber (XSBR), (other than latex), synthetic, in primary forms or in plates, sheets or strip (400219);

Articles of aluminium n.e.s. (761699);

Copper tubes & pipes, of refined copper (741110); Epoxide resins, in primary forms (390730);

Textile fabrics; otherwise impregnated, coated or covered, painted canvas being theatrical scenery, studio back-cloths or the like (590700);

Industrial fatty alcohols (382370);

Electric accumulators, i.e. Lead-acid, (other than for starting piston engines), including separators, whether or not rectangular (including square) (850720);

Mechanical appliances, for projecting, dispersing or spraying liquids or powders, for agricultural or horticultural use, whether or not hand-operated (842481);

Ceramic articles n.e.s., other than of porcelain or china (691490);

Other footwear, other than with uppers of leather/composition leather/textile materials (640590);

Used/new rags, scrap twine, cordage, rope and cables and worn out articles of twine, cordage, rope or cables, of textile materials; other than sorted (631090);

Paper & paperboard of a kind used for writing, printing or other graphic purposes, in rolls or sheets, coated with kaolin or other inorganic substances only, having more than 10% of mechanically processed fibres, (excluding lightweight paper) (481029):

Doors, windows & their frames and thresholds for doors of iron/steel (730830);

Pepper (genus Piper), neither crushed nor ground (90411);

Whiskies (220830);

Natural rubber (excl. Latex), in smoked sheets (400121);

Combined refrigerator-freezers, fitted with separate external doors, electric or other (841810); Rodenticides &other biocides, n.e.s. (380899); Artificial filament yarn (other than sewing thread), single, of viscose rayon, (including artificial monofilament of less than 67 decitex), untwisted or with a twist not exceeding 120 turns per metre, not for retail sale, not textured, not high tenacity (540331);

Other footwear without outer soles (640399); Bars & rods of stainless steel, n.e.s. (722230); Refractory ceramic goods, containing by weight more than 50% of graphite or other forms of carbon or of a mixture of these products, excluding those of siliceous fossil meals or similar earths (690310):

Ferrous waste & scrap, n.e.s. (720449); Bodies (incl. Cabs), for the motor vehicles (870710); Waste, parings & scrap, of polymers, vinyl chloride (391530);

Beans n.e.s., dried/shelled, including seed (71339); Wood (including strips & friezes for parquet flooring), continuously shaped along any edges, ends or faces, whether or not planed, sanded or end-jointed (440910);

Other printers, copying machines, single-function printing, capable of connecting to an automatic data processing machine or to a network (844332); Frames & forks, & parts thereof (cycles) (871491); Felt, of textile materials other than wool or fine animal hair (excluding needle loom felt and stitchbonded fibre fabrics), not impregnated, coated, covered or laminated (560229);

Prepared additives for cements/mortars/concretes (382440);

Textile spinning machines (844520);

Bars, rods & profiles, of refined copper (740710); Hairpins, curling pins, curling grips, hair curlers and the like and parts thereof (961590); Flat-rolled products of stainless steel, width less than 600mm, nes (722090);

Wood sawn/chipped length wise, sliced or peeled, whether or not planed, sanded or end-jointed, of a thickness exceeding 6mm (440710);

Electro-thermic appliances, for domestic use (excl. Hair-dressing appliances and hand dryers, space heating and soil heating apparatus, water heaters, immersion heaters, smoothing irons, microwave ovens, ovens, cookers, cooking plates, boiling rings, grillers, roasters, coffee makers, tea makers and toasters) (851679);

Paper, paperboard, cellulose wadding and webs of soft cellulose fibres; coated, impregnated, covered, surface-decorated or surface-coloured, n.e.s. (481190):

Articles of wood n.e.s. (442190);

Mixtures of odoriferous substance and mixtures, used as raw materials in industries other than the food or drink industries (330290);

Parts of the machinery & apparatus for soldering, brazing or welding, whether or not capable of cutting, parts thereof (846890);

Adhesives based on polymers or rubber (350691); Float glass & surface ground/polished glass, in sheets, wired glass, whether or not having an absorbent or reflecting layer (700530);

Trunks, suit-cases, vanity-cases, executive-cases, and similar containers, n.e.s., of vulcanised fibre or of paperboard, or wholly or mainly covered with such materials or with paper (420299);

Articles and equipment for general physical exercise, gymnastics or athletics (950691); Polyacetals, in primary forms (390710); Tanned/crust hides & skins of bovine or equine animals, in the dry state (crust), without hair on, (other than full grains, unsplit; grain splits) (410449);

Men's/boys' jackets & blazers (excl.

Knitted/crocheted), of synthetic fibres (620333);

Millstones, grindstones, grinding wheels and the like; of natural stone (680423);

Refractory ceramic goods other than those of siliceous fossil meals or similar earths, containing by weight more than 50% of alumina or of a mixture or compound of alumina and of silica (690320);

Finishing agents, dye carriers to accelerate dyeing or fixing of dye-stuffs, other products and preparations, used in the leather or like (excluding paper and textile) industries, (excluding those with a basis of amylaceous substances) (380993):

Gears & gearing (not toothed wheels, chain sprockets and other transmission elements presented separately); ball or roller screws; gear boxes and other speed changers, including torque converters (848340);

Insulating fittings for electrical purposes, of materials other than ceramics or plastics; electrical conduit tubing and joints therefor, of base metal lined with insulating material (854790);

Baby carriages & parts thereof (871500);

Aluminium foil, backed with paper, paperboard, plastics or similar backing materials, of a thickness (excluding any backing) not exceeding 0.2mm (760720);

Gluconic acid, its salts & esters (291816);

Pigments (incl. Metallic powders & flakes), used in the manufacture of paints, other than stamping foils (321290):

Machinery for preparing, tanning or working hides, skins or leather, other than sewing machines (845310);

Disodium carbonate (283620);

Bearing housings, incorporating ball or roller bearings (848320);

Tubes, pipes & hose fittings (e.g. Joints, elbows, flanges) (391740);

Mountings, fittings & similar articles, suitable for furniture of base metal (830242);

Plywood; consisting only of sheets of wood (not bamboo), each ply 6mm or thinner, with at least one outer ply of non-coniferous wood, (without an outer ply of tropical wood) (441232);

Boring or sinking machinery n.e.s., not self-propelled (843049);

High tenacity yarn other than textured yarn, not for retail sale (540219);

Tubes, pipes & hoses, rigid, of plastics n.e.s. (391729):

Domestic food grinders and mixers and fruit or vegetable juice extractors, with self-contained electric motor (850940);

Tools for boring/broaching, for hand tools, whether or not power-operated (820760); Copper wire, of refined copper, of which the maximum cross-sectional dimension is 6mm or less (740819);

Titanium ores & concentrates (261400);

Silencers (mufflers) & exhaust pipe and parts (vehicle) (870892);

Parts and accessories of office machines, nes (847340);

Leather further prepared after tanning or crusting, including parchment-dressed, of bovine (including buffalo) or equine animals, without hair on, split or not, (other than grain splits and full grains, unsplit) (410719):

Ethylene-vinyl acetate copolymers, in primary forms (390130);

Cartons, boxes & cases, of corrugated paper or paperboard (481910);

Moulds for mineral materials (848060);

Waste & scrap of cast iron (720410);

Plates, sheets, film, foil & strip, of polycarbonates, non-cellular and not reinforced, laminated, supported or similarly combined with other materials (392061);

Carding machines (844511);

Alcohol peroxides, ether peroxides, ketone peroxides and their halogenated, sulphonated, nitrated or nitrosated derivatives (290960); Used/new rags, scrap twine, cordage, rope and cables and worn out articles of twine, cordage, rope or cables, of textile materials; sorted (631010);

Refrigerating/freezing chests, cabinets, display counters, show-cases and the like, for storage and display, n.e.s. (841850);

Phenolic resins, in primary forms (390940); Household and toilet articles, n.e.s., in of porcelain or china (691190):

Vat dyes (incl. Those usable in that state as pigments) and preparations based thereon (320415):

Beech (Fagus spp.), sawn or chipped lengthwise, sliced or peeled, whether or not planed, sanded or finger-jointed, thicker than 6mm (440792); Paper & paperboard of a kind used for writing, printing or other graphic purposes, light-weight coated paper, in rolls or sheets, coated with kaolin or other inorganic substances only, having more than 10% of mechanical or chemi-mechanical processed fibres, (481022);

Semi-finished products of iron/non-alloy steel, containing by weight less than 0.25% of carbon, other than rectangular or square cross-section (720719):

Narrow woven fabrics n.e.s., of textile materials n.e.c., (excl. Labels, badges and similar articles) (580639);

Shutters, blinds (incl. Venetian blinds) and similar articles and parts thereof (392530);

Phosphoric acid and polyphosphoric acids (280920);

Paper & paperboard of a kind used for printing & writing, graphics, coated with kaolin or other inorganic substances, containing not more than

10% by weight of total fibres obtained by mechanical or chemi-mechanical process, in rolls (481014);

Garden/similar umbrellas (660110);

Toasters, electric (851672);

Oil-cake and other solid residues; whether or not ground or in the form of pellets, resulting from the extraction of oils, n.e.s. (230690);

Industrial robots, nes (847950);

Metal furniture of a kind used in office (940310);

 $Coffee, not\ roasted,\ not\ decaffe in a ted\ (90111);$

Zinc, not alloyed, unwrought, containing by weight less than 99.99% of zinc (790112);

Silica sands & quartz sands, whether or not coloured (250510);

Portal/pedestal jib cranes (842630);

Pulley tackle & hoists other than skip hoists or hoists of a kind used for raising vehicles, not powered by an electric motor (842519);

Vegetable oils; castor oil and its fractions, whether or not refined, but not chemically modified (151530);

Wrist-watche, not electrically operated, not automatic winding, whether or not incorporating a stop-watch facility, case of precious metal or of metal clad with precious metal (910129); Filtering/purifying machinery & apparatus

(842121); Machinery for the manufacture of confectionery, cocoa or chocolate (843820);

Tea, green; (not fermented), in immediate packings of a content exceeding 3kg (90220); Castors of base metal (830220);

AC generators, (alternators), of an output exceeding 75kva but not exceeding 375kva (850162);

Pectic substances; pectinates and pectates (130220);

Ethylene polymers; in primary forms, polyethylene having a specific gravity of less than 0.94 (390110);

Hoopwood; split poles; piles, pickets, stakes, pointed but not sawn lengthwise; sticks for umbrellas, tool handles etc, roughly trimmed but not turned or bent (440420);

Plates, sheets, film, foil & strip, of polymers of styrene, cellular (392111);

Glass frit & other glass, in the form of powder, granules or flakes (320740);

Firearms like sporting, hunting or target-shooting rifles n.e.c. (930330);

Nutmeg (90810);

Cast articles of non-malleable cast iron (732510); Men's/boys', anoraks (incl. Ski-jackets), windcheaters, wind-jackets and similar articles; men's or boys', of textile materials (other than wool or fine animal hair, cotton or man-made fibres), not knittedd or crocheted (620199);

Inorganic acids other than hydrogen (281119);

Wood, n.e.s., sawn/chipped lengthwise/sliced lengthwise/sliced/peeled, whether or not planed, sanded or end-jointed, of a thickness exceeding 6mm (440729); Coniferous wood, in the rough, whether/not stripped of bark or sapwood, or roughly squared, untreated (440320); Other copper alloys (other than alloys already covered) (740329); Ash & residues (excl. From the manufacture of iron or steel), containing mainly copper (262030); Tanned/crust hides & skins of bovine or equine animals, in the wet state (including wet blue), without hair on, excluding full grains, unsplit; grain splits (410419); Syringes, with/without needles (901831); Carpets & other textile floor coverings, knotted, of wool or fine animal hair, whether or not made up Unwrought lead, refined (780110); Aluminium waste & scrap (760200); Polyethylene having a sp.gr. Of 0.94 or more (390120)Rwanda Commodities not specified (999999); Articles of plastics, n.e.s. (392690); Precious (other than diamonds) and semi-precious Photosensitive semiconductor device (854140); stones, unworked or simply sawn or roughly shaped, not strung, mounted or set (710310); Parts of Machines & mechanical appliances Waste & scrap of cast iron (720410); of those having individual functions Ores & concentrates, n.e.s. In chapter 26; other (847990); than antimony (261790); Parts and accessories of data processing Coffee, not roasted, not decaffeinated (90111); equipment (847330); Tea, black: (fermented) and partly fermented tea. Ferrous waste & scrap, n.e.s. (720449); in immediate packings of a content exceeding 3kg Printed matter, n.e.s. (491199); (90240)Copper waste & scrap (740400); Precious (other than diamonds) and semiprecious stones, unworked or simply sawn or roughly shaped, not strung, mounted or set (710310); Aluminium waste & scrap (760200); Fats & oils derived from milk (other than butter or dairy spreads) (40590) Senegal Labels, badges & similar articles; woven, of textile Parts & accessories of the machines or materials, in the piece, in strips or cut to shape or auxiliary machinery of heading no. 8444 size, not embroidered (580710); (844820);Telephones for cellular networks or for other Precious (other than diamonds) and semiwireless networks (851712); precious stones, unworked or simply sawn Parts of electric filament or discharge lamps, or roughly shaped, not strung, mounted or sealed beam lamp units, ultra-violet or infra-red set (710310); lamps and arc-lamps, n.e.s. (853990); Other apparatus for transmission reception Discs, tapes, solid state non-volatile storage of voice, images other data, incl. Apparatus devices, smart cards and other media for the for communication in a wired wireless recording of sound of other phenomena, whether network (such as a local wide area network) not recorded, incl. Matrices and masters for the (851769); production of discs, but excl. Products of Ch.37., Zinc alloys, unwrought (790120); Parts of machinery, plant/laboratory other n.e.s. (852380); Mechanical appliances, for projecting, dispersing equipment, for treating materials by a or spraying liquids or powders, for other than process involving a change of temperature agricultural or horticultural use, whether or not (841990); hand-operated (842489); Carbon brushes for electrical purposes (854520);

Recovered (waste & scrap) paper/paperboard, n.e.s., and of unsorted waste and scrap (470790); Mounted piezo-electric crystals (854160);

Tools for drilling, other than rock, for machine or hand tools, whether or not power-operated (820750);

Lubricating preparations (excl. Those for the treatment of textile and similar materials), not containing petroleum oils or oils obtained from bituminous minerals (340399);

Other parts & accessories for the motor vehicles (870899);

Topical wood, n.e.s., in the rough, whether or not stripped of bark or sapwood, or roughly squared, untreated (440349):

DC Electric motors and generators of an output not exceeding 750W (850131);

Other machines and apparatus for electrical machines and apparatus, other than than machines and apparatus for electroplating electrolysis electrophoresis signal generators particle accelerators. (854370);

Locks of a kind used for motor vehicles (key, combination or electrically operated), of base metal (830120);

Mixtures of odoriferous substance and mixtures, used as raw materials in industries other than the food or drink industries (330290);

Regulating or controlling instruments and apparatus; automatic, other than hydraulic or pneumatic (903289);

Acetone (291411);

Helical springs of iron/steel (732020);

Sound signalling equipment; electrical, used on cycles or motor vehicles (851230);

Antibiotics & their derivatives, n.e.s. (294190); Parts (incl. Regulators) of the hydraulic turbines and water wheels (841090);

Other monitors, not of a kind solely principally used in an automatic data processing system of heading 84.71 (852859);

Apples, fresh (80810);

Machine-tools for working metal, n.e.s., other than hydraulic presses (846299);

Paper & paperboard of a kind used for writing, printing or other graphic purposes, light-weight coated paper, in rolls or sheets, coated with kaolin or other inorganic substances only, having more than 10% of mechanical or chemi-mechanical processed fibres, (481022);

Road wheels & parts & accessories (870870); Safety airbags with inflater system (in vehicles) & parts (870895);

Articles of vulcanised rubber other than hard rubber, n.e.s. (401699);

Fixed electrical capacitors, ceramic dielectric, multilayer (excl. Power capacitors) (853224); "Spectacle lenses, unmounted, of materials other than glass

(900150);"

Recovered (waste & scrap)

paper/paperboard, n.e.s., and of unsorted waste and scrap (470790);

Articles of vulcanised rubber other than hard rubber, n.e.s. (401699);

Cylindrical roller bearings nes (848250); Leather further prepared after tanning or crusting, including parchment-dressed, of bovine (including buffalo) or equine animals, without hair on, split or not, (other than grain splits and full grains, unsplit) (410719);

Parts of centrifuges, including centrifugal dryers (842191);

Parts of electrical signalling, safety or traffic control equipment, n.e.s. (853090); Plates, sheets, film, foil & strip, of polyethylene terephthalate, non-cellular and not reinforced, laminated, supported or similarly combined with other materials (392062);

Beans n.e.s., dried/shelled, including seed (71339);

Zinc waste & scrap (790200);

Leather further prepared after tanning or crusting, including parchment-dressed leather, without hair on, whether or not split, of goats and kids (411310);

Waste & scrap of stainless steel (720421); Waste & scrap of alloy steel other than stainless steel (720429);

Manganese ores and concentrates, including manganiferous iron ores and concentrates with a manganese content of 20% or more, calculated on the dry weight (260200);

Waste & scrap of cast iron (720410); Lac; natural gums, gum resins, resins and balsams, n.e.s. (130190);

Flexible intermediate bulk containers, of a kind used for the packing of goods, of manmade textile materials (630532);

DC motors and generators of an output exceeding 375kw (850134);

Unwrought lead other than refined, not containing by weight antimony as the principal other element (780199); Tanned/crust hides & skins of bovine or equine animals, in the wet state (including wet blue), without hair on, excluding full

grains, unsplit; grain splits (410419); Aluminium waste & scrap (760200); Copper waste & scrap (740400) Multimeters for measuring or checking voltage, current, resistance or power, without a recording device (903031);

Parts of microphones, loudspeakers, headphones, earphones or audio-frequency electric amplifiers, n.e.s. (851890);

Mountings, fittings & similar articles, for motor vehicles, of base metal (830230);

Suspension systems & parts thereof (vehicles) (870880):

Buttons of plastics, not covered with textile material (960621);

Filtering/purifying machinery & apparatus, other than intake air filters for internal combustion engines (842139);

Apparatus based on use of x-rays and similar; parts and accessories (x-ray generators, tubes, high tension generators, control panels and desks, screens, examination or treatment tables, chairs and like (902290);

Steering wheels, steering columns and steering boxes & parts thereof (870894);

Parts of electrical lighting or signalling equipment, windscreen wipers, defrosters and demisters of a kind used for motor vehicles, n.e.s. (851290); Ceramic sinks, wash basins, wash basin pedestals, baths, bidets, water closet pans, flushing cisterns, urinals and similar sanitary fixtures; of porcelain or china (691010);

Flat-rolled products of stainless steel, width 600mm or more, n.e.s. (721990);

Other printers, copying machines, single-function printing, capable of connecting to an automatic data processing machine or to a network (844332); Rear-view mirrors for vehicles (700910);

Magnetic media for the recording of sound of other phenomena, but excl. Products of Ch. 37., cards incorporating a magnetic stripe (852321); Bumpers & parts thereof of the motor vehicles (870810);

Radiators & parts thereof for the motor vehicles (870891);

Zinc waste & scrap (790200);

Sawing/cutting-off machines working by removing metal, sintered metal carbides or cermets (846150);

Wood, in the rough, n.e.s., whether or not stripped of bark or sapwood, or roughly squared, untreated, (440399):

Unglazed ceramic flags & paving/hearth/wall tiles (690790);

Ignition wiring sets and other wiring sets for vehicles, aircraft or ships (854430);

Paints & varnishes (incl. Enamels & lacquers), (based on polymers other than acrylic or vinyl), dispersed or dissolved in an aqueous medium (320990);

Fixed electrical capacitors, tantalum (excl. Power capacitors) (853221);

Disodium carbonate (283620);

Parts of aeroplanes/helicopters, others (880330); Vitamins & their derivatives, unmixed (293629); Chromatographs and electrophoresis instruments (902720):

Baths of iron other than cast iron/steel (732429); Parts & accessories of bodies (other than safety seat belts) (870829);

Other printers, copying machines, single-function printing, not capable of connecting to an automatic data processing machine or to a network (844339); Electrical lighting or visual signalling equipment for motor vehicles (excl. Lamps of heading 8539) (851220);

Wood, n.e.s., sawn/chipped lengthwise/sliced lengthwise/sliced/peeled, whether or not planed, sanded or end-jointed, of a thickness exceeding 6mm (440729);

Parts of calendering/other rolling machines, for other than metal or glass (842099);

Parts of the vehicles of 87.09 (self-propelled works trucks) (870990);

Kraft paper (excl. Kraftliner & sack kraft paper), weighing 150 g/m2 or less (480439);

Parts of electric sound or visual signalling apparatus, n.e.s. (853190);

Speed indicators & tachometers; stroboscopes (902920);

Machine-tools for wood, cork or hard plastic, etc. N.e.s. (846599);

Articles of apparel & clothing accessories (including gloves, mittens and mitts) (392620); Tea, black; (fermented) and partly fermented tea, in immediate packings of a content exceeding 3kg (90240):

Builders' ware of plastics, n.e.s. (392590); Safety seat belts of the motor vehicles and parts (870821);

Parts of the seats of 94.01 (seats in vehicles, aircrafts, convertible to bed, etc.) (940190); Flat-rolled products of iron/non-alloy steel, not in coils, of a width 600mm or more, hot-rolled, without patterns in relief, of a thickness exceeding 10mm (720851);

Fittings for furniture/coachwork/the like, of plastic (392630);

Fixed electrical capacitors, dielectric of paper or plastics (excl. Power capacitors) (853225);

Shaving, hair, nail, eyelash and other toilet brushes for use on the person, including brushes as parts of appliances (960329);

Human blood; animal blood prepared for therapeutic, prophylactic or diagnostic uses; cultures of micro-organisms (excluding yeast) and similar products (300290);

Palm oil, other than crude, & its fractions, whether or not refined, but not chemically modified (151190);

Raw hides & skins of bovine (including buffalo) and equine animals, fresh, salted or preserved, but

not tanned, parchment dressed or further prepared, whether or not dehaired or split (410190); Waste & scrap of tinned iron/steel (720430); Copper wire, of refined copper, of which the maximum cross-sectional dimension is 6mm or less (740819);

Unwrought lead other than refined, not containing by weight antimony as the principal other element (780199)

Sierra Leone

Storage units of ADP (847170);

Petroleum oils and oils from bituminous minerals, not containing biodiesel, not crude, not waste oils; preparations n.e.s. (271019);

Printed matter, n.e.s. (491199);

Other Parts suitable for use solely or principally with the apparatus of headings 8525 to 8528 (reception and transmission apparatus, monitors, etc.) (852990);

Other reception apparatus for television, whether not incorporating radio broadcast receivers sound video recording reproducing apparatus, black and white other monochrome. (852873);

Electrical apparatus like switches n.e.c. In heading no. 8536, for a voltage not exceeding 1000 volts (853650);

Tableware & kitchenware, of plastic (392410); Other Insulated electric conductors; for a voltage not exceeding 1000 volts, fitted with connectors (854442):

Plates, sheets, strip, rods and profile shapes, of vulcanised rubber other than hard rubber (400829); Other printers, copying machines, single-function printing, capable of connecting to an automatic data processing machine or to a network (844332); Toothed wheels, chain sprockets and other transmission elements presented separately; parts of transmission components (848390); Waste & scrap of stainless steel (720421); Parts of telephone sets, incl. Telephones for cellular networks for other wireless networks other apparatus for the transmission reception of voice, images other data, incl. Apparatus for communication in a wired wireless network (851770);

Containers (including containers for transport of fluids) specially designed and equipped for carriage by one or more modes of transport (860900);

Gears & gearing (not toothed wheels, chain sprockets and other transmission elements presented separately); ball or roller screws; gear boxes and other speed changers, including torque converters (848340);

Mixtures of odoriferous substance & and mixtures; of a kind used in the food or drink industries (330210);

Pumps and compressor for air/vaccum/other gas (841480);

Polypropylene, in primary forms (390210);

Processing units for automatic data processing machines, whether or not containing in the same housing one or two of the following types of unit storage units, input units, output units (847150);

Other apparatus for transmission reception of voice, images other data, incl. Apparatus for communication in a wired wireless network (such as a local wide area network) (851769);

Apparatus based on use of x-rays and similar; parts and accessories (x-ray generators, tubes, high tension generators, control panels and desks, screens, examination or treatment tables, chairs and like (902290);

Fixed electrical capacitors (excl. Tantalum, aluminium electrolytic, ceramic, paper, plastic and power capacitors) (853229); Safety or relief valves, for pipes, boiler shells, tanks, vats or the like (848140); Other tubes, pipes and hollow profiles (for example, open seam or welded, riveted or similarly closed), of iron or steel (730690); Inductors (excl. Inductors for discharge lamps or tubes) (850450);

Electro-diagnostic apparatus (including apparatus for functional exploratory examination or for checking physiological parameters), not electro-cardiographs (901819);

Other Electronic integrated circuits, other than Amplifiers Memories Processors and controllers (854239);

Electrical apparatus; n.e.c., for switching or protecting electrical circuits, for a voltage not exceeding 1000 volts (853690); Precious (other than diamonds) and semi-precious stones, unworked or simply sawn or roughly shaped, not strung, mounted or set (710310):

Filtering/purifying machinery & apparatus, other than intake air filters for internal combustion engines (842139);

Parts & accessories n.e.s. In Ch.90 (903300);

Waste & scrap of stainless steel (720421); Builders' ware of plastics, n.e.s. (392590); Electronic integrated circuits; processors and controllers, whether or not combined with memories, converters, logic circuits,

Parts and accessories of printing machinery used amplifiers, clock and timing circuits, or for printing by means of plates, cylinders and other other circuits (854231); printing components of heading 8442 (844391); Articles for the conveyance/packing of Compressors of a kind used in refrigerating goods, n.e.s. (392390); Towers & lattice masts of iron/steel equipment (841430); Articles & equipment for sports, for outdoor (730820);games and recreation n.e.c. (950699); Diamonds; non-industrial, unworked or T-shirts, singlets & other vests, of textile materials simply sawn, cleaved or bruted, but not (other than cotton), knittedd or crocheted mounted or set (710231); (610990); Railway/tramway track construction Trade advertising material, commercial catalogue material of iron and steel, n.e.s. (730290); and the like (491110) Copper waste & scrap (740400); Aluminium waste & scrap (760200) Somalia Aerials & aerial reflectors (852910); Polyethylene having a sp.gr. Of 0.94 or Clutches & parts thereof for the motor vehicles more (390120): (870893); Precious (other than diamonds) and semiprecious stones, unworked or simply sawn Intake air filters for internal combustion engines or roughly shaped, not strung, mounted or (842131);Precious (other than diamonds) and semi-precious set (710310); stones, unworked or simply sawn or roughly Coral & similar materials, unworked shaped, not strung, mounted or set (710310); (50800): Beans of the species vigna mungo (L.) Hepper or Polymers of propylene/of other olefin vigna radiata (L.) Wilczek, shelled, whether or not polymers, n.e.s. (390290); skinned or split, dried (71331); Tanned/crust skins of sheep/lambs, without Lac; natural gums, gum resins, resins and balsams, wool on, whether or not split, but not n.e.s. (130190); further prepared, in the wet state (including Tanned/crust hides & skins of bovine or equine wet-blue) (410510); Raw skins of sheep/lambs, fresh, or salted, animals, in the wet state (including wet blue), without hair on, excluding full grains, unsplit; dried, limed, pickled or otherwise grain splits (410419); preserved, (but not tanned, parchment-Coral & similar materials, unworked (50800); dressed or further prepared), with wool on Protein: concentrates and textured protein (410210): substances (210610): Whole bovine (incl. Buffalo)/equine Whole bovine (incl. Buffalo)/equine animals, of a animals, of a weight per skin not exceeding weight per skin not exceeding 8kg when simply 8kg when simply dried, 10kg when drydried, 10kg when dry-salted or 16kg when fresh, salted or 16kg when fresh, wet-salted or wet-salted or otherwise preserved (410120); otherwise preserved (410120); Tanned/crust skins of sheep/lambs, without wool Lac; natural gums, gum resins, resins and on, whether or not split, but not further prepared, balsams, n.e.s. (130190); in the wet state (including wet-blue) (410510); Raw skins of sheep/lambs Tanned/crust hides & skins, n.e.s., (410691); (fresh/salted/dried/limed/otherwise, Tanned/crust skins of sheep/lambs, without wool preserved other than by pickling, (but not on, whether or not split, but not further prepared, tanned, parchment-dressed or further in the dry state (410530); preserved), without wool on (410229); Tanned/crust hides & skins of goats or kids, Raw hides & skins of bovine (including without wool or hair on, whether or not split, but buffalo) and equine animals, fresh, salted or not further prepared, in the wet state (including preserved, but not tanned, parchment wet blue) (410621); dressed or further prepared, whether or not Raw skins of sheep/lambs, fresh, or salted, dried, dehaired or split (410190) limed, pickled or otherwise preserved, (but not tanned, parchment-dressed or further prepared), with wool on (410210) Sudan Limestone flux: limestone and other calcareous Unrefined copper; copper anodes for stone, of a kind used for the manufacture of lime electrolytic refining (740200); Wooden furniture (other than for office, or cement (252100); Commodities not specified (999999); kitchen or bedroom use), nesoi (940360); Recovered (waste & scrap) paper/paperboard, Beans of the species vigna mungo (L.) n.e.s., and of unsorted waste and scrap (470790); Hepper or vigna radiata (L.) Wilczek, Carboys, bottles, flasks & similar articles, for the shelled, whether or not skinned or split, conveyance or packing of goods (392330); dried (71331);

Raw skins of sheep/lambs, pickled (but not tanned, parchment-dressed or further preserved), without wool on (410221); Tanned/crust hides & skins of reptiles (410640); Kraft paper (excl. Kraftliner & sack kraft paper), weighing 225g/m2 or more (480459); Tanned/crust hides & skins of bovine or equine animals, in the wet state (including wet blue), without hair on, excluding full grains, unsplit; grain splits (410419); Zinc waste & scrap (790200); Chamotte/dinas earths (250870); Aluminium waste & scrap (760200); Chickpeas (garbanzos), dried, shelled, whether or not skinned or split (71320); Paper&paperboard, not containing fibre greater than 10%, weighing 40g/m2 to 150g/m2, in rolls (480255);Waste & scrap of tinned iron/steel (720430); Bars, rods & profiles, of copper alloys (other than copper-zinc base alloys) (740729); Cotton, not carded/combed (520100); Recovered (waste & scrap) unbleached kraft paper or paperboard or of corrugated paper or paperboard (470710): Copper waste & scrap (740400); Tanned/crust hides & skins of goats or kids,

Tanned/crust hides & skins of goats or kids, without wool or hair on, whether or not split, but not further prepared, in the wet state (including wet blue) (410621);

Lead waste & scrap (780200);

Leather further prepared after tanning or crusting, including parchment-dressed leather, of sheep or lamb, without wool on, whether or not split (411200):

Whole bovine (incl. Buffalo)/equine animals, of a weight per skin not exceeding 8kg when simply dried, 10kg when dry-salted or 16kg when fresh, wet-salted or otherwise preserved (410120);

Lac; natural gums, gum resins, resins and balsams, n.e.s. (130190); Food preparations; of flour, meal, starch, malt extract or milk products, for uses n.e.s. (190190);

Small red (Adzuki) beans (Phaseolus (71332); Waste & scrap of cast iron (720410); Aluminium ores & concentrates (260600); Oil seeds & oleaginous fruits, n.e.s., whether or not broken (120799);

Soya beans, whether/not broken (120100)

Siliceous fossil meals (e.g., kiese) (251200); Lac; natural gums, gum resins, resins and balsams, n.e.s. (130190); Bars, rods & profiles, of copper alloys (other than copper-zinc base alloys) (740729); Aluminium waste & scrap (760200); Waste & scrap of cast iron (720410); Tanned/crust hides & skins of bovine or equine animals, in the wet state (including wet blue), without hair on, full grains, unsplit; grain splits (410411); Leather further prepared after tanning or crusting, including parchment-dressed leather, without hair on, whether or not split, of goats and kids (411310); Copper waste & scrap (740400); Tanned/crust skins of sheep/lambs, without wool on, whether or not split, but not further prepared, in the dry state (410530); Raw skins of sheep/lambs, fresh, or salted, dried, limed, pickled or otherwise preserved, (but not tanned, parchmentdressed or further prepared), with wool on (410210);

Chickpeas (garbanzos), dried, shelled, whether or not skinned or split (71320); Lead waste & scrap (780200); Unwrought lead, refined (780110); Recovered (waste & scrap) unbleached kraft paper or paperboard or of corrugated paper or paperboard (470710);

Kidney beans, incl. White pea beans (phaseolus vulgaris), dried, shelled, whether or not skinned or split (71333); Gum Arabic (130120):

Whole bovine (incl. Buffalo)/equine animals, of a weight per skin not exceeding 8kg when simply dried, 10kg when drysalted or 16kg when fresh, wet-salted or otherwise preserved (410120)

Tanzania

Parts & accessories equally suitable (847350); Relays for a voltage > 60 V but <= 1.000 V (853649):

Telephones for cellular networks or for other wireless networks (851712);

Tubes, pipes & hoses, of vulcanised rubber, not reinforced or otherwise combined with other materials, without fittings (400911); Clays (excluding expanded clays of heading no. 6806); n.e.s. In heading no. 2508, whether or not calcined (250840);

Marble, travertine and alabaster; simply cut or sawn, with a flat or even surface (680221);

Parts & accessories n.e.s. In Ch.90 (903300):

Parts of the steam turbines & other vapour turbines (840690);

Insulating fittings for electrical purposes, of plastics (854720);

Machinery, plant & equip., n.e.s., for treating materials by change of temperature,

Other monitors, of a kind solely principally used in an automatic data processing system (852851); Ignition wiring sets and other wiring sets for vehicles, aircraft or ships (854430);

Parts suitable for use solely or principally with electric motors and generators, electric generating sets and rotary converters, n.e.s. (850300);

Moulds for rubber or plastics injection or

Moulds for rubber or plastics, injection or compression types (848071);

Electric accumulators n.e.s., including separators, whether or not rectangular (including square) (850780);

Taps, cocks, valves and similar appliances; for pipes, boiler shells, tanks, vats or the like, including thermostatically controlled valves (848180);

Parts suitable for use solely or principally with the apparatus of heading 8535, 8536 or 8537, n.e.s. (excl. Boards, panels, consoles, desks, cabinets and other bases for the goods of heading 8537, not equipped with their apparatus) (853890); Coal; (other than anthracite and bituminous), whether or not pulverised but not agglomerated (270119);

Electro-diagnostic apparatus (including apparatus for functional exploratory examination or for checking physiological parameters), not electrocardiographs (901819);

Gaskets, washers & other seals of vulcanised rubber, of non-cellular rubber (401693); Parts of the air-conditioning machines, with motor driven fan and elements for temperature control (841590):

Clasps, frames with clasps, buckles, buckle clasps; of base metal, of a kind used for clothing, footwear, awnings, handbags, travel goods or other made up articles, including parts (830890); Electronic integrated circuits; processors and controllers, whether or not combined with memories, converters, logic circuits, amplifiers, clock and timing circuits, or other circuits (854231);

Ethylene polymers; in primary forms, polyethylene having a specific gravity of less than 0.94 (390110);

Diodes (excl. Photosensitive or light emitting diodes) (854110);

Video games of a kind used with a television receiver (950410);

Epoxide resins, in primary forms (390730); Palm oil and its fractions, crude, not chemically modified (151110);

Pumps and compressors; parts, of air or vacuum pumps, air or other gas compressors and fans, ventilating or recycling hoods incorporating a fan (841490);

Iron ores and concentrates; non-agglomerated (260111);

Parts of the book-binding machinery, including book-sewing machines, parts thereof (844090);

other than for making hot drinks or cooking or heating food (841989);

Auminium plates, sheets & strip, rectangular (including square), of thickness exceeding 0.2mm, alloys (760612); Natural graphite, in powder/flakes (250410);

Machinery for moulding or forming, for working rubber or plastics or for the manufacture of products from these materials, n.e.c. (847759);
Raw skins of sheep/lambs (fresh/salted/dried/limed/otherwise, preserved other than by pickling, (but not tanned, parchment-dressed or further preserved), without wool on (410229);
Waste & scrap of stainless steel (720421);
Other vehicles, not mechanically propelled (871680):

Machines for sorting, screening, separating or washing earth, stone, ores or other mineral substances (847410);

Ferro-silicon, containing by weight 55% or less of silicon (720229);

Parts of metal-rolling mills, other rolls (845590);

Used/new rags, scrap twine, cordage, rope and cables and worn out articles of twine, cordage, rope or cables, of textile materials; other than sorted (631090);

Marble and travertine, merely cut, by sawing or otherwise, into blocks or slabs of a rectangular (including square) shape, having a specific gravity of 2.5 or more (251512);

Zinc, not alloyed, unwrought, containing by weight less than 99.99% of zinc (790112); Tanned/crust hides & skins of bovine or equine animals, in the dry state (crust), without hair on, (other than full grains, unsplit; grain splits) (410449); Pens and pencils n.e.s.; duplicating stylos, pen-holders, pencil-holders, and similar holders, parts of the articles (960899); Leather further prepared after tanning incl. Parchment-dressed, of bovine (including buffalo) or equine animals, no hair, and whole hides and skins, and sides, (full grains, unsplit and grain splits) (410799); Rolls for rolling mills (845530); Sheets for veneering (including those obtained by slicing laminated wood), of Dark Red Meranti, Light Red Meranti and Meranti Bakau, sawn lengthwise, sliced or peeled, planed or not, sanded or end-jointed, not thicker than 6mm (440831); Nuts, n.e.s., fresh or dried, whether or not shelled or peeled (80290);

Waste (incl. Noils, yarn waste and garnetted

stock), of synthetic fibres (550510);

Other printers, copying machines, single-function printing, capable of connecting to an automatic data processing machine or to a network (844332); Tricycles, scooters, pedal cars and similar wheeled toys; dolls' carriages; dolls; other toys; reduced-size (scale) models and similar recreational models, working or not; puzzles of all kinds (950300);

Parts of the slide fasteners (960720);

Tubular/bifurcated rivets, of base metal (830820); Ball bearings (848210);

Chemical products&preparations of the chemical or allied industries, n.e.s. (382490);

Press-fasteners, snap-fasteners, press-studs and parts (960610);

Storage units of ADP (847170);

Buttons, of base metal, not covered with textile material (960622);

Flat-rolled products of iron/non-alloy steel, hot-rolled, of a width 600mm or more, n.e.s. (720890); Television cameras, digital cameras and video camera recorders (852580);

Dates, fresh/dried (80410);

Electric ovens, cookers, cooking plates and boiling rings, electric grillers and roasters, for domestic use (excl. Space heating stoves and microwave ovens) (851660);

Engines parts for internal combustion piston engines (excluding spark-ignition) (840999); Preparations of a kind used in animal feeding, put up for retail sale (except dog or cat food) (230990);

Knitting Machines, Others, And Machines For Making Gimped Yarn, Tulle, Lace, Embroidery, Trimmings, Braid Or Net And Machines For Tufting (844790);

Other parts and accessories of looms, auxiliary machinery, n.e.s. (844849);

Iron or steel tubes, pipes and hollow profiles, seamless, n.e.s. (730490);

Plates, sheets, film, foil & strip, of polyethylene terephthalate, non-cellular and not reinforced, laminated, supported or similarly combined with other materials (392062);

Parts of the machinery for treating textile fabrics (845190);

Parts & accessories of regulating or controlling instruments and apparatus; automatic (903290); Isocyanates (292910);

Road wheels & parts & accessories (870870); Aromatic cyclic alcohols & their derivatives, other than benzyl alcohol (290629);

Plywood, veneered panels and similar laminated wood (other than blockboard, laminboard and battenboard, other than of bamboo, and other than plywood consisting only of sheets of wood each ply 6mm or thinner) (441299);

Paper & paperboard, coated with inorganic substances only, whether or not surface-coloured,

Ferrous waste & scrap, n.e.s. (720449); Recovered (waste & scrap) paper/paperboard, n.e.s., and of unsorted waste and scrap (470790);

Manganese ores and concentrates, including manganiferous iron ores and concentrates with a manganese content of 20% or more, calculated on the dry weight (260200); Essential oils (terpeneless/not), including concretes and absolutes, n.e.s. (330190); Moulds for mineral materials (848060); Ash & residues (excl. From the manufacture of iron or steel), containing mainly zinc, hard zinc spelter (262019);

Sheets n.e.s. For veneer or plywood, other wood sawn lengthwise, sliced or peeled, whether or not planed, sanded or endjointed, not thicker than 6mm (440890);

Seeds of coriander (90920);

Aluminium waste & scrap (760200); Zinc waste & scrap (790200);

Soya beans, whether/not broken (120100); Articles of natural/cultured pearls (711610); Fuel wood, in

logs/billets/twigs/faggots/similar forms, whether or not agglomerated, coniferous (440110);

Tanned/crust hides & skins, n.e.s., (410691);

Plants & parts of plants, including seeds and fruits, n.e.s., used primarily in perfumery, pharmacy or for insecticidal, fungicidal purposes; fresh or dried, whether or not cut, crushed or powdered (121190);

Lead waste & scrap (780200);

Nuts (excl. Ground-nuts), incl. Mixtures, whether or not containing added sugar, other sweetening matter or spirit (200819); Gum Arabic (130120);

Sugars; n.e.s., including invert sugar and other sugar and sugar syrup blends containing, in the dry state, 50% by weight of fructose (170290);

Tanned/crust hides & skins of bovine or equine animals, in the wet state (including wet blue), without hair on, excluding full grains, unsplit; grain splits (410419); Unwrought lead other than refined, not containing by weight antimony as the principal other element (780199); Beech (Fagus spp.), sawn or chipped lengthwise, sliced or peeled, whether or not planed, sanded or finger-jointed, thicker than 6mm (440792)

surface-decorated or printed, n.e.s., in rolls or sheets (481099);

Parts and accessories for instruments/apparatus measuring or checking the flow, level, pressure or other variables of liquids or gases (902690); Tubes, pipes & hollow profiles of circular cross-section (excl. Stainless steel), (excluding cold-drawn or cold-rolled) (730459);

Fuel/lubricating/cooling medium pumps for internal combustion piston engines (841330); Nitrile-function compounds, n.e.s. (292690); Tube/pipe fittings of iron/steel, n.e.s., other than stainless steel (730799);

Moulds for rubber or plastics, other than injection or compression types (848079);

Filtering/purifying machinery & apparatus, for filtering or purifying liquids, n.e.s. (842129); Ferrous turnings, shavings, chips, milling waste, sawdust, fillings, trimmings and stampings, whether or not in bundles (720441);

Self-adhesive plates, sheets, film, foil, tape, strip, other flat shapes thereof, other than in rolls of a width not exceeding 20cm (391990);

Parts of mechanical appliances (parts of machines projecting, dispersing or spraying liquids or powders, whether or not hand-operated) (842490); Machines and appliances for testing the hardness, strength, compressibility, elasticity or other mechanical properties of metals (902410); Synthetic/reconstructed precious/unworked/simply sawn, roughly shaped, (not strung, mounted or set) precious or semi-precious stones, temporarily strung for transport convenience (710420);

Plywood; consisting only of sheets of wood (not bamboo), each ply 6mm or thinner, with at least one outer ply of tropical wood (441231);

Lace mechanically made in the piece, in strips or in motifs, of textile materials (580429);

Polyamides, n.e.s., in primary forms (390890); Polyamide-6/-11/-12/-6,6/-6,9/-6,10 or -6,12, in primary forms (390810);

Rare gases other than argon (280429);

Polyethylene having a sp.gr. Of 0.94 or more (390120):

Flat-rolled products of iron/non-alloy steel, in coils, width 600mm or more, cold-rolled, of a thickness exceeding 1mm but less than 3mm (720916);

Aromatic polycarboxylic acids, n.e.s. (291739); Particle board & similar board of ligneous material, n.e.s. (441090);

Newsprint, in rolls/sheets (480100);

Parts of the filtering/purifying machinery for filtering or purifying liquids or gases, n.e.s. (842199);

Clutches & parts thereof for the motor vehicles (870893);

Recovered (waste & scrap) paper/paperboard, n.e.s., and of unsorted waste and scrap (470790);

Loudspeakers; not mounted in their enclosures (851829);

Pumps n.e.s. In 84.13 (841381);

Wallpaper & similar wall coverings, coated or covered on the face side, with a grained, embossed, coloured, design-printed or otherwise decorated layer of plastics (481420);

Marble, travertine and alabaster; simply cut or sawn, with a flat or even surface (680221); Micro-crystalline petroleum wax, slack wax, lignite wax, ozokerite, peatwax other mineral waxes and similar products obtained by synthesis

waxes and similar products obtained by synthesis or by other processes, whether or not coloured (271290);

Overhead travelling cranes on fixed support (842611);

Stationery n.e.s., made of paper or paperboard (482090);

Starter motors and dual purpose starter-generators, of a kind used for spark or compression-ignition internal combustion engines (851140);

Plates, sheets, film, foil & strip, other than cellular (392190);

Refrigerating/freezing chests, cabinets, display counters, show-cases and the like, for storage and display, n.e.s. (841850);

Bicycles and other cycles; including delivery tricycles, not motorised (871200);

Raw skins of sheep/lambs, pickled (but not tanned, parchment-dressed or further preserved), without wool on (410221):

Particle board of wood, whether/not agglomerated with resins or other organic binding substances (441011);

Apples, fresh (80810);

Articles of iron/steel, n.e.s., forged/stamped, but not further worked (732619);

Almonds, in shell (80211);

Other tubes and pipes (for example, welded, riveted or similarly closed), having circular cross-sections, the external diameter of which exceeds 406.4 mm, of iron or steel (730590);

Compound plasticisers for rubber/plastics (381220);

Silicones, in primary forms (391000);

Hydraulic power engines & motors other than linear acting (cylinders) (841229);

Anionic surface-active agents, whether or not put up for retail sale (340211);

Ignition magnetos, magneto-dynamos and magnetic flywheels, for spark-ignition or compression-ignition internal combustion engines (851120);

Cyclic hydrocarbons; n.e.s. (290290);

Waste & scrap of stainless steel (720421);

Organic surface-active agents, whether or not put up for retail sale (340219);

Reaction initiators, reaction accelerators and catalytic preparations, unsupported, n.e.s. Or included (381590);

Iodine (280120);

Sacks & bags, of a kind used for the packing of goods, of jute or of other textile bast fibres (630510);

Paper, paperboard, cellulose wadding and webs of soft cellulose fibres; coated, impregnated, covered, surface-decorated or surface-coloured, n.e.s. (481190);

Wool, not carded/combed, greasy, (including fleece-washed wool), not carded or combed (510119);

Poppy seeds, whether/not broken (120791); Builders' ware of plastics, n.e.s. (392590);

Peas (Pisum sativum), dried, shelled, whether or not skinned or split (71310);

Vegetable, mineral carving material and articles of these materials, moulded or carved articles of wax, stearin, natural gums, resins or modelling pastes, worked unhardened gelatin (960200);

Parts of the boilers of central heating, n.e.s. (840390);

Carpets and other textile floor coverings; tufted, of nylon or other polyamides, whether or not made up (570320);

Polystyrene other than expansible, in primary forms (390319);

Articles of plaster/of compositions based on plaster; n.e.s. (680990);

Penicillins & their derivatives with a penicillanic acid structure; salts thereof (294110);

Butt welding fittings of iron/steel, other than stainless steel (730793);

Threaded elbows, bends & sleeves of material other than stainless steel (730792);

Rails of iron/steel (730210);

Marble and travertine, merely cut, by sawing or otherwise, into blocks or slabs of a rectangular (including square) shape, having a specific gravity of 2.5 or more (251512);

Relays, for a voltage not >60V (853641);

Tubes, pipes & hoses, of vulcanised rubber (other than hard rubber), reinforced or otherwise combined with materials other than metal or textiles, with fittings (400942);

Natural rubber (excl. Latex), in smoked sheets (400121):

Spirits obtained by distilling grape wine or grape marc (220820);

Printed books, brochures, leaflets, n.e.s. (490199); Bars & rods, hot-rolled, in irregularly wound coils, n.e.s. (722790);

Instruments and apparatus for measuring or checking voltage, current, resistance or power, with a recording device (excluding multimeters) (903039):

Bars & rods of other alloy steel, n.e.s. (722860); Glassine & other glazed transparent or translucent papers, in rolls or sheets (480640);

Poly(ethylene terephthalate), in primary forms, having a viscosity of 78ml/g or higher (390760);

Welded angles, shapes & sections of Iron or steel (730120);

Other n.e.s., flat-rolled, width 600mm or more, n.e.s., (other than plated or coated with zinc) (722599);

Carbon; carbon blacks and other forms of carbon n.e.s. (280300);

Book-binding machinery, including book-sewing machines (844010);

Machine-tools for working metal/cermets/sintered metal carbide, without removing material, n.e.s. In heading no. 8463 (846390);

Parts of the machinery/apparatus/equipment (excluding machine-tools) for preparing or making printing components and parts thereof (844240); Parts of railway/tramway rolling stock n.e.c. (860799):

Mountings, fittings & similar articles, suitable for furniture of base metal (830242);

Mould bases for metal, metal carbides, glass, mineral materials, rubber or plastics (848020); Toughened (tempered) safety glass, (not of a size and shape suitable for incorporation in vehicles, aircraft, spacecraft or vessels) (700719); Semi-finished products of stainless steel, other than of rectangular cross-section (721899); Flat-rolled products of iron/non-alloy steel, in coils, width 600mm or more, cold-rolled, of a thickness of less than 0.5mm (720918); Machines for making cartons, boxes, cases, tubes, drums or similar containers (other than by moulding), of paper pulp, paper or paperboard (844130);

Human blood; animal blood prepared for therapeutic, prophylactic or diagnostic uses; cultures of micro-organisms (excluding yeast) and similar products (300290);

Stoppers, caps & lids, of base metal (830990); Parts and accessories of machines or auxiliary machinery, Spindles, spindle flyers, spinning rings and ring travellers (844833);

Containers for compressed/liquefied gas, of iron or steel (731100);

Self-propelled mechanical shovels, shovel loaders; n.e.s. (842959);

Synthetic/reconstructed precious/semi precious stones, worked or graded or not, (but not strung, mounted or set), temporarily strung for transport convenience (710490);

Other wire of other alloy steel, other than high speed or silico-manganese steel (722990); Fresh orchids (60313);

Topical wood, n.e.s., in the rough, whether or not stripped of bark or sapwood, or roughly squared, untreated (440349):

Flat-rolled products of iron/non-alloy steel, width 600mm or more, cold-rolled, n.e.s. (720990); Anchors, grapnels & parts thereof, of iron/steel (731600);

Moulding boxes for metal foundry (848010);

Parts of locomotives, n.e.s. (860791);

Tanned/crust hides & skins of bovine or equine animals, in the dry state (crust), without hair on, (other than full grains, unsplit; grain splits) (410449);

Rollers for lawns or sports-grounds (843280); Printing type, blocks, plates, cylinders and other printing components, lithographic stones prepared for printing purposes (for example, planed, grained or polished) (844250);

Sodium sulphites (283210);

Lentils, dried, shelled, whether/not skinned or split (71340):

Synthetic organic tanning substance (320210);

Waste & scrap of tinned iron/steel (720430);

Wire of stainless steel (722300);

Unwrought lead other than refined, not containing by weight antimony as the principal other element (780199);

Wood (including strips & friezes for parquet flooring), continuously shaped along any edges, ends or faces, whether or not planed, sanded or end-jointed (440910);

Saw blades, n.e.s. (820299);

Manganese ores and concentrates, including manganiferous iron ores and concentrates with a manganese content of 20% or more, calculated on the dry weight (260200);

Swivel seats with variable height adjustment, excluding medical, surgical, dental, veterinary or barber furniture (940130);

Flat-rolled products of iron/non-alloy steel, width 600mm or more, plated or coated with aluminium, other than plated or coated with aluminium zinc-alloys (721069);

Files, rasps & similar tools (820310);

Aluminium waste & scrap (760200);

Granules of pig iron, spiegeleisen, iron or steel (720510);

Non-coniferous Wood (including strips&friezes for parquet flooring), continuously shaped along any edges, ends or faces, whether or not planed, sanded or end-jointed (440929);

Flax yarn, single (530610);

Vehicles with only compression-ignition internal combustion piston engine (diesel or semi-diesel), cylinder capacity over 1500 but not over 2500cc (870332);

Tanned/crust skins of sheep/lambs, without wool on, whether or not split, but not further prepared, in the wet state (including wet-blue) (410510); Lithium oxide & hydroxide (282520);

Bread, pastry, cakes, biscuits & other products,

n.e.s., whether or not containing cocoa; communion wafers (190590);

Anti-freezing preparations & prepared de-icing fluids (382000);

Terpenic oils produced by the treatment of coniferous woods, crude dipentene, sulphite

turpentine and other crude para-cymene, n.e.s. (380590): Lead ores & concentrates (260700); Sack kraft paper, uncoated, n.e.s. (480429); Zinc waste & scrap (790200); Ferrous waste & scrap, n.e.s. (720449); Apple juice, of a Brix value exceeding 20, unfermented, not containing added spirit, whether or not containing added sugar or other sweetening matter (200979); Lead waste & scrap (780200); Medium density of fibreboard of wood, of a thickness exceeding 5 mm but not exceeding 9 mm (441113): Tanned/crust hides & skins, n.e.s., (410691); Cobalt ores & concentrates (260500): Ash & residues (excl. From the manufacture of iron or steel), containing mainly zinc, hard zinc spelter (262019); Tanned/crust hides & skins of bovine or equine animals, in the wet state (including wet blue), without hair on, excluding full grains, unsplit; grain splits (410419); Zinc oxide; zinc peroxide (281700); Formic acid (291511); Rolls for rolling mills (845530); White cement, whether/not artificially coloured (252321)Other parts & accessories for the motor vehicles Regulating or controlling instruments and Togo (870899); apparatus; automatic, other than hydraulic Ball bearings (848210); or pneumatic (903289); Textile fabrics: impregnated, coated, covered or Polymers of ethylene, in primary forms, laminated with plastics, (excluding polyvinyl n.e.s. (390190): Multimeters for measuring or checking chloride, polyurethane, polyamides polyesters) (590390): voltage, current, resistance or power, Other Insulated electric conductors; for a voltage without a recording device (903031); not exceeding 1000 volts, fitted with connectors Wood, n.e.s., sawn/chipped lengthwise/sliced lengthwise/sliced/peeled, (854442); Parts of Electrical transformers, static converters whether or not planed, sanded or endand inductors (850490); jointed, of a thickness exceeding 6mm Instruments and appliances used in medical or (440729): veterinary sciences, nes (901890); Waste & scrap of stainless steel (720421); Other parts and accessories for printing machinery Recovered (waste & scrap) excl. 844391 (844399); paper/paperboard, n.e.s., and of unsorted Toothed wheels, chain sprockets and other waste and scrap (470790); transmission elements presented separately; parts Petroleum oils and oils obtained from of transmission components (848390); bituminous minerals, crude (270900); Parts of aeroplanes/helicopters, others (880330); Raw hides & skins of bovine (including Gears & gearing (not toothed wheels, chain buffalo) and equine animals, fresh, salted or sprockets and other transmission elements preserved, but not tanned, parchment presented separately); ball or roller screws; gear dressed or further prepared, whether or not boxes and other speed changers, including torque dehaired or split (410190); converters (848340): Zinc waste & scrap (790200): Pumps and compressors; parts, of air or vacuum Waste & scrap of cast iron (720410); pumps, air or other gas compressors and fans, Waste & scrap of alloy steel other than ventilating or recycling hoods incorporating a fan stainless steel (720429); Ash & residues (excl. From the manufacture (841490); Spark-ignition reciprocating/rotary internal of iron or steel), containing mainly zinc, combustion piston engines for aircrafts (840710); hard zinc spelter (262019);

Newsprint, in rolls/sheets (480100);

Aluminium waste & scrap (760200);

Structures and parts thereof of iron or steel, n.e.s. (730890);

Valves for oleohydraulic/pneumatic transmissions (848120);

Doors, windows & their frames & thresholds for doors (392520);

Tubes, pipes & hoses, rigid, of plastics n.e.s. (391729);

Dried leguminous vegetables, n.e.c., shelled, whether or not skinned or split (71390); Parts & accessories of motorcycles (mopeds), other than saddles (871419);

Plates, sheets, film, foil & strip, other than cellular (392190);

Recovered (waste & scrap) paper/paperboard, n.e.s., and of unsorted waste and scrap (470790); Other Parts suitable for use solely or principally with the apparatus of headings 8525 to 8528 (reception and transmission apparatus, monitors, etc.) (852990);

Wood sawn/chipped length wise, sliced or peeled, whether or not planed, sanded or end-jointed, of a thickness exceeding 6mm (440710);

Beans of the species vigna mungo (L.) Hepper or vigna radiata (L.) Wilczek, shelled, whether or not skinned or split, dried (71331);

Pumps n.e.s. In 84.13 (841381);

Mechanical appliances, for projecting, dispersing or spraying liquids or powders, for agricultural or horticultural use, whether or not hand-operated (842481);

Plates, sheets, film, foil & strip, of plastics n.e.s., non-cellular and not reinforced, laminated, supported or similarly combined with other materials (392099);

Iron or steel tubes, pipes and hollow profiles, seamless, n.e.s. (730490);

Chocolate & other food preparations containing cocoa; n.e.s. (180690);

Filtering/purifying machinery & apparatus, for filtering or purifying liquids, n.e.s. (842129); Ribbons for typewriters and the like, inked or otherwise prepared, for giving impressions, whether or not on spools or in cartridges (961210); Manganese ores and concentrates, including manganiferous iron ores and concentrates with a manganese content of 20% or more, calculated on the dry weight (260200);

Other printers, copying machines, single-function printing, capable of connecting to an automatic data processing machine or to a network (844332); Agarbatti & other odoriferous preparations, which operate by burning (including those used during religious rites) (330741);

Electric discharge lamps (excluding ultra-violet), fluorescent, hot cathode (853931); Synthetic filament tow, acrylic/modacrylic

(550130); Zinc waste & scrap (790200);

Aluminium waste & scrap (760200);

Cotton, not carded/combed (520100); Copper waste & scrap (740400); Cashew nuts, in shell (80131); Ferrous waste & scrap, n.e.s. (720449); Raw skins of sheep/lambs (fresh/salted/dried/limed/otherwise, preserved other than by pickling, (but not tanned, parchment-dressed or further preserved), without wool on (410229) Waste & scrap of tinned iron/steel (720430); Flat-rolled products of iron/non-alloy steel, in coils, without patterns in relief, of a width 600mm or more, hot-rolled, of a thickness of less than 3mm (720839);

Hairpins, curling pins, curling grips, hair curlers and the like and parts thereof (961590); Projectors other than of a kind solely or principally used in an automatic data processing system (852861):

Cashew nuts, in shell (080131)

Uganda

Steering wheels, steering columns and steering boxes & parts thereof (870894);

Copper plates, sheets & strip, of a thickness exceeding 0.15mm, of refined copper, not in coils (740919);

Microtomes and parts & accessories thereof (902790);

Parts of the seats of 94.01 (seats in vehicles, aircrafts, convertible to bed, etc.) (940190); Discs, tapes, solid state non volatile storage devices, smart cards and other media for the recording of sound of other phenomena, whether not recorded, incl. Matrices and masters for the production of discs, but excl. Products of Ch.37., other n.e.s. (852380);

Glass mirrors other than rear-view (700991); Gears & gearing (not toothed wheels, chain sprockets and other transmission elements presented separately); ball or roller screws; gear boxes and other speed changers, including torque converters (848340);

Mountings, fittings & similar articles, for motor vehicles, of base metal (830230);

Other parts and accessories of looms, auxiliary machinery, n.e.s. (844849);

Instruments and apparatus for measuring or checking variables of liquids or gases (excluding pressure or the flow and level of liquids and others) (902680);

Filtering/purifying machinery & apparatus, for filtering or purifying liquids, n.e.s. (842129); Non-threaded cotters & cotter-pins, of iron/steel (731824);

Plates, sheets, film, foil & strip, of polyesters n.e.s., non-cellular and not reinforced, laminated, supported or similarly combined with other materials (392069);

Wood, in the rough, n.e.s., whether or not stripped of bark or sapwood, or roughly squared, untreated, (440399):

Ferrous waste & scrap, n.e.s. (720449); Taps, cocks, valves and similar appliances; for pipes, boiler shells, tanks, vats or the like, including thermostatically controlled valves (848180);

"Balances of a sensitivity of 5cg/better, with or without weights (901600);"

Toothed wheels, chain sprockets and other transmission elements presented separately; parts of transmission components (848390); Parts and accessories for instruments, appliances and machines measuring or checking (903190);

Aluminium foil, backed with paper, paperboard, plastics or similar backing materials, of a thickness (excluding any backing) not exceeding 0.2mm (760720); Moulded/pressed articles of paper pulp (482370);

Thermostats, automatic types (903210); Aluminium waste & scrap (760200); Machinery, plant & equip., n.e.s., for treating materials by change of temperature, other than for making hot drinks or cooking or heating food (841989);

Chickpeas (garbanzos), dried, shelled, whether or not skinned or split (71320); Marble and travertine, merely cut, by sawing or otherwise, into blocks or slabs of a rectangular (including square) shape, having a specific gravity of 2.5 or more (251512);

Anti-knock preparations other than those based on lead compounds (381119); Precious (other than diamonds) and semi-precious stones, unworked or simply sawn or roughly shaped, not strung, mounted or set (710310);

Ceramic sinks, wash basins, wash basin pedestals, baths, bidets, water closet pans, flushing cisterns, urinals and similar sanitary fixtures; of porcelain or china (691010);

Instruments, appliances and machines; for measuring or checking n.e.c. (903180); Parts of the machinery of heading 8422 (842290);

Other printers, copying machines, singlefunction printing, not capable of connecting to an automatic data processing machine or to a network (844339);

Sanitary ware & parts thereof, of copper (741820);

Waste & scrap of stainless steel (720421); Parts of the steam turbines & other vapour turbines (840690); Textile furnishing articles, of cotton, not knittedd or crocheted (excluding bedspreads and other articles of heading no. 9404) (630492); Fuel/lubricating/cooling medium pumps for internal combustion piston engines (841330); Diagnostic/laboratory reagents on a on a backing and prepared diagnostic or laboratory reagents whether or not on a backing; certified reference material (382200);

Machinery, plant & equip., n.e.s., for treating materials by change of temperature, other than for making hot drinks or cooking or heating food (841989);

Pumps n.e.s. In 84.13 (841381);

Instruments and apparatus for measuring or checking the flow or level of liquids (902610); Ball bearings (848210);

Tubes, pipes & hoses, of vulcanised rubber (other than hard rubber), reinforced or otherwise combined with materials other than metal or textiles, with fittings (400942);

Plates, sheets, film, foil & strip, other than cellular (392190);

Permanent magnets of metal and articles intended to become permanent magnets after magnetization (excl. Chucks, clamps and similar holding devices) (850511);

Articles of plastics, n.e.s. (392690);

Lubricating preparations (excl. Those for the treatment of textile and similar materials), not containing petroleum oils or oils obtained from bituminous minerals (340399);

Electrical lighting or visual signalling equipment for motor vehicles (excl. Lamps of heading 8539) (851220);

Festive, carnival or other entertainment articles including novelty jokes and conjuring tricks; other than Christmas festivity articles (950590); Synthetic rubber & factice derived, in primary forms or in plates, sheets or strip (400299); Other parts & accessories for the motor vehicles (870899);

Parts of Machinery for working rubber or plastics (847790):

Vacuum cleaners, other than with a self-contained electric motor (850860);

Fluxes & other auxiliary preparations; for soldering, brazing or welding, or preparations used as cores or coatings for welding electrodes or rods (381090);

Precious (other than diamonds) and semi-precious stones, unworked or simply sawn or roughly shaped, not strung, mounted or set (710310); Organic comps. N.e.s. (294200);

Flat knitting machines; stitch-bond (844720); Paper, paperboard, cellulose wadding and webs of soft cellulose fibres; coated, impregnated, covered, surface-decorated or surface-coloured, n.e.s. (481190); Live animals, n.e.s. (10690);

Glassware of a kind used for table/kitchen purposes n.e.s., other than of lead crystal (701399);

Transmission shafts (including cam shafts and crank shafts) and cranks (848310); Unsaturated acyclic hydrocarbons, n.e.s (290129);

Parts And Accessories Of Instruments And Apparatus For Measuring, Checking Or Detecting Electrical Quantities, Or Ionizing Radiations, Others (903090);

Mixtures of odoriferous substance & and mixtures; of a kind used in the food or drink industries (330210);

Offset printing machinery, used for printing by means of plates, cylinders and other printing components of heading 84.42, n.e.s. (844319);

Wood, in the rough, n.e.s., whether or not stripped of bark or sapwood, or roughly squared, untreated, (440399);

Parts & accessories of hydrometers and similar floating instruments, barometers, hygrometers, psychrometers, thermometers, pyrometers; recording or not, any combination of these instruments (902590); Prepared enzymes, n.e.s., oother than rennet and concentrates thereof (350790); Coniferous wood, in the rough, whether/not stripped of bark or sapwood, or roughly squared, untreated (440320);

Unwrought lead other than refined, not containing by weight antimony as the principal other element (780199);

Tanned/crust hides & skins of bovine or equine animals, in the dry state (crust), without hair on, (other than full grains, unsplit; grain splits) (410449);

Jacks & hoists, (other than hydraulic), of a kind used for raising vehicles (842549); Wood sawn/chipped length wise, sliced or peeled, whether or not planed, sanded or end-jointed, of a thickness exceeding 6mm (440710);

Polyesters, n.e.s., saturated, in primary forms (390799);

Containers for compressed/liquefied gas, of iron or steel (731100);

Selenium (280490);

Motorcycles (incl. Mopeds) & cycles (871120):

Plants & parts of plants, including seeds and fruits, n.e.s., used primarily in perfumery, pharmacy or for insecticidal, fungicidal purposes; fresh or dried, whether or not cut, crushed or powdered (121190);

Nickel ores & concentrates (260400); Raw hides & skins of bovine (including buffalo) and equine animals, fresh, salted or Tube/pipe fittings of iron/steel, n.e.s,, other than stainless steel (730799);

Oil/petrol-filters for internal combustion engines (842123);

Lenses, prisms, mirrors & other optical elements, mounted, being parts or fittings for instruments or apparatus, of any material (excluding elements of glass not optically worked) (900290);

Threaded screws & bolts of iron/steel, n.e.s., whether or not with their nuts or washers (731815);

Articles & equipment for sports, for outdoor games and recreation n.e.c. (950699); Articles for funfair, table or parlour games, including pintables, special tables for casino games, automatic bowling alley equipment, n.e.c. (950490);

Radio navigational aid apparatus (852691); Parts and accessories for machine tools, for laser operation, metalworking machining centers, lathes and drilling machines, etc., others (846693); Preparations of a kind used in animal feeding, put up for retail sale (except dog or cat food) (230990);

Lubricating preparations (other than for the treatment of textile and similar materials) containing containing less than 70% (by weight) of petroleum oils or oils obtained from bituminous minerals (340319);

Slide fasteners, fitted with chain scoops of base metal (960711);

Tanned/crust hides & skins of bovine or equine animals, in the dry state (crust), without hair on, (other than full grains, unsplit; grain splits) (410449).

Footwear with outer soles of leather, uppers consisting of leather straps across instep and around the big toe (640320);

Parts of metal-rolling mills, other rolls (845590); Instruments and apparatus for physical or chemical analysis, for measuring or checking viscosity, porosity, expansion, surface tension or quantities of heat, sound or light, n.e.c. (902780); Surgical gloves of vulcanised rubber (401511); Bearings; n.e.c., including combined ball/roller (848280);

Newsprint, in rolls/sheets (480100);

Ash & residues (excl. From the manufacture of iron or steel), containing mainly zinc, hard zinc spelter (262019);

Worn clothing & other worn articles (630900); Wood, n.e.s., sawn/chipped lengthwise/sliced lengthwise/sliced/peeled, whether or not planed, sanded or end-jointed, of a thickness exceeding 6mm (440729):

Machines and appliances for testing the hardness, strength, compressibility, elasticity or other mechanical properties of materials other than metals (902480);

preserved, but not tanned, parchment dressed or further prepared, whether or not dehaired or split (410190);

Beans of the species vigna mungo (L.) Hepper or vigna radiata (L.) Wilczek, shelled, whether or not skinned or split, dried (71331);

Vegetable saps and extracts; n.e.s. In item no. 1302.1 (130219);

Kidney beans, incl. White pea beans (phaseolus vulgaris), dried, shelled, whether or not skinned or split (71333); Parts of the machinery, used in the

Parts of the machinery, used in the industrial preparation or manufacture of food or drink (843890);

Wood, n.e.s., sawn/chipped lengthwise/sliced lengthwise/sliced/peeled, whether or not planed, sanded or endjointed, of a thickness exceeding 6mm (440729);

Cotton, not carded/combed (520100); Tanned/crust hides & skins of bovine or equine animals, in the wet state (including wet blue), without hair on, excluding full grains, unsplit; grain splits (410419); Brewery machinery (843840);

Topical wood, n.e.s., in the rough, whether or not stripped of bark or sapwood, or roughly squared, untreated (440349); Dried leguminous vegetables, n.e.c., shelled, whether or not skinned or split (71390);

Beech (Fagus spp.), sawn or chipped lengthwise, sliced or peeled, whether or not planed, sanded or finger-jointed, thicker than 6mm (440792);

Sawdust & wood waste & scrap, whether or not agglomerated in logs, briquettes, pellets or similar forms (440130)

vigna radiata (L.) Wilczek, shelled, whether or not skinned or split, dried (71331); Machines and mechanical appliances; having individual functions, n.e.c. (847989); Oil seeds & oleaginous fruits, n.e.s., whether or not broken (120799); Wood sawn/chipped length wise, sliced or peeled, whether or not planed, sanded or end-jointed, of a thickness exceeding 6mm (440710); Raw skins of sheep/lambs (fresh/salted/dried/limed/otherwise, preserved other than by pickling, (but not tanned, parchmentdressed or further preserved), without wool on (410229): Waste, parings & scrap, of plastics (391590); Topical wood, n.e.s., in the rough, whether or not stripped of bark or sapwood, or roughly squared, untreated (440349); Wood, n.e.s., sawn or chipped lengthwise, sliced or peeled, whether or not planed, sanded or fingerjointed, thicker than 6mm (440799); Machinery for sugar manufacture (843830); Essential oils, n.e.s., (terpeneless or not), including concretes and absolutes (330129); Waste & scrap of cast iron (720410); Hydrometers and similar floating instruments, barometers, hygrometers, psychrometers, thermometers, pyrometers; recording or not, any combination of these instruments (excluding thermometers and barometers not combined with other instruments) (902580); Tanned/crust skins of sheep/lambs, without wool on, whether or not split, but not further prepared. in the wet state (including wet-blue) (410510); Synthetic fibre other than high tenacity/textured, filament, monofilament (less than 67 decitex), multiple (folded) or cabled, n.e.s., not for retail sale, not sewing thread (540269); Parts of the machinery, used in the industrial preparation or manufacture of food or drink (843890);Bars & rods, hot-rolled, in irregularly wound coils, n.e.s., of circular cross-section measuring 14mm or more in diameter (721399); Textile doubling/twisting machines (844530); Tanned/crust hides & skins of bovine or equine animals, in the wet state (including wet blue), without hair on, excluding full grains, unsplit; grain splits (410419); Whole bovine (incl. Buffalo)/equine animals, of a weight per skin not exceeding 8kg when simply dried, 10kg when dry-salted or 16kg when fresh, wet-salted or otherwise preserved (410120); Cobalt ores & concentrates (260500) Zambia Artists' brushes, writing brushes and similar Optical fibre cables made up of individually brushes for the application of cosmetics (960330); sheathed fibres, whether or not containing Plates, sheets & strip, of non-cellular rubber electric conductors or fitted with connectors (400821);(854470);

Beans of the species vigna mungo (L.) Hepper or

Ethylene polymers; in primary forms, polyethylene having a specific gravity of less than 0.94 (390110);

Electric lamps & lighting fittings, nec (940540); Hand-/foot-operated air pumps (841420); Electrical apparatus; n.e.c., for switching or protecting electrical circuits, for a voltage not exceeding 1000 volts (853690);

Ballasts for discharge lamps/tubes (850410); Parts of the furniture of 94.03 (metal/wooden furniture etc. For office, kitchen uses, etc.) (940390);

Spring washers & other lock washers of iron and steel, non-threaded (731821);

Moulds for rubber or plastics, other than injection or compression types (848079);

Ferrous waste & scrap, n.e.s. (720449);

Articles of jewellery & parts thereof, of precious metal (excluding silver) whether or not plated or clad with precious metal (711319);

Jacks (excl. Of 8425.41) & hoists, hydraulic, of a kind used for raising vehicles (842542);

Cashew nuts, in shell (80131);

Chandeliers and other electric ceiling or wall light fittings; excluding those used for lighting public open spaces or thoroughfares (940510);

New pneumatic tyres, of rubber (excl. Those with a herring-bone or similar tread), of a kind used on light industrial vehicles (401199);

Waste & scrap of cast iron (720410);

DC motors and generators of an output exceeding 75kw but not exceeding 375kw (850133);

Tanned/crust hides & skins of bovine or equine animals, in the dry state (crust), without hair on, (other than full grains, unsplit; grain splits) (410449);

Wood marquetry & inlaid wood; caskets and cases for jewellery or cutlery and similar articles of wood, wooden articles of furniture (442090); Lac; natural gums, gum resins, resins and balsams, n.e.s. (130190);

Unwrought lead other than refined, not containing by weight antimony as the principal other element (780199);

Whole bovine (incl. buffalo)/equine animals, of a weight per skin not exceeding 8kg when simply dried, 10kg when dry-salted or 16kg when fresh, wet-salted or otherwise preserved (410120);

Precious and semi-precious stones (excl. Diamonds, rubies, sapphires and emeralds), worked other than simply sawn or roughly shaped, not strung, mounted or set (710399);

Wattle extract (320120);

Leather further prepared after tanning or crusting, including parchment-dressed, of bovine (including buffalo) or equine animals, without hair on, split or not, (other than grain splits and full grains, unsplit) (410719);

Bars, rods & profiles, of refined copper (740710);

Articles of jewellery & parts thereof, of silver, whether or not plated or clad with other precious metal (711311);

Parts of machinery, plant/laboratory equipment, for treating materials by a process involving a change of temperature (841990);

New pneumatic tyres, of rubber, of a kind used on bicycles (401150);

Silica sands & quartz sands, whether or not coloured (250510);

Ferrous waste & scrap, n.e.s. (720449); Parts of Machinery for working rubber or plastics (847790);

Pliers (incl. Cutting pliers), pincers, tweezers and similar tools (820320); Interchangeable tools for hand tool, whether or not power-operated, for screw-driving or uses n.e.c (820790);

Ceramic tableware, kitchenware, household articles and toilet articles, other than of porcelain or china (691200);

Machinery for moulding or forming, for working rubber or plastics or for the manufacture of products from these materials, n.e.c. (847759);

Instruments and apparatus for measuring or checking voltage, current, resistance or power, with a recording device (excluding multimeters) (903039);

Tools for boring/broaching, for hand tools, whether or not power-operated (820760); Electrical insulators (excl. Those of glass or ceramics and insulating fittings) (854690); Electrical apparatus for switching or protecting electrical circuits, or for making connections to or in electrical circuits, for a voltage > 1.000 V (excl. Fuses, automatic circuit breakers, isolating switches, makeand-break switches, lightning arresters, voltage limiters, surge suppressors and control desks, cabinets, panels etc. Of heading 8537) (853590);

Textile furnishing articles, of textile materials other than cotton or synthetic fibres, not knittedd or crocheted (excluding bedspreads and other articles of heading no. 9404) (630499);

Waste & scrap of alloy steel other than stainless steel (720429);

Lecithins and other phosphoaminolipids, whether or not chemically defined (292320);

Instruments, appliances and machines; for measuring or checking n.e.c. (903180); Ferro-manganese, containing by weight 2% or less of carbon (720219);

Parts & accessories of the machines or auxiliary machinery for preparing textile fibres, other than card clothing (844832); Tanned/crust hides & skins of bovine or equine animals, in the wet state (including wet blue), without hair on, excluding full grains, unsplit; grain splits (410419);

Lead ores & concentrates (260700);

Copper ores & concentrates (260300);

Tanned/crust skins of sheep/lambs, without wool on, whether or not split, but not further prepared, in the wet state (including wet-blue) (410510); Copper waste & scrap (740400);

Manganese ores and concentrates, including manganiferous iron ores and concentrates with a manganese content of 20% or more, calculated on

Cotton, not carded/combed (520100)

the dry weight (260200):

Weighing machinery, having a maximum weighing capacity exceeding 30kg but not exceeding 5000kg (842382);

Lead waste & scrap (780200);

Tanned/crust hides & skins of bovine or equine animals, in the dry state (crust), without hair on, (other than full grains, unsplit; grain splits) (410449);

New pneumatic tyres, of rubber (excl. Those with a herring-bone or similar tread), of a kind used on light industrial vehicles (401199):

Precious metal ores & concentrates (excl. Silver) (261690);

Tools for drilling, other than rock, for machine or hand tools, whether or not power-operated (820750);

Manganese oxides other than manganese dioxide (282090);

Leather further prepared after tanning incl. Parchment-dressed, of bovine (including buffalo) or equine animals, no hair, and whole hides and skins, and sides, (full grains, unsplit and grain splits) (410799); Raw hides & skins of bovine (including buffalo) and equine animals, fresh, salted or preserved, but not tanned, parchment dressed or further prepared, whether or not dehaired or split (410190);

Parts of the machinery, used in the industrial preparation or manufacture of food or drink (843890);

Mixtures of odoriferous substance and mixtures, used as raw materials in industries other than the food or drink industries (330290);

Parings & other waste of leather/composition leather; not suitable for the manufacture of leather articles; leather dust, powder and flour (411520); Tanned/crust hides & skins of bovine or equine animals, in the wet state (including wet blue), without hair on, excluding full grains, unsplit; grain splits (410419); Ash & residues (excl. From the manufacture of iron or steel), containing mainly copper (262030):

Whole bovine (incl. Buffalo)/equine animals, of a weight exceeding 16 kg, dried, fresh, wet-salted or otherwise preserved (410150);

Manganese ores and concentrates, including manganiferous iron ores and concentrates with a manganese content of 20% or more, calculated on the dry weight (260200); Unwrought lead other than refined, not containing by weight antimony as the principal other element (780199)

Source: Own Calculations, UN Comtrade

Table 2: Products identified under Quadrant B for selected Asian countries

Country	2010-2012	2015-2017
Bangladesh	Plaited bands, slings & the like, of iron/steel, not	Tools for pressing/stamping/punching, for
	electrically insulated (731290);	machine or hand tools, whether or not power-
	Parts of taps, cocks, valves and similar	operated (820730);
	appliances (848190);	Tapered roller bearings, incl. Cone and
	Processing units for automatic data processing	tapered roller assemblies (848220);
	machines, whether or not containing in the same	Plates, sticks, tips & the like for tools,
	housing one or two of the following types of unit	unmounted, of sintered metal carbides or
	storage units, input units, output units (847150);	cermets (820900);
	Paints & varnishes (incl. Enamels & lacquers), based on polymers n.e.s., dispersed or dissolved	Line pipe of a kind used for oil/gas, of stainless steel, seamless (730411);
	in a non-aqueous medium (320890);	Air conditioning machines comprising a
	Non-threaded articles, n.e.s., of iron/steel	motor-driven fan and elements for changing
	(731829);	the temperature and humidity, window or wall
	Threaded elbows, bends & sleeves of material	types, self-contained or split-system (841510);
	other than stainless steel (730792);	Non-threaded cotters & cotter-pins, of
	Epoxide resins, in primary forms (390730);	iron/steel (731824);
	Office machines; automatic banknote dispensers,	Parts and accessories for machine tools, for
	coin-sorting machines, coin-counting or	laser operation, metalworking machining
	wrapping machines, pencil-sharpening machines,	centers, lathes and drilling machines, etc.,
	perforating or stapling machines (847290);	others (846693);
	Plates, sheets, film, foil & strip, other than	Parts of the air-conditioning machines, with
	cellular (392190);	motor driven fan and elements for temperature
	Bearings; n.e.c., including combined ball/roller	control (841590);
	(848280);	Parts & accessories of motorcycles (mopeds),
	Organic comps. N.e.s. (294200);	other than saddles (871419);
	Drilling machines for drilling by removing metal, other than numerically controlled	Loudspeakers; not mounted in their enclosures (851829);
	(845929);	Oxidation inhibitors, gum inhibitors, viscosity
	Newsprint, in rolls/sheets (480100);	improvers, anti-corrosive preparations, other
	Parts & accessories n.e.s. In Ch.90 (903300);	prepared additives for mineral oils or liquids
	Parts and accessories for instruments, appliances	used as mineral oils (including gasoline), n.e.s.
	and machines measuring or checking (903190);	(381190);
	Fixed electrical resistors for a power handling	Leaf-springs & leaves therefor, of iron/steel
	capacity <= 20 W (excl. Heating resistors)	(732010);
	(853321);	Paints & varnishes (incl. Enamels & lacquers),
	Parts of cranes, work trucks, shovels, constr	(based on polymers other than acrylic or
	machine (843149);	vinyl), dispersed or dissolved in an aqueous
	Parts of machinery for sorting, screening,	medium (320990);
	separating, washing, crushing etc mineral	Flexible tubing of base metal, with or without
	substances, for agglomerating, shaping or	fittings, of iron or steel (830710);
	moulding solid fuels, ceramic pastes etc, for forming foundry moulds of sand (847490);	Vaccines for human medicine (300220);
	Air compressors mounted on a wheeled chassis	Parts of telephone sets, incl. Telephones for cellular networks for other wireless networks
	for towing (841440);	other apparatus for the transmission reception
	Table/floor/wall/window/ceiling/roof fans, with a	of voice, images other data, incl. Apparatus
	self-contained electric motor of an output not	for communication in a wired wireless
	exceeding 125W (841451);	network (851770);
	Parts suitable for use solely or principally with	Saccharin & its salts (292511);
	electric motors and generators, electric	Electric conductors, for a voltage > 1.000 V,
	generating sets and rotary converters, n.e.s.	insulated, n.e.s. (854460);
	(850300);	Iron or steel threaded articles except screw,
	Diagnostic/laboratory reagents on a on a backing	nut,bolts (731819);
	and prepared diagnostic or laboratory reagents	Other apparatus for transmission reception of
	whether or not on a backing; certified reference	voice, images other data, incl. Apparatus for
	material (382200);	communication in a wired wireless network
		(such as a local wide area network) (851769);

Sewing thread of artificial filaments, whether or not put up for retail sale (540120);

Textile fabrics; otherwise impregnated, coated or covered, painted canvas being theatrical scenery, studio back-cloths or the like (590700);

Electronic integrated circuits; processors and controllers, whether or not combined with memories, converters, logic circuits, amplifiers, clock and timing circuits, or other circuits (854231):

Mechanical seals (848420);

Other Electro-magnets and their parts (excl. Magnets for medical use); electro-magnetic or permanent magnet chucks, clamps and similar holding devices and their parts, n.e.s. (850590); Other parts and accessories of looms, auxiliary machinery, n.e.s. (844849);

Discs, tapes, solid state non volatile storage devices, smart cards and other media for the recording of sound of other phenomena, whether not recorded, incl. Matrices and masters for the production of discs, but excl. Products of Ch.37., other n.e.s. (852380);

Woven fabrics of flax, containing less than 85% flax, other than unbleached or bleached (530929):

Air conditioning machines comprising a motordriven fan and elements for changing the temperature and humidity, window or wall types, self-contained or split-system (841510); Instruments, appliances and machines; for measuring or checking n.e.c. (903180); Parts of electrical ignition or starting equipment,

generators, etc. Of heading 8511, n.e.s. (851190);

Chickpeas (garbanzos), dried, shelled, whether or

not skinned or split (71320); Magnetic or optical readers, machines for transcribing data onto data media in coded form and machines for processing such data, n.e.s.

and machines for processing such data, n.e.s. (847190); Millstones, grindstones, grinding wheels and the

Millstones, grindstones, grinding wheels and the like; of natural stone (680423);

Cast articles of iron (excl. Non-malleable cast iron), excluding grinding balls and similar articles for mills (732599);

Parts & accessories of the machines or auxiliary machinery for preparing textile fibres, other than card clothing (844832);

Parts of aeroplanes/helicopters, others (880330); Articles of iron/steel, n.e.s., forged/stamped, but not further worked (732619);

Parts & accessories of regulating or controlling instruments and apparatus; automatic (903290); Millstones, grindstones, grinding wheels and the like; of other agglomerated abrasives or of ceramics (680422);

Taps, cocks, valves and similar appliances; for pipes, boiler shells, tanks, vats or the like, including thermostatically controlled valves (848180);

Beans n.e.s., dried/shelled, including seed (71339);

Jacks & hoists, (other than hydraulic), of a kind used for raising vehicles (842549); Taps, cocks, valves and similar appliances; for pipes, boiler shells, tanks, vats or the like, including thermostatically controlled valves (848180);

Electrical apparatus for switching or protecting electrical circuits, or for making connections to or in electrical circuits, for a voltage > 1.000 V (excl. Fuses, automatic circuit breakers, isolating switches, make-and-break switches, lightning arresters, voltage limiters, surge suppressors and control desks, cabinets, panels etc. Of heading 8537) (853590):

Machines for the reception, conversion and transmission regeneration of voice, images other data, incl. Switching and routing apparatus (851762);

Gas-operated machinery & apparatus, n.e.s., for soldering, brazing or welding, whether or not capable of cutting (846820);

Bars, rods & profiles, of copper alloys (other than copper-zinc base alloys) (740729); Activated carbon (380210);

Polymers of vinyl chloride/of other halogenated olefin, n.e.s. (390490); Human blood; animal blood prepared for therapeutic, prophylactic or diagnostic uses; cultures of micro-organisms (excluding yeast) and similar products (300290);

Sewing needles, knitting needles, bodkins, crochet hooks and similar articles, for use in the hand, n.e.s., of iron/steel (731990); Polyesters, n.e.s., saturated, in primary forms (390799);

Stranded wire, cables, plaited band and the like, of copper, not electrically insulated (741300);

Parts suitable for use solely or principally with the apparatus of heading 8535, 8536 or 8537, n.e.s. (excl. Boards, panels, consoles, desks, cabinets and other bases for the goods of heading 8537, not equipped with their apparatus) (853890);

Portable automatic data processing (ADP) machines, weighing not more than 10kg, consisting of at least a central processing unit, a keyboard and a display (847130); Copper tubes & pipes, of refined copper (741110):

Dies for drawing/extruding metal, for machine or hand tools, whether or not power-operated (820720):

Ethylene polymers; in primary forms, polyethylene having a specific gravity of less than 0.94 (390110);

Air-conditioning machines, comprising a motor driven fan and elements for changing the temperature and humidity, of a kind used for persons, in motor vehicles (841520);

Transmission belts/belting, n.e.s., of vulcanized rubber (401039);

Electrical measurement instruments nes., without a recording device (903089);

Parts of mechanical appliances (parts of machines projecting, dispersing or spraying liquids or powders, whether or not hand-operated) (842490);

Polyethylene having a sp.gr. Of 0.94 or more (390120);

Tools for drilling, other than rock, for machine or hand tools, whether or not power-operated (820750);

Electrical apparatus like switches n.e.c. In heading no. 8536, for a voltage not exceeding 1000 volts (853650);

Multiple loudspeakers, mounted in the same enclosure (851822);

Organic surface-active agents, surface-active, whether or not containing soap, including auxiliary washing preparations, not for retail sale (340290);

Regulating or controlling instruments and apparatus; automatic, other than hydraulic or pneumatic (903289);

Instruments and appliances used in medical or veterinary sciences, nes (901890);

Ballasts for discharge lamps/tubes (850410); Buckets, shovels, grabs etc for excavating machinery (843141):

Machinery, n.e.s. In heading no. 8441, for making up paper pulp, paper or paperboard (844180):

Parts suitable for use solely or principally with the apparatus of heading 8535, 8536 or 8537, n.e.s. (excl. Boards, panels, consoles, desks, cabinets and other bases for the goods of heading 8537, not equipped with their apparatus) (853890);

Parts of machines and apparatus for soldering or welding or for hot spraying of metals, metal carbides or cermets, n.e.s. (851590); Ball bearings (848210);

Colouring matter & other preparations, n.e.s. (320649):

Electrical lighting or visual signalling equipment for motor vehicles (excl. Lamps of heading 8539) (851220);

Plugs and sockets for a voltage <= 1000 V (excl. Lamp-holders) (853669);

Gummed/adhesive paper,paperboard (excluding self-adhesive), n.e.s. (481149);

Coated electrodes of base metal, for electric arcwelding (831110);

Machinery, plant & equip., n.e.s., for treating materials by change of temperature, other than

Tube/pipe fittings of stainless steel, n.e.s (730729);

Paper & paperboard, coated with inorganic substances only, whether or not surface-coloured, surface-decorated or printed, n.e.s., in rolls or sheets (481099);

Heterocyclic comps. With nitrogen heteroatom only, n.e.s. (293399);

Lace mechanically made in the piece, in strips or in motifs, of man-made fibres (580421); Knitting Machines, Others, And Machines For Making Gimped Yarn, Tulle, Lace, Embroidery, Trimmings, Braid Or Net And Machines For Tufting (844790);

Parts of the sewing machines n.e.s. (845290); Tubes, pipes & hoses, of vulcanised rubber (other than hard rubber), reinforced or otherwise combined with materials other than metal or textiles, with fittings (400942); Pulleys and flywheels, including pulley blocks (848350);

Gaskets and similar joints of metal sheeting combined with other material or two or more layers of metal (848410);

Gears & gearing (not toothed wheels, chain sprockets and other transmission elements presented separately); ball or roller screws; gear boxes and other speed changers, including torque converters (848340); Rubies, sapphires & emeralds, worked (other than simply sawn or roughly shaped), not strung, mounted or set (710391); Ceramic sinks, wash basins, wash basin pedestals, baths, bidets, water closet pans, flushing cisterns, urinals and similar sanitary fixtures; of other than porcelain or china (691090):

Electronic integrated circuits; processors and controllers, whether or not combined with memories, converters, logic circuits, amplifiers, clock and timing circuits, or other circuits (854231);

Non-threaded articles, n.e.s., of iron/steel (731829);

Refrigerators, household type, electric or not, other than compression-type (841829); Monofilament of which any cross-sectional dimension exceeds 1mm, rods, sticks and profile shapes, whether or not surface-worked but not otherwise worked (391690); Machinery for making/repairing footwear, other than sewing machines (845320); Ball bearings (848210); Pneumatic elevators & conveyors (842820);

Aluminium foil, backed with paper, paperboard, plastics or similar backing materials, of a thickness (excluding any backing) not exceeding 0.2mm (760720);

for making hot drinks or cooking or heating food (841989);

Pigments & preparations based thereon (320417);

Spectrometers, spectrophotometers and spectrographs using optical radiations (UV, visible, IR) (902730);

Inductors (excl. Inductors for discharge lamps or tubes) (850450);

Aluminium plates, sheets & strip, (not rectangular or square), of thickness exceeding 0.2mm, alloys (760692);

Brakes & servo-brakes; parts thereof (vehicles) (870830);

Articles of iron/steel wire (732620);

Parts and accessories for instruments/apparatus measuring or checking the flow, level, pressure or other variables of liquids or gases (902690); Hard rubber (e.g., ebonite) in all forms, including waste and scrap, and articles of hard rubber (401700);

Jacks & hoists, (other than hydraulic), of a kind used for raising vehicles (842549);

Parts of Machinery for working rubber or plastics (847790);

Machines for agglomerating, shaping or moulding solid mineral fuels, ceramic paste, unhardened cements, plastering materials in powder or paste form, machines for forming foundry moulds of sand (847480);

Boards, panels, consoles, desks, cabinets and other bases for the goods of heading 8537, not equipped with their apparatus (853810);

Knitting Machines, Others, And Machines For Making Gimped Yarn, Tulle, Lace, Embroidery, Trimmings, Braid Or Net And Machines For Tufting (844790);

Prepared glues & other prepared adhesives, not exceeding a net weight of 1kg (350699);

Clutches and shaft couplings (including universal joints) (848360);

Carding machines (844511);

Loudspeakers; not mounted in their enclosures (851829);

Parts of electric filament or discharge lamps, sealed beam lamp units, ultra-violet or infra-red lamps and arc-lamps, n.e.s. (853990);

Fixed electrical capacitors (excl. Tantalum, aluminium electrolytic, ceramic, paper, plastic and power capacitors) (853229);

Plates, sheets, film, foil & strip, non-cellular and not reinforced, laminated, supported or similarly combined with other materials (392049);

Other centrifuges, incl. Centrifugal dryers (but not clothes-dryers) (842119);

Static converters (850440);

Filtering/purifying machinery & apparatus, for filtering or purifying liquids, n.e.s. (842129);

Pulley tackle & hoists other than skip hoists or hoists of a kind used for raising vehicles, not powered by an electric motor (842519); Disperse dyes & preparations based thereon (320411);

Non-coniferous Wood (including strips & friezes for parquet flooring), continuously shaped along any edges, ends or faces, whether or not planed, sanded or end-jointed (440929):

Tubular metal needles & needles for sutures (901832);

Other colour reception apparatus for television, whether not incorporating radio broadcast receivers sound video recording reproducing apparatus, (852872);

Polycarbonates, in primary forms (390740); Polishes, creams & similar preparations; for footwear or leather (340510);

Microphones and stands therefor (excl. Cordless microphones with built-in transmitter) (851810);

Sharpening (tool/cutter grinding) machines, other than numerically controlled (846039); Parts of taps, cocks, valves and similar appliances (848190);

Electric motors of an output not exceeding 37.5W (850110);

Organic surface-active agents, whether or not put up for retail sale (340219);

Articles of stone/of other mineral substances, n.e.s. (681599);

Optical fibre cables made up of individually sheathed fibres, whether or not containing electric conductors or fitted with connectors (854470);

Parts of machinery, plant/laboratory equipment, for treating materials by a process involving a change of temperature (841990); Machines and appliances for testing the hardness, strength, compressibility, elasticity or other mechanical properties of materials other than metals (902480);

Other printers, copying machines, single-function printing, capable of connecting to an automatic data processing machine or to a network (844332);

Aluminium foil, not backed, not rolled, of a thickness not exceeding 0.2mm (760719); Air compressors mounted on a wheeled chassis for towing (841440);

Parts of the steam turbines & other vapour turbines (840690);

Machines for mixing, kneading, crushing, grinding, screening, sifting, homogenising, emulsifying or stirring (847982);

Other Machine-tools for working stone, ceramics, concrete, asbestos-cement or like mineral materials or for cold working glass, Parts of the machinery, used in the industrial preparation or manufacture of food or drink (843890);

Clasps, frames with clasps, buckles, buckle clasps; of base metal, of a kind used for clothing, footwear, awnings, handbags, travel goods or other made up articles, including parts (830890); Portable automatic data processing (ADP) machines, weighing not more than 10kg, consisting of at least a central processing unit, a keyboard and a display (847130);

Rivets of iron/steel (731823);

Coaxial cable and other coaxial electric conductors, insulated (854420):

Pumps n.e.s. In 84.13 (841381);

Textile spinning machines (844520);

Automatic circuit breakers for a voltage <= 1.000 V (853620);

X-ray tubes (902230);

Parts and accessories of data processing equipment (847330);

DC Electric motors and generators of an output not exceeding 750W (850131);

Filtering/purifying machinery & apparatus (842121);

Centrifugal pumps, n.e.s. In 8413, for liquids (841370);

Other Electronic integrated circuits, other than Amplifiers Memories Processors and controllers (854239);

Parts of fork lift etc., trucks (843120); Containers (including containers for transport of fluids) specially designed and equipped for carriage by one or more modes of transport (860900);

Textile fabrics; impregnated, coated, covered or laminated with poly(vinyl chloride) (590310); Paper, paperboard, cellulose wadding and webs of soft cellulose fibres; coated, impregnated, covered, surface-decorated or surface-coloured, n.e.s. (481190);

Stamping foils (321210);

Parts of tools for working in the hand, parts thereof for other than chain saws and pneumatic tools (846799);

Winding wire for electrical purposes, of copper, insulated (854411);

Other parts & accessories for the motor vehicles (870899);

Instruments and apparatus for physical or chemical analysis, for measuring or checking viscosity, porosity, expansion, surface tension or quantities of heat, sound or light,

n.e.c. (902780);

Parts of Machines & mechanical appliances of those having individual functions (847990); Parts of hoists and winches (843110); Lamps and light fittings; parts thereof, of materials other than glass or plastics (940599);

(other than sawing, grinding or polishing machines) (846490);

Bearings parts, (other than balls, needles and rollers) (848299);

Optical media for the recording of sound of other phenomena, but excl. Products of Ch. 37. (852340);

Dried leguminous vegetables, n.e.c., shelled, whether or not skinned or split (71390); Cases, boxes, crates, drums & similar packings, of wood (441510);

AC motors, multi-phase, of an output > 37.5 W but <= 750 W (850151);

Textile fabrics; otherwise impregnated, coated or covered, painted canvas being theatrical scenery, studio back-cloths or the like (590700);

Locks (key/combination/electrically operated), other than those for motor vehicles or furniture, of base metal (830140); Gummed/adhesive paper,paperboard,

Gummed/adhesive paper,paperboard pressure-sensitive (481141);

Parts of microphones, loudspeakers, headphones, earphones or audio-frequency electric amplifiers, n.e.s. (851890); Organic comps. N.e.s. (294200);

Tubes, pipes & hoses, of vulcanised rubber, reinforced or otherwise combined only with

metal, with fittings (400922); Nickel, not alloyed, unwrought (750210); Tools for milling, for hand tools, whether or not power-operated (820770);

Heterocyclic comps, containing an unfused imidazole ring (whether or not hydrogenated) in the structure, other than hydantoin and its derivatives (293329);

Laboratory/hygienic/pharmaceutical glassware, whether or not graduated or calibrated, of glass n.e.s. (701790); Parts of Machinery for working rubber or

plastics (847790);

Melamine (293361);

Soap in other forms, n.e.s. (340120); Bearings; n.e.c., including combined ball/roller (848280);

Self-adhesive plates, sheets, film, foil, tape, strip, other flat shapes thereof, other than in rolls of a width not exceeding 20cm (391990); Lifts (i.e., passenger elevators) & skip hoists (842810);

Pumps n.e.s. In 84.13 (841381); Synthetic/reconstructed precious/semi

precious stones, worked or graded or not, (but not strung, mounted or set), temporarily strung for transport convenience (710490);

Furniture of cane, osier, or similar materials (other than bamboo or rattan) (940389);

Diamonds, non-industrial, (other than unworked or simply sawn, cleaved or bruted), but not mounted or set (710239);

Parts & accessories of hydrometers and similar floating instruments, barometers, hygrometers, psychrometers, thermometers, pyrometers; recording or not, any combination of these instruments (902590);

Machines for the reception, conversion and transmission regeneration of voice, images other data, incl. Switching and routing apparatus (851762);

Gears & gearing (not toothed wheels, chain sprockets and other transmission elements presented separately); ball or roller screws; gear boxes and other speed changers, including torque converters (848340);

New pneumatic tyres, of rubber, of a kind used on buses or lorries (401120);

Natural/artificial abrasive powder/grain, materials n.e.s., whether or not cut to shape or sewn or otherwise made up (680530);

Glazed ceramic tiles, cubes and similar articles, whether or not rectangular or on a backing, with the largest surface area<7cm square (690810); AC generators, (alternators), of an output not exceeding 75kva (850161);

Mixtures of odoriferous substance & and mixtures; of a kind used in the food or drink industries (330210);

Nonwovens, whether/not

impregnated/coated/covered/laminated, of manmade filaments, (weighing not more than 25g/m2) (560311);

Parts & accessories of the machines for knitting, embroidery, lace machine, other than sinkers, needles and other articles used in forming stitches (844859):

Electric motors of an output not exceeding 37.5W (850110);

Wooden doors & their frames & thresholds (441820);

"Electric Filament lamps (excl. Tungsten halogen filament lamps, lamps of a power <= 200 W and for a voltage > 100 V and ultra-violet or infra-red lamps)

(853929);"

Plates, sheets & strip, of non-cellular rubber (400821);

Other apparatus for transmission reception of voice, images other data, incl. Apparatus for communication in a wired wireless network (such as a local wide area network) (851769); Non-threaded washers, of iron/steel, excluding spring and lock (731822);

Gas/smoke analysis apparatus for physical or chemical analysis (902710);

Ultrasonic scanning apparatus (901812); Parts of electric sound or visual signalling apparatus, n.e.s. (853190);

Aluminium waste & scrap (760200); Other parts and accessories for printing machinery excl. 844391 (844399); Winding wire, other than of copper (854419); Polyacetals, in primary forms (390710); Sugars; lactose and lactose syrup, containing by weight less than 99% lactose, expressed as anhydrous lactose, calculated on the dry matter (170219);

Electric lamps & lighting fittings, nec (940540);

Parts of machines and apparatus for soldering or welding or for hot spraying of metals, metal carbides or cermets, n.e.s. (851590); Insulated electric conductors for a voltage not exceeding 1000 volts, not fitted with connectors (854449):

Paper & paperboard of a kind used for printing & writing, graphics, containing no, or not more than 10% by weight of total fibres obtained by mechanical or chemi-mechanical process, in rolls (481013);

Parts of Machines & mechanical appliances of those having individual functions (847990); Parts & accessories for the metal shaping machine tools (846694);

High tenacity yarn other than textured year, of polyesters, not for retail sale (540220); Floating structures viz. Tanks, coffer-dams, landing stages, buoys and beacons (890790); Plaited bands, slings & the like, of iron/steel, not electrically insulated (731290); Other stemware drinking glasses, other than of lead crystal or glass-ceramics (701328); Wheel rims & spokes for bicycles (871492);

Woven fabrics of synthetic staple fibres, containing 85% or more by weight of acrylic or modacrylic staple fibres, other than unbleached or bleached (551229); Vitamins & their derivatives, unmixed

Vitamins & their derivatives, unmixed (293629);

Kraft paper & paperboard, n.e.s., coated with kaolin or other inorganic substances only, for non-graphic purposes, in rolls or sheets (481039);

Articles of aluminium n.e.s. (761699); Parts of machinery for sorting, screening, separating, washing, crushing etc mineral substances, for agglomerating, shaping or moulding solid fuels, ceramic pastes etc, for forming foundry moulds of sand (847490); Parts of the machinery of heading 8422 (842290);

Clasps, frames with clasps, buckles, buckle clasps; of base metal, of a kind used for clothing, footwear, awnings, handbags, travel goods or other made up articles, including parts (830890);

Other parts and accessories of looms, auxiliary machinery, n.e.s. (844849);

Electric machines and apparatus for laser or other light or photon beam, ultrasonic, electron beam, magnetic pulse or plasma arc Tubes, pipes & hoses, of vulcanised rubber, reinforced or otherwise combined only with metal, without fittings (400921);

Aluminium foil, not backed, not rolled, of a thickness not exceeding 0.2mm (760719); Polyamides, n.e.s., in primary forms (390890); Worn clothing & other worn articles (630900); Insulating fittings for electrical purposes, of plastics (854720);

Pumps for liquids, fitted/designed to be fitted with a measuring device, other than pumps for dispensing fuel or lubricants (841319); Moulds for rubber or plastics, other than injection or compression types (848079); Parts of lifts, skip hoist or escalators (843131); Articles of invellers & parts thereof of precious

Articles of jewellery & parts thereof, of precious metal (excluding silver) whether or not plated or clad with precious metal (711319);

Telephones for cellular networks or for other wireless networks (851712);

Microtomes and parts & accessories thereof (902790);

Transformers having a power handling capacity <= 1 kva (excl. Liquid dielectric transformers) (850431):

Parts and accessories (excl. Pick-up cartridges) suitable for use solely or principally with sound reproducing and recording apparatus and with video equipment for recording and reproducing pictures and sound (excl. Pick-up devices for grooved recording media) (852290);

Polyamide-6/-11/-12/-6,6/-6,9/-6,10 or -6,12, in primary forms (390810);

Poly(vinyl chloride), plasticised, in primary forms, mixed with other substances (390422); Woven fabrics, containing 85% or more by weight of textured polyester filaments, dyed (540752);

Screwdrivers (820540);

Electric ovens, cookers, cooking plates and boiling rings, electric grillers and roasters, for domestic use (excl. Space heating stoves and microwave ovens) (851660);

Apparatus based on use of x-rays and similar; parts and accessories (x-ray generators, tubes, high tension generators, control panels and desks, screens, examination or treatment tables, chairs and like (902290);

Carbon brushes for electrical purposes (854520); Fire extinguishers, whether/not charged (842410);

Lighting sets of a kind used for Christmas trees (940530);

Bakery machinery & machinery for the manufacture of macaroni, spaghetti or similar products (843810);

Press-fasteners, snap-fasteners, press-studs and parts (960610);

Relays, for a voltage not >60V (853641);

welding; electric machines and apparatus for hot spraying of metals, metal carbides or cermets (excl. Metal spray guns specified elsewhere) (851580);

Shearing machines for working metal (846239);

Magnetic resonance imaging apparatus (901813);

Sets of articles of 2/more of the foregoing items (820590):

Lubricating preparations (excl. Those for the treatment of textile and similar materials), not containing petroleum oils or oils obtained from bituminous minerals (340399);
Carboxylic acids, (with aldehyde or ketone function but without other oxygen function), their anhydrides, halides, peroxides, peroxyacids and their derivatives (291830);
Knittedd/crocheted fabrics, n.e.s., and other than those made of wool, fine animal hair, cotton, synthetic or artificial fibres (600690);
Moulds for rubber or plastics, injection or compression types (848071);

Knittedd/crocheted fabrics, n.e.s., of synthetic fibres, dyed (600632);

Other articles of glass, n.e.s. (702000); Cast tube/pipe fittings, other than of non-malleable cast iron (730719);

Machines and mechanical appliances; having individual functions, n.e.c. (847989); Copper tube/pipe fittings (e.g. Couplings, elbows, sleeves) of refined copper (741210);

Aluminium waste & scrap (760200);

Woven fabrics, of artificial filament yarn, containing 85% or more by weight of artificial filament, strip or the like, printed (540824); Lace mechanically made in the piece, in strips or in motifs, of textile materials (580429); Sign-plates, name-plates, address-plates and similar plates, numbers, letters and other symbols; of base metal, excluding those of heading no. 9405 (831000);

Moulds for rubber or plastics, other than injection or compression types (848079); Parts of metal-rolling mills, other rolls (845590);

Furnace burners, for pulverized solid fuel or gas, including combination burners (841620); Refrigerating or freezing equipment; parts thereof, other than furniture (841899); Machines for extruding, drawing, texturing or cutting man-made textile materials (844400); Statuettes & other ornamental articles (392640);

Machine-tools for wood, cork or hard plastic, etc. N.e.s. (846599);

Vacuum pumps (841410);

Melamine resins, in primary forms (390920);

Acrylonitrile-butadiene-styrene (ABS) copolymers, in primary forms (390330); Mechanical appliances, for projecting, dispersing or spraying liquids or powders, for agricultural or

or spraying liquids or powders, for agricultural or horticultural use, whether or not hand-operated (842481);

Machines and mechanical appliances; having individual functions, n.e.c. (847989); Cathode-ray television picture tubes, including video monitor cathode-ray tubes, colour (854011):

Umbrellas & sun umbrellas (excl. Garden and similar umbrellas) (660199);

Pumps for dispensing fuel/lubricant of the type used in filling-stations or in garages, fitted or designed to be fitted with a measuring device (841311);

Paper, paperboard, cellulose wadding or webs of cellulose fibres (482390);

Buttons, other than of base metal or plastics, not covered with textile material (960629); Melamine (293361);

Tanks, casks, drums, cans, boxes and similar containers for any material (excluding compressed or liquefied gas) less than 50l capacity, n.e.s., of iron or steel (731029); Parts of the turbo-jets/turbo-propellers (841191); Moulds for rubber or plastics, injection or compression types (848071);

Hand saws (820210);

Hand tools, other than household tools (820559); Organo-inorganic compounds, n.e.s. (293100); Parts & accessories of motorcycles (mopeds), other than saddles (871419):

Paintings, drawings and pastels; executed entirely by hand (970110);

Tanned/crust skins of sheep/lambs, without wool on, whether or not split, but not further prepared, in the wet state (including wet-blue) (410510); Machinery, plant and equipment, n.e.s., for making hot drinks, cooking or heating food (841981);

Pallets, box pallets & other load boards; pallet collars (441520);

Gummed/adhesive paper,paperboard, pressuresensitive (481141);

Speed indicators & tachometers; stroboscopes (902920);

Parts & accessories for the metal shaping machine tools (846694);

Soap in other forms, n.e.s. (340120);

Parts and accessories of machines or auxiliary machinery, Spindles, spindle flyers, spinning rings and ring travellers (844833);

Plates, sheets, film, foil&strip, of polymers of propylene, non-cellular and not reinforced, laminated, supported or similarly combined with other materials (392020);

Needle roller bearings (848240);

Angles, shapes & sections of iron/non-alloy steel, hot-rolled, hot-drawn or extruded, of a height of 80mm or more (721640);

Fluxes & other auxiliary preparations; for soldering, brazing or welding, or preparations used as cores or coatings for welding electrodes or rods (381090);

Flanges of stainless steel (730721);

Plates, sheets, film, foil & strip, other than cellular (392190);

Flexible tubes, pipes & hoses of plastic, having a minimum burst pressure of 27.6mpa (391731);

Office/school supplies, of plastics (392610); Refractory ceramic goods other than those of siliceous fossil meals or similar earths, containing by weight more than 50% of alumina or of a mixture or compound of alumina and of silica (690320);

Warp knit fabrics, including those made on galloon knitting machines, n.e.s., excluding those made from cotton or synthetic or artificial fibres (600590);

Wooden furniture of a kind used in bedroom (940350):

Aluminium alloys, bars, rods and profiles, other than hollow (760429);

Paints & varnishes (incl. Enamels & lacquers), based on acrylic or vinyl polymers, dispersed or dissolved in an aqueous medium (320910); Organic surface-active agents, surface-active, whether or not containing soap, including auxiliary washing preparations, not for retail sale (340290):

Butt welding fittings of stainless steel (730723);

Articles of apparel & clothing accessories (other than gloves, mittens and mitts), of vulcanized rubber (401590);

Tooth brushes, incl. Dental-plate brushes (960321);

Table/kitchen/other h-hold. Articles and their parts, enamelled, of Iron (excluding cast) or steel (732394);

Compounded rubber (other than with carbon black or silica), vulcanized, in plates, sheets and strip (400591);

Precious and semi precious stones (excl. Diamonds, rubies, sapphires and emeralds), worked other than simply sawn or roughly shaped, not strung, mounted or set (710399); Centrifugal pumps, n.e.s. In 8413, for liquids (841370);

Hooks, eyes & eyelets of a kind used for clothing, footwear, awnings, handbags, travel goods or other made up articles, of base metal (830810);

Woven fabrics of cotton, containing 85% or more by weight of cotton, of yarns of different

High tenacity yarn other than textured yarn, not for retail sale (540219);

Cement clinkers (252310);

Automatic data processing machines; comprising in the same housing at least a central processing unit and an input and output unit, whether or not combined, n.e.c. (Digital computer CPU with some of storage, input-output units) (847141); Articles of iron/steel, n.e.s. (732690); Chandeliers and other electric ceiling or wall light fittings; excluding those used for lighting public open spaces or thoroughfares (940510); Machinery n.e.s. In Ch.84, for industrial preparation or manufacture of food or drink (843880):

Waste & scrap of cast iron (720410); Parts of machinery for making/finishing paper or

paperboard (843999);
Appliances which are worn, carried or implanted

Appliances which are worn, carried or implanted in the body, to compensate for a defect or disability (902190);

Other colour reception apparatus for television, whether not incorporating radio broadcast receivers sound video recording reproducing apparatus, (852872);

Waste oils; of petroleum or obtained from bituminous minerals, not crude and preparations n.e.c., weight 70% or preparations of the same, not containing polychlorinated biphenyls (pcbs), polychorinated terphenyls (pcts) or polybrominated biphenyls (pbbs) (271099); Articles of stone/of other mineral substances, n.e.s. (681599);

Knives & cutting blades, for machines or mechanical appliances, n.e.s. (820890); Flat knitting machines; stitch-bond (844720); Statuettes & other ornamental articles (392640); Furnaces & ovens, non-electric, for the roasting, melting or other heat-treatment of ores, pyrites or of metals, for industrial or laboratory use (841710);

Tubes, pipes & hoses of plastics, n.e.s. (391739); Ignition wiring sets and other wiring sets for vehicles, aircraft or ships (854430); Mountings, fittings & similar articles, of base metal, suitable for buildings (830241); "Spectacle lenses, unmounted, of materials other than glass (900150);"

Engines parts for internal combustion piston engines (excluding spark-ignition) (840999); Woven fabrics, containing 85% or more by weight of textured polyester filaments, unbleached or bleached (540751); Parts of the machinery used for cleaning, sorti

unbleached or bleached (540751); Parts of the machinery used for cleaning, sorting or grading seed, grain or dried leguminous vegetables or for milling or working cereals or dried leguminous vegetables (not farm-type machinery) (843790); colours, plain weave, weighing more than 200g/m2 (520941);

Bars & rods, hot-rolled, in irregularly wound coils (722100);

Boxes, cases, crates and similar articles for the conveyance or packing of goods (392310); Engines parts for internal combustion piston engines (excluding spark-ignition) (840999); Other parts and accessories for printing machinery excl. 844391 (844399); Artificial parts of the body, excluding artificial

joints (902139); Plates, sheets, film, foil & strip, of plastics n.e.s.. non-cellular and not reinforced.

laminated, supported or similarly combined with other materials (392099);

Burglar or fire alarms and similar apparatus (853110);

Machinery for making paper/paperboard (843920);

Flat-rolled products of iron/non-alloy steel, width 600mm or more, (not corrugated), plated or coated with zinc (not electrolytically) (721049);

Containers for compressed/liquefied gas, of iron or steel (731100);

Parts of the furniture of 94.03 (metal/wooden furniture etc. For office, kitchen uses, etc.) (940390);

Glazed ceramic flags & paving/hearth/wall tiles (690890);

Straw/fodder balers, incl. Pick-up balers (843340);

Refills for ball point pens, comprising the ball point and ink-reservoir (960860);

Cashew nuts, in shell (80131);

Machinery, plant & equip., n.e.s., for treating materials by change of temperature, other than for making hot drinks or cooking or heating food (841989);

Cranes; transporter, gantry and bridge cranes (842619);

Aluminium structures (like plates, rods, profiles, tubes and the like) and their parts, n.e.s. (761090);

Bending, folding, straightening or flattening machines (including presses), (other than numerically controlled), for working metal (846229):

Parts and accessories nesoi of bicycles etc. (871499);

Hydrometers and similar floating instruments, barometers, hygrometers, psychrometers, thermometers, pyrometers; recording or not, any combination of these instruments (excluding thermometers and barometers not combined with other instruments) (902580); Packing containters of

paper/paperboard/cellulose wadding/fibres, n.e.s. (481950);

Artificial filament yarn (other tha high tenacity), filament, monofilament (less than 67 decitex), of viscose rayon, multiple (folded) or cabled, n.e.s., not for retail sale, not sewing thread (540349); Parts of Electrical transformers, static converters and inductors (850490);

Textile fabrics; impregnated, coated, covered or laminated with plastics, (excluding polyvinyl chloride, polyurethane, polyamides polyesters) (590390);

Vegetable seeds, of a kind used for sowing (120991);

Additives for lubricating oil, containing petroleum oils or oils obtained from bituminous minerals (381121):

Polypropylene, in primary forms (390210); Kaolin and other kaolinic clays; whether or not calcined (250700);

Carboys, bottles, flasks, jars, pot and other containers of glass, (not ampoules), used for the conveyance or packing of goods (701090); Beauty/make-up preparations & preparations for skin care, n.e.s. (330499);

Other electrical parts of machinery (854890); Seeders, planters & transplanters (843230); Tubular/bifurcated rivets, of base metal (830820);

Pears & quinces, fresh (80820); Hinges of base metal (830210);

Television cameras, digital cameras and video camera recorders (852580):

Parts of the furniture of 94.03 (metal/wooden furniture etc. For office, kitchen uses, etc.) (940390);

Parts of the pumps of 8413.11-8413. (841391); Parts of telephone sets, incl. Telephones for cellular networks for other wireless networks other apparatus for the transmission reception of voice, images other data, incl. Apparatus for communication in a wired wireless network (851770);

Monofilament of which any cross-sectional dimension exceeds 1mm, rods, sticks and profile shapes, whether or not surface-worked but not otherwise worked (391690);

Ginger (91010);

Refractory bricks, blocks, tiles & similar refractory ceramic constructional goods; containing by weight, singly or together, more than 50% of the elements Mg, Ca or Cr (690210);

Lubricating preparations (excl. Those for the treatment of textile and similar materials), not containing petroleum oils or oils obtained from bituminous minerals (340399):

Vegetable oils; coconut (copra) oil and its fractions, other than crude, whether or not refined, but not chemically modified (151319); Buttons of plastics, not covered with textile material (960621);

Packing/wrapping machinery (842240); Ornamental trimmings in the piece, without embroidery, other than knittedd or crocheted (580890);

Handkerchiefs, cleansing/facial tissues and towels (481820);

Compounded rubber, compounded with carbon black or silica, in primary forms or in plates, sheets or strip (400510);

Table/kitchen/other h-hold. Articles and their parts, of Iron or steel, n.e.s. (732399); Metal furniture (other than for office use) (940320):

Hydraulic power engines & motors other than linear acting (cylinders) (841229);

Calcium carbonate (283650):

Catheters, cannulae & the like (901839); Silk waste (including cocoons unsuitable for reeling, yarn waste and garnetted stock) (500300);

Tubes, pipes & hoses of plastics, having a minimum burst pressure of 27.6mpa (391732); Instruments and apparatus for measuring or checking voltage, current, resistance or power, without a recording device (excluding multimeters) (903033);

Articles of iron/steel, n.e.s., forged/stamped, but not further worked (732619); Parts suitable for use solely or principally with electric motors and generators, electric generating sets and rotary converters, n.e.s.

Interchangeable tools for hand tool, whether or not power-operated, for screw-driving or uses n.e.c (820790);

(850300):

Induction or dielectric heating equipment (excl. Ovens and furnaces) (851440); Vegetable waxes (other than triglycerides); whether or not refined or coloured (152110); Parts of the pumps of 8413.11-8413. (841391);

Articulated link chain (other than roller), of iron or steel (731512);

Hubs (excl. Coaster braking hubs & hub brakes) and free-wheel sprocket-wheels (871493);

Machinery for filling, closing, sealing, capsuling or labelling bottles, cans, bags or other containers (842230);

Other monitors, not of a kind solely principally used in an automatic data processing system of heading 84.71 (852859); Buckets, shovels, grabs etc for excavating machinery (843141);

Tubes, pipes & hoses, rigid, of plastics n.e.s. (391729);

Electric accumulators n.e.s., including separators, whether or not rectangular (including square) (850780);

Narrow woven fabrics n.e.s., of man-made fibres (excl. Labels, badges and similar articles) (580632);

Cartons, boxes & cases, of corrugated paper or paperboard (481910);

Plates, separators and other parts of electric accumulators, n.e.s. (850790);

Linear acting (cylinders) pneumatic power engines and motors (841231);

Adhesive dressings & other articles having an adhesive layer, packed for retail sale for medical, surgical, dental or veterinary purposes (300510); Printed books, brochures, leaflets, n.e.s. (490199);

Polyesters, n.e.s., saturated, in primary forms (390799);

Preparations of a kind used in animal feeding, put up for retail sale (except dog or cat food) (230990);

Articulated link chain (other than roller), of iron or steel (731512);

Wirewound variable electrical resistors, incl. Rheostats and potentiometers, for a power handling capacity > 20 W (excl. Heating resistors) (853339);

Toothed wheels, chain sprockets and other transmission elements presented separately; parts of transmission components (848390); Semi conductor media, Smart cards for the recording of sound of other phenomena, but excl. Products of Ch. 37. (852352);

Fuel/lubricating/cooling medium pumps for internal combustion piston engines (841330); Parts of the sewing machines n.e.s. (845290); Containers for compressed/liquefied gas, of iron or steel (731100);

Tubes, pipes & hoses, of vulcanised rubber, not reinforced or otherwise combined with other materials, without fittings (400911);

Machines for crushing or grinding earth, stone, ores or other mineral substances (847420);

Commodities not specified (999999); Ferro-silicon, containing by weight more than

55% of silicon (720221); Office/school supplies, of plastics (392610);

Textile fabrics; impregnated, coated, covered or laminated with polyurethane (590320);

Synthetic monofilament, other than Elastomeric and polypropylene, of 67 decitex or more and of which no cross-sectional dimension exceeds 1mm (540419);

Toasters, electric (851672);

Slide fasteners, other than those fitted with chain scoops of base metal (960719);

Input or output units for automatic data processing machines, whether or not containing storage units in the same housing (excl. Keyboards) (847160);

Plates, sheets, film, foil&strip, cellular, of plastics n.e.s. (392119);

Machinery parts; not containing electrical connectors, insulators, coils, contacts or other electrical features, n.e.c. In chapter 82 (848790):

Wooden furniture of a kind used in office (940330);

Wooden doors & their frames & thresholds (441820);

Plates, sheets, film, foil & strip, of polyesters n.e.s., non-cellular and not reinforced, laminated, supported or similarly combined with other materials (392069);

Fuel/lubricating/cooling medium pumps for internal combustion piston engines (841330); Woven fabrics of synthetic staple fibres, containing 85% or more by weight of polyester staple fibres, other than unbleached or bleached (551219);

Other semi conductor media, for the recording of sound of other phenomena, but excl. Products of Ch. 37., other than Smart Cards and Solid state non volatile storage devices (852359);

Parts of the engines & motors of 8412 (841290):

Trunks, suit-cases, vanity-cases, executive-cases, and similar containers, of vulcanised fibre or of paperboard, or wholly or mainly covered with such materials or with paper (420219);

Parts of electric sound or visual signalling apparatus, n.e.s. (853190);

Bars & rods of iron/non-alloy steel, hot-rolled, hot-drawn or hot-extruded, n.e.s. In heading no. 7214, other than of rectangular cross-section (721499);

Aluminous cement, whether/not coloured or in the form of clinkers (252330);

Cartons, boxes & cases, of corrugated paper or paperboard (481910);

Hand-woven tapestries of the type (Gobelins, Flanders, Aubusson, Beauvais and the like) and needle-worked tapestries (e.g. Petit point, cross-stitch) whether or not made up (580500);

Motorcycles (incl. Mopeds) & cycles (871130);

Parts of the slide fasteners (960720);

Television cameras, digital cameras and video camera recorders (852580);

Technically spec. Natural rubber (TSNR), in primary forms or in plates, sheets or strip (excluding latex and smoked sheets) (400122); Boring or sinking machinery n.e.s., not self propelled (843049);

Flax yarn, single (530610);

Jacks (excl. Of 8425.41) & hoists, hydraulic, of a kind used for raising vehicles (842542);

Parts of electric water heaters, immersion heaters, space heating apparatus and soil heating apparatus, hair-dressing apparatus and hand dryers, electro-thermic appliances of a kind used for domestic purposes and electric heating resistors, n.e.s. (851690);

Transmission shafts (including cam shafts and crank shafts) and cranks (848310);

Wheel rims & spokes for bicycles (871492); Colouring matter of vegetable/animal origin, including dyeing extracts not animal black, whether or not chemically defined; preparations based on colouring matter of vegetable or animal origin (320300);

Waste & scrap of alloy steel other than stainless steel (720429);

Parts of the harvesting machinery (including parts of threshing machinery, straw or fodder balers and grass or hay mowers) (843390); Wooden furniture (other than for office, kitchen or bedroom use), nesoi (940360);

Other glassware of a kind used for table or kitchen purposes (not drinking glasses or of glass-ceramics) (701349);

Parts of the agricultural/horticult (843290); Aluminium structures (like plates, rods, profiles, tubes and the like) and their parts, n.e.s. (761090);

Wooden furniture of a kind used in bedroom (940350);

Sanitary towels & tampons, napkins (481840); Ferrous waste & scrap, n.e.s. (720449);

Plates, sticks, tips & the like for tools, unmounted, of sintered metal carbides or cermets (820900);

Woven fabrics of cotton, containing 85% or more by weight of cotton, of yarns of different colours, 3-thread or 4-thread twill, including cross twill, weighing not more than 200g/m2 (520843);

Machinery for filling, closing, sealing, capsuling or labelling bottles, cans, bags or other containers (842230);

New pneumatic tyres, of rubber, of a kind used on motorcycles (401140);

Printed circuits (853400);

Textured yarn, other than of nylon or other polymides, polyesters, or polypropylene, not for retail sale, not sewing thread (540239);

Woven fabrics of cotton, bleached, weighing not more than 200g/m2, n.e.s. (521212);

Machinery parts; not containing electrical connectors, insulators, coils, contacts or other electrical features, n.e.c. In chapter 82 (848790); Other Parts suitable for use solely or principally with the apparatus of headings 8525 to 8528 (reception and transmission apparatus, monitors, etc.) (852990);

Other articles of copper, n.e.s. (741999);

Parts of aircraft engines (spark-ignition reciprocating or rotary internal combustion piston engines) (840910);

Tubes, pipes & hose fittings (e.g. Joints, elbows, flanges) (391740);

Other tubes, pipes and hollow profiles (for example, open seam or welded, riveted or similarly closed), of iron or steel (730690); Machinery for finishing paper/paperboard (843930):

Quicklime (252210);

Chain, of iron/steel, n.e.s., of iron or steel (731589);

Float glass & surface ground/polished glass, in sheets, non-wired, (other than coloured throughout the mass (body tinted), opacified, flashed or merely surface ground) (700529); Ferrous waste & scrap, n.e.s. (720449); Yarn other than high tenacity/textured, filament, monofilament (less than 67 decitex), not partially oriented, single, untwisted or twisted 50 turns or less per metre, not for retail sale, not sewing thread (540247);

Parts of machinery for making/finishing paper or paperboard (843999);

Rivets of iron/steel (731823);

Toilet paper, in rolls (481810);

Spools, cops, bobbins & similar supports, for the conveyance or packing of goods (392340); "Acyclic polycarboxylic acids, their anhydrides.

Halides, peroxides, peroxyacids and their Derivatives, n.e.s. (291719);"

Articles of glass such as statuettes and other ornaments of lamp worked glass, other than imitation jewellery (701890);

Other winches; capstans, not powered by electric motor (842539);

Buttons, other than of base metal or plastics, not covered with textile material (960629); Plates, sheets, film, foil & strip, non-cellular and not reinforced, laminated, supported or similarly combined with other materials (392051);

Woven fabrics of flax, containing less than 85% flax, other than unbleached or bleached (530929);

Buttons of plastics, not covered with textile material (960621);

Parts and accessories of printing machinery used for printing by means of plates, cylinders and other printing components of heading 8442 (844391);

Vegetable fats and oils and their fractions; partly or wholly hydrogenated, inter-esterified, re-esterified or elaidinised, whether or not refined, but not further prepared (151620); Flat-rolled products of iron/non-alloy steel, width 600mm or more, electrolytically plated or coated with zinc (721030);

Wadding; other articles of wadding, of cotton (560121);

Uncoated paper & paperboard, n.e.s., weight 225/m2 or more, in rolls or sheets (480593); Parts of the slide fasteners (960720);

Waste (incl. Noils, yarn waste & garnetted stock), of artificial fibres (550520);

Tableware & kitchenware, of wood (441900); Polymers of propylene/of other olefin polymers, n.e.s. (390290);

Filtering/purifying machinery & apparatus, other than intake air filters for internal combustion engines (842139);

Springs & leaves for springs, n.e.s., of iron/steel (732090);

Glazed ceramic flags & paving/hearth/wall tiles (690890);

Pipe-cutters, bolt croppers, perforating punches and similar tools (820340);

Builders' ware of plastics, n.e.s. (392590); Parts & accessories of the of spinning, combing, drawing machines, n.e.s. (844839);

Paper/paperboard labels of all kind, unprinted (482190);

Buttons, of base metal, not covered with textile material (960622);

"Travel sets for personal toilet, sewing, shoe or clothes cleaning (960500);"

Footwear with outer soles and uppers of rubber or plastics, upper straps or thongs assembled to the sole by plugs (640220);

Spark-ignition reciprocating/rotary internal combustine engines, for marine propulsion, (other than outboard motors), (840729); High tenacity yarn other than textured year, of polyesters, not for retail sale (540220); Tube/pipe fittings of stainless steel, n.e.s (730729);

Floor coverings of plastics (other than polymers of vinyl chloride), whether or not self-adhesive, in rolls or in the form of tiles (391890);

Hand-operated spanners & wrenches, adjustable (820412);

Paraffin wax; containing by weight<0.75% of oil, obtained by synthesis or by other processes, whether or not coloured (271220);

DC motors and generators of an output exceeding 375kw (850134);

Synthetic staple fibres, not carded, combed or otherwise processed for spinning (550320); Natural rubber other than latex/smoked sheets, in primary forms or in plates, sheets or strip

Paper&paperboard, not containing fibre greater than 10%, weighing less than 40g/m2, in rolls or sheets (480254);

Waffles & wafers (190532); Quicklime (252210); Paper&paperboard, not containing fibre greater than 10%, weight 40-150g/m2, n.e.s. (480257);

Other machines for resistance welding of metals, neither fully nor partly automatic (851529);

Other drinking glasses, other than of lead crystal or glass-ceramics (701337);

Semi-finished products of iron/non-alloy steel, containing by weight 0.25% or more of carbon (720720):

Flat-rolled products of iron/non-alloy steel, width 600mm or more, plated or coated with materials n.e.s. (721090):

Press-fasteners, snap-fasteners, press-studs and parts (960610);

Nickel alloys, unwrought (750220);

Woven fabrics of synthetic filament yarn, containing less than 85% by weight of synthetic filaments, mixed mainly or solely with cotton, bleached or unbleached (540781); Narrow woven fabrics n.e.s., of man-made fibres (excl. Labels, badges and similar articles) (580632);

Sculptures and statuary; original, in any material (970300);

Fire-extinguishers; preparations and charges and charged fire-extinguishing grenades (381300);

Constant weight scales & scales for discharging a predetermined weight of material into a bag or container, including hopper scales (842330);

Bars & rods of iron/non-alloy steel, hot-rolled, hot-drawn or hot-extruded, containing indentations, ribs, grooves or other deformations produced during the rolling process or twisted after rolling (721420); Poly(vinyl chloride), plasticised, in primary forms, mixed with other substances (390422); Other type of rubber thread & cord, textile covered; textile yarn, and strip and the like, impregnated/coated/covered/sheathed with rubber or plastics (560490);

Waste (incl. Noils, yarn waste & garnetted stock), of artificial fibres (550520); Button moulds and other parts of buttons; button blanks (960630);

Glazed ceramic tiles, cubes and similar articles, whether or not rectangular or on a backing, with the largest surface area<7cm square (690810);

Doors, windows & their frames & thresholds, of aluminium (761010);

Date, numbering, sealing stamps and the like (including devices for printing or embossing labels), designed for operating by hand; hand operated composing sticks and printing sets (961100);

Textile winding (incl. Weft-winding), or reeling machines (844540);

Table/kitchen/other h-hold. Articles and their parts, of Iron or steel, n.e.s. (732399); Sauces and preparations therefor, n.e.s.; mixed condiments and mixed seasonings (210390); Video recording or reproducing apparatus, whether or not incorporating a video tuner (excl. Magnetic tape-type and video camera recorders) (852190):

Electrical apparatus for switching or protecting electrical circuits, or for making connections to or in electrical circuits, for a voltage > 1.000 V (excl. Fuses, automatic circuit breakers, isolating switches, make-and-break switches, lightning arresters, voltage limiters, surge suppressors and control desks, cabinets, panels etc. Of heading 8537) (853590);

Contact lenses (900130);

Non-electrical lamps & light fittings (940550); Waste, parings & scrap, of polymers ethylene (391510);

Compounded rubber (other than with carbon black or silica), vulcanized, in primary forms (other than plates, sheets, strip, solutions or dispersions) (400599);

Woven fabrics of polyester staple fibres n.e.s., mixed mainly or solely with viscose rayon staple fibres (551511);

Polysulphides, polysulphones & other products of chemical synthesis n.e.s.; in primary forms (391190).

Gear boxes & parts thereof, of vehicles (870840):

Vehicles with only spark-ignition internal combustion reciprocating piston engine, cylinder capacity over 1000 but not over 1500cc (870322);

Bamboos (140110);

Electric accumulators, i.e. Lead-acid, (other than for starting piston engines), including separators, whether or not rectangular (including square) (850720);

Cotton sewing thread, put up for RS, containing less than 85% by weight of cotton (520420); Woven fabrics of synthetic staple fibres, containing 85% or more by weight of polyester staple fibres, unbleached or bleached (551211); Parts of the machinery for treating textile fabrics (845190);

Sugars; n.e.s., including invert sugar and other sugar and sugar syrup blends containing, in the dry state, 50% by weight of fructose (170290); Polymers of styrene, in primary forms, n.e.s. (390390):

Furniture for medical, surgical, veterinary use (e.g. Operating tables, examination tables, hospital beds with mechanical fittings) and parts thereof (940290);

Organic surface-active products & preparations for washing the skin, in the form of liquid or cream and put up for retail sale, whether or not containing soap (340130); Instruments and apparatus for measuring or checking voltage, current, resistance or power, with a recording device (excluding multimeters) (903039);

Tubes & pipes of aluminium, not alloyed (760810):

Parts And Accessories Of Instruments And Apparatus For Measuring, Checking Or Detecting Electrical Quantities, Or Ionizing Radiations, Others (903090);

Parts & accessories n.e.s. In Ch.90 (903300); Towers & lattice masts of iron/steel (730820); Seeders, planters & transplanters (843230); Wooden furniture (other than for office, kitchen or bedroom use), nesoi (940360); Table/kitchen/other h-hold. Articles and their parts, of stainless steel (732393); Cocoa; powder, containing added sugar or other sweetening matter (180610); Glass fibres (incl. Glass wool) & articles thereof, n.e.s. (701990);

Textile fabrics; impregnated, coated, covered or laminated with plastics, (excluding polyvinyl chloride, polyurethane, polyamides polyesters) (590390);

Synthetic staple fibres, n.e.c., not carded, combed or otherwise processed for spinning (550390);

Artificial guts (sausage casings) of hardened protein or of cellulosic materials (391710); Metal furniture of a kind used in office (940310);

Bars & rods, hot-rolled, in irregularly wound coils, n.e.s., of circular cross-section measuring less than 14mm in diameter (721391);

Flanges of iron/steel (excl. Of cast iron), of material other than stainless steel (730791); Machinery for the extraction or preparation of animal or fixed vegetable fats or oils (847920);

Bedspreads other than knittedd/crocheted (630419);

Printed books, brochures, leaflets, n.e.s. (490199);

Surface-active preparations, Washing and cleaning preparations, whether or not containing soap, put up for retail sale (340220);

Iron oxides & hydroxides (282110); Cast articles of iron (excl. Non-malleable cast iron), excluding grinding balls and similar articles for mills (732599);

Apparatus based on use of x-rays and similar; parts and accessories (x-ray generators, tubes, high tension generators, control panels and

Tubes, pipes & hoses, rigid, of plastics n.e.s. (391729);

Waste, parings & scrap, of polymers, vinyl chloride (391530);

Lace mechanically made in the piece, in strips or in motifs, of textile materials (580429);

DC motors and generators of an output exceeding 750W but not exceeding 75kw (850132);

Tanned/crust hides & skins of goats or kids, without wool or hair on, whether or not split, but not further prepared, in the wet state (including wet blue) (410621);

Ethylene polymers; in primary forms, polyethylene having a specific gravity of less than 0.94 (390110);

Chocolate & other food preparations containing cocoa; n.e.s. (180690);

Polymers of ethylene, in primary forms, n.e.s. (390190);

Reservoirs, tanks, vats & similar containers of a capacity exceeding 300 litres (392510);

Yarn other than high tenacity/textured, filament, monofilament (less than 67 decitex), single, untwisted or twisted 50 turns or less per metre, n.e.s., not for retail sale, not sewing thread (540249);

Collages and similar decorative plaques (artwork) (970190);

Food preparations, n.e.s. (210690);

Dyed Woven fabrics of synthetic, polyester staple fibres (less than 85% by weight), mixed mainly or solely with cotton, not exceeding 170g/m2 (551323);

Hypochlorites, Chlorites, And Hypobromites, n.e.s. (282890);

Spools, cops, bobbins & similar supports, for the conveyance or packing of goods (392340); Carbonates, n.e.s. (283699);

Ceramic sinks, wash basins, wash basin pedestals, baths, bidets, water closet pans, flushing cisterns, urinals and similar sanitary fixtures; of other than porcelain or china (691090);

Wooden furniture of a kind used in office (940330):

Waste of wool/of fine animal hair, including yarn waste, but excluding garnetted stock and noils of wool or of fine animal hair (510320);

Paper/paperboard labels of all kind, printed (482110);

Cotton yarn, single (excl. Sewing thread), of uncombed fibres, 85% or more by weight of cotton, measuring 714.29 decitex or more, (not exceeding 14 metric number), not for retail sale (520511);

Made up fishing nets of man-made textile materials (560811);

Vitamins & their derivatives, unmixed (293629);

desks, screens, examination or treatment tables, chairs and like (902290);

Vehicles with only spark-ignition internal combustion reciprocating piston engine, cylinder capacity over 1000 but not over 1500cc (870322);

Uncoated paper & paperboard, n.e.s., weight 150g/m2 or less, in rolls or sheets (480591); Instruments and appliances used in medical or veterinary sciences, nes (901890);

Silk yarn & yarn spun from silk waste; put up for retail sale, and silk-worm gut (500600); Parts of the machinery of 84.41, for making up paper pulp, paper or paperboard, including cutting machines of all kinds (844190); Baths, shower-baths, sinks & wash-basins (392210);

Shutters, blinds (incl. Venetian blinds) and similar articles and parts thereof (392530); Surveying equipment; articles n.e.s., including hydrographic, oceanographic, hydrological, meteorological or geophysical instruments and appliances (excluding compasses) (901580); Automatic circuit breakers for a voltage > 1000 V but < 72,5 kv (853521); Urea, whether/not in aqueous solution (310210);

Woven fabrics, containing 85% or more by weight of textured polyester filaments, unbleached or bleached (540751);

Transformers, having a power handling capacity > 1 kva but <= 16 kva (excl. Liquid dielectric transformers) (850432);

Sauces and preparations therefor, n.e.s.; mixed condiments and mixed seasonings (210390); Pumps for dispensing fuel/lubricant of the type used in filling-stations or in garages, fitted or designed to be fitted with a measuring device (841311);

Wool tops & other combed wool other than in fragments (510529);

Buttons, of base metal, not covered with textile material (960622);

Float glass & surface ground/polished glass, in sheets, wired glass, whether or not having an absorbent or reflecting layer (700530);

Petroleum oils and oils from bituminous minerals, not containing biodiesel, not crude, not waste oils; preparations n.e.s. (271019); Synthetic fibre other than high

tenacity/textured, filament, monofilament (less than 67 decitex), multiple (folded) or cabled, n.e.s., not for retail sale, not sewing thread (540269);

Plates, sheets, film, foil & strip, non-cellular and not reinforced, laminated, supported or similarly combined with other materials (392049);

Hat-shapes, plaited or made by assembling strips of any material, neither blocked to

"Brooms & brushes, consisting of twigs or other vegetable materials bound together, with or without handles

(960310);"

Glassine & other glazed transparent or translucent papers, in rolls or sheets (480640); Bars & rods, hot-rolled, in irregularly wound coils, n.e.s., of circular cross-section measuring less than 14mm in diameter (721391); Articles of plastics, n.e.s. (392690); Orange juice, not frozen, of a Brix value exceeding 20, unfermented, not containing added spirit, whether or not containing added sugar or other sweetening matter (200919);

Gelatin, incl. Gelatin in rectangular sheets, whether or not surface-worked or coloured) and gelatin derivatives; isinglass; other glues of animal origin (350300);

Particle board of wood, whether/not agglomerated with resins or other organic binding substances (441011);

Woven fabrics of cotton, containing 85% or more by weight of cotton, of yarns of different colours, weighing more than 100g/m2 but not more than 200g/m2 (520842);

Cutlery, other than plated with precious metal (821599);

Yarn other than sewing thread, of synthetic staple fibres, mixed mainly or solely with fibres (other than wool, fine animal hair or cotton), n.e.s., not put up for retail sale (550999); Petroleum gases and other gaseous hydrocarbons; in gaseous state, other than natural gas (271129);

Ash & residues (excl. From the manufacture of iron or steel), containing mainly zinc, hard zinc spelter (262019);

Spray guns & similar appliances (842420); Bars, rods & profiles, of refined copper (740710);

Long pile fabrics, knittedd/crocheted (600110); Parts & accessories of compound optical microscopes (including those for photomicrography, cinephotomicrography or microprojection) (901190);

Knittedd/crocheted fabrics, n.e.s., of synthetic fibres, of yarns of different colours (600633); Synthetic fibre other than high tenacity/textured, filament, monofilament (less than 67 decitex), single, twisted more than 50 turns per metre, n.e.s., not for retail sale, not sewing thread (540259);

Parts and accessories of printing machinery used for printing by means of plates, cylinders and other printing components of heading 8442 (844391);

Candles, tapers&the like (340600); Angles, shapes & sections of iron/non-alloy steel, n.e.s. (721699); shape, nor with made brims, nor lined, nor trimmed (650200);

Nonwovens, whether/not

impregnated/coated/covered/laminated, not of man-made filaments, (weighing more than 25g/m2 but not more than 70g/m2) (560392); Tubes, pipes & hoses, rigid, of polymers of ethylene (391721);

Tanned/crust hides & skins of goats or kids, without wool or hair on, whether or not split, but not further prepared, in the wet state (including wet blue) (410621);

White cement, whether/not artificially coloured (252321);

Cooking appliances & plate warmers, for liquid fuel, of iron or steel (732112); Articles of plastics, n.e.s. (392690); Outer soles & heels, of rubber/plastics (640620):

Builders' ware of plastics, n.e.s. (392590); Flat-rolled products of iron/non-alloy steel, width 600mm or more, painted, varnished or coated with plastics (721070);

Waste oils; of petroleum or obtained from bituminous minerals, not crude and preparations n.e.c., weight 70% or preparations of the same, not containing polychlorinated biphenyls (pcbs), polychorinated terphenyls (pcts) or polybrominated biphenyls (pbbs) (271099); Tubes, pipes & hoses of plastics, n.e.s. (391739):

Vices, clamps & the like (820570); Threaded elbows, bends & sleeves of stainless steel (730722);

Residues of starch manufacture and similar residues; whether or not in the form of pellets (230310);

Paper&paperboard, not containing fibre greater than 10%, weight 40-150g/m2, in sheets 435mm or less by 297mm or less (unfolded) (480256);

Knittedd/crocheted fabrics of a exceeding 30 cm, other than pile or terry fabrics, containing by weight 5% or more of elastomeric yarn or rubber thread, (600490);

Woven fabrics of cotton, containing 85% or more by weight of cotton, of yarns of different colours, weighing more than 100g/m2 but not more than 200g/m2 (520842);

Plates, sheets, film, foil & strip, of cellulose acetate, non-cellular and not reinforced, laminated, supported or similarly combined with other materials (392073);

Ceramic wares for laboratory, chemical or other technical uses, of porcelain or china (690911);

Printed books, brochures, leaflets, in single sheets, whether or not folded (490110);

Plates, sheets, film, foil & strip, of acrylic polymers (excluding polymethyl methacrylate), non-cellular and not reinforced, laminated, supported or similarly combined with other materials (392059);

Harrows, other than disc harrows, scarifiers, cultivators, weeders and hoes (843229); Anti-freezing preparations & prepared de-icing fluids (382000);

Stoppers, lids, caps and other closures, for the conveyance or packing of goods (392350); Other monitors, of a kind solely principally used in an automatic data processing system (852851);

Articles of apparel & clothing accessories (other than gloves, mittens and mitts), of vulcanized rubber (401590);

Labels, badges & similar articles; woven, of textile materials, in the piece, in strips or cut to shape or size, not embroidered (580710); Metal furniture (other than for office use) (940320);

Parts & accessories n.e.s., of wood, plastic machine tools (846692);

Other hosiery articles n.e.s., knittedd/crocheted, without applied soles, of textile materials n.e.s. (611599);

Parts of machinery, plant/laboratory equipment, for treating materials by a process involving a change of temperature (841990);

Raw hides&skins (fresh,/salted, dried, limed, pickled or otherwise preserved), but not tanned, parchment-dressed or further prepared, whether or not dehaired or split (410390);

Glass microspheres not >1 mm in diameter (701820):

Knittedd/crocheted fabrics, n.e.s., of cotton, unbleached or bleached (600621);

Sports footwear other than ski-boot, snowboard boots or cross-country ski footwear, with outer soles and uppers of rubber or plastics (640219); Pile fabrics & terry fabrics (excluding long pile and loop pile), of textile materials (other than cotton or man-made fibres), knittedd or crocheted (600199);

Tubes, pipes & hoses, rigid, of polymers of ethylene (391721);

Woven fabrics of vegetable textile fibres n.e.s., woven fabrics of paper yarn (531100);

Carpets and other textile floor coverings; n.e.s., whether or not made up (570500);

Sugar confectionery; chewing gum, whether or not sugar-coated, not containing cocoa (170410); Zinc waste & scrap (790200);

Refills for ball point pens, comprising the ball point and ink-reservoir (960860);

Festive, carnival or other entertainment articles including novelty jokes and conjuring tricks; other than Christmas festivity articles (950590);

Wood marquetry & inlaid wood; caskets and cases for jewellery or cutlery and similar articles of wood, wooden articles of furniture (442090):

Bidets, lavatory pans, flushing cisterns and similar sanitary ware, n.e.s. (392290); Hand-/foot-operated air pumps (841420); Woven fabrics of synthetic filament yarn, containing less than 85% by weight of synthetic filaments, mixed mainly or solely with cotton, dyed (540782);

Knittedd/crocheted fabrics, n.e.s., of cotton, of yarns of different colours (600623); Polymers of ethylene, in primary forms, n.e.s. (390190):

Equipment for

moving/grading/levelling/scraping/excavating earth minerals or ores, not self propelled n.e.s. (843069);

Raw hides & skins of bovine (including buffalo) and equine animals, fresh, salted or preserved, but not tanned, parchment dressed or further prepared, whether or not dehaired or split (410190);

Woven fabrics of cotton, containing less than 85% by weight of cotton, mixed mainly or solely with man-made fibres, weighing more than 200g/m2, of yarns of different colours, denim (521142);

Narrow woven fabrics n.e.s., of cotton (excl. Labels, badges and similar articles) (580631); Hand tools, other than household tools (820559):

Hard rubber (e.g., ebonite) in all forms, including waste and scrap, and articles of hard rubber (401700);

Parts of liquid elevators (841392); Float glass & surface ground/polished glass, in sheets, non-wired, having an absorbent reflecting or non-reflecting layer (700510); Leather further prepared after tanning or crusting, including parchment-dressed leather, of bovine (including buffalo) or equine animals, without hair on, whole hides and skins, full grain, unsplit (410711); Cane molasses (170310); Soap & organic surface-active products; in the

form of bars, cakes, moulded shapes, and paper, wadding, felt and nonwovens, impregnated, coated or covered with soap or detergent, not for toilet use (340119); Parts And Accessories For Surveying instrument etc Nesoi (901590); Seeds of coriander (90920);

Trunks, suit-cases, vanity-cases, executive-cases, and similar containers, n.e.s., of vulcanised fibre or of paperboard, or wholly or mainly covered with such materials or with paper (420299);

Swivel seats with variable height adjustment, excluding medical, surgical, dental, veterinary or barber furniture (940130);

Fuel wood, in logs/billets/twigs/faggots/similar forms, whether or not agglomerated, coniferous (440110);

Primary cells and primary batteries, electric (excl. Spent, and those of silver oxide, mercuric oxide, manganese dioxide, lithium and air-zinc) (850680);

Footwear with outer soles of leather, uppers consisting of leather straps across instep and around the big toe (640320);

Glycerol (290545);

Clothing accessories, of leather/composition leather, n.e.s. (420340);

"Cuttle fish (Sepia officinalis, Rossiamacrosoma, Septiola spp.) And squid (Ommastrephes spp., Loligo spp., Nototodarusspp., Sepioteuthis spp.) (30741);"

Raw hides & skins of bovine (including buffalo) and equine animals, fresh, salted or preserved, but not tanned, parchment dressed or further prepared, whether or not dehaired or split (410190);

Machines and apparatus of heading 8486; parts and accessories (848690);

Carpets and other textile floor coverings; tufted, of wool or fine animal hair, whether or not made up (570310);

Woven fabrics of carded wool/carded fine animal hair, mixed mainly or solely with man-made filaments (511120);

Woven fabrics of cotton, containing 85% or more by weight of cotton, dyed, plain weave, weighing more than 100g/m2 but not more than 200g/m2 (520832);

Copper wire, of refined copper, of which the maximum cross-sectional dimension exceeds 6mm (740811);

Knittedd/crocheted fabrics, n.e.s., and other than those made of wool, fine animal hair, cotton, synthetic or artificial fibres (600690);

Woven fabrics of synthetic filament yarn, adhesive or thermal bonded (540730);

Melamine resins, in primary forms (390920); Other footwear, other than with uppers of leather/composition leather/textile materials

(640590); Dried vegetables, n.e.s.; mixtures of vegetables, whole, cut, sliced, broken or in powder, but not

further prepared (71290); Metal furniture of a kind used in office (940310);

Syringes, with/without needles (901831); Vegetable materials of a kind used primarily for plaiting, n.e.s. (140190);

Ice & snow (220190);

Prefabricated buildings (940600);

Other articles of glass, n.e.s. (702000);

Cast glass & rolled glass, in non-wired sheets, (excluding those coloured throughout the mass (body tinted) opacified, flashed or having an absorbent or reflecting layer) (700319); Parings & other waste of leather/composition leather; not suitable for the manufacture of leather articles; leather dust, powder and flour (411520);

Slide fasteners, other than those fitted with chain scoops of base metal (960719); Parts of the other gas turbines (excluding turbo-jets and turbo-propellers) (841199); Woven fabrics of cotton, dyed, weighing not more than 200g/m2, n.e.s. (521213); Transformers having a power handling capacity > 16 kva but <= 500 kva (excl. Liquid dielectric transformers) (850433); Woven fabrics of cotton, containing 85% or more by weight of cotton, printed, plain weave, weighing more than 100g/m2 but not more than 200g/m2 (520852);

Pumps for liquids, fitted/designed to be fitted with a measuring device, other than pumps for dispensing fuel or lubricants (841319); Sewing thread of synthetic filaments, whether or not put up for retail sale (540110); Paper&paperboard, not containing fibre greater than 10%, weighing 40g/m2 to 150g/m2, in rolls (480255);

Yarn of combed wool, not put up for retail sale (510710);

Paper&paperboard, not containing fibre greater than 10%, weighing less than 40g/m2, in rolls or sheets (480254);

Woven fabrics of cotton, containing 85% or more by weight of cotton, printed, of weaves n.e.s., weighing not more than 200g/m2 (520859);

Palmitic acid, stearic acid, their salts and esters (291570);

Copper tube/pipe fittings (e.g. Couplings, elbows, sleeves) of copper alloys (741220); Articles of iron/steel, n.e.s. (732690); Paper/paperboard labels of all kind, unprinted (482190);

Waste & scrap of cast iron (720410); Stoppers, lids, caps and other closures, for the conveyance or packing of goods (392350); Woven fabrics of cotton, containing 85% or more by weight of cotton, dyed, plain weave, weighing more than 100g/m2 but not more than 200g/m2 (520832);

Waffles & wafers (190532);

Transformers having a power handling capacity > 500 kva (excl. Liquid dielectric transformers) (850434);

Threaded Screw hooks & screw rings, of iron/steel (731813);

Induction or dielectric furnaces and ovens (851420);

Quilted textile products in the piece, composed of one or more layers of textile materials assembled with padding by stitching or otherwise (excluding embroidery in strips or motifs) (581100);

Woven fabrics of cotton, containing less than 85% by weight of cotton, mixed mainly or solely with man-made fibres, weighing more than 200g/m2, dyed, plain weave (521131); Edible parts of plants, prepared or preserved, whether or not containing added sugar, other sweetening matter or spirit, n.e.s. (200899); Waste (incl. Noils, yarn waste and garnetted stock), of synthetic fibres (550510); Plates, sheets, film, foil&strip, not self-adhesive, of polymers of ethylene, non-cellular and not reinforced, laminated, supported or similarly combined with other materials (392010); Boxes, cases, crates and similar articles for the conveyance or packing of goods (392310);

conveyance or packing of goods (392310); Ceramic tableware, kitchenware, household articles and toilet articles, other than of porcelain or china (691200);

Toilet paper, in rolls (481810);

Knittedd/crocheted fabrics of a exceeding 30 cm, other than pile or terry fabrics, containing by weight 5% or more of elastomeric yarn or rubber thread, (600490);

Woven fabrics, of artificial filament yarn, containing 85% or more by weight of artificial filament, strip or the like, dyed (540822); Printed books, brochures, leaflets, in single sheets, whether or not folded (490110); Woven fabrics of cotton, unbleached, weighing not more than 200g/m2, n.e.s. (521211); Head-bands, linings, covers, hat foundations, hat frames, peaks and chinstraps; for headgear (650700);

Synthetic fibre other than high tenacity/textured, filament, monofilament (less than 67 decitex), multiple (folded) or cabled, n.e.s., not for retail sale, not sewing thread (540269);

Tanned/crust hides & skins of bovine or equine animals, in the wet state (including wet blue), without hair on, excluding full grains, unsplit; grain splits (410419);

Unwrought lead other than refined, not containing by weight antimony as the principal other element (780199);

Woven fabrics of cotton, containing 85% or more by weight of cotton, dyed, plain weave, weighing more than 200g/m2 (520931); Liquid dielectric transformers, having a power handling capacity <= 650 kva (850421); Woven fabrics obt. From strip/the like, of synthetic textile materials (540720); Doors, windows & their frames & thresholds, of aluminium (761010);

Safety headgear, whether/not lined/trimmed (650610);

Lifting machinery n.e.s. In 84.26, self-propelled derricks and cranes, not on tyres (842649);

Slide fasteners, fitted with chain scoops of base metal (960711);

Clothing accessories, of leather/composition leather, n.e.s. (420340);

Dyed woven fabrics of artificial staple fibres (85% or more in weight) (551612);

Woven fabrics of artificial staple fibres (85% or more in weight), of yarns of different colours (551613);

Particle board of wood, whether/not agglomerated with resins or other organic binding substances (441011);

Hydrogen chloride (hydrochloric acid) (280610);

Woven fabrics of cotton, containing 85% or more by weight of cotton, dyed, plain weave, weighing more than 200g/m2 (520931); Belts & bandoliers, of leather/composition leather (420330);

Woven fabrics of cotton, containing 85% or more by weight of cotton, unbleached, plain weave, weighing more than 200g/m2 (520911);

Liquid dielectric transformers, having a power handling capacity > 650 kva but <= 10000 kva (850422);

Poly(ethylene terephthalate), in primary forms, having a viscosity of 78ml/g or higher (390760);

Chocolate & other food preparations containing cocoa; n.e.s. (180690); Parts of boring or sinking machinery (843143);

Titanium waste & scrap (810830);

Labels, badges & similar articles; woven, of textile materials, in the piece, in strips or cut to shape or size, not embroidered (580710); Waste (incl. Noils, yarn waste and garnetted stock), of synthetic fibres (550510);

Vegetables and mixtures of vegetables (excluding potatoes), prepared or preserved otherwise than by vinegar or acetic acid, frozen (200490);

Anhydrous ammonia (281410);

Woven fabrics of cotton, containing 85% or more by weight of cotton, bleached, plain weave, weighing more than 200g/m2 (520921);

Safety headgear, whether/not lined/trimmed (650610);

Labels, badges & similar articles; (other than woven), of textile materials, in the piece, in strips or cut to shape or size, not embroidered (580790);

Cotton yarn, single (excl. Sewing thread), of combed fibres, 85% or more by weight of cotton, not for retail sale (520521);

Juice of any single citrus fruit (excluding orange or grapefruit), of a Brix value exceeding 20, unfermented, not containing added spirit, whether or not containing added sugar or other sweetening matter (200939);

Woven fabrics of cotton, containing 85% or more by weight of cotton, dyed, 3-thread or 4-thread twill, including cross twill, weighing not more than 200g/m2 (520833);

Flax, raw/retted (530110);

Other articles of zinc, n.e.s. (790700); Plates, sheets & strip, of cellular rubber (400811).

Carpets and other textile floor coverings; woven, (not tufted or flocked), of man-made textile materials, of pile construction, not made up (570232);

Woven fabrics of cotton, containing less than 85% by weight of cotton, mixed mainly or solely with man-made fibres, weighing more than 200g/m2, of yarns of different colours, plain weave (521141);

Cotton sewing thread, not put up for retail sale, containing less than 85% by weight of cotton (520419);

Tanned/crust hides & skins of bovine or equine animals, in the wet state (including wet blue), without hair on, full grains, unsplit; grain splits (410411);

Woven fabrics of cotton, printed, weighing not more than 200g/m2, n.e.s. (521215);

Narrow woven fabrics n.e.s., of textile materials n.e.c., (excl. Labels, badges and similar articles) (580639);

Slide fasteners, fitted with chain scoops of base metal (960711):

Knittedd/crocheted fabrics, n.e.s., of cotton, printed (600624);

Vegetable materials of a kind used primarily for plaiting, n.e.s. (140490);

Other tubes, pipes & hollow profile (not seamless), welded, of square or rectangular cross-section, n.e.s. (730661);

Non-alcoholic beverages; other than non-alcoholic beer, n.e.s., not including fruit or vegetable juices (220290);

Woven fabrics of cotton, unbleached, plain weave, containing 85% or more by weight of cotton, weighing not more than 100g/m2 (520811);

Woven fabrics obt. From high tenacity yarn, of nylon, other polyamides or of polyesters (540710);

Handkerchiefs, cleansing/facial tissues and towels (481820);

Anhydrous ammonia (281410);

Synthetic staple fibres, n.e.c., carded/combed or otherwise processed for spinning (550690);

Turbo-jets, of a thrust not >25kn (841111); Tanned/crust hides & skins, n.e.s., of animals other than equine, ovine, bovine, goats or kids, swine and reptiles; without wool or hair on, whether or not split, but not further prepared, in the dry state (crust) (410692);

Woven fabrics of cotton, containing 85% or more by weight of cotton, bleached, of weaves n.e.s., weighing more than 200g/m2 (520929); Nonwovens, whether/not

impregnated/coated/covered/laminated, not of man-made filaments, (weighing more than 70g/m2 but not more than 150g/m2) (560393); Industrial monocarboxylic fatty acids; acid oils from refining; (other than stearic acid, oleic acid or tall oil fatty acids) (382319); Transmission apparatus for radio broadcasting television (852550);

Semi-finished products of iron/non-alloy steel, containing by weight less than 0.25% of carbon, other than rectangular or square cross-section (720719);

Tall oil fatty acids (382313);

Potato starch (110813);

Waste & scrap of stainless steel (720421); Ground-nuts, prepared/preserved, whether or not containing added sugar, other sweetening matter or spirit (200811);

Paper/paperboard labels of all kind, printed (482110);

Copper waste & scrap (740400);

Woven fabrics of cotton, containing 85% or more by weight of cotton, of yarns of different colours, weighing more than 200g/m2, denim (520942):

Woven fabrics of acrylic/modacrylic staple fibres n.e.s., mixed mainly or solely with fibres n.e.s. (551529);

Woven fabrics of synthetic filament yarn, bleached or unbleached, n.e.s. (540791); Orange juice, not frozen, of a Brix value exceeding 20, unfermented, not containing added spirit, whether or not containing added sugar or other sweetening matter (200919); Tanned/crust hides & skins of bovine or equine animals, in the wet state (including wet blue), without hair on, full grains, unsplit; grain splits (410411);

Cement clinkers (252310);

Waste, parings & scrap, of rubber (other than hard rubber) and powders and granules obtained therefrom (400400);

Hand-made paper & paperboard (480210); Whole bovine (incl. Buffalo)/equine animals, of a weight per skin not exceeding 8kg when simply dried, 10kg when dry-salted or 16kg when fresh, wet-salted or otherwise preserved (410120);

Zinc waste & scrap (790200);

Mixtures of juices, unfermented, not containing added spirit, whether or not containing added sugar or other sweetening matter (200990); Cotton sewing thread, not put up for retail sale, containing 85% or more by weight of cotton (520411);

Woven fabrics of cotton, containing 85% or more by weight of cotton, bleached, of weaves n.e.s., weighing more than 200g/m2 (520929); Woven fabrics of cotton, containing 85% or more by weight of cotton, printed, of weaves n.e.s., weighing not more than 200g/m2 (520859);

Refractory bricks, blocks, tiles & similar refractory ceramic constructional goods, n.e.s. (690290):

Other Waterproof footwear with rubber or plastic outer soles and uppers (not assembled by stitch, rivet, nail, screw, plug or similar) (640199); Tableware & kitchenware, of plastic (392410); Natural rubber (excl. Latex), in smoked sheets (400121);

Light petroleum oils & preparations (not crude), made from petroleum and bitum oils, etc. (271011);

Table linen, knittedd/crocheted (630240);

Sewing machine needles (845230);

Nickel waste & scrap (750300);

Doors, windows & their frames & thresholds for doors (392520);

Textured yarn other than sewing thread, of polyesters, not for retail sale (540233);

Ground-nuts, prepared/preserved, whether or not containing added sugar, other sweetening matter or spirit (200811);

Particle board & similar board of ligneous material, n.e.s. (441090);

Waste & scrap of stainless steel (720421); Yarn other than sewing thread, of polyester staple fibres, mixed mainly or solely with artificial staple fibres, not put up for retail sale (550951);

Leather further prepared after tanning or crusting, including parchment-dressed, of bovine (including buffalo) or equine animals, without hair on, split or not, (other than grain splits and full grains, unsplit) (410719);

Ice cream and other edible ice; whether or not containing cocoa (210500);

Woven fabrics of cotton, containing less than 85% by weight of cotton, mixed mainly or solely with man-made fibres, weighing more than 200g/m2, bleached (521120);

Seeds, n.e.s., of a kind used for sowing (120999); Woven fabrics of cotton, containing less than 85% by weight of cotton, mixed mainly or solely with man-made fibres, unbleached, of weaves n.e.s., weighing more than 200g/m2 (521119); Woven fabrics of cotton, containing 85% or more by weight of cotton, bleached, of weaves

Vessels n.e.c. For the transport of goods and other vessels for the transport of both persons and goods (890190);

Copper-zinc base alloys (brass), unwrought (740321);

Leather further prepared after tanning or crusting, including parchment-dressed, of bovine (including buffalo) or equine animals, without hair on, split or not, (other than grain splits and full grains, unsplit) (410719); Sports footwear; tennis shoes, basketball shoes, gym shoes, training shoes and the like, with outer soles of rubber or plastics and uppers of textile materials (640411); Apparatus based on the use of alpha, beta or gamma radiations, including radiography or

radiotherapy apparatus; for medical, surgical, dental or veterinary uses (902221); Food preparations; pasta, uncooked (excluding

Food preparations; pasta, uncooked (excluding that containing eggs), not stuffed or otherwise prepared (190219);

Light petroleum oils & preparations (not crude), made from petroleum and bitum oils, etc. (271011);

Woven fabrics of cotton, dyed, weighing more than 200g/m2, n.e.s. (521223);

AC generators, (alternators), of an output exceeding 375kva but not exceeding 750kva (850163);

Parts of garments/clothing accessories, ne.s., (not knittedd or crocheted) (621790);
Dairy produce; buttermilk, curdled milk or cream, kephir, fermented or acidified milk or cream, whether or not concentrated or containing added sweetening, flavouring, fruit or cocoa (excluding yoghurt) (40390);
Twine, cordage, ropes, cables of synthetic fibres other than polyethylene or polypropylene, whether or not plaited, braided or impregnated, coated, covered or sheathed with rubber or plastics (560750);
Nuts (excl. Ground-nuts), incl. Mixtures, whether or not containing added sugar, other sweetening matter or spirit (200819);

stainless steel (720429); Reclaimed rubber in primary forms/in plates,

Waste & scrap of alloy steel other than

sheets or strip (400300); Reservoirs, tanks, vats & similar containers of a capacity exceeding 300 litres (392510);

Woven fabrics of cotton, containing 85% or more by weight of cotton, of yarns of different colours, 3-thread or 4-thread twill, including cross twill, weighing more than 200g/m2 (520943):

Sacks & bags (incl. Cones), of plastics other than ethylene polymers, for the conveyance or packing of goods (392329);

Liquid dielectric transformers, having a power handling capacity <= 650 kva (850421);

n.e.s., weighing not more than 200g/m2 (520829);

Natural rubber latex, whether/not pre-vulcanised, in primary forms or in plates, sheets or strip (400110);

Copper waste & scrap (740400);

Seeds of coriander (90920);

Mattresses of cellular rubber or plastics, whether or not covered (940421);

Sawdust & wood waste & scrap, whether or not agglomerated in logs, briquettes, pellets or similar forms (440130);

Woven fabrics of cotton, containing 85% or more by weight of cotton, bleached, plain weave, weighing not more than 100g/m2 (520821); Zinc powders & flakes (790390);

Zinc alloys, unwrought (790120);

Sugar confectionery; (excluding chewing gum, including white chocolate), not containing cocoa (170490);

Fruit, edible; guavas, mangoes and mangosteens, fresh or dried (80450);

Footwear (excl. Waterproof) with outer soles and uppers of leather, covering the ankle (640351); Woven fabrics of cotton, containing less than 85% by weight of cotton, mixed mainly or solely with man-made fibres, bleached, plain weave, weighing 200g/m2 or less (521021);

Float glass & surface ground/polished glass, in sheets, non-wired, having an absorbent reflecting or non-reflecting layer (700510);

Frozen fish fillets & other fish meat (whether or not minced) (30429);

Gloves, mittens & mitts, knittedd/crocheted (other than impregnated, coated or covered with plastics or rubber), of textile materials (other than wool or fine animal hair, cotton or synthetic fibres) (611699);

Sacks & bags, of a kind used for the packing of goods, of textile materials (other than jute, cotton or man-made textile materials) (630590);

Woven fabrics of cotton, containing 85% or more by weight of cotton, dyed, plain weave, weighing not more than 100g/m2 (520831); Tanned/crust hides & skins of swine, without hair on, whether or not split, but not further prepared, in the dry state (crust) (410632); Potatoes, prepared or preserved otherwise than by vinegar or acetic acid, frozen (200410); Other Knittedd/crocheted fabrics of pile and terry fabrics, with width not exceeding 30 cm, containing by weight 5% or more of elastomeric yarn or rubber thread (600290);

New pneumatic tyres, of rubber, of a kind used on bicycles (401150);

Cotton yarn, single (excl. Sewing thread), of uncombed fibres, 85% or more by weight of cotton, less than 714.29 but not less than 232.56 decitex (exceeding 14 but not exceeding 43 metric number), not for retail sale (520612);

Tanned/crust hides & skins of bovine or equine animals, in the wet state (including wet blue), without hair on, excluding full grains, unsplit; grain splits (410419);

Tanned/crust hides & skins of bovine or equine animals, in the dry state (crust), without hair on, (other than full grains, unsplit; grain splits) (410449);

Tubes, pipes & hoses, rigid, of polymers of vinyl chloride (391723);

Ski suits, knittedd/crocheted (611220); Tableware & kitchenware, of plastic (392410); Men's/boys' jackets & blazers (excl.

Knitted/crocheted), of wool or fine animal hair (620331);

Vegetables, fruit, nuts and other edible parts of plants, prepared or preserved by vinegar or acetic acid (excluding cucumbers and gherkins) (200190);

Non-alcoholic beverages; other than non-alcoholic beer, n.e.s., not including fruit or vegetable juices (220290);

Panty hose, tights, stockings, socks, etc., of textile materials (other than synthetic fibres), knittedd or crocheted (611529);

Cruise ships, excursion boats and similar vessels, principally designed for the transport of persons, ferry boats of all kinds (890110); Webs, mattresses, boards & similar non-woven products, of glass fibres, excluding mats and thin sheets (701939);

Knittedd/crocheted fabrics, n.e.s., of cotton, dyed (600622);

Natural honey (40900);

Oil-cake and other solid residues; whether or not ground or in the form of pellets, resulting from the extraction of oils, n.e.s. (230690); Lecithins and other phosphoaminolipids, whether or not chemically defined (292320);

Zinc alloys, unwrought (790120);

Food preparations, n.e.s. (210690);

Doors, windows & their frames & thresholds for doors (392520);

Fresh fruit, n.e.s. (81090);

Men's/boys' jackets & blazers,

knittedd/crocheted, of textile materials (other than wool or fine animal hair, cotton or synthetic fibres) (610339);

Trunks, suit-cases, vanity-cases, executive-cases, and similar containers, with outer surface of plastics or of textile materials (420212):

Flat-rolled products of iron/non-alloy steel, width 600mm or more, corrugated, plated or coated with zinc (not electrolytically) (721041);

Household articles & toilet articles, of plastic (392490);

Sweet biscuits (190531);

Women's/girls' blouses, shirts & shirt blouses, of silk or silk waste (not knittedd or crocheted) (620610);

Organic surface-active products & preparations for washing the skin, in the form of liquid or cream and put up for retail sale, whether or not containing soap (340130);

Wadding; other articles of wadding, other than of cotton or man-made fibres (560129);

Food preparations; of flour, meal, starch, malt extract or milk products, for uses n.e.s. (190190); Hat-shapes, plaited or made by assembling strips of any material, neither blocked to shape, nor with made brims, nor lined, nor trimmed (650200):

Leather further prepared after tanning incl. Parchment-dressed, of bovine (including buffalo) or equine animals, no hair, and whole hides and skins, and sides, (full grains, unsplit and grain splits) (410799);

Toilet/facial tissue stock, towel/napkin/similar articles, for household or sanitary use, cellulose, wadding webs of cellulose fibres (480300); Tubes, pipes & hoses, rigid, of polymers of vinyl chloride (391723);

Woven fabrics of other synthetic staple fibers (less than 85% in weight), mixed mainly or solely with cotton, exceeding 170g/m2 (551449); "Trees, shrubs and bushes, grafted or not, of kinds,

Which bear edible fruits or nuts (60220);" Woven fabrics, containing 85% or more by weight of synthetic filaments (excluding nylon or other polyamides and polyesters), printed (540774);

Tulles and other net fabrics (excluding woven, knittedd or crocheted fabrics) (580410); Jams, fruit jellies, marmalades, purees and pastes; of fruit or nuts, n.e.s. Cooked preparations (excluding homogenised), whether or not containing added sugar or other sweetening matter (200799);

Garlic, fresh/chilled (70320);

Gauze, other than narrow fabrics (580300); Household articles & toilet articles, of plastic (392490);

"Frozen crustaceans, cooked by steaming or by boiling in water, incuding flours, meals and pellets, of

Crustaceans, fit for human consumption (30619);"

Knotted netting of twine, cordage/rope, for other than fishing, of man-made textile materials (560819);

Seeds of forage plants; other than lucerne, clover, fescue, Kentucky blue grass, and rye grass seeds, of a kind used for sowing (120929); Copper-tin base alloys (bronze), unwrought (740322);

Natural rubber (excl. Latex), in smoked sheets (400121);

Hand pumps for liquids, other than those specified in 841311 and 841319, to be fitted with a measuring device (841320)

Bromochlorodifluoromethane,

bromotrifluoromethane and

dibromotetrafluoroethanes (290346);

Hats & other headgear, plaited/made by assembling strips of any material, whether or not lined or trimmed (650400);

Used/new rags, scrap twine, cordage, rope and cables and worn out articles of twine, cordage, rope or cables, of textile materials; other than sorted (631090):

Hat-forms, hat bodies & hoods of felt, neither blocked to shape nor with made brims; plateaux and manchons (including slit manchons) of felt (650100):

Tanned/crust hides & skins of bovine or equine animals, in the dry state (crust), without hair on, (other than full grains, unsplit; grain splits) (410449);

Woven fabrics, of artificial filament yarn, containing 85% or more by weight of artificial filament, strip or the like, printed (540824); Food preparations; of flour, meal, starch, malt extract or milk products, suitable for infants or young children, put up for retail sale (190110); Knittedd/crocheted fabrics, n.e.s., of cotton, dyed (600622);

Whole bovine (incl. Buffalo)/equine animals, of a weight per skin not exceeding 8kg when simply dried, 10kg when dry-salted or 16kg when fresh, wet-salted or otherwise preserved (410120)

Bhutan

Chemical products & preparations of the chemical or allied industries, n.e.s. (382490); Pebbles, gravel, broken or crushed stone; commonly used for concrete aggregates, for road metalling or for railway or other ballast, shingle and flint, whether or not heat-treated (251710); Furniture of cane, osier, or similar materials (other than bamboo or rattan) (940389); Wooden furniture (other than for office, kitchen or bedroom use), nesoi (940360);

Plywood; consisting only of sheets of wood (not bamboo), each ply 6mm or thinner, not containing an outer ply of non-coniferous or tropical wood (441239);

Soap & organic surface-active products; in the form of bars, cakes, moulded shapes, and paper, wadding, felt and nonwovens, impregnated, coated or covered with soap or detergent, not for toilet use (340119)

Instruments and appliances used in medical or veterinary sciences, nes (901890);

Chemical products & preparations of the chemical or allied industries, n.e.s. (382490); Threaded screws & bolts of iron/steel, n.e.s., whether or not with their nuts or washers (731815);

Packing/wrapping machinery (842240); Tanks, casks, drums, cans, boxes and similar containers for any material (excluding compressed or liquefied gas) less than 50l capacity, n.e.s., of iron or steel (731029); Preparations of a kind used in animal feeding, put up for retail sale (except dog or cat food) (230990);

Ferro-manganese, containing by weight 2% or less of carbon (720219);

Iron or steel threaded articles except screw, nut,bolts (731819);

Folding cartons, boxes & cases, of non-corrugated paper or paperboard (481920); Parts of cranes, work trucks, shovels, constr machine (843149);

Doors, windows & their frames and thresholds for doors of iron/steel (730830); Food preparations, n.e.s. (210690); Household articles & toilet articles, of plastic (392490);

Articles of plastics, n.e.s. (392690);

Plywood; consisting only of sheets of wood (not bamboo), each ply 6mm or thinner, with at least one outer ply of non-coniferous wood, (without an outer ply of tropical wood) (441232);

Polymers of ethylene, in primary forms, n.e.s. (390190);

Natural graphite, in powder/flakes (250410); Vodka (220860);

Fresh fruit, n.e.s. (81090);

Jams, fruit jellies, marmalades, purees and pastes; of fruit or nuts, n.e.s. Cooked preparations (excluding homogenised), whether or not containing added sugar or other sweetening matter (200799);

Apple juice, of a Brix value exceeding 20, unfermented, not containing added spirit, whether or not containing added sugar or other sweetening matter (200979);

Glaziers' putty, grafting putty, resin cements, caulking compounds and other mastics; painters' fillings (321410);

Carboys, bottles, flasks & similar articles, for the conveyance or packing of goods (392330); Essential oils (terpeneless/not), including concretes and absolutes, n.e.s. (330190); Orange juice, frozen, unfermented & not containing added spirit, whether or not containing added sugar or other sweetening matter (200911);

Non-alcoholic beverages; other than non-alcoholic beer, n.e.s., not including fruit or vegetable juices (220290);

Vegetable materials of a kind used primarily for plaiting, n.e.s. (140490);

Portland cement (excl. White cement), whether or not artificially coloured (252329); Liqueurs & cordials (220870);

Edible parts of plants, prepared or preserved, whether or not containing added sugar, other sweetening matter or spirit, n.e.s. (200899); Plates, sheets, film, foil & strip, of polymers of propylene, non-cellular and not reinforced, laminated, supported or similarly combined with other materials (392020);

Ethyl alcohol and other spirits, denatured, of any strength (220720);

Undenatured ethyl alcohol of an alcoholic strength by volume of 80% vol. Or higher (220710);

Spirits obtained by distilling grape wine or grape marc (220820);

Tubes, pipes & hose fittings (e.g. Joints, elbows, flanges) (391740);

Rum & tafia (220840);

Ice & snow (220190);

Orange juice, not frozen, of a Brix value not>20, unfermented, not containing added spirit, whether or not containing added sugar or other sweetening matter (200912)

Cambodia

Non-threaded articles, n.e.s., of iron/steel (731829);

Threaded nuts of iron/steel (731816); Interchangeable tools for hand tool, whether or not power-operated, for screw-driving or uses n.e.c (820790);

Tube/pipe fittings of iron/steel, n.e.s,, other than stainless steel (730799);

Regulating or controlling instruments and apparatus; automatic, other than hydraulic or pneumatic (903289);

Machines for the reception, conversion and transmission regeneration of voice, images other data, incl. Switching and routing apparatus (851762);

Paints and varnishes; (including enamels, lacquers and distempers), prepared water pigments of a kind used for finishing leather (321000):

Primary cells and primary batteries, electric (excl. Spent, and those of silver oxide, mercuric oxide, manganese dioxide, lithium and air-zinc) (850680);

Sauces and preparations therefor, n.e.s.; mixed condiments and mixed seasonings (210390); Parts of Machines & mechanical appliances of those having individual functions (847990); Parts of telephone sets, incl. Telephones for cellular networks for other wireless networks other apparatus for the transmission reception of voice, images other data, incl. Apparatus for communication in a wired wireless network (851770):

Threaded coach screws of iron/steel (731811); Parts of Electrical transformers, static converters and inductors (850490);

Apparatus based on use of x-rays and similar; parts and accessories (x-ray generators, tubes, high tension generators, control panels and desks, screens, examination or treatment tables, chairs and like (902290);

Self-adhesive plates, sheets, film, foil, tape, strip, other flat shapes thereof, other than in rolls of a width not exceeding 20cm (391990);

Pulleys and flywheels, including pulley blocks (848350):

Sunglasses (900410);

Other apparatus for transmission reception of voice, images other data, incl. Apparatus for communication in a wired wireless network (such as a local wide area network) (851769); Paper, paperboard, cellulose wadding or webs of cellulose fibres (482390);

Parts and accessories nesoi of bicycles etc. (871499);

Coaxial cable and other coaxial electric conductors, insulated (854420);

Discs, tapes, solid state non volatile storage devices, smart cards and other media for the recording of sound of other phenomena, whether Taps, cocks, valves and similar appliances; for pipes, boiler shells, tanks, vats or the like, including thermostatically controlled valves (848180):

Lubricating preparations (other than for the treatment of textile and similar materials) containing less than 70% (by weight) of petroleum oils or oils obtained from bituminous minerals (340319);

Drive-axles with differential, whether or not provided with other transmission components, and non-driving axles; parts thereof (870850); Winding wire for electrical purposes, of copper, insulated (854411);

Fixed electrical resistors for a power handling capacity > 20 W (excl. Heating resistors) (853329);

Check (nonreturn) valves, for pipes, boiler shells, tanks, vats or the like (848130); Fuses, for a voltage not >1000V (853610); Bearing housings, incorporating ball or roller bearings (848330);

Printed books, brochures, leaflets, in single sheets, whether or not folded (490110); Engines parts for internal combustion piston engines (excluding spark-ignition) (840999); Fixed electrical capacitors (excl. Tantalum, aluminium electrolytic, ceramic, paper, plastic and power capacitors) (853229); New pneumatic tyres, of rubber, of a kind used on buses or lorries (401120); Sharpening (tool/cutter grinding) machines, other than numerically controlled (846039);

other than numerically controlled (846039); Locks of a kind used for motor vehicles (key, combination or electrically operated), of base metal (830120);

Parts and accessories for instruments, appliances and machines measuring or checking (903190);

Human blood; animal blood prepared for therapeutic, prophylactic or diagnostic uses; cultures of micro-organisms (excluding yeast) and similar products (300290);

Carboys, bottles, flasks, jars, pot and other containers of glass, (not ampoules), used for the conveyance or packing of goods (701090); Plates, sheets, film, foil & strip, of plastics n.e.s., non-cellular and not reinforced, laminated, supported or similarly combined with other materials (392099);

Silencers (mufflers) & exhaust pipe and parts (vehicle) (870892);

Brakes & servo-brakes; parts thereof (vehicles) (870830);

Engines; parts, suitable for use solely or principally with spark-ignition internal combustion piston engines (for other than aircraft) (840991);

Wire of iron/non-alloy steel, plated or coated with zinc (721720);

not recorded, incl. Matrices and masters for the production of discs, but excl. Products of Ch.37., other n.e.s. (852380);

Transformers having a power handling capacity <= 1 kva (excl. Liquid dielectric transformers) (850431);

Springs & leaves for springs, n.e.s., of iron/steel (732090);

Human blood; animal blood prepared for therapeutic, prophylactic or diagnostic uses; cultures of micro-organisms (excluding yeast) and similar products (300290);

Parts & accessories n.e.s. In Ch.90 (903300); Food preparations; pasta, uncooked (excluding that containing eggs), not stuffed or otherwise prepared (190219);

Articles of wood n.e.s. (442190);

Instruments and appliances used in medical or veterinary sciences, nes (901890);

Colouring matter & other preparations, n.e.s. (320649);

Parts and accessories of data processing equipment (847330);

Parts suitable for use solely or principally with the apparatus of heading 8535, 8536 or 8537, n.e.s. (excl. Boards, panels, consoles, desks, cabinets and other bases for the goods of heading 8537, not equipped with their apparatus) (853890);

Wooden furniture (other than for office, kitchen or bedroom use), nesoi (940360);

Essential oils (terpeneless/not), including concretes and absolutes, n.e.s. (330190); Paper/paperboard labels of all kind, printed (482110):

Articles of iron/steel, n.e.s. (732690);

Live animals, n.e.s. (10690);

Portable automatic data processing (ADP) machines, weighing not more than 10kg, consisting of at least a central processing unit, a keyboard and a display (847130);

Thermometers and pyrometers; (other than liquid filled, for direct reading), not combined with other instruments (902519);

Trade advertising material, commercial catalogue and the like (491110);

Other machines and apparatus for electrical machines and apparatus, other than than machines and apparatus for electroplating electrolysis electrophoresis signal generators particle accelerators. (854370);

Inductors (excl. Inductors for discharge lamps or tubes) (850450);

Television cameras, digital cameras and video camera recorders (852580):

Parts of the slide fasteners (960720);

Articles of plastics, n.e.s. (392690);

Electric conductors, for a voltage > 1.000 V, insulated, n.e.s. (854460);

Boards, panels, consoles, desks, cabinets and other bases for the goods of heading 8537, not equipped with their apparatus (853810);

Tubes, pipes & hoses, of vulcanised rubber (other than hard rubber), reinforced or otherwise combined only with textile materials, with fittings (400932);

Polypropylene, in primary forms (390210); Machines for the reception, conversion and transmission regeneration of voice, images other data, incl. Switching and routing apparatus (851762);

Polishes & creams, scouring pastes, n.e.s. (340590);

Electrical apparatus for switching or protecting electrical circuits, or for making connections to or in electrical circuits, for a voltage > 1.000 V (excl. Fuses, automatic circuit breakers, isolating switches, make-and-break switches, lightning arresters, voltage limiters, surge suppressors and control desks, cabinets, panels etc. Of heading 8537) (853590);

Springs & leaves for springs, n.e.s., of iron/steel (732090);

Machines for sorting, screening, separating or washing earth, stone, ores or other mineral substances (847410);

Pumps and compressor for air/vaccum/other gas (841480);

Wheel rims & spokes for bicycles (871492); Other printers, copying machines, singlefunction printing, capable of connecting to an automatic data processing machine or to a network (844332);

Road wheels & parts & accessories (870870); Starter motors and dual purpose startergenerators, of a kind used for spark or compression-ignition internal combustion engines (851140);

Medicaments, (not containing antibiotics, hormones, alkaloids or their derivatives), for therapeutic or prophylactic uses, (not packaged for retail sale) (300390); Parts of the air-conditioning machines, with motor driven fan and elements for temperature

control (841590);

Tubes, pipes & hoses, of vulcanised rubber (other than hard rubber), reinforced or otherwise combined only with textile materials, without fittings (400931); Sound signalling equipment; electrical, used on cycles or motor vehicles (851230); Bumpers & parts thereof of the motor vehicles

(870810); Storage units of ADP (847170); Parts And Accessories Of Instruments And Apparatus For Measuring, Checking Or

Detecting Electrical Quantities, Or Ionizing Radiations, Others (903090);

Parts and accessories for machine tools, for laser operation, metalworking machining centers, lathes and drilling machines, etc., others (846693):

Topical wood, n.e.s., in the rough, whether or not stripped of bark or sapwood, or roughly squared, untreated (440349);

Clasps, frames with clasps, buckles, buckle clasps; of base metal, of a kind used for clothing, footwear, awnings, handbags, travel goods or other made up articles, including parts (830890); Furniture of cane, osier, or similar materials (other than bamboo or rattan) (940389);

Handbags, whether/not with shoulder strap (including those without handle), with outer surface of vulcanised fibre or of paperboard (420229):

Parts of machinery, plant/laboratory equipment, for treating materials by a process involving a change of temperature (841990);

Handbags, whether/not with shoulder strap (including those without handle), with outer surface of leather, of composition leather or of patent leather (420221);

Buttons of plastics, not covered with textile material (960621);

Ceramic articles n.e.s., other than of porcelain or china (691490);

Baby carriages & parts thereof (871500); Metal furniture (other than for office use) (940320);

Dried leguminous vegetables, n.e.c., shelled, whether or not skinned or split (71390);

Clocks (excluding those with watch movements and instrument panel clocks), other than alarm or wall clocks, other than electrically operated (910599):

Tanned/crust hides & skins of bovine or equine animals, in the wet state (including wet blue), without hair on, excluding full grains, unsplit; grain splits (410419);

Personal weighing machines, incl. Baby scales (842310);

Ethylene polymers; in primary forms, polyethylene having a specific gravity of less than 0.94 (390110);

Pins other than safety pins, of iron/steel (731930);

Articles of precious metal/metal clad with precious metal, other than that of platinum (711590):

Parts & accessories of the machines or auxiliary machinery of heading no. 8444 (844820); Other articles of copper, n.e.s. (741999); Knives & cutting blades, for machines or mechanical appliances, n.e.s. (820890); Prefabricated buildings (940600); Sanitary towels & tampons, napkins (481840); Slide fasteners, other than those fitted with chain scoops of base metal (960719);

Navigational instruments and appliances for navigation other than aeronautical or space navigation (excluding direction finding compasses) (901480);

Distributors and ignition coils of a kind used for spark-ignition or compression-ignition internal combustion engines (851130);

Paper/paperboard labels of all kind, unprinted (482190);

Pulleys and flywheels, including pulley blocks (848350):

Filtering/purifying machinery & apparatus (842121);

Steering wheels, steering columns and steering boxes & parts thereof (870894);

Electrical lighting or visual signalling equipment for motor vehicles (excl. Lamps of heading 8539) (851220);

Suspension systems & parts thereof (vehicles) (870880);

Transmission belts/belting, n.e.s., of vulcanized rubber (401039);

Wooden furniture of a kind used in office (940330);

Moulds for rubber or plastics, other than injection or compression types (848079); Regulating or controlling instruments and apparatus; automatic, other than hydraulic or pneumatic (903289);

Intake air filters for internal combustion engines (842131);

Radio-broadcast receivers not capable of operating without an external source of power, of a kind used in motor vehicles, incl. Apparatus capable of also receiving radio-telephony or radio-telegraphy, not combined with sound recording or reproducing apparatus (852729);

Instruments and apparatus for measuring or checking voltage, current, resistance or power, without a recording device (excluding multimeters) (903033);

Sunglasses (900410);

Pliers (incl. Cutting pliers), pincers, tweezers and similar tools (820320);

Articles of vulcanised rubber other than hard rubber, n.e.s. (401699);

Semi conductor media, Smart cards for the recording of sound of other phenomena, but excl. Products of Ch. 37. (852352);

Milling machines (excl. Knee-type), other than numerically controlled (845969);

Parts and accessories nesoi of bicycles etc. (871499):

Clasps, frames with clasps, buckles, buckle clasps; of base metal, of a kind used for clothing, footwear, awnings, handbags, travel goods or other made up articles, including parts (830890);

Articles of plastics, n.e.s. (392690);

Wooden furniture of a kind used in bedroom (940350);

Stone; natural (excluding marble, travertine, alabaster, other calcareous stone or granite), monumental or building stone, (other than simply cut or sawn, with a flat or even surface) (680299);

Mattresses of cellular rubber or plastics, whether or not covered (940421);

Articles of yarn/strip/the like of synthetc or artificial monofilament, twine, cordage, rope or cables n.e.s. Or included (560900);

Articles of stone/of other mineral substances, n.e.s. (681599);

Collages and similar decorative plaques (artwork) (970190);

Parts of the air-conditioning machines, with motor driven fan and elements for temperature control (841590);

Knittedd/crocheted fabrics of a exceeding 30 cm, other than pile or terry fabrics, containing by weight 5% or more of elastomeric yarn or rubber thread, (600490);

Labels, badges & similar articles; woven, of textile materials, in the piece, in strips or cut to shape or size, not embroidered (580710); Narrow woven fabrics (excluding pile fabrics),

containing by weight 5% or more of elastomeric yarn or rubber thread (580620);

Ceramic tableware, kitchenware, household articles and toilet articles, other than of porcelain or china (691200);

Chemical products & preparations of the chemical or allied industries, n.e.s. (382490); Office/school supplies, of plastics (392610); Paper/paperboard labels of all kind, unprinted (482190);

Worn clothing & other worn articles (630900); Filtering/purifying machinery & apparatus (842121);

Stationery n.e.s., made of paper or paperboard (482090);

Automatic circuit breakers for a voltage <= 1.000 V (853620);

Line telephone sets with cordless handsets (851711);

Sewing machines (excl. H-hold. Type), automatic units (845221);

Other monitors, of a kind solely principally used in an automatic data processing system (852851);

Machine-tools for wood, cork or hard plastic, etc. N.e.s. (846599);

Women's/girls' blouses, shirts & shirt blouses, of silk or silk waste (not knittedd or crocheted) (620610):

Hooks, eyes & eyelets of a kind used for clothing, footwear, awnings, handbags, travel goods or other made up articles, of base metal (830810);

Instruments and apparatus for measuring or checking variables of liquids or gases (excluding pressure or the flow and level of liquids and others) (902680);

AC motors, single-phase of an output > 37.5 W (850140);

Other articles of copper, n.e.s. (741999); Parts of Machines & mechanical appliances of those having individual functions (847990); Hooks, eyes & eyelets of a kind used for clothing, footwear, awnings, handbags, travel

clothing, footwear, awnings, handbags, travel goods or other made up articles, of base metal (830810);

Inner tubes, of rubber, n.e.s. (401390); Interchangeable tools for hand tool, whether or not power-operated, for screw-driving or uses n.e.c (820790);

Unused postage, revenue/similar stamps of current or new issue in the country in which they have, or will have, a recognised face value; stamp-impressed paper; cheque forms; banknotes, stock, share or bond certificates and the like of similar title (490700);

Dental instruments and appliances; other than dental drill engines (901849);

Multimeters for measuring or checking voltage, current, resistance or power, without a recording device (903031);

Tanned/crust hides & skins of bovine or equine animals, in the dry state (crust), without hair on, (other than full grains, unsplit; grain splits) (410449);

Threaded articles of copper, n.e.s. (741539); Brakes, incl. Coaster braking hubs and hubbrakes, and parts thereof (871494); Hubs (excl. Coaster braking hubs & hub brakes) and free-wheel sprocket-wheels

(871493); Monoculars & other optical telescopes and astronomical instruments, excluding instruments for radio-astronomy (900580); Machines and appliances for testing the hardness, strength, compressibility, elasticity

hardness, strength, compressibility, elasticity or other mechanical properties of materials other than metals (902480);

Cut flowers & flower buds of a kind suitable for bouquets or ornamental purposes, fresh, other than roses, carnations, orchids, chrysanthemums or lillies (60319);

Relays, for a voltage not >60V (853641); Handbags, whether/not with shoulder strap (including those without handle), with outer surface of vulcanised fibre or of paperboard (420229);

Pneumatic power engines & motors other than linear acting (cylinders) (841239);

Multimeters with a recording device (903032); New pneumatic tyres, of rubber, of a kind used on bicycles (401150);

Parts & accessories n.e.s. In Ch.90 (903300);

Ornamental trimmings in the piece, without embroidery, other than knittedd or crocheted (580890):

Sculptures and statuary; original, in any material (970300);

Prepared enzymes, n.e.s., other than rennet and concentrates thereof (350790);

Pepper (genus Piper), neither crushed nor ground (90411);

Raw hides & skins of bovine (including buffalo) and equine animals, fresh, salted or preserved, but not tanned, parchment dressed or further prepared, whether or not dehaired or split (410190):

Cashew nuts, in shell (80131);

Articles & equipment for sports, for outdoor games and recreation n.e.c. (950699);
Babies' garments & clothing accessories, of textile materials (other than cotton or synthetic fibres), knittedd or crocheted (611190);
Safety headgear, whether/not lined/trimmed (650610);

Articles and equipment for general physical exercise, gymnastics or athletics (950691); Statuettes & other ornaments, of base metal plated with precious metal (830621);

Wood marquetry & inlaid wood; caskets and cases for jewellery or cutlery and similar articles of wood, wooden articles of furniture (442090); Iron/steel wool; pot scourers & scouring or polishing pads, gloves and the like, of iron/steel (732310);

Statuettes & other ornaments, of base metal other than plated with precious metal (830629); Palm oil and its fractions, crude, not chemically modified (151110);

Other footwear, other than with uppers of leather/composition leather/textile materials (640590);

Natural rubber (excl. Latex), in smoked sheets (400121);

Turmeric (curcuma) (91030);

Men's/boys' jackets & blazers (excl.

Knitted/crocheted), of cotton (620332);

Sheets for veneering (including those obtained by slicing laminated wood), for plywood or similar laminated wood and other wood, sawn lengthwise, sliced or peeled, planed or not, sanded, spliced or end-jointed, not over 6 mm thick (440810);

Men's/boys' shirts (excl. Knitted/crocheted), of textile materials n.e.s. (620590);

Sheets n.e.s. For veneer or plywood, other wood sawn lengthwise, sliced or peeled, whether or not planed, sanded or end-jointed, not thicker than 6mm (440890):

Sports footwear other than ski-boot, snowboard boots or cross-country ski footwear, with outer soles of rubber, plastics, leather or composition leather and uppers of leather (640319);

Sewing needles, knitting needles, bodkins, crochet hooks and similar articles, for use in the hand, n.e.s., of iron/steel (731990); Articles and equipment for general physical exercise, gymnastics or athletics (950691); Knittedd/crocheted fabrics of synthetic fibres n.e.s., of a width not exceeding 30 cm (600330);

Narrow woven fabrics (excluding pile fabrics), containing by weight 5% or more of elastomeric yarn or rubber thread (580620); Orthopaedic/fracture appliances (902110); Wrist-watches, electrically operated, with or without a stop-watch, (other than those of heading no. 9101) with mechanical display only (910211);

Drilling machines for drilling by removing metal, other than numerically controlled (845929);

Leather further prepared after tanning or crusting, including parchment-dressed leather, of bovine (including buffalo) or equine animals, without hair on, whole hides and skins, full grain, unsplit (410711); Other hosiery articles n.e.s., knittedd/crocheted, without applied soles, of textile materials n.e.s. (611599);

Ophthalmic instruments and appliances (901850);

Articles & equipment for sports, for outdoor games and recreation n.e.c. (950699); Imitation jewellery, of base metal, whether or not plated with precious metal (711719); Leather further prepared after tanning or crusting, including parchment-dressed leather, of animals (other than sheep and lambs, goats and kids, swine and reptiles), without wool or hair on, whether or not split, n.e.s. (411390); Aluminium alloys, unwrought (760120); Sheets for veneering (including those obtained by slicing laminated wood), for plywood or similar laminated wood and other wood, sawn lengthwise, sliced or peeled, planed or not, sanded, spliced or end-jointed, not over 6 mm thick (440810);

Knittedd/crocheted fabrics of a width exceeding 30 cm, other than pile or terry fabrics, containing by weight 5% or more of elastomeric yarn but not containing rubber thread (600410);

Sewing thread of artificial filaments, whether or not put up for retail sale (540120); Nuts, n.e.s., fresh or dried, whether or not shelled or peeled (80290);

Fully or partly automatic machines for resistance welding of metals (851521); Sewing thread of synthetic filaments, whether or not put up for retail sale (540110); Lace mechanically made in the piece, in strips or in motifs, of man-made fibres (580421);

Technically spec. Natural rubber (TSNR), in primary forms or in plates, sheets or strip (excluding latex and smoked sheets) (400122); Clothes hangers, of wood (442110); Men's/boys' jackets & blazers (excl. knittedd/crocheted), of wool or fine animal hair (620331);

Statuettes & other ornaments, of wood (442010); Men's/boys' trousers, bib & brace overalls, of textile materials (other than wool, fine animal hair, cotton or synthetic fibres), (not knittedd or crocheted) (620349);

Men's/boys' jackets & blazers, knittedd/crocheted, of cotton (610332); Women's/girls' blouses, shirts & shirt blouses, of man-made fibres (not knittedd or crocheted) (620640) Technically spec. Natural rubber (TSNR), in primary forms or in plates, sheets or strip (excluding latex and smoked sheets) (400122); Plywood, veneered panels and similar laminated wood (other than blockboard, laminboard and battenboard, other than of bamboo, and other than plywood consisting only of sheets of wood each ply 6mm or thinner) (441299);

Hand tools, n.e.s., of a kind used in agriculture, horticulture or forestry (820190); Motorcycles (incl. Mopeds) & cycles (871120);

Women's/girls' slips & petticoats (excl. knittedd/crocheted), of textile material other than man-made fibres (620819); Women's/girls' slips & petticoats, of textile materials (other than man-made fibres),

knittedd or crocheted (610819); Leather further prepared after tanning or crusting, including parchment-dressed leather, without hair on, whether or not split, of goats and kids (411310);

Knittedd/crocheted fabrics, n.e.s., of cotton, unbleached or bleached (600621); Plywood; consisting only of sheets of wood (not bamboo), each ply 6mm or thinner, with at least one outer ply of tropical wood (441231);

Outer soles & heels, of rubber/plastics (640620):

Games, operated by coins, banknotes, bank cards, tokens or by other means of payment, other than billiard articles and accessories, and automatic bowling alley equipment (950430):

Panty hose, tights, stockings, socks, etc., of textile materials (other than synthetic fibres), knittedd or crocheted (611529);

Women's/girls' jackets & blazers (excl. Knitted/crocheted), of textile materials (620439);

Wood, in the rough, n.e.s., whether or not stripped of bark or sapwood, or roughly squared, untreated, (440399);

Men's/boys' trousers, bib & brace overalls, of textile materials (other than wool, fine animal hair, cotton or synthetic fibres), (not knittedd or crocheted) (620349);

Women's/girls' slips & petticoats (excl. Knitted/crocheted), of man-made fibres (620811):

Topical wood, n.e.s., in the rough, whether or not stripped of bark or sapwood, or roughly squared, untreated (440349);

Sports footwear other than ski-boot, snowboard boots or cross-country ski footwear, with outer soles of rubber, plastics, leather or composition leather and uppers of leather (640319);

		Food preparations, n.e.s. (210690); Men's/boys' shirts (excl. Knitted/crocheted), of textile materials n.e.s. (620590); Men's/boys' jackets & blazers, knittedd/crocheted, of textile materials (other than wool or fine animal hair, cotton or synthetic fibres) (610339); Turmeric (curcuma) (091030)
East Timor	Pumps and compressors; parts, of air or vacuum pumps, air or other gas compressors and fans, ventilating or recycling hoods incorporating a fan (841490); Static converters (850440); Interchangeable tools for hand tool, whether or not power-operated, for screw-driving or uses n.e.s. (820790); Ferrous waste & scrap, n.e.s. (720449); Furniture for medical, surgical, veterinary use (e.g. Operating tables, examination tables, hospital beds with mechanical fittings) and parts thereof (940290); Aluminium waste & scrap (760200); Parts of telephone sets, incl. Telephones for cellular networks for other wireless networks other apparatus for the transmission reception of voice, images other data, incl. Apparatus for communication in a wired wireless network (851770); Relays, for a voltage not >60V (853641); Commodities not specified (999999); Chemical products & preparations of the chemical or allied industries, n.e.s. (382490)	Electric lamps & lighting fittings, nec (940540); Boards, cabinets and similar combinations of apparatus for electric control or the distribution of electricity, for a voltage <= 1.000 V (853710); Aluminium waste & scrap (760200); Parts of Machines & mechanical appliances of those having individual functions (847990); Baths of iron other than cast iron/steel (732429)
Laos	Electrical apparatus like switches n.e.c. In heading no. 8536, for a voltage not exceeding 1000 volts (853650); Other parts and accessories for printing machinery excl. 844391 (844399); Diagnostic/laboratory reagents on a on a backing and prepared diagnostic or laboratory reagents whether or not on a backing; certified reference material (382200); Machine-tools; parts and accessories, dividing heads and other special attachments for machine-tools (846630); Packing/wrapping machinery (842240); Machinery & apparatus for soldering, brazing or welding, whether or not capable of cutting, n.e.s. (846880); Taps, cocks, valves and similar appliances; for pipes, boiler shells, tanks, vats or the like, including thermostatically controlled valves (848180); Ferrous waste & scrap, n.e.s. (720449); Air compressors mounted on a wheeled chassis for towing (841440); Filtering/purifying machinery & apparatus (842121); Tubes, pipes & hollow profiles (not seamless), welded, of circular cross-section, n.e.s. (730630);	Protein; concentrates and textured protein substances (210610); Food preparations; of flour, meal, starch, malt extract or milk products, for uses n.e.s. (190190); Electric lamps & lighting fittings, nec (940540); Winding wire, other than of copper (854419); Unused postage, revenue/similar stamps of current or new issue in the country in which they have, or will have, a recognised face value; stamp-impressed paper; cheque forms; banknotes, stock, share or bond certificates and the like of similar title (490700); Human blood; animal blood prepared for therapeutic, prophylactic or diagnostic uses; cultures of micro-organisms (excluding yeast) and similar products (300290); Aromatic polycarboxylic acids, n.e.s. (291739); Grinding balls & similar cast articles for mills, other than of non-malleable cast iron (732591); Imitation jewellery, of base metal, whether or not plated with precious metal (711719); Agarbatti & other odoriferous preparations, which operate by burning (including those used during religious rites) (330741);

Sawing machines for working wood, cork, bone, hard rubber, hard plastics or similar hard materials (846591);

Clasps, frames with clasps, buckles, buckle clasps; of base metal, of a kind used for clothing, footwear, awnings, handbags, travel goods or other made up articles, including parts (830890); Hydraulic presses for working metal (846291); Dies for drawing/extruding metal, for machine or hand tools, whether or not power-operated (820720);

Reservoirs, tanks, vats and similar containers; for any material (excluding compressed or liquefied gas), of iron or steel, capacity exceeding 300l, whether or not lined or heat insulated (730900); Liquid dielectric transformers, having a power handling capacity <= 650 kva (850421); Forging/die-stamping machines (including presses) and hammers, for working metal (846210);

Parts of taps, cocks, valves and similar appliances (848190);

Human blood; animal blood prepared for therapeutic, prophylactic or diagnostic uses; cultures of micro-organisms (excluding yeast) and similar products (300290) Unwrought lead, refined (780110); Gold (incl. Gold plated with platinum, unwrought, for non-monetary purposes (excl. Gold in powder form) (710812);

Topical wood, n.e.s., in the rough, whether or not stripped of bark or sapwood, or roughly squared, untreated (440349);

Heat exchange units, whether/not electrically heated, for domestic purposes (841950); Coniferous wood, in the rough, whether/not stripped of bark or sapwood, or roughly squared, untreated (440320);

Plywood; consisting only of sheets of wood (not bamboo), each ply 6mm or thinner, with at least one outer ply of tropical wood (441231);

Copper ores & concentrates (260300); Wood, in the rough, n.e.s., whether or not stripped of bark or sapwood, or roughly squared, untreated, (440399); Sheets n.e.s. For veneer or plywood, other wood sawn lengthwise, sliced or peeled, whether or not planed, sanded or end-jointed,

not thicker than 6mm (440890)

Myanmar

Gear boxes & parts thereof, of vehicles (870840);

Copper tube/pipe fittings (e.g. Couplings, elbows, sleeves) of copper alloys (741220); Glass mirrors other than rear-view (700991); Mechanical appliances, for projecting, dispersing or spraying liquids or powders, for other than agricultural or horticultural use, whether or not hand-operated (842489);

Conveyor belts/belting, reinforced only with metal, of vulcanised rubber (401011); Packing containters of

paper/paperboard/cellulose wadding/fibres, n.e.s. (481950);

Boards, panels, consoles, desks, cabinets and other bases for the goods of heading 8537, not equipped with their apparatus (853810); Lenses, prisms, mirrors & other optical elements, mounted, being parts or fittings for instruments or apparatus, of any material (excluding elements of glass not optically worked) (900290); Dentifrices, in individual retail packages

(330610); Sound recording or reproducing apparatus; n.e.c.

(851989); Insulating fittings for electrical purposes, of

plastics (854720); Fixed electrical resistors for a power handling capacity <= 20 W (excl. Heating resistors) (853321);

Machines and mechanical appliances; having individual functions, n.e.c. (847989);

Electric ovens, cookers, cooking plates and boiling rings, electric grillers and roasters, for domestic use (excl. Space heating stoves and microwave ovens) (851660);

Tools for milling, for hand tools, whether or not power-operated (820770);

Parts of electrical ignition or starting equipment, generators, etc. Of heading 8511, n.e.s. (851190);

Artificial flowers, foliage & fruit, artificial, parts thereof and their plastic articles (670210);

Mountings, fittings and similar articles; suitable for other than buildings or furniture, of base metal (830249);

Pens and pencils n.e.s.; duplicating stylos, pen-holders, pencil-holders, and similar holders, parts of the articles (960899); Textile spinning machines (844520); Photographic film in rolls, sensitised and unexposed (370210);

Lamps and light fittings; parts thereof, of materials other than glass or plastics (940599); Pulleys and flywheels, including pulley blocks (848350);

Human blood; animal blood prepared for therapeutic, prophylactic or diagnostic uses; cultures of micro-organisms (excluding yeast) and similar products (300290);

Chandeliers and other electric ceiling or wall light fittings; excluding those used for lighting public open spaces or thoroughfares (940510); Cotton, not carded/combed (520100);

Parts of the vehicles of 87.16 (Trailers, semitrailers and other vehicles not mechanically propelled) (871690);

Pulleys and flywheels, including pulley blocks (848350);

Processing units for automatic data processing machines, whether or not containing in the same housing one or two of the following types of unit storage units, input units, output units (847150); Tubes, pipes & hoses of plastics, n.e.s. (391739); "Therapeutic respiration apparatus such as ozone, oxygen, aerosol therapy apparatus; artificial respiration or other therapeutic respiration apparatus (901920);"

Parts and accessories (excl. Pick-up cartridges) suitable for use solely or principally with sound reproducing and recording apparatus and with video equipment for recording and reproducing pictures and sound (excl. Pick-up devices for grooved recording media) (852290); Parts of mechanical appliances (parts of

Parts of mechanical appliances (parts of machines projecting, dispersing or spraying liquids or powders, whether or not hand-operated) (842490);

Urea, whether/not in aqueous solution (310210); Generators of a kind used for internal combustion engines (excl. Magneto dynamos and dual purpose starter-generators) (851150); Knives & cutting blades, for machines or mechanical appliances, n.e.s. (820890); Other units of automatic data processing (847180);

Hand-operated spanners & wrenches, non-adjustable (820411);

Surface-active preparations, Washing and cleaning preparations, whether or not containing soap, put up for retail sale (340220);

Electronic integrated circuits; processors and controllers, whether or not combined with memories, converters, logic circuits, amplifiers, clock and timing circuits, or other circuits (854231);

Tubes, pipes & hoses, of vulcanised rubber (other than hard rubber), reinforced or otherwise combined only with textile materials, without fittings (400931);

Table/floor/wall/window/ceiling/roof fans, with a self-contained electric motor of an output not exceeding 125W (841451);

Parts of taps, cocks, valves and similar appliances (848190);

Other monitors, not of a kind solely principally used in an automatic data processing system of heading 84.71 (852859);

Microtomes and parts & accessories thereof (902790);

Ceramic sinks, wash basins, wash basin pedestals, baths, bidets, water closet pans, flushing cisterns, urinals and similar sanitary

Plugs and sockets for a voltage <= 1000 V (excl. Lamp-holders) (853669);

Tubes, pipes & hoses, rigid, of polymers of ethylene (391721);

Other vehicles, not mechanically propelled (871680);

Air compressors mounted on a wheeled chassis for towing (841440);

Turbo-jets, of a thrust >25 kn (841112); Spark ignition reciprocating piston engines, of a kind used for the propulsion of vehicles of chapter 87, of a cylinder capacity exceeding 1000cc (840734);

Calcium carbonate (283650):

Recovered (waste & scrap) paper/paperboard, n.e.s., and of unsorted waste and scrap (470790);

Surface-active preparations, Washing and cleaning preparations, whether or not containing soap, put up for retail sale (340220);

Parts of aeroplanes/helicopters, others (880330);

Poly(vinyl chloride), non-plasticised, in primary forms, mixed with other substances (390421):

Domestic food grinders and mixers and fruit or vegetable juice extractors, with selfcontained electric motor (850940);

Artificial waxes & prepared waxes (other than of chemically modified lignite or polyethylene glycol) (340490);

Stoppers, lids, caps and other closures, for the conveyance or packing of goods (392350); Parts and accessories for string musical instruments other than keyboard instruments (920992):

Parts & accessories of the machines or auxiliary machinery of heading no. 8444 (844820);

Instruments and apparatus for physical or chemical analysis, for measuring or checking viscosity, porosity, expansion, surface tension or quantities of heat, sound or light, n.e.c. (902780);

Activated carbon (380210);

Machine-tools for wood, cork or hard plastic, etc. N.e.s. (846599);

Statuettes & other ornaments, of base metal plated with precious metal (830621); Mucilages & thickeners, whether or not modified, derived from vegetable products, n.e.s. (130239);

Granite; articles thereof, (other than simply cut or sawn, with a flat or even surface) (680293);

String musical instruments, played other than with a bow (e.g. Guitars and harps) (920290); Liquid dielectric transformers, having a power handling capacity <= 650 kva (850421);

fixtures; of other than porcelain or china (691090);

Headphones, earphones and combined microphone/speaker sets (excl. Telephone sets, hearing aids and helmets with built-in headphones, whether or not incorporating a microphone) (851830);

Transmission belts/belting, n.e.s., of vulcanized rubber (401039);

Drilling machines for drilling by removing metal, other than numerically controlled (845929);

Other colour reception apparatus for television, whether not incorporating radio broadcast receivers sound video recording reproducing apparatus, (852872);

Moulds for rubber or plastics, injection or compression types (848071);

Hat-racks, hat-pegs, brackets & similar fixtures, of base metal (830250);

AC motors, multi-phase, of an output > 750 W but <= 75 kw (850152);

Parts of the filtering/purifying machinery for filtering or purifying liquids or gases, n.e.s. (842199):

Transistors with a dissipation rate ≥ 1 W (excl. Photosensitive transistors) (854129);

Trunks, suit-cases, vanity-cases, executive-cases and similar containers, with outer surface of leather, of composition leather or of patent leather (420211):

Parts of electric water heaters, immersion heaters, space heating apparatus and soil heating apparatus, hair-dressing apparatus and hand dryers, electro-thermic appliances of a kind used for domestic purposes and electric heating resistors, n.e.s. (851690);

Engines; parts, suitable for use solely or principally with spark-ignition internal combustion piston engines (for other than aircraft) (840991);

Instruments and appliances used in medical or veterinary sciences, nes (901890);

Clocks (excluding those with watch movements and instrument panel clocks), other than alarm or wall clocks, other than electrically operated (910599);

Diodes (excl. Photosensitive or light emitting diodes) (854110);

Bumpers & parts thereof of the motor vehicles (870810);

Compressors of a kind used in refrigerating equipment (841430);

Storage units of ADP (847170);

Instruments and apparatus for measuring or checking the flow or level of liquids (902610); Parts & accessories n.e.s. In Ch.90 (903300); Registers, account books, notebooks, order books, diaries and similar articles (482010); Sunglasses (900410);

Tennis, badminton and similar racquets; whether or not strung (950659); Parts of the turbo-jets/turbo-propellers (841191):

Sound recording or reproducing apparatus; n.e.c. (851989);

Containers for compressed/liquefied gas, of iron or steel (731100);

Inductors (excl. Inductors for discharge lamps or tubes) (850450);

Parts of the machinery used for cleaning, sorting or grading seed, grain or dried leguminous vegetables or for milling or working cereals or dried leguminous vegetables (not farm-type machinery) (843790);

Raw skins of sheep/lambs, pickled (but not tanned, parchment-dressed or further preserved), without wool on (410221); Chromium ores & concentrates (261000); Aluminium alloys, unwrought (760120); Other parts, trimmings & accessories of um, brellas and walking sticks n.e.c. (660390); Zinc, not alloyed, unwrought, containing by weight less than 99.99% of zinc (790112); Wood (including strips & friezes for parquet flooring), continuously shaped along any edges, ends or faces, whether or not planed, sanded or end-jointed (440910); Artists' brushes, writing brushes and similar brushes for the application of cosmetics

(960330):

Tanned/crust hides & skins of goats or kids, without wool or hair on, whether or not split. but not further prepared, in the wet state (including wet blue) (410621);

Safety airbags with inflater system (in vehicles) & parts (870895);

Seeds of coriander (90920);

Articles of goldsmiths'/silversmith and their parts, , of base metal clad with precious metal (711420);

Binoculars (900510);

Seeds of cumin (90930);

Unwrought lead other than refined, not containing by weight antimony as the principal other element (780199);

Zinc, not alloyed, unwrought, containing by weight 99.99% or more of zinc (790111);

Plywood; consisting only of sheets of wood (not bamboo), each ply 6mm or thinner, with at least one outer ply of non-coniferous wood, (without an outer ply of tropical wood)

Raw hides & skins of bovine (including buffalo) and equine animals, fresh, salted or preserved, but not tanned, parchment dressed or further prepared, whether or not dehaired or split (410190);

New pneumatic tyres, of rubber, of a kind used on motor cars (including station wagons and racing cars) (401110);

Parts and accessories of data processing equipment (847330);

Drive-axles with differential, whether or not provided with other transmission components, and non-driving axles; parts thereof (870850); Taps, cocks, valves and similar appliances; for pipes, boiler shells, tanks, vats or the like, including thermostatically controlled valves (848180):

Engines parts for internal combustion piston engines (excluding spark-ignition) (840999); Parts of tools for working in the hand, parts thereof for other than chain saws and pneumatic tools (846799);

Glassware of a kind used for table/kitchen purposes n.e.s., other than of lead crystal (701399);

Articles of precious metal/metal clad with precious metal, other than that of platinum (711590);

Festive, carnival or other entertainment articles including novelty jokes and conjuring tricks; other than Christmas festivity articles (950590); Intake air filters for internal combustion engines (842131);

Photosensitive semiconductor device (854140); Twine, cordage, ropes, cables of synthetic fibres other than polyethylene or polypropylene, whether or not plaited, braided or impregnated, coated, covered or sheathed with rubber or plastics (560750);

Filtering/purifying machinery & apparatus, other than intake air filters for internal combustion engines (842139);

Machines for the reception, conversion and transmission regeneration of voice, images other data, incl. Switching and routing apparatus (851762);

Parts of the machinery & apparatus for soldering, brazing or welding, whether or not capable of cutting, parts thereof (846890);

Fixed electrical capacitors (excl. Tantalum, aluminium electrolytic, ceramic, paper, plastic and power capacitors) (853229);

Offset printing machinery, used for printing by means of plates, cylinders and other printing components of heading 84.42, n.e.s. (844319); Parts and accessories for machine tools, for laser operation, metalworking machining centers, lathes and drilling machines, etc., others (846693);

Electrical apparatus; n.e.c., for switching or protecting electrical circuits, for a voltage not exceeding 1000 volts (853690);

Parts of the industrial/laboratory furnaces and ovens (including incinerators) (841790);

Bells, gongs & the like, non-electric, of base metal (830610);

Builders' joinery&carpentry of wood (441890);

Parings & other waste of leather/composition leather; not suitable for the manufacture of leather articles; leather dust, powder and flour (411520);

Trunks, suit-cases, vanity-cases, executive-cases, and similar containers, n.e.s., of vulcanised fibre or of paperboard, or wholly or mainly covered with such materials or with paper (420299);

Articles of goldsmiths'/silversmith and their parts, whether or not plated or clad with other precious metal (711411);

Topical wood, n.e.s., in the rough, whether or not stripped of bark or sapwood, or roughly squared, untreated (440349);

Aeroplanes and other aircraft of an unladen weight exceeding 15,000kg (880240); Wood sawn/chipped length wise, sliced or peeled, whether or not planed, sanded or endjointed, of a thickness exceeding 6mm (440710);

Unwrought lead, refined (780110); Raw skins of sheep/lambs (fresh/salted/dried/limed/otherwise, preserved other than by pickling, (but not tanned, parchment-dressed or further preserved), without wool on (410229); Chickpeas (garbanzos), dried, shelled, whether

or not skinned or split (71320)

Other machines and apparatus for electrical machines and apparatus, other than than machines and apparatus for electroplating electrolysis electrophoresis signal generators particle accelerators. (854370);

Coal; (other than anthracite and bituminous), whether or not pulverised but not agglomerated (270119);

Non-threaded articles, n.e.s., of iron/steel (731829);

Other parts and accessories of looms, auxiliary machinery, n.e.s. (844849);

Gaskets, washers & other seals of vulcanised rubber, of non-cellular rubber (401693);

Table, kitchen/other household articles and parts thereof, (other than pot scourers and scouring or polishing pads, gloves and the like), of copper (741819);

Ballasts for discharge lamps/tubes (850410); Parts of the machinery of 84.41, for making up paper pulp, paper or paperboard, including cutting machines of all kinds (844190);

Rodenticides&other biocides, n.e.s. (380899); Hinges of base metal (830210);

Fuel/lubricating/cooling medium pumps for internal combustion piston engines (841330); Surveying equipment; articles n.e.s., including hydrographic, oceanographic, hydrological, meteorological or geophysical instruments and appliances (excluding compasses) (901580); Blankets (excl. Electric) & travelling rugs, n.e.s. (630190);

Plates, sheets, film, foil & strip, of plastics n.e.s., non-cellular and not reinforced, laminated, supported or similarly combined with other materials (392099);

Jacks & hoists, (other than hydraulic), of a kind used for raising vehicles (842549);

Seats of a kind used for motor vehicles (940120); Other tubes, pipes and hollow profiles (for example, open seam or welded, riveted or similarly closed), of iron or steel (730690); Sign-plates, name-plates, address-plates and similar plates, numbers, letters and other symbols; of base metal, excluding those of

heading no. 9405 (831000); Builders' ware of plastics, n.e.s. (392590); Insulated electric conductors for a voltage not exceeding 1000 volts, not fitted with connectors (854449);

Static converters (850440);

Hydraulic power engines & motors other than linear acting (cylinders) (841229);

Parts of the sewing machines n.e.s. (845290); Tanned/crust hides & skins of bovine or equine animals, in the dry state (crust), without hair on, (other than full grains, unsplit; grain splits) (410449);

Pumps and compressor for air/vaccum/other gas (841480);

Parts & accessories for the metal shaping machine tools (846694);

Belts & bandoliers, of leather/composition leather (420330);

Telephones for cellular networks or for other wireless networks (851712);

Pumps and compressors; parts, of air or vacuum pumps, air or other gas compressors and fans, ventilating or recycling hoods incorporating a fan (841490):

Tools for boring/broaching, for hand tools, whether or not power-operated (820760); Discs, tapes, solid state non volatile storage devices, smart cards and other media for the recording of sound of other phenomena, whether not recorded, incl. Matrices and masters for the production of discs, but excl. Products of Ch.37., other n.e.s. (852380);

Paints & varnishes (incl. Enamels & lacquers), (based on polymers other than acrylic or vinyl), dispersed or dissolved in an aqueous medium (320990);

Self-adhesive plates, sheets, film, foil, tape, strip, other flat shapes thereof, other than in rolls of a width not exceeding 20cm (391990);

Prepared enzymes, n.e.s., oother than rennet and concentrates thereof (350790);

Tools for drilling, other than rock, for machine or hand tools, whether or not power-operated (820750);

Threaded nuts of iron/steel (731816); Moulds for rubber or plastics, other than injection or compression types (848079); Tricycles, scooters, pedal cars and similar wheeled toys; dolls' carriages; dolls; other toys; reduced-size (scale) models and similar recreational models, working or not; puzzles of all kinds (950300);

Printed books, brochures, leaflets, in single sheets, whether or not folded (490110); Other Insulated electric conductors; for a voltage not exceeding 1000 volts, fitted with connectors (854442);

Catheters, cannulae & the like (901839); Metal furniture of a kind used in office (940310); Mountings, fittings and similar articles; suitable for other than buildings or furniture, of base metal (830249);

Iron or steel threaded articles except screw, nut, bolts (731819);

Aluminium alloys, bars, rods and profiles, other than hollow (760429);

Parts and accessories for instruments, appliances and machines measuring or checking (903190); Electrical insulators (excl. Those of glass or ceramics and insulating fittings) (854690); Recovered (waste & scrap) paper/paperboard, n.e.s., and of unsorted waste and scrap (470790); Video recording or reproducing apparatus; magnetic tape-type (852110);

Articles of iron/steel, n.e.s. (732690); Spark-ignition reciprocating/rotary internal combustion piston engines for aircrafts (840710); Tanks, casks, drums, cans, boxes and similar containers for any material (excluding compressed or liquefied gas) less than 50l capacity, n.e.s., of iron or steel (731029); Parts of locks, of base metal (830160); Parts and accessories nesoi of bicycles etc. (871499):

Articles of plastics, n.e.s. (392690); Coated rods & cored wire, of base metal, for soldering, brazing or welding by flame (831130); Refrigerating or freezing equipment; parts thereof, other than furniture (841899); Loudspeakers; not mounted in their enclosures (851829);

Marble and travertine, merely cut, by sawing or otherwise, into blocks or slabs of a rectangular (including square) shape, having a specific gravity of 2.5 or more (251512);

Machines which perform two or more of the functions of printing, copying or facsimile transmission, capable of connecting to an automatic data processing machine or to a network (844331);

Bearings; n.e.c., including combined ball/roller (848280);

Universal AC/DC motors of an output exceeding 37.5W (850120);

Chandeliers and other electric ceiling or wall light fittings; excluding those used for lighting public open spaces or thoroughfares (940510); Ceramic wares; pots, jars and similar articles of a kind used for the conveyance or packing of goods and ceramic troughs, tubs and similar receptacles used in agriculture (690990); Acrylonitrile-butadiene-styrene (ABS) copolymers, in primary forms (390330); Handbags, whether/not with shoulder strap (including those without handle), with outer surface of leather, of composition leather or of patent leather (420221);

Suspension systems & parts thereof (vehicles) (870880);

Mixtures of odoriferous substance and mixtures, used as raw materials in industries other than the food or drink industries (330290);

Aerials & aerial reflectors (852910);

Cuff-links & studs of base metal, whether or not plated with precious metal (711711);

Printed circuits (853400);

Parts suitable for use solely or principally with the apparatus of heading 8535, 8536 or 8537, n.e.s. (excl. Boards, panels, consoles, desks, cabinets and other bases for the goods of heading 8537, not equipped with their apparatus) (853890):

Natural/artificial abrasive powder/grain, on a base of paper or paperboard only, whether or not

cut to shape or sewn or otherwise made up (680520);

Other printers, copying machines, singlefunction printing, capable of connecting to an automatic data processing machine or to a network (844332);

Articles of a kind normally carried in the pocket or in the handbag, with outer surface of leather or composition leather or of patent leather (420231):

Vegetable, mineral carving material and articles of these materials, moulded or carved articles of wax, stearin, natural gums, resins or modelling pastes, worked unhardened gelatin (960200); Acrylic polymers (other than polymethyl methacrylate), in primary forms (390690); Plates, sheets & strip, of non-cellular rubber (400821);

Parts of the slide fasteners (960720); Springs & leaves for springs, n.e.s., of iron/steel (732090);

Clasps, frames with clasps, buckles, buckle clasps; of base metal, of a kind used for clothing, footwear, awnings, handbags, travel goods or other made up articles, including parts (830890); Gears & gearing (not toothed wheels, chain sprockets and other transmission elements presented separately); ball or roller screws; gear boxes and other speed changers, including torque converters (848340);

Multiple loudspeakers, mounted in the same enclosure (851822);

Oxidation inhibitors, gum inhibitors, viscosity improvers, anti-corrosive preparations, other prepared additives for mineral oils or liquids used as mineral oils (including gasoline), n.e.s. (381190);

Other drinking glasses, other than of lead crystal or glass-ceramics (701337);

Folding cartons, boxes & cases, of non-corrugated paper or paperboard (481920); Plates, sheets, film, foil & strip, other than cellular (392190);

Paints & varnishes (incl. Enamels & lacquers), based on polymers n.e.s., dispersed or dissolved in a non-aqueous medium (320890); Sewing needles, knitting needles, bodkins, crochet hooks and similar articles, for use in the hand, n.e.s., of iron/steel (731990); Cathode-ray television picture tubes, including video monitor cathode-ray tubes, colour (854011):

Carpets and other textile floor coverings; tufted, of textile materials (other than man-made, wool or fine animal hair), whether or not made up (570390):

Ignition wiring sets and other wiring sets for vehicles, aircraft or ships (854430);

Boards, cabinets and similar combinations of apparatus for electric control or the distribution of electricity, for a voltage <= 1.000 V (853710); Textile fabrics; impregnated, coated, covered or laminated with plastics, (excluding polyvinyl chloride, polyurethane, polyamides polyesters) (590390);

Slide fasteners, other than those fitted with chain scoops of base metal (960719);

Pigments (incl. Metallic powders & flakes), used in the manufacture of paints, other than stamping foils (321290);

Pistachios (80250);

Iron or steel tubes, pipes and hollow profiles, seamless, n.e.s. (730490);

Signal generators (854320);

Ball bearings (848210);

Other glassware of a kind used for table or kitchen purposes (not drinking glasses or of glass-ceramics) (701349);

Paper, paperboard, cellulose wadding and webs of soft cellulose fibres; coated, impregnated, covered, surface-decorated or surface-coloured, n.e.s. (481190);

Fused magnesia; dead-burned (sintered), whether or not containing small quantities of other oxides added before sintering, other magnesium oxide, whether or not pure, (not natural magnesium carbonate) (251990);

Electrical apparatus like switches n.e.c. In heading no. 8536, for a voltage not exceeding 1000 volts (853650);

Bearing housings, incorporating ball or roller bearings (848330):

Coaxial cable and other coaxial electric conductors, insulated (854420);

Parts of the pumps of 8413.11-8413. (841391); Marble, travertine and alabaster; simply cut or sawn, with a flat or even surface (680221); Parts & accessories of bodies (other than safety

Flanges of iron/steel (excl. Of cast iron), of material other than stainless steel (730791); Articles of vulcanised rubber other than hard rubber, n.e.s. (401699);

Other telephone sets (851718);

seat belts) (870829);

Ferrous waste & scrap, n.e.s. (720449);

Regulating or controlling instruments and apparatus; automatic, other than hydraulic or pneumatic (903289);

Baths of iron other than cast iron/steel (732429); Textile fabrics; impregnated, coated, covered or laminated with poly(vinyl chloride) (590310); Articles of iron/steel, n.e.s., forged/stamped, but not further worked (732619);

Threaded screws & bolts of iron/steel, n.e.s., whether or not with their nuts or washers (731815);

Road wheels & parts & accessories (870870);

Electric motors of an output not exceeding 37.5W (850110);

Polypropylene, in primary forms (390210);

Parts & accessories of the of spinning, combing, drawing machines, n.e.s. (844839);

Unwrought lead, refined (780110);

Glass beads, imitation pearls, imitation precious or semi-precious stones and similar glass smallwares (701810);

Anti-oxidising preparations and other compound stabilisers; for rubber or plastics, n.e.s. Or included (381230);

Office/school supplies, of plastics (392610); Carboys, bottles, flasks, jars, pot and other containers of glass, (not ampoules), used for the conveyance or packing of goods (701090); Glycerol (290545);

Metal furniture (other than for office use) (940320);

Refrigerating/freezing equip. N.e.s (841869); Residual products of the chemical/allied industries, n.e.s. (382590);

Non-threaded washers, of iron/steel, excluding spring and lock (731822);

Projectors, not of a kind solely principally used in an automatic data processing system of heading 84.71 (852869);

Other monitors, of a kind solely principally used in an automatic data processing system (852851);

Clutches and shaft couplings (including universal joints) (848360);

Commodities not specified (999999);

Handkerchiefs, of textile materials other than cotton (not knittedd or crocheted) (621390); Other apparatus for transmission reception of

voice, images other data, incl. Apparatus for communication in a wired wireless network (such as a local wide area network) (851769);

Electro-diagnostic apparatus (including apparatus for functional exploratory examination or for checking physiological parameters), not electro-cardiographs (901819);

Coffee, not roasted, not decaffeinated (90111); Pumps n.e.s. In 84.13 (841381);

Stranded wire, ropes & cables of iron/steel, not electrically insulated (731210);

Aluminium foil, not backed, not rolled, of a thickness not exceeding 0.2mm (760719);

Plugs and sockets for a voltage <= 1000 V (excl. Lamp-holders) (853669);

Parts suitable for use solely or principally with electric motors and generators, electric generating sets and rotary converters, n.e.s. (850300):

Tailors' dummies and other lay figures; automata and other animated displays used for shop window dressing (961800);

Prepared additives for cements/mortars/concretes (382440);

Synthetic staple fibres, not carded, combed or otherwise processed for spinning (550320); Drills of all kinds, for working in the hand, with self-contained electric motor (846721); Furniture for medical, surgical, veterinary use (e.g. Operating tables, examination tables, hospital beds with mechanical fittings) and parts thereof (940290);

Tubes, pipes & hoses, of vulcanised rubber (other than hard rubber), reinforced or otherwise combined with materials other than metal or textiles, without fittings (400941); Woven fabrics, containing 85% or more by weight of textured polyester filaments, dved

weight of textured polyester filaments, dyed (540752); Parts of the air-conditioning machines, with

Parts of the air-conditioning machines, with motor driven fan and elements for temperature control (841590);

Apples, fresh (80810);

Electric accumulators n.e.s., including separators, whether or not rectangular (including square) (850780);

Tanned/crust skins of sheep/lambs, without wool on, whether or not split, but not further prepared, in the dry state (410530);

Single yarn other than sewing thread, of synthetic staple fibres, containing 85% or more by weight of polyester, not put up for retail sale (550921);

Unglazed ceramic tiles, cubes and similar articles, whether or not rectangular, the largest surface area of which is capable of being enclosed in a square with side<7 cm (690710); Ethylene polymers; in primary forms, polyethylene having a specific gravity of less than 0.94 (390110);

Other parts & accessories for the motor vehicles (870899);

Organic surface-active agents, surface-active, whether or not containing soap, including auxiliary washing preparations, not for retail sale (340290);

Textured yarn other than sewing thread, of polyesters, not for retail sale (540233); Aluminium waste & scrap (760200); Plates, sheets, film, foil&strip, cellular, of plastics n.e.s. (392119);

Winding wire, other than of copper (854419); Other parts, trimmings & accessories of um,brellas and walking sticks n.e.c. (660390); Preparations of a kind used in animal feeding, put up for retail sale (except dog or cat food) (230990);

Leather further prepared after tanning incl. Parchment-dressed, of bovine (including buffalo) or equine animals, no hair, and whole hides and skins, and sides, (full grains, unsplit and grain splits) (410799);

Tableware & kitchenware, of porcelain or china (691110);

Parts of the seats of 94.01 (seats in vehicles, aircrafts, convertible to bed, etc.) (940190); Wine other than sparkling wine of fresh grapes, in containers holding 2 litres or less (220421); Brakes & servo-brakes; parts thereof (vehicles) (870830);

Oil/petrol-filters for internal combustion engines (842123);

Agarbatti & other odoriferous preparations, which operate by burning (including those used during religious rites) (330741);

Tools (other than saws and drill) for working in the hand, with with self-contained electric motor (846729);

Iron/steel wool; pot scourers & scouring or polishing pads, gloves and the like, of iron/steel (732310):

Glutamic acid & its salts (292242);

Pictures, designs & photographs, n.e.s. (491191); Hand-made paper & paperboard (480210);

Silicones, in primary forms (391000);

Tubes, pipes & hoses, of vulcanised rubber (other than hard rubber), reinforced or otherwise combined with materials other than metal or textiles, with fittings (400942);

Pigments & preparations based on titanium dioxide, containing less than 80% by weight of titanium dioxide calculated on the dry weight (320619);

Parts of cranes, work trucks, shovels, constr machine (843149);

Saturated acyclic monocarboxylic acid; anhydrides, halides, peroxides, peroxyacids and halogenated, sulphonated, nitrated or nitrosated derivatives, n.e.s. (291590);

Sanitary towels & tampons, napkins (481840); Leaf-springs & leaves therefor, of iron/steel (732010);

Paper/paperboard labels of all kind, printed (482110);

Electric ovens, cookers, cooking plates and boiling rings, electric grillers and roasters, for domestic use (excl. Space heating stoves and microwave ovens) (851660);

Other Electro-magnets and their parts (excl. Magnets for medical use); electro-magnetic or permanent magnet chucks, clamps and similar holding devices and their parts, n.e.s. (850590); Tubes & pipes of aluminium alloys (760820); Spray guns & similar appliances (842420); Medium density fibreboard (MDF) of wood, of a thickness not exceeding 5 mm (441112); Adhesives based on polymers or rubber (350691);

Calcium carbonate (283650);

Floor-cloths, dish-cloths, dusters and similar cleaning cloths (630710);

Cashew nuts, in shell (80131);

Hard rubber (e.g., ebonite) in all forms, including waste and scrap, and articles of hard rubber (401700);

Plates, sheets, film, foil&strip, not self-adhesive, of polymers of ethylene, non-cellular and not reinforced, laminated, supported or similarly combined with other materials (392010); Electric conductors, for a voltage > 1.000 V, insulated, n.e.s. (854460);

Radiators & parts thereof for the motor vehicles (870891):

Spectacles, goggles & the like (other than sunglasses), corrective, protective or other (900490);

Densified wood, in blocks/plates/strips/profile shapes (441300);

Hand tools, other than household tools (820559); Tableware & kitchenware, of plastic (392410); Worked monumental/building stone of granite; articles thereof, simply cut or sawn, with a flat or even surface (680223);

Unwrought lead other than refined, not containing by weight antimony as the principal other element (780199);

Trunks, suit-cases, vanity-cases, executive-cases, and similar containers, n.e.s., of vulcanised fibre or of paperboard, or wholly or mainly covered with such materials or with paper (420299); Hairpins, curling pins, curling grips, hair curlers and the like and parts thereof (961590);

Activated carbon (380210);

Mountings, fittings & similar articles, for motor vehicles, of base metal (830230);

Composite paper & paperboard (made by sticking layers together with an adhesive), not surface-coated or impregnated, whether or not internally reinforced, in rolls or sheets (480700); Plates, sheets, film, foil&strip, of polymers of propylene, non-cellular and not reinforced, laminated, supported or similarly combined with other materials (392020);

Ultra-violet or infra-red lamps (853949); Sewing machines (excl. H-hold. Type),

automatic units (845221);

Polyacetals, in primary forms (390710); Bearing housings, incorporating ball or roller bearings (848320);

Armoured/reinforcededed safes, strong-boxes, doors and safe deposit lockers for strong-rooms, cash or deed boxes and the like, of base metal (830300);

Boards, cabinets and similar combinations of apparatus for electric control or the distribution of electricity, for a voltage > 1.000 V (853720); Toothed wheels, chain sprockets and other transmission elements presented separately; parts of transmission components (848390); Injection-moulding machines (847710); Prepared foods obt. By the swelling or roasting

of cereals or cereal products (190410);

319 | Page

Shampoos (330510);

Table/kitchen/other h-hold. Articles and their parts, of aluminium (761519);

Sacks & bags (excl. Those having a base of a width of 40cm or more) (481940);

Narrow woven pile fabrics (incl. Terry towelling and similar terry fabrics) and chenille fabrics (excl. Labels, badges and similar articles) (580610);

Video recording or reproducing apparatus, whether or not incorporating a video tuner (excl. Magnetic tape-type and video camera recorders) (852190);

Articulated link chain (other than roller), of iron or steel (731512);

Yarn other than high tenacity/textured, filament, monofilament (less than 67 decitex), single, untwisted or twisted 50 turns or less per metre, not for retail sale, not sewing thread (540245); Sparking plugs of a kind used for spark-ignition or compression-ignition internal combustion engines (851110);

Other cooking appliances & plate warmers, for solid fuel and fuels other than gas or liquid, of iron or steel (732119);

Ties, bow ties & cravats, of silk or silk waste (not knittedd or crocheted) (621510); Valves for oleohydraulic/pneumatic

Valves for oleohydraulic/pneumati transmissions (848120);

Potatoes, prepared or preserved otherwise than by vinegar or acetic acid, frozen (200410); Woven fabrics, containing 85% or more by weight of filaments of nylon or other polyamides, dyed (540742);

Cloves (whole fruit, cloves & stems) (90700); Petroleum oils and oils from bituminous minerals, not containing biodiesel, not crude, not waste oils; preparations n.e.s. (271019);

Pickling preparations for metal surfaces; soldering, brazing or welding powders and pastes consisting of metal and other materials (381010); Articles of iron/steel wire (732620);

Candles, tapers&the like (340600);

Candles, tapers&the like (540000);

Rear-view mirrors for vehicles (700910);

Cinnamon and cinnamon-tree flowers, other than cinnamon (Cinnamomum zeylanicum Blume), neither crushed nor ground (90619);

Tin, not alloyed, unwrought (800110);

Machinery for working rubber/plastic (847780); Non-ionic surface-active agents, whether or not put up for retail sale (340213);

Woven fabrics of cotton, containing less than 85% by weight of cotton, mixed mainly or solely with man-made fibres, weighing 200g/m2 or less, dyed, plain weave (521031);

Air compressors mounted on a wheeled chassis for towing (841440);

Plates, sheets, film, foil & strip, of polyurethanes, cellular (392113);

Tanned/crust hides & skins of bovine or equine animals, in the wet state (including wet blue), without hair on, excluding full grains, unsplit; grain splits (410419);

Articles of vulcanised rubber (other than hard rubber), moulded rubber mats and mats of non-rectangular shape made by cutting from the piece, of cellular rubber (401610);

Apparatus based on use of x-rays and similar; parts and accessories (x-ray generators, tubes, high tension generators, control panels and desks, screens, examination or treatment tables, chairs and like (902290);

Carriages for disabled persons, mechanically propelled (871390);

Raw hides&skins (fresh,/salted, dried, limed, pickled or otherwise preserved), but not tanned, parchment-dressed or further prepared, whether or not dehaired or split (410390);

Internal combustion piston engines (diesel or semi-diesel engines), of a kind used for other than marine propulsion or the vehicles of chapter 87 (840890);

Sugar confectionery; (excluding chewing gum, including white chocolate), not containing cocoa (170490);

Laboratory/hygienic/pharmaceutical glassware, whether or not graduated or calibrated, of glass n.e.s. (701790);

Chloroprene (chlorobutadiene) rubber (CR), (other than latex), synthetic, in primary forms or in plates, sheets or strip (400249);

Plasters (consisting of calcined gypsum or calcium sulphate), whether or not coloured, with or without small quantities of accelerators or retarders (252020);

Self-propelled mechanical shovels & shovel loaders, with a 360 degree revolving super structure (842952);

Sets of assorted kitchen/tableware articles (e.g. Spoons, forks, ladles, skimmers, cake-servers, fish-knives, butter knives, sugar tongs and similar), not plated with precious metal (821520);

Glazed ceramic flags & paving/hearth/wall tiles (690890);

Protein; concentrates and textured protein substances (210610);

Yachts & other vessels for pleasure or sports, rowing boats and canoes, nec (890399); Woven fabrics of cotton, containing 85% or more by weight of cotton, dyed, plain weave, weighing more than 100g/m2 but not more than 200g/m2 (520832);

Chocolate & other food preparations containing cocoa; n.e.s. (180690);

Cutlery, other than plated with precious metal (821599);

DC Electric motors and generators of an output not exceeding 750W (850131);

Percussion musical instruments (e.g. Drums, xylophones, cymbals, castanets, maracas) (920600);

Compounded rubber, compounded with carbon black or silica, in primary forms or in plates, sheets or strip (400510);

Magnetic media for the recording of sound of other phenomena, but excl. Products of Ch. 37., other than cards incorporating a magnetic stripe (852329):

Particle board & similar board of ligneous material, n.e.s. (441090);

Rangefinders (901510);

Spark-ignition reciprocating/rotary internal combustion engines, outboard motors for marine propulsion, (840721);

Electric accumulators, i.e. Lead-acid, (other than for starting piston engines), including separators, whether or not rectangular (including square) (850720);

String musical instruments, played other than with a bow (e.g. Guitars and harps) (920290); Rosin & resin acids (380610);

Equipment for scaffolding/shuttering/pit-propping, of iron/steel (730840);

Peas (Pisum sativum), dried, shelled, whether or not skinned or split (71310);

Rolled/flaked grains of oats (110412);

Vegetable fats and oils and their fractions; partly or wholly hydrogenated, inter-esterified, reesterified or elaidinised, whether or not refined, but not further prepared (151620);

Products suit. For use as glues/adhesives, put up for retail sale as glues or adhesives, not exceeding a net weight of 1kg (350610); Linear acting (cylinders) hydraulic power

engines and motors (841221);

Seeds of anise/badian (90910);

Containers (including containers for transport of fluids) specially designed and equipped for carriage by one or more modes of transport (860900);

Dark Red Meranti, Light Red Meranti and meranti bakau, wood in the rough, whether or not stripped of bark or sapwood, or roughly squared, untreated (440341);

Foliage, branches and other parts of plants, without flowers or flower buds, and grasses, suitable for bouquets or for ornamental purposes, fresh/dried/dyed/bleached/impregnated/otherwise prepared (60499);

Oil-cake and other solid residues; whether or not ground or in the form of pellets, resulting from the extraction of sunflower seed oils (230630); Woven fabrics of cotton, containing 85% or more by weight of cotton, dyed, plain weave, weighing not more than 100g/m2 (520831); Tanned/crust hides & skins of bovine or equine animals, in the wet state (including wet blue),

without hair on, full grains, unsplit; grain splits (410411): Shaving, hair, nail, eyelash and other toilet brushes for use on the person, including brushes as parts of appliances (960329); Colouring matter of vegetable/animal origin, including dyeing extracts not animal black, whether or not chemically defined; preparations based on colouring matter of vegetable or animal origin (320300); Hydrogen peroxide, whether or not solidified with urea (284700); Rotary positive displacement pumps, n.e.s. In 8413, for liquids (841360): Cane/beet sugar & chemically pure sucrose, in solid form, not containing added flavouring or colouring matter (170199); Drawing, marking-out or mathematical calculating instruments (901720); Oriented strand board (OSB) of wood, whether or not agglomerated with resins or other organic binding substances (441012); Antimony oxides (282580); Bars & rods of stainless steel, cold-formed or cold-finished (722220); Wood marquetry & inlaid wood; caskets and cases for jewellery or cutlery and similar articles of wood, wooden articles of furniture (442090) Nepal Carboys, bottles, flasks, jars, pot and other Parts of Machinery for working rubber or containers of glass, (not ampoules), used for the plastics (847790); conveyance or packing of goods (701090); Automatic circuit breakers for a voltage <= Gear boxes & parts thereof, of vehicles 1.000 V (853620): (870840); Acrylic polymers (other than polymethyl Ball bearings (848210); methacrylate), in primary forms (390690); Newsprint, in rolls/sheets (480100); Hard rubber (e.g., ebonite) in all forms, Hand-operated spanners & wrenches, nonincluding waste and scrap, and articles of hard adjustable (820411); rubber (401700); Engines; parts, suitable for use solely or Spectrometers, spectrophotometers and principally with spark-ignition internal spectrographs using optical radiations (UV, combustion piston engines (for other than visible, IR) (902730); aircraft) (840991); Tubes, pipes & hoses, of vulcanised rubber (other than hard rubber), reinforced or Hinges of base metal (830210); Pumps and compressor for air/vaccum/other gas otherwise combined with materials other than (841480): metal or textiles, with fittings (400942); Parts and accessories nesoi of bicycles etc. Kiwifruit, fresh (81050); (871499); Polymers of vinyl acetate/of other vinylesters, Parts of the air-conditioning machines, with n.e.s., in primary forms, other than motor driven fan and elements for temperature copolymers (390599); control (841590); Parts of the filtering/purifying machinery for Wheat other than durum wheat; mesli (100190); filtering or purifying liquids or gases, n.e.s. Instruments and apparatus for measuring or (842199); checking voltage, current, resistance or power, Electric accumulators; lead-acid, of a kind without a recording device (excluding used for starting piston engines, including multimeters) (903033); separators, whether or not rectangular Lubricating preparations (excl. Those for the (including square) (850710); treatment of textile and similar materials), not Pumps and compressor for air/vaccum/other containing petroleum oils or oils obtained from gas (841480); bituminous minerals (340399); Non-threaded articles, n.e.s., of iron/steel Parts of Machines & mechanical appliances of (731829);

those having individual functions (847990);

Compressors of a kind used in refrigerating equipment (841430);

Tube/pipe fittings of iron/steel, n.e.s,, other than stainless steel (730799);

Semi conductor media, solid state non-volatile storage devices, for the recording of sound of other phenomena, but excl. Products of Ch. 37. (852351);

Moulds for rubber or plastics, injection or compression types (848071);

Amino-resins, n.e.s., in primary forms (390930); Other printers, copying machines, singlefunction printing, not capable of connecting to an automatic data processing machine or to a network (844339);

Transmission belts/belting, n.e.s., of vulcanized rubber (401039);

Beans of the species vigna mungo (L.) Hepper or vigna radiata (L.) Wilczek, shelled, whether or not skinned or split, dried (71331);

Parts & accessories of regulating or controlling instruments and apparatus; automatic (903290); Epoxide resins, in primary forms (390730); Digital Automatic Data Processing Machines And Units Thereof Presented In The Form Of Systems, N.e.s.o.i. (847149);

Vehicles with only compression-ignition internal combustion piston engine (diesel or semi-diesel), cylinder capacity over 1500 but not over 2500cc (870332);

Used/new rags, scrap twine, cordage, rope and cables and worn out articles of twine, cordage, rope or cables, of textile materials; other than sorted (631090);

Sewing machines of the household type (845210);

Refrigerating or freezing equipment; parts thereof, other than furniture (841899); Paints & varnishes (incl. Enamels & lacquers),

based on polymers n.e.s., dispersed or dissolved in a non-aqueous medium (320890);

Cutting machines for paper/pulp/paperboard (844110);

Locks (key/combination/electrically operated), other than those for motor vehicles or furniture, of base metal (830140);

Textile fabrics; impregnated, coated, covered or laminated with plastics, (excluding polyvinyl chloride, polyurethane, polyamides polyesters) (590390);

Perfumes & toilet waters (330300);

Rolls for rolling mills (845530);

Machinery for filling, closing, sealing, capsuling or labelling bottles, cans, bags or other containers (842230);

Silica sands & quartz sands, whether or not coloured (250510);

Chocolate & other food preparations containing cocoa; n.e.s. (180690);

Sewing machines, not household or automatic unit type (845229);

Electrical measurement instruments nes., without a recording device (903089); Poly(ethylene terephthalate), in primary forms, having a viscosity of 78ml/g or higher (390760);

Organic comps. N.e.s. (294200);

Pumps and compressors; parts, of air or vacuum pumps, air or other gas compressors and fans, ventilating or recycling hoods incorporating a fan (841490);

Electric motors of an output not exceeding 37.5W (850110);

Self-adhesive plates, sheets, film, foil, tape, strip, other flat shapes thereof, other than in rolls of a width not exceeding 20cm (391990); Lifting machinery n.e.s. In 84.26, self-propelled derricks and cranes, not on tyres (842649);

Other parts and accessories for printing machinery excl. 844391 (844399); Bamboos (140110);

Gears & gearing (not toothed wheels, chain sprockets and other transmission elements presented separately); ball or roller screws; gear boxes and other speed changers, including torque converters (848340); Tubes, pipes & hoses of plastics, n.e.s. (391739);

Tube/pipe fittings of iron/steel, n.e.s,, other than stainless steel (730799);

Filtering/purifying machinery & apparatus, other than intake air filters for internal combustion engines (842139);

Gaskets and similar joints of metal sheeting combined with other material or two or more layers of metal (848410);

Ferrous waste & scrap, n.e.s. (720449); Parts & accessories of the machines for knitting, embroidery, lace machine, other than sinkers, needles and other articles used in

forming stitches (844859); Plates, separators and other parts of electric accumulators, n.e.s. (850790);

Parts of the machinery of heading 8422 (842290);

Toothed wheels, chain sprockets and other transmission elements presented separately; parts of transmission components (848390); Marble, travertine and alabaster; simply cut or sawn, with a flat or even surface (680221); Leather further prepared after tanning or crusting, including parchment-dressed, of bovine (including buffalo) or equine animals, without hair on, split or not, (other than grain splits and full grains, unsplit) (410719); Pile fabrics & terry fabrics (excluding long pile and loop pile), of man-made fibres, knittedd or crocheted (600192);

Mixtures of odoriferous substance and mixtures, used as raw materials in industries other than the food or drink industries (330290);

Parts of machines and apparatus for soldering or welding or for hot spraying of metals, metal carbides or cermets, n.e.s. (851590);

Parts of machinery, plant/laboratory equipment, for treating materials by a process involving a change of temperature (841990);

Parts of machinery for making/finishing paper or paperboard (843999);

Particle board & similar board of ligneous material, n.e.s. (441090);

Surface-active preparations, Washing and cleaning preparations, whether or not containing soap, put up for retail sale (340220);

Knittedd/crocheted fabrics, n.e.s., and other than those made of wool, fine animal hair, cotton, synthetic or artificial fibres (600690);

Polyethylene having a sp.gr. Of 0.94 or more (390120);

Diodes (excl. Photosensitive or light emitting diodes) (854110);

Knittedd/crocheted fabrics, n.e.s., of synthetic fibres, dyed (600632);

Prefabricated buildings (940600);

Instruments, apparatus and models designed for demonstrational purposes (in education or exhibitions), unsuitable for other uses (902300); Plates, sheets, film, foil & strip, of plastics n.e.s., non-cellular and not reinforced, laminated, supported or similarly combined with other materials (392099);

Paper&paperboard, not containing fibre greater than 10%, weight 40-150g/m2, n.e.s. (480257); Electric generating sets with compressionignition internal combustion piston engines (diesel or semi-diesel engines), of an output not exceeding 75kva (850211);

Organic surface-active agents, surface-active, whether or not containing soap, including auxiliary washing preparations, not for retail sale (340290);

Conveyor belts/belting, reinforced only with metal, of vulcanised rubber (401011);

Threaded elbows, bends & sleeves of material other than stainless steel (730792);

Resinoids (330130);

Relays for a voltage > 60 V but $\leq 1.000 \text{ V}$ (853649);

Kraft paper (excl. Kraftliner & sack kraft paper), weighing more than 150 g/m2 but less than 225 g/m2 (480449);

Plates, sheets, film, foil & strip, other than cellular (392190);

Vegetable seeds, of a kind used for sowing (120991);

Bars, rods & profiles, of copper-zinc base alloys (brass) (740721);

Moulds for rubber or plastics, injection or compression types (848071);

Flat-rolled products of iron/non-alloy steel, not in coils, of a width 600mm or more, hot-rolled, without patterns in relief, of a thickness exceeding 10mm (720851);

Machine tools for planing, milling or moulding (by cutting) machines, for working wood, cork, bone, hard rubber, hard plastics or similar hard materials (846592);

Aluminium structures (like plates, rods, profiles, tubes and the like) and their parts, n.e.s. (761090);

Auminium plates, sheets & strip, (not rectangular or square), of thickness exceeding 0.2mm, alloys (760692);

Parts and accessories of printing machinery used for printing by means of plates, cylinders and other printing components of heading 8442 (844391);

Vegetable oils; soya-bean oil and its fractions, crude, whether or not degummed, not chemically modified (150710);

Builders' ware of plastics, n.e.s. (392590);

Parts of Electrical transformers, static converters and inductors (850490);

Refractory bricks, blocks, tiles & similar refractory ceramic constructional goods, n.e.s. (690290);

Parts of the engines & motors of 8412 (841290);

Waffles & wafers (190532);

Industrial monocarboxylic fatty acids; acid oils from refining; (other than stearic acid, oleic acid or tall oil fatty acids) (382319); Products suit. For use as glues/adhesives, put up for retail sale as glues or adhesives, not exceeding a net weight of 1kg (350610); Plywood; consisting only of sheets of wood (not bamboo), each ply 6mm or thinner, with at least one outer ply of tropical wood (441231);

Yarn of jute/of other textile bast fibres, single (530710);

Parts of the machinery of 84.41, for making up paper pulp, paper or paperboard, including cutting machines of all kinds (844190);
Parts of machinery for making/finishing paper

or paperboard (843999); Parts of hoists and winches (843110); Surface-active preparations, Washing and

cleaning preparations, whether or not containing soap, put up for retail sale (340220):

Flat-rolled products of iron/non-alloy steel, in coils, without patterns in relief, of a width 600mm or more, hot-rolled, of a thickness exceeding 10mm (720836);

Fresh fruit, n.e.s. (81090);

Boat/dock fenders, of vulcanised rubber, whether/not inflatable, of non-cellular rubber (401694);

Lace mechanically made in the piece, in strips or in motifs, of textile materials (580429); Aluminium structures (like plates, rods, profiles, tubes and the like) and their parts, n.e.s. (761090);

Other footwear with outer soles & uppers of rubber or plastics, (other than just covering the ankle) (640299);

Narrow woven fabrics (excluding pile fabrics), containing by weight 5% or more of elastomeric varn or rubber thread (580620):

Floor coverings of plastics (other than polymers of vinyl chloride), whether or not self-adhesive, in rolls or in the form of tiles (391890);

Other machines for resistance welding of metals, neither fully nor partly automatic (851529); Parts of the machinery, used in the industrial preparation or manufacture of food or drink (843890);

Copper tube/pipe fittings (e.g. Couplings, elbows, sleeves) of copper alloys (741220); Carpets and other textile floor coverings; tufted, of nylon or other polyamides, whether or not made up (570320);

Other parts, trimmings & accessories of um, brellas and walking sticks n.e.c. (660390); Tubes, pipes & hoses of plastics, having a minimum burst pressure of 27.6mpa (391732); Stranded wire, ropes & cables of iron/steel, not electrically insulated (731210);

Machinery parts; not containing electrical connectors, insulators, coils, contacts or other electrical features, n.e.c. In chapter 82 (848790); Tubes, pipes & hose fittings (e.g. Joints, elbows, flanges) (391740);

Articles of vulcanised rubber other than hard rubber, n.e.s. (401699);

Kraft paper (excl. Kraftliner & sack kraft paper), weighing 150 g/m2 or less (480439);

Whole bovine (incl. Buffalo)/equine animals, of a weight per skin not exceeding 8kg when simply dried, 10kg when dry-salted or 16kg when fresh, wet-salted or otherwise preserved (410120); Footwear (excl. Waterproof) with outer soles and uppers of leather (not covering the ankle) (640359);

Other articles of zinc, n.e.s. (790700); Sterile surgical catgut, similar sterile suture materials (including sterile absorbable surgical or dental yarns) and sterile tissue adhesives for surgical wound closure; sterile laminaria and sterile laminaria tents; sterile absorbable surgical or dental haemostatics; sterile surgical or dental adhesion barriers, whether or not absorbable (300610); Kidney beans, incl. White pea beans (phaseolus vulgaris), dried, shelled, whether or not skinned or split (71333);

Armoured/reinforcededed safes, strong-boxes, doors and safe deposit lockers for strong-rooms, cash or deed boxes and the like, of base metal (830300);

Dried leguminous vegetables, n.e.c., shelled, whether or not skinned or split (71390); Monofilament of which any cross-sectional dimension exceeds 1mm, rods, sticks and profile shapes, whether or not surface-worked but not otherwise worked (391690); Machines and appliances for testing the

hardness, strength, compressibility, elasticity or other mechanical properties of materials other than metals (902480);

Particle board of wood, whether/not agglomerated with resins or other organic binding substances (441011);

Machine-tools; parts and accessories, dividing heads and other special attachments for machine-tools (846630);

Automatic data processing machines; comprising in the same housing at least a central processing unit and an input and output unit, whether or not combined, n.e.c. (Digital computer CPU with some of storage, inputoutput units) (847141);

Footwear with outer soles and uppers of rubber or plastics, upper straps or thongs assembled to the sole by plugs (640220); Powders, whether/not compressed, for skin care, whether or not compressed (excluding lip, eye, manicure or pedicure preparations) (330491);

Structures and parts thereof of iron or steel, n.e.s. (730890);

Parts of lifting handling machinery n.e.s. (843139);

Poly(vinyl acetate), other than in aqueous dispersion, in primary forms (390519); Parts of machinery for sorting, screening, separating, washing, crushing etc mineral substances, for agglomerating, shaping or moulding solid fuels, ceramic pastes etc, for forming foundry moulds of sand (847490); Lecithins and other phosphoaminolipids, whether or not chemically defined (292320); AC generators, (alternators), of an output exceeding 75kva but not exceeding 375kva (850162):

Containers for compressed/liquefied gas, of iron or steel (731100);

Doors, windows & their frames & thresholds, of aluminium (761010);

Equipment for

moving/grading/levelling/scraping/excavating earth minerals or ores, not self-propelled n.e.s. (843069);

Medicaments, consisting of mixed or unmixed products n.e.s., for therapeutic or prophylactic uses, packaged for retail sale (300490);

Tools of 2/more of heading no. 8202 to 8205, put up in sets for retail sale (820600);

Reservoirs, tanks, vats & similar containers of a capacity exceeding 300 litres (392510);

Sanitary towels & tampons, napkins (481840); Flat-rolled products of iron/non-alloy steel, width 600mm or more, electrolytically plated or coated with zinc (721030);

Induction or dielectric heating equipment (excl. Ovens and furnaces) (851440);

Paper&paperboard, of which >10 % by weight of mechanical or chemi-mechanical processed fibre, in rolls (480261);

Self-adhesive plates, sheets, film, foil, tape, strip, other flat shapes thereof, other than in rolls of a width not exceeding 20cm (391990);

Waterproof footwear with rubber or plastic outer soles and uppers (not assembled by stitch, rivet, nail, screw, plug or similar), covering the ankle (but not the knee) (640192);

Cullet & other waste & scrap of glass, glass in the mass (700100);

Articles for the conveyance/packing of goods, n.e.s. (392390);

Nonwovens, whether/not

impregnated/coated/covered/laminated, of manmade filaments, (weighing more than 25g/m2 but not more than 70g/m2) (560312);

Footwear with outer soles of leather, uppers consisting of leather straps across instep and around the big toe (640320);

Footwear with outer soles and uppers of rubber or plastics, upper straps or thongs assembled to the sole by plugs (640220);

Textile fabrics; impregnated, coated, covered or laminated with poly(vinyl chloride) (590310); Boxes, cases, crates and similar articles for the conveyance or packing of goods (392310); Sanitary ware & parts thereof, of aluminium (761520);

Men's/boys' singlets & other vests, bathrobes and similar articles, of textile materials other than cotton (not knittedd or crocheted) (620799); Machines for preparing textile fibres, n.e.s. (844519);

Narrow woven fabrics n.e.s., of man-made fibres (excl. Labels, badges and similar articles) (580632);

Articles of apparel & clothing accessories (including gloves, mittens and mitts) (392620); Synthetic staple fibres, n.e.c., not carded, combed or otherwise processed for spinning (550390);

Vehicles with only spark-ignition internal combustion reciprocating piston engine, cylinder capacity over 1000 but not over 1500cc (870322);

Other articles of zinc, n.e.s. (790700); Articles for the conveyance/packing of goods, n.e.s. (392390);

Perfumes & toilet waters (330300);

Articles of vulcanised rubber other than hard rubber, n.e.s. (401699);

Sanitary ware & parts thereof, of copper (741820);

Plates, sheets, film, foil&strip, of polymers of propylene, non-cellular and not reinforced, laminated, supported or similarly combined with other materials (392020);

Sports footwear other than ski-boot, snowboard boots or cross-country ski footwear, with outer soles and uppers of rubber or plastics (640219);

Tubes, pipes & hoses, rigid, of plastics n.e.s. (391729);

Resinoids (330130);

Plates, sheets, film, foil & strip, other than cellular (392190);

Plates, sheets, film, foil & strip, of polyesters n.e.s., non-cellular and not reinforced, laminated, supported or similarly combined with other materials (392069);

Other footwear with outer soles & uppers of rubber or plastics, (other than just covering the ankle) (640299);

Woven fabrics of polyester staple fibres n.e.s., mixed mainly or solely with viscose rayon staple fibres (551511);

Vehicles with only spark-ignition internal combustion reciprocating piston engine, cylinder capacity over 1000 but not over 1500cc (870322):

Paints & varnishes (incl. Enamels & lacquers), based on polymers n.e.s., dispersed or dissolved in a non-aqueous medium (320890); Fittings for furniture/coachwork/the like, of plastic (392630);

DC motors and generators of an output exceeding 375kw (850134);

Mattresses of cellular rubber or plastics, whether or not covered (940421);

Ground-nuts, prepared/preserved, whether or not containing added sugar, other sweetening matter or spirit (200811);

Tubes, pipes & hose fittings (e.g. Joints, elbows, flanges) (391740);

Liquid dielectric transformers, having a power handling capacity <= 650 kva (850421); Wheel rims & spokes for bicycles (871492);

Parts of the machinery, used in the industrial preparation or manufacture of food or drink (843890);

Coral & similar materials, unworked (50800); Woven fabrics of cotton, dyed, weighing not more than 200g/m2, n.e.s. (521213); Turmeric (curcuma) (91030);

Preparations of a kind used in animal feeding, put up for retail sale (except dog or cat food) (230990);

Kraft paper (excl. Sack kraft paper), creped or crinkled, whether or not embossed or perforated, in rolls or sheets (480830);

Wheel rims & spokes for bicycles (871492); Spark-ignition reciprocating/rotary internal combustion piston engines for aircrafts (840710); Dried vegetables, n.e.s.; mixtures of vegetables, whole, cut, sliced, broken or in powder, but not further prepared (71290);

Household articles & toilet articles, of plastic (392490);

Rape/colza oil (excl. Low erucic acid rape or colza oil and its fractions), other than crude, but not chemically modified (151491);

Apparatus based on the use of x-rays; including radiography or radiotherapy apparatus, for medical, surgical or veterinary uses, not dental uses, excluding computed tomography apparatus (902214);

Lentils, dried, shelled, whether/not skinned or split (71340);

Parts & accessories of machines and appliances testing hardness, strength, compressibility, elasticity or other mechanical properties of materials (e.g. Metal, wood, textiles, paper, plastics) (902490);

Other Waterproof footwear with rubber or plastic outer soles and uppers (not assembled by stitch, rivet, nail, screw, plug or similar) (640199); Unwrought products of refined copper, n.e.s. (740319):

Tanks, casks, drums, cans, boxes &similar containers, for any material (excluding compressed or liquefied gas), 50l or more capacity but not exceeding 300l (731010); Sacks & bags (incl. Cones), of plastics other than ethylene polymers, for the conveyance or packing of goods (392329);

Turmeric (curcuma) (91030);

Artificial filament yarn (other tha high tenacity), filament, monofilament (less than 67 decitex), of viscose rayon, multiple (folded) or cabled, not for retail sale, not sewing thread (540341); Woven fabrics obt. From high tenacity yarn, of nylon, other polyamides or of polyesters (540710);

Shampoos (330510);

Copper-zinc base alloys (brass), unwrought (740321):

Sports footwear; tennis shoes, basketball shoes, gym shoes, training shoes and the like, with outer soles of rubber or plastics and uppers of textile materials (640411);

Essential oils of citrus fruit, n.e.s., (terpeneless or not), including concretes and absolutes (330119); Containers for compressed/liquefied gas, of iron or steel (731100);

Sanitary ware & parts thereof, of aluminium (761520);

Nonwovens, whether/not

impregnated/coated/covered/laminated, of man-made filaments, (weighing not more than 25g/m2) (560311);

Sacks & bags (incl. Cones), of Ethylene polymers, for the conveyance or packing of goods (392321);

Synthetic fibre other than high tenacity/textured, filament, monofilament (less than 67 decitex), multiple (folded) or cabled, n.e.s., not for retail sale, not sewing thread (540269):

Parts of the harvesting machinery (including parts of threshing machinery, straw or fodder balers and grass or hay mowers) (843390); Plywood, veneered panels and similar laminated wood; of bamboo (441210); Soap & organic surface-active products; in the form of bars, cakes, moulded shapes, and paper, wadding, felt and nonwovens, impregnated, coated or covered with soap or detergent, not for toilet use (340119); Woven fabrics of silk/silk waste, n.e.s. (500790);

Copper plates, sheets & strip, of a thickness exceeding 0.15mm, of copper-zinc base alloys (brass), not in coils (740929);

DC motors and generators of an output exceeding 75kw but not exceeding 375kw (850133);

Table/kitchen/other h-hold. Articles and their parts, of stainless steel (732393);

Yarn other than high tenacity/textured, filament, monofilament (less than 67 decitex), not partially oriented, single, untwisted or twisted 50 turns or less per metre, not for retail sale, not sewing thread (540247);

Recovered (waste & scrap) paper/paperboard, n.e.s., and of unsorted waste and scrap (470790);

Sauces and preparations therefor, n.e.s.; mixed condiments and mixed seasonings (210390); Jams, fruit jellies, marmalades, purees and pastes; of fruit or nuts, n.e.s. Cooked preparations (excluding homogenised), whether or not containing added sugar or other sweetening matter (200799);

Sugar confectionery; chewing gum, whether or not sugar-coated, not containing cocoa (170410);

Flat-rolled products of iron/non-alloy steel, in coils, width 600mm or more, cold-rolled, of a thickness exceeding 1mm but less than 3mm (720916);

Preparations of a kind used in animal feeding, put up for retail sale (except dog or cat food) (230990);

Nonwovens, whether/not

impregnated/coated/covered/laminated, of manmade filaments, (weighing not more than 25g/m2) (560311);

Musical instruments (other than keyboard), the sound of which is produced or must be amplified electrically (920790);

Flat-rolled products of iron/non-alloy steel, cold-rolled, of a width less than 600mm, containing by weight 0.25% or more of carbon (721129); Floor coverings of polymers of vinyl chloride, whether or not self-adhesive, in rolls or in the form of tiles (391810);

Sewing thread of synthetic staple fibres, whether or not put up for retail sale (550810);

Food preparations, n.e.s. (210690);

Onions, dried,

whole/cut/sliced/broken/powdered, but not further prepared (71220);

Cheese, n.e.s., not grated, powdered or processed (40690);

Office/school supplies, of plastics (392610); Live plants, n.e.s., incl. Their roots (60290); Pile fabrics & terry fabrics (excluding long pile and loop pile), of textile materials (other than cotton or man-made fibres), knittedd or crocheted (600199);

Sheets n.e.s. For veneer or plywood, other wood sawn lengthwise, sliced or peeled, whether or not planed, sanded or end-jointed, not thicker than 6mm (440890);

Cinnamon and cinnamon-tree flowers, other than cinnamon (Cinnamomum zeylanicum Blume), neither crushed nor ground (90619);

Uppers & parts thereof (excl. Stiffeners) (640610);

Self-adhesive plates, sheets, film, foil, other flat shapes thereof, in rolls of a width not exceeding 20cm (391910);

Winding wire for electrical purposes, of copper, insulated (854411);

Long pile fabrics, knittedd/crocheted (600110); Carpets and other textile floor coverings; tufted, of man-made textile materials (other than polyamides), whether or not made up (570330); Other footwear with outer soles & uppers of rubber or plastics, covering the ankle (640291); Dried fruit, n.e.s. (81340);

Plates, sheets, film, foil&strip, not self-adhesive, of polymers of ethylene, non-cellular and not reinforced, laminated, supported or similarly combined with other materials (392010); Tanned/crust hides & skins of bovine or equine animals, in the wet state (including wet blue), without hair on, excluding full grains, unsplit; grain splits (410419);

Preparations for use on the hair, n.e.s. (330590); Cotton, carded/combed (520300):

Blockboard, laminboard and battenboard (not bamboo, and other than plywood consisting only

Cloth, grill, netting & fencing of iron/steel, coated with plastic (731442);

Plywood; consisting only of sheets of wood (not bamboo), each ply 6mm or thinner, not containing an outer ply of non-coniferous or tropical wood (441239);

Lifting machinery n.e.s. In 84.26, for other than mounting on road vehicles, (842699); Plates, sheets, film, foil&strip, not self-adhesive, of polymers of ethylene, non-cellular and not reinforced, laminated, supported or similarly combined with other materials (392010);

Men's/boys' suits (excl. Knittedd/crocheted), of wool or fine animal hair (620311); Yarn other than sewing thread, of synthetic staple fibres, mixed mainly or solely with fibres (other than wool, fine animal hair or cotton), n.e.s., not put up for retail sale (550999);

Flat-rolled products of iron/non-alloy steel, width 600mm or more, (not corrugated), plated or coated with zinc (not electrolytically) (721049);

Household articles & toilet articles, of plastic (392490);

Furniture of plastics (940370);

Tubes, pipes & hoses, rigid, of polymers of vinyl chloride (391723);

Floor coverings of plastics (other than polymers of vinyl chloride), whether or not self-adhesive, in rolls or in the form of tiles (391890):

Waterproof footwear with rubber or plastic outer soles and uppers (not assembled by stitch, rivet, nail, screw, plug or similar), covering the ankle (but not the knee) (640192);

Tanned/crust hides & skins of bovine or equine animals, in the wet state (including wet blue), without hair on, excluding full grains, unsplit; grain splits (410419);

Sanitary towels & tampons, napkins (481840);

Carpets and other textile floor coverings; tufted, of man-made textile materials (other than polyamides), whether or not made up (570330);

Tubes, pipes & hoses, rigid, of polymers of ethylene (391721);

Other Waterproof footwear with rubber or plastic outer soles and uppers (not assembled by stitch, rivet, nail, screw, plug or similar) (640199);

Table/kitchen/other h-hold. Articles and their parts, of aluminium (761519);

Copper plates, sheets & strip, of a thickness exceeding 0.15mm, of copper-tin base alloys, not in coils (740939);

of sheets of wood each ply 6mm or thinner) (441294);

Bread, pastry, cakes, biscuits & other products, n.e.s., whether or not containing cocoa; communion wafers (190590);

Yarn other than high tenacity/textured, filament, monofilament (less than 67 decitex), single, untwisted or twisted 50 turns or less per metre, n.e.s., not for retail sale, not sewing thread (540249):

Collages and similar decorative plaques (artwork) (970190);

Stuffed pasta, whether/not cooked/otherwise prepared (190220);

Grill, netting & fencing of iron/steel, welded at the intersection, n.e.s., (other than plated or coated with zinc) (731439);

Fresh fruit, n.e.s. (81090);

Granules, chippings & powder, of stone, whether or not heat-treated (251749);

Cloth, grill, netting & fencing of iron/steel, plated or coated with zinc (731441); Plates, sheets, film, foil&strip, cellular, of plastics n.e.s. (392119);

Furniture of plastics (940370);

Plates, sheets, film, foil&strip, of polymers of propylene, non-cellular and not reinforced, laminated, supported or similarly combined with other materials (392020);

Barbed wire of iron/steel; twisted hoop or single flat wire and loosely twisted double wire, used for fencing (731300);

Fats & oils derived from milk (other than butter or dairy spreads) (40590):

Women's/girls' nightdresses & pyjamas, of manmade fibres (not knittedd or crocheted) (620822); Fresh (unripened/uncured) cheese, including whey cheese, not fermented, and curd (40610); Articles of goldsmiths'/silversmith and their parts, , of base metal clad with precious metal (711420);

Yarn other than sewing thread, of synthetic staple fibres, mixed mainly or solely with fibres (other than wool, fine animal hair or cotton), n.e.s., not put up for retail sale (550999); Juice of any single fruit/vegetable, n.e.s., unfermented, not containing added spirit, whether or not containing added sugar or other sweetening matter (200980);

Cinnamon (Cinnamomum zeylanicum Blume), neither crushed nor ground (090611); Copper plates, sheets & strip, of a thickness exceeding 0.15mm, of copper-zinc base alloys

(brass), not in coils (740929)

Yemen

Polyurethanes, in primary forms (390950); Organic comps. N.e.s. (294200); Printed circuits (853400); Felt, (excluding needleloom felt and stitch-bonded fibre fabrics), whether or not Plates, sheets, film, foil&strip, cellular, of plastics n.e.s. (392119);

Women's/girls' jackets & blazers (excl. Knitted/crocheted), of synthetic fibres (620433);

Stoppers, lids, caps and other closures, for the conveyance or packing of goods (392350); Gum/wood/sulphate turpentine oils (380510); Electric generating sets with spark-ignition internal combustion piston engines (850220); Men's/boys' jackets & blazers (excl. Knitted/crocheted), of synthetic fibres (620333);

"Brooms & brushes, consisting of twigs or other vegetable materials bound together, with or without handles (960310):"

Yarn other than sewing thread, of artificial staple fibres, containing 85% or more by weight of artificial staple fibres, not put up for retail sale (551011);

Cinnamon(Cinnamomum zeylanicum Blume), neither crushed nor ground (90611);

Articles of plastics, n.e.s. (392690);

Pencils & crayons, with leads encased in rigid sheath (960910);

Nuts, n.e.s., fresh or dried, whether or not shelled or peeled (80290);

Cereals, n.e.s. (100890);

Carboys, bottles, flasks & similar articles, for the conveyance or packing of goods (392330); Yarn of jute/of other textile bast fibres, multiple (folded) or cabled (530720)

Other Parts suitable for use solely or principally with the apparatus of headings 8525 to 8528 (reception and transmission apparatus, monitors, etc.) (852990); Winding wire, other than of copper (854419);

impregnated, coated, covered or laminated (560290);

Other Parts suitable for use solely or principally with the apparatus of headings 8525 to 8528 (reception and transmission apparatus, monitors, etc.) (852990);

Moulds for rubber or plastics, other than injection or compression types (848079); Parts and accessories for machine tools, for laser operation, metalworking machining centers, lathes and drilling machines, etc., others (846693):

Lubricating preparations (excl. Those for the treatment of textile and similar materials), not containing petroleum oils or oils obtained from bituminous minerals (340399);

Refractory cements, mortars, concretes and similar composition (381600);

Injection-moulding machines (847710); Other printers, copying machines, singlefunction printing, not capable of connecting to an automatic data processing machine or to a network (844339);

Other footwear, other than with uppers of leather/composition leather/textile materials (640590);

Essential oils, n.e.s., (terpeneless or not), including concretes and absolutes (330129); Parts of machines and apparatus for soldering or welding or for hot spraying of metals, metal carbides or cermets, n.e.s. (851590); Almonds, in shell (80211);

Footwear (excl. Waterproof) with outer soles of rubber or plastics and uppers of textile materials (640419):

Pulleys and flywheels, including pulley blocks (848350);

Hand tools, other than household tools (820559); Engines parts for internal combustion piston engines (excluding spark-ignition) (840999); Other monitors, of a kind solely principally used in an automatic data processing system (852851);

Knittedd/crocheted fabrics, n.e.s., of synthetic fibres, dyed (600632);

New pneumatic tyres, of rubber, of a kind used on motor cars (including station wagons and racing cars) (401110);

Presses, crushers & similar machinery used in the manufacture of wine, cider, fruit juices or similar beverages (843510);

Tubes, pipes & hoses of plastics, n.e.s. (391739); Other footwear with outer soles & uppers of rubber or plastics, (other than just covering the ankle) (640299):

Organic surface-active agents, whether or not put up for retail sale (340219);

Sewing machines, not household or automatic unit type (845229);

New pneumatic tyres, of rubber (excl. those with a herring-bone or similar tread), of a kind used on light industrial vehicles (401199):

Moulds for rubber or plastics, other than injection or compression types (848079); Commodities not specified (999999);

Paper, paperboard, cellulose wadding and webs of soft cellulose fibres; coated, impregnated, covered, surface-decorated or surface-coloured, n.e.s. (481190); Dates, fresh/dried (80410);

Calcium carbonate (283650);

Handkerchiefs, cleansing/facial tissues and towels (481820);

Kidney beans, incl. White pea beans (phaseolus vulgaris), dried, shelled, whether or not skinned or split (71333);

Recovered (waste & scrap) paper/paperboard, n.e.s., and of unsorted waste and scrap (470790);

Petroleum oils and oils obtained from bituminous minerals, crude (270900); Unwrought lead, refined (780110); Tanned crust skins of sheep/lambs, wi

Tanned/crust skins of sheep/lambs, without wool on, whether or not split, but not further prepared, in the wet state (including wet-blue) (410510);

Lac; natural gums, gum resins, resins and balsams, n.e.s. (130190); Ball point pens (960810);

Tanned/crust hides & skins of goats or kids, without wool or hair on, whether or not split, but not further prepared, in the wet state (including wet blue) (410621);

Lead waste & scrap (780200);

Natural gas, liquefied (271111);

Waste and scrap of primary cells, primary batteries and electric accumulators; spent primary cells, spent primary batteries and spent electric accumulators (854810); Whole bovine (incl. Buffalo)/equine animals, of a weight per skin not exceeding 8kg when simply dried, 10kg when dry-salted or 16kg

Containers for compressed/liquefied gas, of iron or steel (731100);

Paper/paperboard labels of all kind, unprinted (482190);

Tanned/crust hides & skins of bovine or equine animals, in the wet state (including wet blue), without hair on, excluding full grains, unsplit; grain splits (410419);

Zinc waste & scrap (790200);

Tanned/crust hides & skins of bovine or equine animals, in the wet state (including wet blue), without hair on, full grains, unsplit; grain splits (410411);

Lac; natural gums, gum resins, resins and balsams, n.e.s. (130190); Natural barium sulphate (barytes) (251110); Lead waste & scrap (780200); Unwrought lead, refined (780110); Recovered (waste & scrap) paper/paperboard, n.e.s., and of unsorted waste and scrap (470790); Oils of fish; fish-liver oils and their fractions, whether or not refined, but not chemically modified (150410)

Source: Own Calculations, UN Comtrade